Applying SCET to Parton Showers

by

Claudio Marcantonini

Doctor of Philosophy

ARCHIVES

at the

MASSACHUSETTS INSTITUTE OF TECHNOLOGY

June 2010

© Massachusetts Institute of Technology 2010. All rights reserved.

Author

Department of Physics May 13, 2010

Certified by.....

Iain W. Stewart Associate Professor of Physics Thesis Supervisor

///

Accepted by

Krishna Rajagopal Associate Department Head for Education

MASSACHUOFTTO INOTES			
19103	OF TECHNOLOGY	Ē	
Г			
	NOV 1 8 2010		
LIBRARIES			
LIBRARIES			

.

Applying SCET to Parton Showers

by

Claudio Marcantonini

Submitted to the Department of Physics on May 13, 2010, in partial fulfillment of the requirements for the degree of Doctor of Philosophy

Abstract

In this thesis we study corrections to parton showers in the context of soft collinear effective theory (SCET). Monte Carlo event generators like Pythia or Herwig are heavily used by experimentalists to simulate events and they are indispensable tools to make exclusive theoretical predictions. They are based on a leading log parton shower algorithm that allows to resum the dominant contributions in the soft and collinear radiation. In this work we construct a framework to classify corrections to the parton shower that can be used to systematically improve event generators. We formulate parton showers as a standard matching procedure between a tower of soft collinear effective field theories called $SCET_i$. We find two different kinds of corrections: hard-scattering corrections and jet-structure corrections. To relate these different effective field theories we make use of an important symmetry of SCET. called reparametrization invariance. In order to systematically study this symmetry, we construct operators that are invariant under reparametrization and we use them to find a minimal basis of operators that are homogeneous in the power counting. Complete basis of operators are constructed for pure glue operators for deep inelasting scattering at twist-4, for production of two and three jets from e^+e^- and for production of two jets via gluon fusion.

Thesis Supervisor: Iain W. Stewart Title: Associate Professor of Physics

To my wife Chiara

Acknowledgements

I would like to thank the many people have been close to me in the years I have spent at MIT. First of all, I would like to thank my advisor Prof. Iain Stewart. I'm very grateful to him for his precious advise in the physics field and beyond, and for all the patience he has had with me. Thanks to all my collaborators, in particular to Dr.Matthew Baumgart with whom I have done part of this work, to Dr.Wouter J. Waalewijn, Dr.Frank J.Tackmann and Dr.Carola F.Berger. Thanks to my committee members Prof. Krishna Rajagopal and Prof. Peter Fisher, and to Prof. Robert Jaffe and Prof. Alan Guth for their valuable advise. I have enjoyed my years here at the Center for Theoretical Physics. I am grateful for the opportunity I have had to build so many relationships with my colleagues, in particular thanks to Eric Fitzgerald, Vijay Kumar, David Guarrera, Ambar Jain, Christian Arnesen, Keith S. M. Lee, Riccardo Abbate, Antonello Scardicchio, Guido Festuccia, Mauro Brigante, Erasmo Coletti, Andrea de Simone and Massimiliano Procura.

I will be always grateful to Prof. John Parson for everything he thought me. I want also to thank all the members of the Focolare Movement in Boston, they have been like a family for me. In particular I want to say thanks to Francesco Mazzini, Michael James, Peter L. Risech, George Saikali and Marc Bacuyag for being close to me in crucial moments of my life. Thanks to Thibault Provost and Mihai Anton for their friendship and the patience they had for enduring me as their housemate. I want to thank my friends in Italy that, even if we leave far apart, I have felt close to me. I am grateful to my family, I have always felt their love and support for me. Thanks to my wife Chiara, for bringing me closer to God.

Contents

1	Introduction		17
	1.1	QCD	7
	1.2	m EFT	8
	1.3	Parton Showers	9
	1.4	Outline	5
2	Soft	Collinear Effective Field Theory 24	7
	2.1	Introduction to SCET	7
		2.1.1 Gauge Invariant Field Products	1
	2.2	Reparametrization invariance	3
3	Rep	rametrization Invariant Collinear Operators 3'	7
	3.1	Introduction $\ldots \ldots 3$	7
	3.2	Convolutions	0
	3.3	Construction of RPI and Gauge Invariant objects	4
3.4 Reducing the Operator Basis		Reducing the Operator Basis	8
	3.5	Extension to Massive Collinear Fields	2
	3.6 Determination of R_n and Expansion of Ψ_n and $\mathcal{G}_n^{\mu\nu}$		3
	3.7 Applications		9
		3.7.1 Scalar Current	9
		3.7.2 General Quark and Gluon Operators	0
		3.7.3 Deep Inelastic Scattering for Quarks at Twist-4	4
		B.7.4 Deep Inelastic Scattering for Gluons at Twist-4	3

		3.7.5	Two Jet production: $n-n'$ operators $\ldots \ldots \ldots \ldots \ldots$	76
		3.7.6	Three Jet Production: n_1 - n_2 - n_3 operators	79
		3.7.7	Two Jets from Gluon Fusion: $gg \rightarrow q\bar{q}$ operators	83
	3.8	Conclu	1sion	87
4	Par	ton Sh	owers to NLO	89
	4.1	Introd	uction	89
	4.2	Obtair	ning the Parton Shower with SCET	96
		4.2.1	Bauer-Schwartz Method	96
		4.2.2	Using $SCET_i$	101
	4.3	Partor	1 Shower in SCET via Operator Replacement	108
		4.3.1	LO Shower Revisited	110
		4.3.2	Soft emissions	120
		4.3.3	Summary LO Parton Showers	123
	4.4	SCET	Power Corrections to the Shower	124
		4.4.1	Hard-Scattering Corrections	125
		4.4.2	Jet-Structure Corrections	135
		4.4.3	Operator Running	138
		4.4.4	From Operators Toward a Corrected Shower	143
	4.5	Correc	etion Map at NLO/NLL	148
	4.6	Conclu	ision	151
5	Cor	clusio	ns	157
A	Inva	ariance	e to the choice of hard-vector q^{μ}	161
в	Ma	tching	\mathbf{SCET}_i to \mathbf{SCET}_{i+1}	163
	B.1	Finite	RPI	163
	B.2	Match	ing QCD to $SCET_1$	170
	B.3	Match	$\operatorname{ing} \operatorname{SCET}_1 \operatorname{to} \operatorname{SCET}_2 \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots \ldots $	176
		B.3.1	One-Gluon Emission	176
		B.3.2	Two-Gluon Emissions	184

B.4 Matching $SCET_2$ to $SCET_3$ to $SCET_N$	199
C $\mathcal{O}(\alpha_s^2)$ Correction to Splitting Function	205

List of Figures

1-1	Parton branching	21
1-2	$p\bar{p}$ collision with final shower	23
2-1	SCET collinear cone	28
2-2	SCET operators	32
4-1	$SCET_i$ operators	104
4-2	$SCET_i$ matching	109
4-3	Finite RPI	110
4-4	Kinimatic variables for one-gluon and two-gluons emission	112
4-5	Matching $SCET_1$ to $SCET_2$	113
4-6	Parton showers	119
4-7	Matching QCD to $SCET_1$ to $SCET_2$ for one emission $\ldots \ldots \ldots$	127
4-8	Plot of the ratios of the QCD and SCET_2 amplitudes squared $\ \ldots$.	130
4-9	Plot of $\Theta(x)_{\Lambda,a}$	131
4-10	Merging of the two-jet and three-jet amplitudes squared $\ .$	132
4-11	Matching QCD to $SCET_1$ to $SCET_2$ to $SCET_3$ for two-gluon emission	133
4-12	Amplitude squared for the LO SCET_N operator $\ldots \ldots \ldots \ldots$	145
4-13	Contribution to the amplitude squared of the jet-structure piece at NLC	0147
B-1	Kinematic variable for one gluon emissions	166
B-2	Matching QCD to $SCET_1$ for the two-jet configuration $\ldots \ldots \ldots$	172
B-3	Merging of the two-jet and three-jet amplitudes squared	184
B-4	Plot of the $SCET_2$ amplitudes square up to NLO	185

B-5	Matching $SCET_1$ to $SCET_2$ to $SCET_3$ for two emissions $\ldots \ldots \ldots$	186
B-6	Feynman diagrams for two emissions in SCET ₁ from the operator $\mathcal{O}_1^{(0)}$.	189
B-7	Feynman diagrams for two emissions in SCET ₁ from the operator $\mathcal{O}_1^{(1)}$.	192
C-1	Box and crossed contributions	206
C-2	Kinematics for double gluon emission	206
C-3	Single emission, one-loop contributions to $P_{qq}^{(1)}$	218

List of Tables

2.1	RPI transformations	35
4.1	Ingredients for a LO parton shower algorithm	124
4.2	Ingredients for a NLO parton shower algorithm	149
C.1	Purely finite contributions to ${\mathcal J}$	209
C.2	Contributions to $\mathcal{J} _{E,F}$	212
C.3	Contributions to crossed amplitude squared diagram	216
C.4	Contributions to box amplitude squared diagram	217

Chapter 1

Introduction

1.1 QCD

Quantum Chromodynamics (QCD) is the theory of the strong interaction which is one of the four fundamental forces. It describes the interactions between quarks and gluons and how they bind together to form particles, called hadrons, such as the proton, neutron and pion. It is also one of the building blocks of the Standard Model (SM) of particles physics. QCD emerged in the 1970s and since then its predictions have been verified by a huge number of experiments. With the Large Hadron Collider (LHC) at CERN, we will probe nature at energies never experimentally reached before. The discovery of the Higgs boson, the last piece of the SM that has not been detected yet, and possibly of physics beyond the Standard Model are the major goals of the LHC. Detecting a signal at the LHC of Higgs particle or of new physics, is not easy because of the huge background of events that depend on SM processes. For this reason it is extremely important to improve our prediction of QCD events as much as possible.

One of the key predictions of QCD is asymptotic freedom [57]. This means that the coupling constant, α_s , becomes large at low energy and small at high energy. This makes it possible to use at high energy "fixed order perturbative perturbation theory", where we can expand QCD in powers of the small α_s . When we study a physical event, in general, several energy scales are involved. This poses two main problems to perturbative QCD. First it is fundamental to disentangle the physics that happens at high energies, where we can use perturbation theory, from the physics at lower energies that is non-perturbative. This is achieved using "factorization theorems" [37]. Second, even in the regime where α_s is small, in the perturbative expansion, α_s often appears multiplied by factor of $\log(\mu_1/\mu_2)$, where μ_1 and μ_2 are two scales that are present in the process. In general a generic observable (\mathcal{O}) has the following schematic expansion in perturbation theory

$$\mathcal{O} = 1 + \alpha_s L^2 + \alpha_s^2 L^4 + \alpha_s^3 L^6 + \cdots$$

$$\alpha_s L + \alpha_s^2 L^3 + \alpha_s^3 L^5 + \cdots$$

$$\alpha_s + \alpha_s^2 L^2 + \alpha_s^3 L^4 + \cdots,$$
:
$$(1.1)$$

where 1 denotes the tree level result and $L = \log(\mu_1/\mu_2)$. Even if $\alpha_s \ll 1$, for $\mu_1 \gg \mu_2$ we may have $\alpha_s \log^2(\mu_1/\mu_2) \sim 1$, and the perturbative expansion breaks down. In this case the large logarithms need to be resummed. Resumming the first row of Eq. (1.1) is called leading logarithmic (LL) resumation, resumming the second row of Eq. (1.1) is called next-to-leading logarithmic (NLL) resumation and so on. In this thesis we study large logarithms that arise in collinear and soft emissions using an effective field theory approach.

1.2 EFT

In nature different phenomena happen at different length, time, or energy scales. The idea behind effective field theory (EFT) is that the physics at a lower scale should not depend on the details of the physics at a higher scale [76, 88]. For example at the length scale of everyday life, $\sim 10^{0}$ meter (m), classical mechanics is a very successful theory, but we know that at the scale of atoms, $\sim 10^{-10} m$, classical mechanics breaks down and the theory that describes the motion of objects is quantum mechanics. We can consider classical mechanics as an EFT of quantum mechanics at length scales of

 $\sim 10^0\,m.$ Another example is Newtonian gravity. Planetary motion (orbital radius $\sim 10^6 m$) can be well explained using Newtonian gravity but we know that the more fundamental theory that describes the gravitational force is general relativity (GR). We can recover Newtonian gravity from a perturbative expansion of GR in the ratio $\phi = G_N M/(c^2 R)$, where M is the mass of the sun and R is the typical orbital radius. Newtonian gravity is an EFT of GR in the limit of small ϕ . An EFT is an approximate theory of an underlying more fundamental theory, that includes only the appropriate degrees of freedom to describe physical phenomena occurring at a chosen length scale. The appropriate parameters are those that are at the same scale as the physical quantities we are interested in studying, and the rest of the parameters are either too small or too large for the description. In some cases an EFT can be obtained from the underlyine theory by doing a perturbative expansion in the small parameters or in the inverse of the large parameters. Using an EFT makes it possible to do calculations, and to make predictions, that would be much harder to do using the underlying theory. For example it is possible in principle to study phenomena like the motion of a bullet using quantum mechanics, but it is in practice computationally prohibitive.

There are many EFTs for QCD; each one is suitable to describe phenomena at a particular scale with particular degrees of freedom. Some examples are: chiral perturbation theory [97], heavy quark effective field theory [79] and non relativistic QCD [88]. In this work will use soft collinear effective field theory (SCET) [8, 10, 14, 18]. SCET is the theory that describes soft and collinear particles that we define in the next section. We give an introduction of SCET in chapter 2.

1.3 Parton Showers

The LHC is a proton-proton collider. In a typical proton-proton collision, two partons, quarks or gluons, are extracted from the protons and interact in a hard collision. The collision may produce leptons, such as electrons, muons and neutrinos, and hadrons. There are at least two energetic scales involved: Q, the energy at which the pro-

tons interact, and Λ_{QCD} the scale where perturbation theory is not valid anymore. Q is called a hard scale and in an energetic collision it is high enough that we can use perturbation theory. Λ_{QCD} is called the hadronization scale because it is the energy scale governing how quarks and gluons bind together to form hadrons. Using the fundamental theory of QCD, fixed order perturbative calculations have been implemented only at next-to-leading order (NLO), α_s^2 , in many cases and next-nextto-leading order (NNLO), α_s^3 , in few cases. At tree level, the number of Feynman diagrams to calculate grows factorially with the number of partons in the final state. These calculations have only been performed for a relatively low number of external particles. Even tree-level expressions have only been worked out for $\mathcal{O}(10)$ particles [55, 69, 75, 83]. At one-loop the frontier is four external particles processes [26, 27, 67] and at two loops is three [2, 3, 30]. These limitation show that a direct computation of QCD processes with many external particles is not currently feasible. However, high energy colliders produce events with thousands of particles in each event. Besides, there are regions of phase space in which high-order terms are enhanced and cannot be neglected. If Q is a hard scale in the process then, in these regions, the amplitude gets enhanced so that its coefficient is $(\alpha_s \ln^2(Q/q))^m$, where $q \ll Q$ refers to a small scale that is induced by the choice of observables.

Instead of studying a process for a precise prediction to some order in perturbation theory, a different approach is to seek an approximate result where we capture the dominant contributions taking into account such enhanced terms at all orders. Enhanced higher-order regions come from kinematic configurations where the relevant QCD matrix element becomes large. In particular, this is associated with emissions of soft or collinear particles. To be more precise, let us consider a particle *i* with momentum q_i coming from some hard scattering that happens at energy scale Q, branching into two particles *j* and *k* with momenta q_j and q_k , see Fig. 1-1-(*A*). Because of the internal propagator, the amplitude (*A*) of this process is proportional to $1/q_i^2$ and in the limit where the mass of the particles *j* and *k* is zero, we have

$$A \sim \frac{1}{q_i^2} = \frac{1}{2E_j E_k \left(1 - \cos\theta\right)}$$
(1.2)



Figure 1-1: (A): branching of particle *i* to particles jk; (B): strongly ordered limit where $q_0^2 \gg q_1^2 \gg q_2^2 \gg \ldots$, this implies that each emission is more collinear than the previous one.

where E_j and E_k are the energies of the particles j and k, and θ is the angle between $\vec{q_j}$ and $\vec{q_k}$. From Eq. (1.2) we see that the amplitude gets enhanced when E_j or $E_k \to 0$, or $\theta \to 0$. In the first case the particles emitted are called soft because of the low energy, while in the second case they are called collinear because the particles are close to each other. A key characteristic is that in the collinear limit the cross section factorizes

$$d\sigma_{X+jk} = dz \frac{dq_i}{q_i^2} P_{i \to jk} d\sigma_{X+i}, \qquad (1.3)$$

where $d\sigma_{X+i}$ is the cross section to produce the particle *i* inside some larger process X, $d\sigma_{X+jk}$ is the cross section to produce the particles jk, and P is called "splitting function" and represents the probability of the particle *i* to split into particle jk and can be calculated perturbatively in α_s . As an example, for a quark splitting to a quark and a gluon $(q \to qg)$ after averaging and summing over spins, at leading order in α_s , this is

$$P_{q \to qg}^{(0)} = \frac{\alpha_s}{2\pi} C_F \frac{1+z^2}{1-z},$$
(1.4)

where z is the momentum fraction of the emitted quark with respect to the parent quark and $C_F = 4/3$. If we integrated Eq. (1.3) over q_i^2 , we would get a large logarithm $\log(Q/q)$. We can extend this argument to the emission of an arbitrary number of particles $(q_0, q_1, q_2, ...)$ and we get large logarithms in the "strongly ordered" region where

$$q_0^2 \gg q_1^2 \gg q_2^2 \gg \dots$$
 (1.5)

Condition Eq. (1.5) is equivalent to $q_{0\perp} \gg q_{1\perp} \gg q_{2\perp} \gg \ldots$ where \perp refers to the perpendicular component of the momentum with respect to the momentum of the mother particle. This means that in the strongly ordered limit, each emission is more collinear than the previous one, as depicted in Fig. 1-1-(*B*).

We saw that $P^{(0)}$ gives the probability for a parton to split. Using the Altarelli-Parisi equation it is possible to prove that the "Sudakov Factor", $\Delta(q^2, q_0^2)$, gives the probability of a parton to evolve from q_0^2 to q^2 without branching [44], where

$$\Delta(q^2, q_0^2) = \exp\left[-\int_{q_0^2}^{q^2} \frac{dq'^2}{q'^2} \int dx \frac{\alpha_s}{2\pi} P_{jk}^{(0)}(x)\right].$$
 (1.6)

We can use the splitting function and the Sudakov factor to construct an algorithm that branches a parton *i* into two partons *jk* and then iterates this process to produce an arbitrary number of partons in the final state. This process is called a parton shower and can be implemented in a Monte Carlo simulation as an event generator. Whereas a fixed order calculation is based on the perturbative expansion of α_s , the parton showers is defined in the soft-collinear limit and uses a probabilistic Markov chain of $1 \rightarrow 2$ particle splittings to recursively generate partons. In this way we resum the LL contributions by systematically treating real parton radiation.

To study an event in a hadron collider both fixed order calculations and parton shower event generators are used. We generate an event in three phases [40]. First we select the hard process at the parton level with a probability proportional to its production cross section, calculated using standard fixed order perturbation theory. Second, the produced partons, which are taken to be highly off-shell at the hard scale Q, radiate additional partons using an event generator and evolve down until their off-shellness reaches the hadronization scale $\Lambda_{\rm QCD}$. Finally, all the partons hadronize using a confinement model. We illustrate these three phases in Fig. 1-2 for



Figure 1-2: $p\bar{p}$ collision with final shower: two partons taken from the protons (red blobs) interact at scale Q. parton showers is used to emit radiations from Q down to the scale $\Lambda_{\rm QCD}$ where the partons hadronize (blue blobs). There is also radiation from the initial partons and from the remnants of the protons (not shown).

a $p\bar{p}$ collision. Two partons taken from the protons (red blobs) interact at scale Q producing two partons. A parton shower event generator is used to emit radiation from Q down to the scale $\Lambda_{\rm QCD}$ where the partons hadronize (blue blobs). There is also radiation from the initial partons and from the remnants (not shown); these are also described using a parton showers approach. Monte Carlo event generators like Pythia [91, 92] or Herwig [5, 38] are heavily used by experimentalists to simulate these types of events and have proved indispensable for making exclusive theoretical predictions.

There have been several improvements to LL parton showers such as MC@NLO [52], or CKKW [31], but a systematic way to resum NLL is missing in the literature and there is not even a clear method to catalog all the necessary corrections. The main problem to include NLL corrections is that we have to take into account emissions that are not strongly ordered, where $q_i^2 \gg q_{i+i}^2$ and where the factorization formula Eq. (1.3) is no longer valid. This means that we have to consider interference between different amplitudes as well as spin and color correlations.

In this thesis we set up a rigorous framework to study corrections to parton showers

using SCET to pave the way for an implementation of a NLL parton shower algorithm. SCET is an appropriate effective theory for studying parton showers because it is designed to reproduce exactly the limit of soft and collinear particles. Moreover, SCET is organized in an expansion of a power counting parameter that makes it possible to classify all corrections to a known order in this parameter. The first work on parton showers using SCET was Ref. [17], where the authors proved how the splitting function and the Sudakov factor emerge naturally in SCET. They reproduced the LL parton showers using SCET but they introduce choices and approximations at several points which makes prohibitive to calculate corrections.

In our work, we describe the parton shower using a tower of independent but related effective field theories that we call $SCET_i$. Each $SCET_i$ is a soft-collinear effective field theory. The difference between $SCET_i$ and $SCET_{i+1}$ is that $SCET_{i+1}$ describes collinear particles with virtuality that is much smaller than the virtuality of a collinear particle in $SCET_i$, that is if q_{i+1} is a generic collinear particle in $SCET_{i+1}$ and q_i a generic collinear particle in SCET_i we have that $q_{i+1}^2 \ll q_i^2$. We describe each emission in the strongly ordered region using a different $SCET_i$. Even if we have many EFTs, we will use only a single power counting parameter that we call λ . The LO operator in λ for n partons describes n emissions in the strongly ordered region. To resum NLL we need to calculate the NLO operators in λ as well as correction in α_s . In this work we will focus on corrections of the power counting, but in our framework we can also calculate corrections in α_s . We find two different kinds of corrections at NLO in λ : hard-scattering corrections and jet-structure corrections. The hardscattering corrections depend on the hard-scattering process being investigated. The jet-structure corrections are independent from what happens at the scale Q, hence they are universal in the sense that they are the same for each process we want to study. The $SCET_i$ picture, besides defining a clear method to calculate NLO corrections, has another important advantage in that we have factorization between emissions also at NLO. This characteristic is a crucial ingredient that gives hope for a future construction of a NLL parton shower algorithm.

In order to relate these different $SCET_i$ we will make use of an important symmetry

of SCET, called reparametrization invariance (RPI). Part of this thesis is devoted to study RPI. Symmetries are fundamental tools in all fields of physics. Knowing that a quantity is invariant under a set of transformations allows us to make some predictions. For example, if we have a quantity that depends only on two vectors, p^{μ} and q^{μ} , and we know it is a Lorentz scalar, we can immediately say that it can be only a function of q^2 , p^2 and $p \cdot q$. Similarly, knowing that SCET is invariant under RPI, we are allowed to express all quantities in SCET using a complete a set of RPI invariant objects. In this thesis we construct operators that are invariant under reparametrization. Using RPI operators will turn out to be very powerful method to find a minimal basis that is homogeneous in the power counting in particular for processes with multiple jets.

1.4 Outline

In chapter 2 we give a brief review of SCET. After describing the main ingredients of SCET, we define RPI in this context. In chapter 3 we construct a set of operators that are invariant under reparametrization. It is based on Ref. [80]. We can use this set to reduce the number of operators in SCET. We construct a minimum basis of pure glue operators for DIS at twist-4, for production of two and three jets from e^+e^- , and for production of 2 jets from gluon fusion. Chapter 4 is dedicated to parton showers and is based on Ref. [21]. We formulate the shower emissions as a standard matching procedure between different SCET_i, namely SCET_i \rightarrow SCET_{i+1}. We use this formulation to classify and compute various corrections to the shower. Conclusions are given in chapter 5. We leave many of the technical details to the appendices.

Chapter 2

Soft Collinear Effective Field Theory

2.1 Introduction to SCET

Soft Collinear Effective Theory (SCET) is an effective field theory of QCD that describes the interactions of collinear and soft particles [8, 10, 14, 18]. The momentum pof any particle can be decomposed along two light-cone vectors, n and \bar{n} , with $n^2 = 0$, $\bar{n}^2 = 0$ and $n \cdot \bar{n} = 2$, as

$$p^{\mu} = n \cdot p \frac{\bar{n}^{\mu}}{2} + \bar{p} \frac{n^{\mu}}{2} + p_{\perp}^{\mu} , \qquad (2.1)$$

where $\bar{p} = \bar{n} \cdot p$ and the particle's invariant mass is $p^2 = p^+ \bar{p} + p_{\perp}^2$. We use a Minkowskian notation for $p_{\perp}^2 = -\vec{p}_{\perp}^2$, where \vec{p}_{\perp} is Euclidean. SCET's degrees of freedom include collinear and soft fields.¹

A particle is collinear to a direction n if its momentum scales as:

$$(n \cdot p, \,\bar{p}, \,p_{\perp}) \sim (\lambda^2, \,1, \,\lambda) \,\bar{p} \,, \tag{2.2}$$

¹Our primary interest here is the perturbative structure of jets, so we use what in literature is often called $SCET_I$ theories with collinear and ultrasoft modes. For simplicitly we will always use the phrase soft in place of ultrasoft.



Figure 2-1: In SCET, a particle with momentum p is collinear to the direction n if \vec{p} is inside a cone centered in \vec{n} and of angle λ .

where $\bar{p} \sim Q$ is the hard scale, and $\lambda \ll 1$ is the power counting parameter of the SCET. Pictorially, we can think of a particle collinear to the direction n as having its three-momentum \vec{p} is inside a cone centered on \vec{n} and of angle λ , see Fig. (2-1). A particle is soft if

$$(n \cdot p, \bar{p}, p_{\perp}) \sim (\lambda^2, \lambda^2, \lambda^2) Q.$$
 (2.3)

Both Eqs. (2.2) and (2.3) imply $p^2 \lesssim Q\lambda^2$.

We obtain SCET from QCD by expanding in powers of λ , integrating out the hard modes and dividing the quark and gluon fields into separate soft and collinear modes. The soft fields are $\xi_s(x)$ and $A_s(x)$ where $\partial^{\mu} \xi_s(x) \sim \partial^{\mu} A_s(x) \sim \lambda^2$. For the collinear case, we introduce a momentum-space lattice and we give each field an index-label describing which lattice vector it is collinear to. To divide the QCD fields in this way, we split the momentum of a collinear particle into a "large" part \tilde{p}^{μ} and a residual one $k^{\mu} \sim \lambda^2$

$$p^{\mu} = \tilde{p}^{\mu} + k^{\mu}$$
, where $\tilde{p}^{\mu} \equiv n \cdot p \frac{n^{\mu}}{2} + p_{\perp}^{\mu}$. (2.4)

We can remove the large momenta \tilde{p} from the fermion field by the Fourier transform

$$\psi(x) = \sum_{\tilde{p},n} e^{-i\tilde{p}\cdot x} \psi_{n,\tilde{p}} \,. \tag{2.5}$$

For a collinear particle along n, $\partial^{\mu} \psi_{n,\tilde{p}}(x) \sim \lambda^2$. The four component field, $\psi_{n,\tilde{p}}$, has two large components $\xi_{n,\tilde{p}}$, and two small components $\xi_{\bar{n},\bar{p}}$, that are defined using the following Dirac-space projectors:

$$\psi_{n,\tilde{p}} = \frac{\hbar \bar{n}}{4} \psi_{n,\tilde{p}} + \frac{\bar{n} \bar{n}}{4} \psi_{n,\tilde{p}} \equiv \xi_{n,\tilde{p}} + \xi_{\bar{n},\tilde{p}}.$$
(2.6)

These satisfy the relations

We can also similarly define a collinear gluon field $A^{\mu}_{n,\tilde{q}}(x)$ whose fluctuations are characterized by the scale of its residual momenta, $q^2 \sim \lambda^2$. Pictorially, we can think of $\xi_{n,\tilde{p}}$ and $A^{\mu}_{n,\tilde{q}}(x)$ as fields that create a particle whose three-momentum lies inside a cone with opening angle $\sim \lambda$ about the three-direction \vec{n} . We define \mathcal{P}^{μ}_{n} as the momentum operator that picks up the large component of the momentum: $\mathcal{P}^{\mu}_{n} \xi_{n,\tilde{p}}(x) = \tilde{p}^{\mu} \xi_{n,\tilde{p}}(x)$. Collinear fields always appear with a sum over \tilde{p} , so it is useful to redefine the fields as

$$\xi_n = \sum_{\tilde{p}} e^{i\tilde{p} \cdot x} \xi_{n,\tilde{p}} , \quad A_n^{\mu} = \sum_{\tilde{q}} e^{i\tilde{q} \cdot x} A_{n,\tilde{q}}^{\mu} .$$
 (2.8)

The leading order SCET Lagrangian is

$$\mathcal{L}_{\text{SCET}}^{(0)} = \mathcal{L}_s^{(0)} + \sum_{n \in \{n_i\}} \mathcal{L}_n^{(0)}$$
(2.9)

where $\mathcal{L}_n^{(0)}$ has only collinear interactions among particles collinear to the same n. Particles collinear to different directions can interact either by the exchange of soft modes in $\mathcal{L}_s^{(0)}$, or from their coupling to other sectors in external operators. Two collinear sectors in SCET, n_1 and n_2 , are distinct if [9]

$$n_1 \cdot n_2 \gg \lambda^2 \,, \tag{2.10}$$

so any particle is collinear in at most one direction within a given SCET.

Thus the key defining concepts of an SCET-theory are its hard-scale Q, its collinear sectors $\{[n_i]\}$, and its power counting parameter λ which governs the importance of operators and size of the collinear sectors.

We define collinear covariant derivatives as

$$i\bar{n}\cdot D_n = \overline{\mathcal{P}}_n + g\bar{n}\cdot A_{n,p}, \quad iD_n^{\perp\mu} = \mathcal{P}_{n\perp}^{\mu} + gA_{n,p}^{\perp\mu}, \quad in\cdot D_n = in\cdot\partial + gn\cdot A_{n,p}, \quad (2.11)$$

where \mathcal{P}_n is the momentum operator that picks up the large component of the momentum, that is $\mathcal{P}_n \xi_{\tilde{p},n} = \tilde{p} \xi_{p,n}$. When integrating out hard offshell fluctuations and constructing gauge invariant structures in SCET, it is necessary to include collinear Wilson lines, W_n , defined by [14]

$$W_n(x) = \left[\sum_{\text{perms}} \exp\left(\frac{-g}{\overline{\mathcal{P}}_n} \,\bar{n} \cdot A_{n,p}(x)\right)\right].$$
(2.12)

The collinear fields $A_{n,p}^{\mu}$ are defined with the zero-bin procedure [78]. To couple soft degrees of freedom we define an soft covariant derivative

$$iD_s^{\mu} = i\partial^{\mu} + gA_s^{\mu} \,, \tag{2.13}$$

that can act on collinear fields. At lowest order the coupling to *n*-collinear fields involves $n D_s$ and can be removed from the Lagrangian by the BPS field redefinition [14]

$$\xi_{n,p}(x) \to Y_n(x)\xi_{n,p}(x), \qquad A_{n,q}(x) \to Y_n(x)A_{n,q}(x)Y_n^{\dagger}(x), \qquad (2.14)$$

with the soft Wilson line

$$Y_n(x^{\mu}) = \Pr \exp \left(i g \int_{-\infty}^0 ds \ n \cdot A_s(x^{\mu} + sn^{\mu}) \right).$$
 (2.15)

This field redefinition allows us to organize power corrections as gauge invariant products of collinear and soft fields as we discuss in the next subsection.

The SCET collinear lagrangian, \mathcal{L}_n is derived from the QCD lagrangian by inte-

grating out the field, $\xi_{\bar{n}}$. At LO, we have [10, 18]

$$\mathcal{L}_{n}^{(0)} = \bar{\xi}_{n} \left(in \cdot \partial + g \, n \cdot A_{n} + i \not\!\!\!D_{n\perp} W_{n} \frac{1}{\mathcal{P}_{n}} W_{n}^{\dagger} i \not\!\!\!D_{n\perp} \right) \frac{\not\!\!\!/}{2} \xi_{n} \,, \tag{2.16}$$

where we intrinsically sum over the large, label momenta, \tilde{p} , as well as the collinear index, n, which we have kept explicit as a label. The LO collinear Lagrangian for gluons is given in [14].

Operators are formed from products of the above fields, and the power counting for an operator is determined by adding up contributions from its constituents. The power counting for the fields and derivatives in SCET is

$$\xi_n \sim \lambda, \qquad (n \cdot A_n, \bar{n} \cdot A_n, A_n^{\perp}) \sim (\lambda^2, 1, \lambda), \qquad q_s \sim \lambda^3, \qquad A_s \sim \lambda^2,$$
$$i\partial^{\mu} \sim \lambda^2, \qquad (in \cdot \partial, \bar{n} \cdot \mathcal{P}, \mathcal{P}_{n\perp}) \sim (\lambda^2, 1, \lambda), \qquad W_n \sim Y_n \sim \lambda^0. \tag{2.17}$$

2.1.1 Gauge Invariant Field Products

To build operators in SCET we want to use structures which are gauge invariant and homogeneous in the power counting. For SCET a convenient set of gauge invariant structures are:

$$\chi_n \equiv W_n^{\dagger} \xi_n, \qquad \mathcal{D}_n^{\mu} \equiv W_n^{\dagger} D_n^{\mu} W_n \,, \tag{2.18}$$

together with the \mathcal{P}_n^{μ} label momentum operator and derivative operator $i\partial^{\mu}$ acting on these structures. The collinear fields in Eq. (2.18) are the ones obtained after the field redefinition in Eq. (2.14). It is convenient to be able to switch the collinear derivatives multiplied by Wilson lines for gauge invariant field strengths, for which we use

$$i\mathcal{D}_{n}^{\perp\mu} = \mathcal{P}_{n\perp}^{\mu} + g\mathcal{B}_{n\perp}^{\mu}, \qquad i\overleftarrow{\mathcal{D}}_{n}^{\perp\mu} = -\mathcal{P}_{n\perp}^{\dagger\mu} - g\mathcal{B}_{n\perp}^{\mu},$$
$$in \cdot \mathcal{D}_{n} = in \cdot \partial + gn \cdot \mathcal{B}_{n}, \qquad in \cdot \overleftarrow{\mathcal{D}}_{n} = in \cdot \overleftarrow{\partial} - gn \cdot \mathcal{B}_{n}, \qquad (2.19)$$



Figure 2-2: Final state with a quark (q_1) , antiquark $(p_{\bar{q}})$, and gluon (k_1) . Different kinematic configurations are described by different SCET operators. In (I), the quark and the gluon are collinear to the direction n_0 , represented by their sharing a cone. In (II), the vectors q_1 and k_1 are too far apart to be collinear. The Feynman diagrams below show that collinear fields can come from the Lagrangian vertices, but noncollinear ones arise from higher-multiplicity operators.

and note that $\bar{n} \cdot \mathcal{D}_n = \bar{\mathcal{P}}_n$. Here the field strength tensors are

$$g\mathcal{B}_{n\perp}^{\mu} \equiv \left[\frac{1}{\overline{\mathcal{P}}_{n}}[i\bar{n}\cdot\mathcal{D}_{n},i\mathcal{D}_{n}^{\perp\mu}]\right], \qquad gn\cdot\mathcal{B}_{n} \equiv \left[\frac{1}{\overline{\mathcal{P}}_{n}}[i\bar{n}\cdot\mathcal{D}_{n},in\cdot\mathcal{D}_{n}]\right], \qquad (2.20)$$

where the label operators and derivatives act only on fields inside the outer square brackets, and $g\mathcal{B}_{n\perp}^{\mu}$ and $gn\cdot\mathcal{B}_n$ are Hermitian.

We can construct gauge invariant operators using the fields defined above. Since the collinear fields carry a label referring to a specific light-cone vector, these operators describe particles in a specific region of phase space. SCET therefore distinguishes situations with the same number particle but different kinematics using different operators. For example, one can take an amplitude for three external particles: a quark (q_1) , a gluon (k_1) and an antiquark $(p_{\bar{q}})$. We can consider two different configurations that we call $|q_{n_0}g_{n_0}\bar{q}_{\bar{n}}\rangle$ and $|q_{n_1}g_{n'_1}\bar{q}_{\bar{n}}\rangle$. In the first, shown in Fig. 2-2(*I*), the quark and the gluon are n_0 -collinear, and the antiquark is collinear to a different direction, \bar{n} . Here the amplitude is described by operators with two distinct directions, say

$$\bar{\chi}_{n_0}\Gamma\chi_{\bar{n}} \sim \lambda^2, \quad \bar{\chi}_{n_0}g\mathcal{B}^{\mu}_{n_0\perp}\Gamma'\chi_{\bar{n}} \sim \lambda^3,$$
(2.21)

(where the form of the Dirac structures Γ and Γ' are not central to our discussion here). The first operator in (2.21) can emit $\bar{n} \cdot A_{n_0}$ gluons from the Wilson line in χ_{n_0} but requires a Lagrangian insertion to emit an $A_{n_0}^{\perp}$ gluon. Schematically the amplitude for a transverse gluon has contributions

$$A^{I} = \int d^{4}x \left\langle 0 \left| T \mathcal{L}_{n_{0}}^{(0)}(x) \, \bar{\chi}_{n_{0}} \Gamma \chi_{\bar{n}}(0) \right| q_{n_{0}} g_{n_{0}} \bar{q}_{\bar{n}} \right\rangle + \left\langle 0 \left| \bar{\chi}_{n_{0}} g \mathcal{B}_{n_{0} \perp}^{\mu} \Gamma' \chi_{\bar{n}}(0) \right| q_{n_{0}} g_{n_{0}} \bar{q}_{\bar{n}} \right\rangle.$$
(2.22)

In Fig. 2-2(*II*) each of the particles is collinear in a distinct direction, so no cone of size $\sim \lambda$ fits two of the momenta. In this case, the amplitude can only come from an operator with three distinct directions, such as $\bar{\chi}_{n_1} \mathcal{B}^{\mu}_{n'_1 \perp} \Gamma' \chi_{\bar{n}}$:

$$A^{II} = \langle 0 | \bar{\chi}_{n_1} g \mathcal{B}^{\mu}_{n'_{\perp} \perp} \Gamma'' \chi_{\bar{n}} | q_{n_1} g_{n'_1} \bar{q}_{\bar{n}} \rangle .$$
(2.23)

2.2 Reparametrization invariance

When a set of fields have their largest momentum component in a light-like or timelike direction then the structure of operators built from these fields is constrained by reparametrization invariance. This invariance appears due to the ambiguity in the decomposition of momenta in terms of basis vectors and in terms of large and small components, in other words reparametrization constraints arise because the decomposition in Eq. (2.1) is not unique. We can shift n by a small amount and still have a suitable basis vector for the particle. We also have a large amount of freedom in the choice of \bar{n} . For each $\{n, \bar{n}\}$ pair the most general set of RPI transformations which preserves the relations $n^2 = 0$, $\bar{n}^2 = 0$, and $n \cdot \bar{n} = 2$ are

(I)
$$\begin{cases} n_{\mu} \to n_{\mu} + \Delta_{\mu}^{\perp} \\ \bar{n}_{\mu} \to \bar{n}_{\mu} \end{cases}$$
 (II)
$$\begin{cases} n_{\mu} \to n_{\mu} \\ \bar{n}_{\mu} \to \bar{n}_{\mu} + \varepsilon_{\mu}^{\perp} \end{cases}$$
 (III)
$$\begin{cases} n_{\mu} \to (1+\alpha) n_{\mu} \\ \bar{n}_{\mu} \to (1-\alpha) \bar{n}_{\mu} \end{cases}$$
 (2.24)

where the five infinitesimal parameters are $\{\Delta_{\mu}^{\perp}, \epsilon_{\mu}^{\perp}, \alpha\}$, and satisfy $\bar{n} \cdot \epsilon^{\perp} = n \cdot \epsilon^{\perp} = \bar{n} \cdot \Delta^{\perp} = n \cdot \Delta^{\perp} = 0$. The transformations (I), (II) and (III) in Eqs. (2.24) are called RPI-I, RPI-II and RPI-III. To ensure that n provides an equivalent physical description of the collinear direction for these particles requires the power counting $\{\Delta_{\mu}^{\perp}, \epsilon_{\mu}^{\perp}, \alpha\} \sim \{\lambda^{1}, \lambda^{0}, \lambda^{0}\}$ [77]. Thus n can only be shifted by a small amount, while parametrically large values of α and $\varepsilon_{\mu}^{\perp}$ are allowed. This is because the vector n has physical meaning, \vec{n} is the direction where most of the momentum is allocated, that is the direction \vec{p} is inside a cone centered on \vec{n} with on opening angle $\sim \lambda$. The RPI-I transformations moves n inside this collinear cone. \bar{n} does not carry any real physical meaning and it is only needed to decompose the momentum in (2.1). The collinear sectors $\{n_i\}$ in SCET are really equivalence classes of null vectors, $\{[n_i]\}$, where an equivalence class $[n_j]$ is defined as

$$[n_j] = \{ n \in [n_j] | n \cdot n_j \lesssim \lambda^2 \}.$$

$$(2.25)$$

The class $[n_j]$ consists of all light-like vectors connected to n_j^{μ} by a type-I RPI transformation $n_j^{\mu} \rightarrow n_j^{\mu} + \Delta_{n_j \perp}^{\mu}$.

The type-III boost simply ensures that $(\#Nn_i) - (\#N\bar{n}_i) - (\#Dn_i) + (\#D\bar{n}_i) = 0$ for each *i*, where $(\#Nn_i)$ counts the number of n_i factors in the numerator of an operator, $(\#D\bar{n}_i)$ counts the numbers of \bar{n}_i factors in the denominator, etc. With three collinear directions an example of a type-III RPI invariant parameter is

$$\frac{n_1 \cdot \bar{n}_2 \ \bar{n}_1 \cdot \bar{n}_3}{\bar{n}_2 \cdot \bar{n}_3} \ . \tag{2.26}$$

The type-I and type-II transformations of collinear objects are more interesting and are summarized in Table 2.1, which we take from Ref. [77]. Since the factors induced

Type (I)	Type (II)
$n \rightarrow n + \Delta^{\perp}$	$n \rightarrow n$
$\bar{n} \rightarrow \bar{n}$	$\bar{n} \rightarrow \bar{n} + \varepsilon^{\perp}$
$n \cdot D_n \to n \cdot D_n + \Delta^\perp \cdot D_n^\perp$	$n \cdot D_n \rightarrow n \cdot D_n$
$D_{n\perp}^{\mu} \rightarrow D_{n\perp}^{\mu} - \frac{\Delta_{\perp}^{\mu}}{2} \bar{n} \cdot D_n - \frac{\bar{n}^{\mu}}{2} \Delta^{\perp} \cdot D_{n\perp}$	$D_{n\perp}^{\mu} \rightarrow D_{n\perp}^{\mu} - \frac{\varepsilon_{\perp}^{\mu}}{2} n \cdot D_n - \frac{n^{\mu}}{2} \varepsilon^{\perp} \cdot D_n^{\perp}$
$\bar{n} \cdot D_n \rightarrow \bar{n} \cdot D_n$	$\bar{n} \cdot D_n \rightarrow \bar{n} \cdot D_n + \bar{\varepsilon}^\perp \cdot D_n^\perp$
$\xi_n \to \left(1 + \frac{1}{4} \not \Delta^{\perp} \vec{\eta}\right) \xi_n$	$\xi_n \rightarrow \left(1 + \frac{1}{2} \not\in^{\perp} \frac{1}{\bar{n} \cdot D_n} \not\!\!\!D_n^{\perp}\right) \xi_n$
$W \rightarrow W$	$W \rightarrow \left[\left(1 - \frac{1}{\bar{n} \cdot D_n} \ \epsilon^{\perp} \cdot D_n^{\perp} \right) W \right]$

Table 2.1: Summary of infinitesimal type I and II transformations from Ref. [77]. With multiple collinear directions these transformations exist for each $\{n_i, \bar{n}_i\}$ pair.

by these transformations occur at different orders in λ , demanding overall invariance of a physical process provides connections between the Wilson coefficients of operators at different orders in the expansion.

When we couple collinear and soft particles there is another ambiguity, associated with the decomposition of a collinear momentum into large and small pieces. If the total momentum P^{μ} of a collinear particle is decomposed into the sum of a large collinear p^{μ} and a small soft momentum k^{μ} :

$$P^{\mu} = p^{\mu} + k^{\mu} = \frac{n^{\mu}}{2} \bar{n} \cdot (p+k) + \frac{\bar{n}^{\mu}}{2} n \cdot k + (p_{\perp} + k_{\perp})^{\mu}, \qquad (2.27)$$

then operators must be invariant under a transformation that takes $\bar{n} \cdot p \to \bar{n} \cdot p + \bar{n} \cdot \ell$, $p_{\perp}^{\mu} \to p_{\perp}^{\mu} + \ell_{\perp}^{\mu}$, $\bar{n} \cdot k \to \bar{n} \cdot k - \bar{n} \cdot \ell$, and $k_{\perp}^{\mu} \to k_{\perp}^{\mu} - \ell_{\perp}^{\mu}$. To construct invariant objects that have nice gauge transformation properties we use the combined covariant derivatives [15, 25],

$$iD_{n\perp}^{\mu} + W_n iD_{s\perp}^{\mu} W_n^{\dagger}, \qquad \qquad i\bar{n} \cdot D_n + W_n i\bar{n} \cdot D_s W_n^{\dagger}. \qquad (2.28)$$

This can be implemented by taking

$$i\mathcal{D}_{n\perp}^{\mu} \to i\mathcal{D}_{\text{full}}^{\perp\mu} = i\mathcal{D}_{n\perp}^{\mu} + iD_{\text{s}}^{\perp\mu}, \qquad \overline{\mathcal{P}}_n \to i\bar{n}\cdot\mathcal{D}_{\text{full}} = \overline{\mathcal{P}}_n + i\bar{n}\cdot D_{\text{s}}, \qquad (2.29)$$

and then expanding in λ . The results in Eq. (2.29) give powerful relations as they relate the coefficients of operators involving collinear fields to those involving soft fields. These relations are quite easy to derive order by order in λ . Note that reparametrization constraints associated with transformation of the soft Wilson line Y_n are automatically enforced by the other constraints.²

²For example, prior to the field redefinition only the combination $in \cdot D = in \cdot \partial + gn \cdot A_s + gn \cdot A_n$ appears acting on collinear fields. A type-I transformation connects this to a \mathcal{D}_n^{\perp} , and Eq. (2.29) then connects this to the same iD_s^{\perp} that one would find by direct transformation of Y_n .
Chapter 3

Reparametrization Invariant Collinear Operators

3.1 Introduction

To study a process using SCET, the standard procedure is to take the QCD current, J_{QCD} , underlying that event and to expand it in terms of SCET operators using an operator expansion:

$$J_{\rm QCD} = \sum_{i} C_i O_i \,, \tag{3.1}$$

where C_i are the Wilson coefficients describing the physics at the hard scale, and O_i are the SCET operators that reproduce the infra-red (IR) behavior.¹ The process of calculating the Wilson coefficients is called matching. All the operators in SCET have a power counting in λ , and the OPE is organized as an expansion in λ . In order to fully reproduce J_{QCD} , we have to match it to an infinite tower of SCET operators with higher and higher power counting, but at a given power of λ , the number of operators is finite, and we only match J_{QCD} to SCET operators up to a fixed order in λ . To construct the expansion (3.1), the standard procedure is to build

¹ We will see in the next paragraph that the product of Wilson coefficients and operators in Eq. (3.1) is actually a convolution.

a gauge invariant basis of operators with a definite power counting, using the gauge invariant fields defined in Section (2.1.1). We call leading order (LO) operators, $O_i^{\rm LO}$, the operators in (3.1) with the lowest power counting, next-to-leading-order (NLO) operators, $O_i^{\rm NLO}$, the operators with the next higher power counting and so forth. Thus we can write (3.1) as

$$J_{\text{QCD}} = \sum_{i} C_{i}^{\text{LO}} O_{i}^{\text{LO}} + \sum_{i} C_{i}^{\text{NLO}} O_{i}^{\text{NLO}} + \dots$$
(3.2)

SCET is invariant under reparametrization invariance, thus we have

$$\delta_{\rm RPI} \left(\sum_{i} C_i O_i \right) = 0 , \qquad (3.3)$$

where with δ_{RPI} we indicate the set of transformations in table (2.1). We can solve the Eqs. (3.3) order by order in λ and find relations among Wilson coefficients, and because the transformations in (2.1) occur at different order, Eqs. (3.3) allow us to relate Wilson coefficients at different order. In other words, we use RPI transformations to reduce the basis of operators that we need for the matching. Because the reparametrization invariant transformations depend on the collinear direction n, if we have operators with different directions, we have a different set of transformations for each n. Thus when dealing with operators with multiple directions, solving Eqs. (3.3) becomes hard, if not prohibitive.

In this chapter we will construct RPI operators \mathbf{Q}^{i} , which are reparametrization invariant, that is $\delta_{\text{RPI}}(\mathbf{Q}^{i}) = \mathbf{Q}^{i}$. The results of the chapter were presented in Ref. [80]. The operators \mathbf{Q}^{i} are made of reparametrization invariant fermion fields Ψ_{n} , and gluon fields $\mathcal{G}^{\mu\nu}$, that we call superfields. The superfields are made gauge invariant using a reparametrization invariant Wilson line \mathcal{W}_{n} that is the generalization of the usual W_{n} . These objects do not have a definite power counting order, in particular we will know the order in the λ -expansion where they start, but they will contain terms at all higher orders as well. We build a basis with these RPI and gauge invariant objects, which is made minimal using equations of motion and kinematic constraints as discussed below in Section 3.4. Each element of this basis is assigned a Wilson coefficient, and then the elements are expanded to find the final basis with elements of a definite power counting. In this way we immediately obtain relations between Wilson coefficients of operators at different orders. Once we expand and check for redundancy, the number of independent Wilson coefficients is equal to the number of independent RPI operators in the reduced basis. We will apply RPI operators to construct the minimal basis of operator for several processes.

In hard-scattering processes, DIS provides a familiar context where the construction of a minimal operator basis requires judicial use of the quark and gluon equations of motion, and an invariance under reparametrizations of a light-like direction [41, 42, 62, 63, 86], for a review see [64]. The invariance under reparametrizations becomes more valuable at higher orders in the expansion, being particularly constraining on the basis of twist-4 operators derived in Refs. [41, 42, 62, 63]. We derive RPI constraints for collinear operators in DIS and compare to these classic results as a test of our setup. For DIS the minimization of the basis of RPI operators is quite similar to the reduction of operators in Ref. [63]. On the other hand the basis of SCET operators are comprised entirely of analogs of "good" quark and gluon fields, namely a two-component quark field χ_n and just two components of the gluon field strength, $\mathcal{B}_{n\perp}^{\mu}$. These objects both incorporate Wilson lines, and for these operators it is easier to find a minimal basis. The RPI relations provide Lorentz invariance connections between the Wilson coefficients in this basis. These constraints carry a process independence, they depend on the type of operators being considered, but not on the precise process in which they will be used. It should be emphasized that when matrix elements are considered for a particular process, a further reduction in the number of independent hadronic functions becomes possible. For twist-4 quark operators in DIS this type of further reduction was discussed in detail in Ref. [42] and for inclusive B-decays in. [95], but this type of reduction is not our focus here.

Our construction is general enough that it applies not just to DIS like processes, but to operators with multiple collinear directions, which are useful for processes with multiple hadrons and jets. These operator bases provide a starting point for deriving appropriate factorization theorems for different processes. The invariant operator procedure becomes more and more efficient as the number of directions grows.

The outline of this chapter is as follows. In subsection 3.2 we study the convolution between Wilson coefficient and operator. We divide hard interactions into two categories, those with an external hard leptonic reference vector q^{μ} , and those where the hard interaction is between strongly interacting particles. Since most SCET applications focus on the former case, we address some of the additional notational complications that occur for the latter. A set of RPI invariant collinear objects is constructed in subsection 3.3, followed by a summary of identities that can be used to reduce the operator basis in subsection 3.4. The inclusion of mass effects is considered in subsection 3.5, and the expansion of the RPI objects is carried out in subsection 3.6. Applications for constructing operators are considered in subsection 3.7. In subsection 3.7.1 we verify that our approach provides a simple way to reproduce the known RPI result for the chiral-even scalar current given in Ref. [58]. In subsection 3.7.2 we construct a general basis of field structures involving up to four active quark or gluon operators, and with up to four distinct collinear directions. In subsection 3.7.3 we consider the special case of quark operators for DIS at twist-4 with one collinear direction, and compare with the literature. In subsection 3.7.4 we derive a basis of operators for pure gluon scattering in DIS up to twist-4. Finally we apply the formalism to jet production. In subsection 3.7.5 we demonstrate that very little information is gained about the operator basis describing $e^+e^- \rightarrow 2$ jets. In subsection 3.7.6 we show that RPI turns out to be quite powerful for constraining the $e^+e^- \rightarrow 3$ jet operators. Finally we show that RPI is also useful for two jet production from gluon-fusion, $gg \rightarrow q\bar{q}$, and we construct a basis of operators for this process in subsection 3.7.7. Conclusions are given in subsection 3.8.

3.2 Convolutions

In the presence of collinear fields, a hard interaction can introduce convolutions in variables ω_i between the perturbatively calculable Wilson coefficient $C(Q^2, \omega_i)$ and the matrix element of the collinear operators. In this case the amplitude, cross-section, or decay rate has the form

$$A = \int [d\omega_1 \cdots d\omega_k] C(Q^2, \omega_i) \langle O(\omega_i) \rangle.$$
(3.4)

The convolutions occur because a component of the hard momentum and of one or more collinear momenta are $\mathcal{O}(\lambda^0)$. The exchange of momentum between the hard and collinear components yields a convolution in variables ω_i , where the number of such variables is constrained by gauge invariance and by momentum conservation in the matrix element. A gauge invariant momentum from the collinear fields can be picked out by a delta function acting on one of the collinear objects in Eq. (2.18), such as $[\delta(\omega - \bar{n} \cdot \mathcal{P}_n)\chi_n]$, and traditionally in SCET a subscript notation is used for these products,

$$\chi_{n,\omega} \equiv \left[\delta\left(\omega - \overline{\mathcal{P}}_n\right) \chi_n \right], \qquad (i\mathcal{D}_{n\perp}^{\mu})_{\omega} \equiv \left[i\mathcal{D}_{n\perp}^{\mu} \delta\left(\omega - \overline{\mathcal{P}}_n^{\dagger}\right) \right], (g\mathcal{B}_{n\perp}^{\mu})_{\omega} \equiv \left[g\mathcal{B}_{n\perp}^{\mu} \delta\left(\omega - \overline{\mathcal{P}}_n^{\dagger}\right) \right], \qquad (gn \cdot \mathcal{B}_n)_{\omega} \equiv \left[gn \cdot \mathcal{B}_n \delta\left(\omega - \overline{\mathcal{P}}_n^{\dagger}\right) \right].$$
(3.5)

We will refer to these as homogeneous objects since they have a definite order in λ , and call the operators build from these objects homogeneous operators. As an example we have the bilinear scalar operator,

$$O(\omega_1, \omega_2) = \bar{\chi}_{n,\omega_1} \,\frac{\not{n}}{2} \,\chi_{n,\omega_2} \,. \tag{3.6}$$

When we consider RPI it will be convenient to use different δ functions and convolution variables $\hat{\omega}$, that are type-III invariant. Essentially each $\overline{\mathcal{P}}_n = \overline{n} \cdot \mathcal{P}_n$ must be multiplied by a scalar transforming as n under RPI type-III. There are two cases to consider:

- i) situations where there is a reference vector q^{μ} for the hard interaction, $|q^2| = Q^2 \gg \Lambda_{\text{QCD}}$, which is external to the QCD dynamics,
- ii) situations where the hard interactions are purely from strongly interacting par-

ticles.

Case i) applies to examples such as DIS where q^{μ} is the momentum transfer from the virtual photon, or $e^+e^- \rightarrow j$ ets where q^{μ} is the four momentum of the e^+e^- pair. Here we can use $n \cdot q \sim \lambda^0$ to make the δ -function type-III invariant for *n*-collinear fields. Since $Q^2 \gg \Delta \Lambda_{\rm QCD} \gg \Lambda_{\rm QCD}^2$ we know that $n \cdot q \gg n \cdot p$, where *p* is the momentum of a collinear particle in the jet. Thus we use a variable $\hat{\omega}$ with mass dimension two, and will find δ -functions of the form²

$$\delta(\hat{\omega} - n \cdot q \,\overline{\mathcal{P}}_n) \,. \tag{3.7}$$

We also introduce a subscript notation with hatted variables,

$$\chi_{n,\hat{\omega}} \equiv \left[\delta \left(\hat{\omega} - n \cdot q \,\overline{\mathcal{P}}_n \right) \chi_n \right], \qquad (i\mathcal{D}_{n\perp}^{\mu})_{\hat{\omega}} \equiv \left[i\mathcal{D}_{n\perp}^{\mu} \,\delta \left(\hat{\omega} - n \cdot q \,\overline{\mathcal{P}}_n^{\dagger} \right) \right], \\ (g\mathcal{B}_{n\perp}^{\mu})_{\hat{\omega}} \equiv \left[g\mathcal{B}_{n\perp}^{\mu} \,\delta \left(\hat{\omega} - n \cdot q \,\overline{\mathcal{P}}_n^{\dagger} \right) \right], \qquad (gn \cdot \mathcal{B}_n)_{\hat{\omega}} \equiv \left[gn \cdot \mathcal{B}_n \,\delta \left(\hat{\omega} - n \cdot q \,\overline{\mathcal{P}}_n^{\dagger} \right) \right]. \tag{3.8}$$

Since $\delta(\hat{\omega} - n \cdot q\overline{\mathcal{P}}_n) \sim \lambda^0$, it is leading order in the power counting. Furthermore, we have $\delta(\hat{\omega} - n \cdot q\overline{\mathcal{P}}) = \delta(\hat{\omega}/n \cdot q - \overline{\mathcal{P}})/n \cdot q$, so identifying $\hat{\omega} = n \cdot q \omega$ there is no real change to the structure of Eq. (3.4). An operator built out of the components given in Eq. (3.8) has multiple labels, $O(\hat{\omega}_1, \hat{\omega}_2, \ldots)$, and the Wilson coefficient for the operator will be a function of the same parameters, $C(\hat{\omega}_1, \hat{\omega}_2, \ldots)$, yielding Eq. (3.4) with $\hat{\omega}$'s replacing ω 's.

For processes in case ii) there is no analog of the external q^{μ} . Examples here include $pp \rightarrow \text{jets}$, or any other hard process that does not involve external leptons or photons. The key difference with case i) is that here the hard interaction must involve two or more collinear directions, so we are guaranteed that there are scalar products $n_i \cdot n_j \sim \lambda^0$. For this type of reaction the type-III invariant δ -functions which are convoluted with Wilson coefficients always involve large momenta for two different

² For *B*-decays these type-III invariant δ -functions were used in Ref. [85], with $q^{\mu} \simeq m_b v^{\mu}$, $\delta(\hat{\omega} - n \cdot q \,\overline{\mathcal{P}}_n) = \delta(\hat{\omega} - m_b n \cdot v \,\overline{\mathcal{P}}_n) = 1/m_b \,\delta(\hat{\omega}' - n \cdot v \,\overline{\mathcal{P}}_n)$, where $\hat{\omega} = m_b \hat{\omega}'$. This form of invariant δ -function was also quite useful for analyzing the factorization theorem for $e^+e^- \to J/\psi X$ in Ref. [49].

collinear directions,

$$\Delta_{ij} \equiv \delta \left(\hat{\omega}_{ij} - \frac{1}{2} n_i \cdot n_j \,\overline{\mathcal{P}}_{n_i} \overline{\mathcal{P}}_{n_j} \right). \tag{3.9}$$

Here $\overline{\mathcal{P}}_{n_i}$ acts on a gauge invariant block of n_i -collinear fields, and $\overline{\mathcal{P}}_{n_j}$ acts on a block of n_j -collinear fields. Since this δ -operator does not act on a single block of collinear fields we will not use a subscript notation like Eq. (3.8) for $\hat{\omega}_{ij}$. In this case the structure of the factorization theorem between operators and Wilson coefficients is a bit different than in Eq. (3.4). For example, consider an operator with collinear objects for four directions, where the convolution is

$$\int \left[\prod_{ij} d\hat{\omega}_{ij}\right] C(\hat{\omega}_{ij}) \left[\prod_{km} \Delta_{km}\right] \bar{\chi}_{n_1}(g\mathcal{B}_{n_3}^{\perp}) (g\mathcal{B}_{n_4}^{\perp}) \chi_{n_2}.$$
(3.10)

Here the products are over the six unique pairs ij with $i \neq j$, and $\overline{\mathcal{P}}_{n_i}$ in the Δ_{km} acts on the n_i -collinear field(s). The convolutions in Eq. (3.10) can be manipulated into the form of Eq. (3.4) by inserting four factors of $1 = \int d\omega_i \, \delta(\omega_i - \overline{\mathcal{P}}_{n_i})$, writing $\hat{\delta}_{ij} = \delta(\hat{\omega}_{ij} - n_i \cdot n_j \, \omega_i \omega_j/2)$ and carrying out the integrals over the six $\hat{\omega}_{ij}$'s to give

$$\int \left[d\omega_1 \cdots d\omega_4 \right] C \left(n_i \cdot n_j \, \omega_i \omega_j \right) \bar{\chi}_{n_1,\omega_1} (g \mathcal{B}_{n_3,\omega_3}^{\perp}) (g \mathcal{B}_{n_4,\omega_4}^{\perp}) \chi_{n_2,\omega_2} \,. \tag{3.11}$$

Here the RPI-III transformation of the measure cancels against that of the δ -functions in the operator, and RPI has constrained the Wilson coefficients to only depend on invariant products $n_1 \cdot n_2 \omega_1 \omega_2$, $n_1 \cdot n_3 \omega_1 \omega_3$, etc.

Due to the simplicity of the soft-collinear coupling at leading order in SCET a further factorization of the EFT matrix element can be made into collinear pieces J, and soft pieces S at each order in the power counting:

$$\langle O(\omega_i) \rangle = \int dk_j J(\omega_i, k_j) S(k_j).$$
 (3.12)

However it is the factorization in Eq. (3.4) that will be central to our discussion of reparametrization invariant operators.

3.3 Construction of RPI and Gauge Invariant objects

We now construct reparametrization invariant objects in SCET whose leading terms give the fields in Eq. (2.17). These are then generalized to objects that are simultaneously RPI and gauge invariant whose leading terms give the objects in Eqs. (2.18,3.8). For simplicity only collinear objects are considered in this section. Pulling out the large phases from the collinear quark field and gluon field strength, and decomposing the full theory field into independent collinear sectors we have at tree level,

$$\psi(x) = \sum_{n} e^{-ix \cdot \mathcal{P}_n} \psi_n(x), \qquad G^{\mu\nu}(x) = \sum_{n} e^{-ix \cdot \mathcal{P}_n} G_n^{\mu\nu}(x). \tag{3.13}$$

Full Lorentz invariance act on the fields $\psi(x)$ and $G^{\mu\nu}(x)$, but the RPI transformations that we are interested acts independently on each collinear sector labeled by n. Two sectors i, j are independent if $n_i \cdot n_j \gg \lambda^2$, and the sums in Eq. (3.13) are really over equivalence classes, $\{n\}$, where a class consists of vectors related by RPI. From the discussion in section 2.2 the *n*-reparametrization invariant collinear quark and field strength are easy to identify

$$\psi_n \equiv \left(1 + \frac{1}{\bar{n} \cdot D_n} \not\!\!\!D_n^{\perp} \frac{\not\!\!\!n}{2}\right) \xi_n, \qquad ig \, G_{\mu\nu}^n \equiv \left[i D_{\mu}^n, i D_{\nu}^n\right]. \qquad (3.14)$$

Under the transformations in Table 2.1 for $\{n, \bar{n}\}$, the quark field ψ_n remains invariant [77], while the gluon tensor is invariant because the vector D_n^{μ} is invariant. To make the fields in Eq. (3.14) invariant under the additional reparametrization transformations that link collinear and soft derivatives we replace $in \cdot D_n \to in \cdot D_n + gn \cdot A_s$, $iD_{n\perp}^{\mu} \to iD_{n\perp}^{\mu} + W_n i D_{s\perp}^{\mu} W_n^{\dagger}$, and $i\bar{n} \cdot D_n \to i\bar{n} \cdot D_n + W_n i\bar{n} \cdot D_s W_n^{\dagger}$. After this replacement the decoupling field redefinitions in Eq. (2.14) can be made. In Eq. (3.14) $\#\xi_n = 0$, and the term in ψ_n with a \perp -covariant derivative corresponds to the two components of the full fermion field that are small when $p_{\perp}/\bar{n} \cdot p \ll 1$. Since $\#\psi_n \neq 0$, the ψ_n field does not provide a definite power counting for operators. For example, $\bar{\psi}_n \vec{\eta} \psi_n \sim \lambda^0$ whereas $\bar{\psi}_n \eta \psi_n \sim \lambda^4$.

We also need reparametrization invariant δ -functions whose expansions reproduce Eqs. (3.7) and (3.9) at lowest order. For example, these are needed to construct an RPI operator which when expanded gives $\bar{\chi}_{n,\omega_1} \vec{\pi} \chi_{n,\omega_2}$ at lowest order. For situations where there is an external hard vector q^{μ} the invariant δ -function is

$$\hat{\Delta}_i \equiv \delta(\hat{\omega} - 2q \cdot i\partial_{n_i}) = \delta(\hat{\omega} - n_i \cdot q \,\overline{\mathcal{P}}_{n_i}) + \dots , \qquad (3.15)$$

where as described in section 2.1.1, q^{μ} is a parameter specific to the kinematics of the process being studied. Notice that $\delta(\hat{\omega} - 2q \cdot i\partial_n)$ starts at $\mathcal{O}(\lambda^0)$, is RPI, and is gauge invariant when acting on singlet operators. Here

$$i\partial_n^{\mu} \equiv \frac{n^{\mu}}{2}\overline{\mathcal{P}}_n + \mathcal{P}_{n\perp}^{\mu} + \frac{\bar{n}^{\mu}}{2}in\cdot\partial_n\,,\qquad(3.16)$$

and functions of $i\partial_n^{\mu} \sim (\lambda^2, 1, \lambda)$ can be expanded in powers of λ . Note that $\overline{\mathcal{P}}_n$ and $\mathcal{P}_{n\perp}^{\mu}$ are only non-zero when they act on *n*-collinear fields. It is useful to extend this property to the full $i\partial_n^{\mu}$, which we can do by distributing an $i\partial^{\mu}$ derivative across all fields that it acts on, writing for example $i\partial^{\mu}\bar{\psi}_{n_1}\psi_{n_2} = (i\partial_{n_1}^{\mu}\bar{\psi}_{n_1})\psi_{n_2} + \bar{\psi}_{n_1}(i\partial_{n_2}^{\mu}\psi_{n_2})$. In some hard processes there is more than one external hard vector, and a natural question arises as to whether q^{μ} provides a unique choice for this construction. For example, in DVCS, $\gamma^* p \to \gamma^{(*)} p'$ we have the momentum q^{μ} of the incoming γ^* and the momentum q'^{μ} of the outgoing $\gamma^{(*)}$. In appendix A we show that as long as $q_{\perp} - q'_{\perp} \sim \lambda$ or smaller, the choice q suffices, since for the purpose of constructing a basis of operators it is equivalent to the choice of any linear combination of q and q'. On the other hand, for situations where there is no external hard vector q^{μ} , the appropriate RPI δ -function is

$$\hat{\Delta}_{ij} \equiv \delta(\hat{\omega}_{ij} - 2i\partial_{n_i} \cdot i\partial_{n_j}) = \delta\left(\hat{\omega}_{ij} - \frac{1}{2}n_i \cdot n_j \,\overline{\mathcal{P}}_{n_i}\overline{\mathcal{P}}_{n_j}\right) + \dots \qquad (3.17)$$

This δ -function operator acts on two independent collinear directions. In general we must include in an operator a set of $\hat{\Delta}_i$ and $\hat{\Delta}_{ij}$ which are linearly independent. Once

we expand, the first term in the series for $\hat{\Delta}_{ij}$ is not independent of the first term from $\hat{\Delta}_i$, so the δ -function shown on the RHS of Eq. (3.17) can always be eliminated, as we did in Eq. (3.11).

We will also make use of a reparametrization invariant Wilson line, \mathcal{W}_n , which has the same gauge transformation properties as W_n ,

$$\mathcal{W}_n = W_n \, e^{-iR_n} \,. \tag{3.18}$$

Here the operator R_n starts with a term at $\mathcal{O}(\lambda)$ and is built of *n*-collinear gluon fields,

$$R_n = R_n \left[\overline{\mathcal{P}}_n \,, \mathcal{P}_{n\perp}^{\mu} \,, g \mathcal{B}_{n\perp}^{\mu} \,, t^{\mu} \right], \qquad (3.19)$$

where the vector t^{μ} is either q^{μ} or $i\partial_{n'}^{\mu}$ with $n \cdot n' \sim \lambda^0$. Furthermore, R_n is Hermitian, dimensionless, and collinear gauge invariant. We leave the explicit construction of R_n to section 3.6 below, and for the remainder of this section take these properties as given.

Under collinear gauge transformations, ψ_n and \mathcal{W}_n transform the same way as ξ_n and W_n , and $G_n^{\mu\nu}$ transforms as a nonabelian field strength. Thus using \mathcal{W}_n we can form analogs of the results in Eq. (3.8) that are simultaneously RPI and gauge invariant, namely the superfields

$$\Psi_n \equiv \mathcal{W}_n^{\dagger} \psi_n \,, \qquad \mathcal{G}_n^{\mu\nu} \equiv \mathcal{W}_n^{\dagger} \mathcal{G}_n^{\mu\nu} \mathcal{W}_n \,. \tag{3.20}$$

For cases with an external q^{μ} we also introduce a subscript notation,

$$\Psi_{n,\hat{\omega}} \equiv \left[\delta(\hat{\omega} - 2q \cdot i\partial_n)\Psi_n\right], \qquad \mathcal{G}_{n,\hat{\omega}}^{\mu\nu} \equiv \left[\mathcal{G}_n^{\mu\nu}\,\delta(\hat{\omega} + 2q \cdot i\,\overleftarrow{\partial_n})\right]. \tag{3.21}$$

Operators built out of the superfields Ψ_n and $\mathcal{G}_n^{\mu\nu}$ are simultaneously RPI and gauge invariant. They are not homogeneous in the power counting, but the superfields reduce to the objects in Eq. (3.8) at lowest order in the λ expansion. For example, the superfield for the fermion

$$\Psi_n = e^{iR_n} W_n^{\dagger} \left(1 + \frac{1}{i\bar{n} \cdot D_n} i D_n^{\perp} \frac{\vec{n}}{2} \right) \xi_n = e^{iR_n[i\partial_n, \mathcal{B}_n]} \left(1 + \frac{1}{\overline{\mathcal{P}}_n} i \mathcal{D}_n^{\perp} \frac{\vec{n}}{2} \right) \chi_n$$

$$= \chi_n + \dots$$
(3.22)

Similarly, $(g_{\nu\nu'}^{\perp}\bar{n}_{\mu})ig(\mathcal{G}_{n}^{\mu\nu'}) = \overline{\mathcal{P}}_{n} g\mathcal{B}_{n\nu}^{\perp} + \dots$ Thus to form a RPI version of the bilinear fermion operator $O(\omega_{1}, \omega_{2})$ in Eq. (3.6) we simply take

$$\mathbf{Q}(\hat{\omega}_1, \hat{\omega}_2) = \bar{\Psi}_{n, \hat{\omega}_1} \not \!\!\! / \Psi_{n, \hat{\omega}_2} , \qquad (3.23)$$

and note that expanding in λ gives $\mathbf{Q}(\hat{\omega}_1, \hat{\omega}_2) = (n \cdot q)^{-1} O(\omega_1, \omega_2) + \dots$

We will also need the equations of motion for the RPI quark and gauge superfields in Eq. (3.20). The *n*-collinear Lagrangian for the quark field is [10]

$$\mathcal{L}_{qn} = \bar{\xi}_n \left(in \cdot D_n + i \not\!\!\!D_n^{\perp} \frac{1}{i\bar{n} \cdot D_n} i \not\!\!\!D_n^{\perp} \right) \frac{\not\!\!/}{2} \xi_n , \qquad (3.24)$$

We can write Eq. (3.24) in terms of ψ_n as a simple Dirac Lagrangian

$$\mathcal{L}_{qn} = \bar{\psi}_n \, i D_n \, \psi_n \,, \tag{3.25}$$

The equation of motion for ψ_n is a simple Dirac equation $\mathcal{D}_n \psi_n = 0$. Using $\mathcal{W}_n \mathcal{W}_n^{\dagger} = 1$, we can write $\mathcal{W}_n^{\dagger} i \mathcal{D}_n \mathcal{W}_n \mathcal{W}_n^{\dagger} \psi_n = 0$, and thus obtain the equation of motion for Ψ_n

$$\hat{\mathcal{P}}_n \Psi_n = 0. \tag{3.26}$$

Here $\hat{\mathcal{D}}^{\mu}_{n}$ is the RPI and gauge invariant derivative

$$\hat{\mathcal{D}}_n^{\mu} \equiv \mathcal{W}_n^{\dagger} D_n^{\mu} \, \mathcal{W}_n = e^{iR_n} \, \mathcal{D}_n^{\mu} \, e^{-iR_n} \,. \tag{3.27}$$

For the gluon field we have the equation of motion $[iD^n_{\nu}, G^{\mu\nu}_n] = igT^A \sum_f \bar{\psi}^f_n T^A \gamma^{\mu} \psi^f_n$

and for the superfield

$$i\hat{\mathcal{D}}^n_{\nu}\mathcal{G}^{\nu\mu}_n = [i\hat{\mathcal{D}}^n_{\nu}, \mathcal{G}^{\nu\mu}_n] = -ig\,T^A\bar{\Psi}^f_nT^A\gamma^{\mu}\Psi^f_n.$$
(3.28)

Note that $ig\mathcal{G}_n^{\mu\nu} = [i\hat{\mathcal{D}}_n^{\mu}, i\hat{\mathcal{D}}_n^{\nu}].$

3.4 Reducing the Operator Basis

In general there are three steps that one can consider to reduce the perturbative and nonperturbative information in the EFT to its minimal form:

- a) Find a minimal basis of homogeneous operators and of RPI operators that suffice at the desired order in λ . The homogeneous operators can be written entirely in terms of χ_n , $\mathcal{B}^{\mu}_{n\perp}$, and $\mathcal{P}^{\mu}_{\perp}$.
- b) Compare the homogeneous and RPI basis to determine which perturbative Wilson coefficients are fixed by RPI.
- c) Consider the decomposition of matrix elements of operators in the homogeneous basis, and derive further relations between the resulting non-perturbative functions.

Generically the relation between the operator basis looks like

$$\sum_{n_i} \sum_{\ell} \int [\prod_j d\hat{\omega}_j] \, \hat{C}_{\ell}(\hat{\omega}_j) \, [\mathbf{Q}_{\ell}(\hat{\omega}_j)] = \sum_{n_i} \sum_{\ell} \int [\prod_j d\omega_j] \, C_{\ell}(\omega_j) \, [O_{\ell}(\omega_j)] + \dots \,, \quad (3.29)$$

where $\mathbf{Q}_{\ell}(\hat{\omega}_j)$ are RPI operators and $O_{\ell}(\omega_j)$ are homogeneous operators, and the ellipse denotes higher order terms in the power expansion. In general our focus in this article is to carry out b) which is still largely process independent. For the most part we give no discussion of item c), which obviously must be considered process by process. In order to consider b) we must first determine a) which is the focus of this Section. We will discuss the equations of motion and other relations that allow a reduction in the basis of operators at each order in λ . First we consider the gauge invariant objects with homogeneous power counting. We would like to demonstrate that all operators can be reduced to a form that only involves the basic building blocks χ_n , $g\mathcal{B}^{\mu}_{n\perp}$, and $\mathcal{P}^{\mu}_{\perp}$. All other homogeneous objects can be reduced to these. For example, one might think that the objects $g\mathcal{B}^{\mu\nu}_{\perp\perp} \equiv [1/\overline{\mathcal{P}} W^{\dagger}[iD^{\mu}_{n\perp}, iD^{\nu}_{n\perp}]W]$ and $g\mathcal{B}^{\mu}_{\perp 2} \equiv [1/\overline{\mathcal{P}} W^{\dagger}[iD^{\mu}_{n\perp}, in \cdot D_n]W]$ are independent. However they are related to the building blocks by

$$g\mathcal{B}_{\perp\perp}^{\mu\nu} = \frac{1}{\overline{\mathcal{P}}}\mathcal{P}_{\perp}^{\mu}(g\mathcal{B}_{\perp}^{\nu}) - \frac{1}{\overline{\mathcal{P}}}\mathcal{P}_{\perp}^{\nu}(g\mathcal{B}_{\perp}^{\mu}) + \frac{1}{\overline{\mathcal{P}}}\left[g\mathcal{B}_{\perp}^{\mu}, g\mathcal{B}_{\perp}^{\nu}\right], \qquad (3.30)$$
$$g\mathcal{B}_{\perp2}^{\mu} = \frac{1}{\overline{\mathcal{P}}}\mathcal{P}_{\perp}^{\mu}(gn\cdot\mathcal{B}) - \frac{1}{\overline{\mathcal{P}}}in\cdot\partial_{n}(g\mathcal{B}_{\perp}^{\mu}) + \frac{1}{\overline{\mathcal{P}}}\left[g\mathcal{B}_{\perp}^{\mu}, gn\cdot\mathcal{B}\right],$$

where we will see below that $n \cdot \mathcal{B}$ and $in \cdot \partial_n \mathcal{B}^{\mu}_{\perp}$ can also be reduced using the gluon equation of motion. For χ_n the equation of motion is

$$in \cdot \partial_n \chi_n = -(gn \cdot \mathcal{B}_n)\chi_n - i\mathcal{P}_n^{\perp} \frac{1}{\overline{\mathcal{P}}_n} i\mathcal{P}_n^{\perp} \chi_n , \qquad (3.31)$$

which allows us to eliminate $in\partial_n$ derivatives on χ_n . To obtain the equations of motion for the gluon objects we consider $-g^2T^A\sum_f \bar{\psi}_n^f W_n T^A W_n^{\dagger} \gamma^{\mu} \psi_n^f = [i\mathcal{D}_{\nu}^n, [i\mathcal{D}_n^{\mu}, i\mathcal{D}_n^{\nu}]].$ Expanding in λ and multiplying on the right with $\delta(\omega - \overline{\mathcal{P}}_n^{\dagger})$ gives three equations

$$\begin{split} &\omega\left(gn\cdot\mathcal{B}\right)_{\omega}=2\mathcal{P}_{\nu}^{\perp}(g\mathcal{B}_{\perp}^{\nu})_{\omega}+\frac{2\omega'}{\omega}\left[\left(g\mathcal{B}_{\perp}^{\nu}\right)_{\omega-\omega'},\left(g\mathcal{B}_{\nu}^{\perp}\right)_{\omega'}\right]-\frac{2}{\omega}\,g^{2}T^{A}\sum_{f}\left[\bar{\chi}_{n}^{f}\,T^{A}\bar{\eta}\chi_{n}^{f}\right]_{\omega},\\ &\omega\left[in\cdot\partial_{n}g\mathcal{B}_{\perp}^{\mu}\right]_{\omega}=-\left[\mathcal{P}_{\nu}^{\perp}\left[g\mathcal{B}_{\perp}^{\mu},g\mathcal{B}_{\perp}^{\nu}\right]\right]_{\omega}-\left[g\mathcal{B}_{\perp}^{\perp},\left[\mathcal{P}_{\perp}^{[\mu}g\mathcal{B}_{\perp}^{\nu]}\right]\right]_{\omega}-\left[g\mathcal{B}_{\nu}^{\perp},\left[g\mathcal{B}_{\perp}^{\mu},g\mathcal{B}_{\perp}^{\nu}\right]\right]_{\omega}\\ &+\frac{\omega}{2}\left[\mathcal{P}_{\perp}^{\mu}gn\cdot\mathcal{B}\right]_{\omega}-\left[\mathcal{P}_{\nu}^{\perp}\mathcal{P}_{\perp}^{[\mu}g\mathcal{B}_{\perp}^{\nu]}\right]_{\omega}+\frac{\omega}{2}\left[g\mathcal{B}_{\perp}^{\mu},gn\cdot\mathcal{B}\right]_{\omega}-\frac{\omega'}{2}\left[\left(gn\cdot\mathcal{B}\right)_{\omega-\omega'},\left(g\mathcal{B}_{\perp}^{\mu}\right)_{\omega'}\right]\\ &-g^{2}T^{A}\sum_{f}\left[\bar{\chi}_{n}^{f}T^{A}\gamma_{\perp}^{\mu}\frac{1}{\overline{\mathcal{P}}}(\mathcal{P}_{\perp}+g\mathcal{B}_{\perp})\frac{\vec{\eta}}{2}\chi_{n}^{f}\right]_{\omega}g^{2}T^{A}\sum_{f}\left[\bar{\chi}_{n}^{f}\frac{\vec{\eta}}{2}(\mathcal{P}_{\perp}^{\dagger}+g\mathcal{B}_{\perp})\frac{1}{\overline{\mathcal{P}}^{\dagger}}T^{A}\gamma_{\perp}^{\mu}\chi_{n}^{f}\right]_{\omega},\\ &g^{2}T^{A}\sum_{f}\left[\bar{\chi}_{n}^{f}(\mathcal{P}_{\perp}^{\dagger}+g\mathcal{B}_{\perp})\frac{1}{\overline{\mathcal{P}}^{\dagger}}T^{A}\frac{1}{\overline{\mathcal{P}}}(\mathcal{P}_{\perp}+g\mathcal{B}_{\perp})\vec{\eta}\chi_{n}^{f}\right]_{\omega}\\ &=\frac{\omega}{2}\left[in\cdot\partial_{n}gn\cdot\mathcal{B}\right]_{\omega}-\left[\left(\mathcal{P}_{\perp}\right)^{2}gn\cdot\mathcal{B}\right]_{\omega}-\left[\mathcal{P}_{\nu}^{\perp}\left[g\mathcal{B}_{\perp}^{\nu},gn\cdot\mathcal{B}\right]\right]_{\omega}-\left[g\mathcal{B}_{\perp\nu},\left[\mathcal{P}_{\perp}^{\nu}gn\cdot\mathcal{B}\right]\right]_{\omega}\\ &+\left[g\mathcal{B}_{\perp\nu},\left[in\cdot\partial_{n}g\mathcal{B}_{\perp}^{\nu}\right]\right]_{\omega}-\left[g\mathcal{B}_{\perp\nu},\left[g\mathcal{B}_{\perp}^{\nu},gn\cdot\mathcal{B}\right]\right]_{\omega}+\left[in\cdot\partial_{n}\mathcal{P}_{\nu}^{\perp}g\mathcal{B}_{\perp}^{\nu}\right]_{\omega}\end{aligned}$$

$$+\frac{\omega'}{2}\left[(gn\cdot\mathcal{B})_{\omega-\omega'},(gn\cdot\mathcal{B})_{\omega'}\right].$$
(3.32)

Here we sum over the color A, over the flavors f, and integrate over the repeated index ω' . In our analysis the first two equations will be used to eliminate $gn \cdot \mathcal{B}_n$ and $in \cdot \partial_n g \mathcal{B}^{\mu}_{\perp}$ respectively. The last relation only becomes relevant at higher orders than those we consider here. The above relations imply that when building a homogeneous basis of operators we do not need to consider the objects

$$in \cdot \partial_n \chi_n$$
, $n \cdot \mathcal{B}_n$, $in \cdot \partial_n \mathcal{B}_{n\perp}^{\mu}$, $\mathcal{B}_{\perp\perp}^{\mu\nu}$, $\mathcal{B}_{\perp\perp}^{\mu}$. (3.33)

Next we derive relations that can be used to reduce RPI operators to a minimal form. Given the definition in Eq. (3.27), we can write $i\hat{D}_n^{\mu} = i\partial_n^{\mu} + [i\hat{D}_n^{\mu}]$, and it is straightforward using Eq. (3.50) below to prove that

$$[q \cdot i\partial_n i\hat{\mathcal{D}}_n^{\nu}] = q_{\mu} ig\mathcal{G}_n^{\mu\nu}, \qquad (3.34)$$

and hence that $q_{\mu}[\hat{\mathcal{D}}_{n}^{\mu}] = 0$. (The results here and below apply equally well for t = qand $t = i\partial_{n'}$ with $n \cdot n' \sim \lambda^{0}$. For simplicity we use the notation with t = q.) Eq. (3.34) can be used to rewrite the quark superfields equation of motion in Eq. (3.26) as

$$i\partial_n \Psi_n = -\left[\frac{1}{q \cdot i\partial_n} q_\mu \gamma_\nu i g \mathcal{G}_n^{\mu\nu}\right] \Psi_n \,, \qquad (3.35)$$

Since $q \cdot i\partial_n \delta(\hat{\omega} - 2q \cdot i\partial_n) = \frac{1}{2}\hat{\omega} \delta(\hat{\omega} - 2q \cdot i\partial_n)$ we also have the result

$$q \cdot i\partial_n \Psi_{n,\hat{\omega}} = \frac{\hat{\omega}}{2} \Psi_{n,\hat{\omega}} .$$
(3.36)

In a similar way, $q \cdot i\partial_n \mathcal{G}_{n,\hat{\omega}}^{\mu\nu} = (-\hat{\omega}/2)\mathcal{G}_{n,\hat{\omega}}^{\mu\nu}$. The collinear gluon equation of motion for $\mathcal{G}_n^{\mu\nu}$ in Eq. (3.28) can be rewritten as

$$[i\partial_{\nu}^{n}\mathcal{G}_{n}^{\nu\mu}] = -ig\,T^{A}\bar{\Psi}_{n}^{f}T^{A}\gamma^{\mu}\Psi_{n}^{f} + \left[\left[\frac{1}{q\cdot i\partial_{n}}q_{\alpha}\mathcal{G}_{n\nu}^{\alpha}\right],\,ig\mathcal{G}_{n}^{\mu\nu}\right].$$
(3.37)

The quark and gluon operators will have $\hat{\omega}$ subscripts, $\Psi_{n,\hat{\omega}}$ and $\mathcal{G}_{n,\hat{\omega}}$, so only the equations of motion in Eqs. (3.35,3.37) should be used to remove derivatives since the $i\partial_n$ derivatives commute with the presence of the δ -function denoted by the $\hat{\omega}$ subscript. The QCD Bianchi identity, $D_{\mu}G_{\nu\sigma} + D_{\nu}G_{\sigma\mu} + D_{\sigma}G_{\mu\nu} = 0$, also gives a relation for $\mathcal{G}_n^{\mu\nu}$, namely $\hat{\mathcal{D}}_n^{\mu}\mathcal{G}_n^{\nu\sigma} + \hat{\mathcal{D}}_n^{\nu}\mathcal{G}_n^{\sigma\mu} + \hat{\mathcal{D}}_n^{\sigma}\mathcal{G}_n^{\mu\nu} = 0$. Rearranging it gives the following relation

$$i\partial_{n}^{\alpha}\mathcal{G}_{n}^{\mu\nu} = q_{\beta}\left\{\left[\left[\frac{ig}{q\cdot i\partial_{n}}\mathcal{G}_{n}^{\alpha\beta}\right],\mathcal{G}_{n}^{\mu\nu}\right] - \left[\left[\frac{ig}{q\cdot i\partial_{n}}\mathcal{G}_{n}^{\beta\mu}\right],\mathcal{G}_{n}^{\beta\mu}\right] + \left[\left[\frac{ig}{q\cdot i\partial_{n}}\mathcal{G}_{n}^{\beta\nu}\right],\mathcal{G}_{n}^{\mu\alpha}\right]\right\} - i\partial_{n}^{[\mu}\mathcal{G}_{n}^{\nu]\alpha}$$

$$(3.38)$$

which implies that $i\partial_n^{\alpha} \mathcal{G}_n^{\mu\nu}$, $i\partial_n^{\mu} \mathcal{G}_n^{\nu\alpha}$, and $i\partial_n^{\nu} \mathcal{G}_n^{\alpha\mu}$ are not all independent. Closing Eq. (3.38) with γ^{μ} allows us to remove $i\partial_n \mathcal{G}_n^{\mu\nu}$, which is how we will choose to use this identity in quark operators. An analog of the Bianchi identity does not occur for the building block $g\mathcal{B}_n^{\mu}$ in homogeneous operators; it easy to verify that when expanded in λ , Eq. (3.38) is trivially satisfied. Eqs. (3.35–3.38) are the RPI equivalent of the results in Eqs. (3.31,3.32), and can be used to reduce the RPI operator basis.

The above results imply that when building an RPI operator basis we do not need to consider the objects

$$i\partial_n \Psi_n$$
, $q \cdot i\partial_n \Psi_{n,\hat{\omega}}$, $[i\partial_\nu^n \mathcal{G}_n^{\nu\mu}]$, $i\partial_n \mathcal{G}_n^{\mu\nu}$, $[q \cdot i\partial_n \mathcal{G}_{n,\hat{\omega}}^{\nu\mu}]$. (3.39)

This list is not exhaustive. By manipulating operators in specific situations further structures can be eliminated using a combination of the above identities. For example, for in Sections 3.7.3 and 3.7.4 below we will see that $q_{\mu}\mathcal{G}_{n}^{\mu\nu}i\partial_{\nu}^{n}$, with the $i\partial_{\nu}^{n}$ acts on a *n*-collinear quark or gluon field, can be eliminated.

In principle one can just count the number of RPI operators and compare to the number of operators in a homogeneous operator basis with definite power counting to determine whether there are any RPI constraints on the Wilson coefficients. The key issue here is that of linear independence, even if one has the the same number of operators in the RPI and homogeneous basis, it could be that two RPI operators constrain the same linear combination of operators in the homogeneous basis.

3.5 Extension to Massive Collinear Fields

Massive collinear quarks in SCET were first studied in Refs. [74, 89]. After the field redefinition in Eq. (2.14) they have the LO Lagrangian

$$\mathcal{L}_{qn} = \bar{\xi}_n \left[in \cdot D_n + (i \not\!\!D_n^{\perp} - m) \frac{1}{i\bar{n} \cdot D_n} (i \not\!\!D_n^{\perp} + m) \right] \frac{\not\!\!n}{2} \xi_n \,. \tag{3.40}$$

The appropriate RPI transformations with massive quarks were determined in Ref. [34]. The only change is in the type-II transformation of the fermion field, where one has to add a mass dependent term:

$$\xi_n \xrightarrow{\mathrm{II}} \left[1 + \frac{\not{\xi}^{\perp}}{2} \frac{1}{\bar{n} \cdot iD_n} (i \not{D}_{n\perp} - m) \right] \xi_n.$$
(3.41)

Under this transformation the Lagrangian in Eq. (2.14) falls into two invariant parts, one fixed by the leading order kinetic term and one whose coefficient encodes the choice of mass scheme. Note that the RPI transformation itself is not modified by the presence of a mass term, the transformation of \bar{n} is still exactly as in Eq. (2.24).

We can now build an analog of the RPI superfield for a massive collinear quark. The reparametrization invariant quark field is

$$\psi_n = \left(1 + \frac{1}{\bar{n} \cdot iD_n} (i \not\!\!D_n^\perp + m) \frac{\not\!\!n}{2}\right) \xi_n.$$
(3.42)

This leads to the modified RPI superfield for a massive collinear quark

$$\Psi_n = e^{iR_n} \left[1 + \frac{1}{\overline{\mathcal{P}}_n} (i\mathcal{D}_n^{\perp} + m) \frac{\tilde{n}}{2} \right] \chi_n \,. \tag{3.43}$$

This result is included for completeness. Our focus in the remainder of the chapter will be on massless collinear quark fields.

3.6 Determination of R_n and Expansion of Ψ_n and $\mathcal{G}_n^{\mu\nu}$

In this Section we derive an expression for R_n appearing in the RPI Wilson line, and then expand the invariant objects Ψ_n , $\mathcal{G}_n^{\mu\nu}$, $\delta(\hat{\omega} - 2q \cdot i\partial_n)$, and $\delta(\hat{\omega}_{12} - 2i\partial_{n_1} \cdot i\partial_{n_2})$. We can define the collinear Wilson line W_n by the equation:

$$[(\bar{n} \cdot D_n)W_n] = 0. (3.44)$$

We define the RPI \mathcal{W}_n generalizing (3.44) to a covariant derivative D_n along a (non light-like) direction t as:

$$[(t \cdot D_n)\mathcal{W}_n] = 0, \qquad (3.45)$$

where t is such that $n \cdot t \sim \lambda^0$. This implies the momentum space representation:

$$\mathcal{W}_n = \left[\sum_{\text{perms}} \exp\left(\frac{-g}{(t \cdot i\partial_n)} t \cdot A_n\right) \right] \,. \tag{3.46}$$

We would like to find R_n such that $\mathcal{W}_n = W_n e^{-iR_n}$. Thus e^{-iR_n} is the operator that rotates W_n from the light-like direction n to the direction t. \mathcal{W}_n is reparametrization invariant to the choice of the basis vector n, which labels the *n*-collinear fields A_n^{μ} , since such reparametrizations cannot change the fact that $n \cdot t \sim \lambda^0$. Recall that the subscript n on \mathcal{W}_n labels the equivalence class $\{n\}$ of vectors that are related by type-I and type-III RPI transformations. For any t such that $n \cdot t \sim \lambda^0$ we have

$$\frac{1}{t \cdot i\partial_n} t \cdot A_n = \frac{1}{\overline{\mathcal{P}}_n} \, \bar{n} \cdot A_n + \dots \,, \tag{3.47}$$

and thus

$$\mathcal{W}_n = W_n + \dots , \qquad (3.48)$$

where the ellipses represent power suppressed terms. In Eq. (3.47) the $n \cdot t$'s in the numerator and denominator cancel out in the leading term, leaving a t independent result.

For situations where we have an external hard vector q^{μ} , we can simply take $t^{\mu} = q^{\mu}$ and use the corresponding \mathcal{W}_n as the RPI invariant Wilson line.

For situations where there is no external q^{μ} , the choice for t^{μ} in \mathcal{W}_n is less obvious since the only available RPI vectors are operators themselves, $i\partial_{n'}^{\mu}$, where n' is a distinct collinear direction from n. In this situation, any choice $t^{\mu} = i\partial_{n'}^{\mu}$ satisfying $n \cdot t = n \cdot n' \overline{\mathcal{P}}_{n'} + \ldots \sim \lambda^0 + \ldots$ is equally good, and the existence of the hard interaction guarantees that such an n' exists. In this case \mathcal{W}_n still yields W_n at lowest order, and hence only behaves like an operator in the n' direction through terms in the power corrections, namely the ellipsis in Eq. (3.48). In these ellipse terms the $i\partial_{n'}$'s appear linearly order by order. Since the derivative $i\partial_{n'}$ does not act on n-collinear fields it behaves just like an external vector q as far as manipulations related to the n-collinear fields are concerned.

In the remainder of this Section we adopt the notation t = q, even though the algebra applies equally well to both cases mentioned above, with the substitution $q \to t = i\partial_{n'}$ in appropriate places. The only complication for the case $t = i\partial_{n'}$ is that the dot product $n \cdot i\partial_{n'}$ must be expanded using

$$2i\partial_{n} \cdot i\partial_{n'} = \frac{n \cdot n'}{2} \overline{\mathcal{P}}_{n} \overline{\mathcal{P}}_{n'} + n' \cdot i\partial_{n\perp} \overline{\mathcal{P}}_{n'} + n \cdot i\partial_{n'\perp} \overline{\mathcal{P}}_{n} + 2i\partial_{n\perp} \cdot i\partial_{n'\perp} + \frac{\overline{n} \cdot n'}{2} n \cdot i\partial_{n} \overline{\mathcal{P}}_{n'} + \frac{\overline{n} \cdot n}{2} n' \cdot i\partial_{n'} \overline{\mathcal{P}}_{n} + \overline{n} \cdot i\partial_{n'}^{\perp} n \cdot i\partial_{n} + \overline{n'} \cdot i\partial_{n'}^{\perp} n' \cdot i\partial_{n'} + \frac{\overline{n} \cdot \overline{n'}}{2} n \cdot i\partial_{n} n' \cdot i\partial_{n'},$$

$$(3.49)$$

where the first term is $\sim \lambda^0$, the next two $\sim \lambda$, the following three are $\sim \lambda^2$, then the next two are $\sim \lambda^3$, and the last one is $\sim \lambda^4$.

Adopting t = q, Eq. (3.45) can be used to prove that

$$(q \cdot iD_n) = \mathcal{W}_n \left(q \cdot i\partial_n \right) \mathcal{W}_n^{\dagger}.$$
(3.50)

To calculate iR_n we exploit Eq. (3.50) and calculate iR_n order by order in λ . Substituting Eq. (3.18) into Eq. (3.50) we find

$$(q \cdot i\mathcal{D}_n) = e^{-iR_n} \left(q \cdot i\partial_n\right) e^{iR_n} . \tag{3.51}$$

Because of the Hermicity of $i\mathcal{D}_n^{\mu}$ and $i\partial_n^{\mu}$, R_n is Hermitian. Applying the Hadamard formula to Eq. (3.51) we obtain

$$(q \cdot i\mathcal{D}_n) = (q \cdot i\partial_n) + \sum_{j=1}^{\infty} \frac{1}{j!} \{\{(q \cdot i\partial_n), (iR_n)^j\}\}, \qquad (3.52)$$

where $\{\!\{A,B\}\!\}=[A,B]$ and

$$\{\{A, B^{j}\}\} = \{\{[A, B], B^{j-1}\}\} = [[\cdots [A, B, B], \dots, B]] .$$
(3.53)

Expanding R_n in terms with $R_n^{(k)} \sim \lambda^k$ we can expand all the objects in Eq. (3.52) in λ and solve the resulting equations order by order for $R_n^{(k)}$. Thus we write

$$iR_{n} = \sum_{k=1}^{\infty} iR_{n}^{(k)},$$

$$(q \cdot i\mathcal{D}_{n}) = \frac{n \cdot q}{2}\overline{\mathcal{P}}_{n} + (q_{\perp} \cdot \mathcal{P}_{n\perp}) + (q_{\perp} \cdot g\mathcal{B}_{n\perp}) + \frac{\bar{n} \cdot q}{2}(n \cdot i\partial_{n}) + \frac{\bar{n} \cdot q}{2}(gn \cdot \mathcal{B}_{n}),$$

$$(q \cdot i\partial_{n}) = \frac{n \cdot q}{2}\overline{\mathcal{P}}_{n} + (q_{\perp} \cdot \mathcal{P}_{n\perp}) + \frac{\bar{n} \cdot q}{2}(n \cdot i\partial_{n}).$$
(3.54)

 $(q \cdot i\partial_n)$ is a derivative operator, so when it acts in a commutator with $(g\mathcal{B}_n^{\mu})$ we have

$$\left[(q \cdot i\partial_n), (g\mathcal{B}_n^{\mu})\right] = \left[q \cdot i\partial_n \left(g\mathcal{B}_n^{\mu}\right)\right], \qquad (3.55)$$

where the last set of square brackets means that the derivative acts only inside. Substituting Eq. (3.54) into (3.52) we can solve for $iR_n^{(k)}$. The first two terms are

$$iR_n^{(1)} = \left[\frac{2}{n \cdot q \,\overline{\mathcal{P}}_n} \, q_\perp \cdot (g\mathcal{B}_n^\perp)\right],\tag{3.56}$$

$$\begin{split} iR_n^{(2)} &= \left[\frac{1}{n \cdot q \,\overline{\mathcal{P}}_n} \left(\bar{n} \cdot q\right) \left(gn \cdot \mathcal{B}_n\right)\right] - \left[\frac{4q_\perp \cdot \mathcal{P}_{n\perp}}{\left(n \cdot q \,\overline{\mathcal{P}}_n\right)^2} \, q_\perp \cdot \left(g\mathcal{B}_n^\perp\right)\right] \\ &+ \left[\frac{2}{n \cdot q \,\overline{\mathcal{P}}_n} \left[\left[\frac{1}{n \cdot q \,\overline{\mathcal{P}}_n} \, q_\perp \cdot \left(g\mathcal{B}_n^\perp\right)\right], \, q_\perp \cdot \left(g\mathcal{B}_n^\perp\right)\right] \right]. \end{split}$$

The $n \cdot \mathcal{B}_n$ term should be further reduced with the equation of motion in Eq. (3.32) to terms involving χ_n and $\mathcal{B}_{n\perp}^{\mu}$. In terms of the $iR_n^{(k)}$ we can determine the λ expansion of the invariant Wilson line

$$\mathcal{W}_n = \sum_{k=0}^{\infty} \mathcal{W}_n^{(k)} \,. \tag{3.57}$$

Using the definition in Eq. (3.18) the first few terms are

$$\mathcal{W}_{n}^{(0)} = W_{n}, \qquad \mathcal{W}_{n}^{(1)} = -W_{n}(iR_{n}^{(1)}), \qquad \mathcal{W}_{n}^{(2)} = \left[\frac{1}{2}(iR_{n}^{(1)})^{2} - (iR_{n}^{(2)})\right].$$
 (3.58)

The expansion of the invariant Wilson line is therefore

$$\mathcal{W}_n = W_n - W_n (iR_n^{(1)}) + W_n \left[\frac{1}{2} (iR_n^{(1)})^2 - (iR_n^{(2)})\right] + \dots$$
(3.59)

Using these $R_n^{(k)}$'s and Table 2.1 it is simple to check explicitly that \mathcal{W}_n is RPI up to order $O(\lambda^3)$. Note that we did not assign a suppression for q_{\perp} anywhere above (ie, we took $q_{\perp} \sim \lambda^0$). Taking $q_{\perp} \sim \lambda$ causes further suppression of some of the terms in Eq. (3.56). For cases where $q_{\perp} = 0$ the expansion of \mathcal{W}_n starts at $O(\lambda^2)$.

We will also need the λ expansion of the invariant δ -functions, $\delta(\hat{\omega} - 2q \cdot i\partial_n)$ and $\delta(\hat{\omega}_{12} - 2i\partial_{n_1} \cdot i\partial_{n_2})$. For the former we have

$$\delta(\hat{\omega} - 2q \cdot i\partial_n) = \delta(\hat{\omega} - n \cdot q \,\overline{\mathcal{P}}_n - 2q_\perp \cdot \mathcal{P}_{n\perp} - \bar{n} \cdot q \, in \cdot \partial_n) \qquad (3.60)$$
$$= \frac{1}{n \cdot q} \left[\left(1 + \sum_{k=1}^{\infty} p_n^{(k)} \right) \delta(\omega - \overline{\mathcal{P}}_n) \right],$$

where the first two terms are

$$p_n^{(1)} = -\frac{2q_{\perp} \cdot \mathcal{P}_{n\perp}}{n \cdot q} \frac{d}{d\omega}, \qquad p_n^{(2)} = 2\left(\frac{q_{\perp} \cdot \mathcal{P}_{n\perp}}{n \cdot q}\right)^2 \frac{d^2}{d\omega^2} - \frac{\bar{n} \cdot q}{n \cdot q} (in \cdot \partial_n) \frac{d}{d\omega}.$$
 (3.61)

When combining the operator with the Wilson coefficient $C(\omega_i)$ we can integrate by parts to move these derivatives onto the $C(\omega_i)$ and leave a simple δ -function in the operator. For the δ -function with two collinear directions we have

$$\delta(\hat{\omega}_{12} - 2i\partial_{n_1} \cdot i\partial_{n_2})$$

$$= \delta\left(\hat{\omega}_{12} - \frac{n_1 \cdot n_2}{2} \overline{\mathcal{P}}_{n_1} \overline{\mathcal{P}}_{n_2} - \overline{\mathcal{P}}_{n_1} n_1 \cdot i\partial_{n_2 \perp} - \overline{\mathcal{P}}_{n_2} n_2 \cdot i\partial_{n_1 \perp} - 2i\partial_{n_1 \perp} \cdot i\partial_{n_2 \perp} - \frac{\overline{n}_1 \cdot n_2}{2} n_1 \cdot i\partial_{n_1} \overline{\mathcal{P}}_{n_2} - \frac{\overline{n}_2 \cdot n_1}{2} n_2 \cdot i\partial_{n_2} \overline{\mathcal{P}}_{n_1} - n_1 \cdot i\partial_{n_1} \overline{n}_1 \cdot i\partial_{n_2 \perp} - n_2 \cdot i\partial_{n_2} \overline{n}_2 \cdot i\partial_{n_1 \perp} - \frac{\overline{n}_1 \cdot \overline{n}_2}{2} n_1 \cdot i\partial_{n_1} n_2 \cdot i\partial_{n_2} \right)$$

$$= \left[\left(1 + \sum_{k=1}^{\infty} p_{n_1 n_2}^{(k)} \right) \delta\left(\hat{\omega}_{12} - \frac{n_1 \cdot n_2}{2} \overline{\mathcal{P}}_{n_1} \overline{\mathcal{P}}_{n_2} \right) \right], \qquad (3.62)$$

where the first two terms are

$$p_{n_{1}n_{2}}^{(1)} = -\left[\overline{\mathcal{P}}_{n_{1}}n_{1}\cdot i\partial_{n_{2}\perp} + \overline{\mathcal{P}}_{n_{2}}n_{2}\cdot i\partial_{n_{1}\perp}\right]\frac{d}{d\omega},$$

$$p_{n_{1}n_{2}}^{(2)} = \left[\overline{\mathcal{P}}_{n_{1}}n_{1}\cdot i\partial_{n_{2}\perp} + \overline{\mathcal{P}}_{n_{2}}n_{2}\cdot i\partial_{n_{1}\perp}\right]^{2}\frac{d^{2}}{d\omega^{2}}$$

$$-\left[2i\partial_{n_{1}\perp}\cdot i\partial_{n_{2}\perp} + \frac{\bar{n}_{1}\cdot n_{2}}{2}n_{1}\cdot i\partial_{n_{1}}\overline{\mathcal{P}}_{n_{2}} + \frac{\bar{n}_{2}\cdot n_{1}}{2}n_{2}\cdot i\partial_{n_{2}}\overline{\mathcal{P}}_{n_{1}}\right]\frac{d}{d\omega}.$$
(3.63)

All terms with $n \cdot i\partial_n$ in Eqs. (3.61) and (3.63) will be further reduced by the equations of motion in Eqs. (3.31) and (3.32) when they appear in operators.

Finally we expand the superfields in Eq. (3.20) in λ , writing

$$\Psi_{n,\hat{\omega}} = \sum_{k=1}^{\infty} \Psi_{n,\hat{\omega}}^{(k)}, \qquad \qquad \mathcal{G}_{n,\hat{\omega}}^{\mu\nu} = \sum_{k=1}^{\infty} \mathcal{G}_{n,\hat{\omega}}^{(k)\mu\nu}, \qquad (3.64)$$

where $\Psi_{n,\hat{\omega}}^{(k)} \sim \lambda^k$ and $\mathcal{G}_{n,\hat{\omega}}^{(k)\mu\nu} \sim \lambda^k$. The expansion of the quark superfield is straight-forward, the first few orders are

$$\Psi_{n,\hat{\omega}}^{(1)} = \frac{1}{n \cdot q} \chi_{n,\omega}, \qquad (3.65)$$

$$\Psi_{n,\hat{\omega}}^{(2)} = \frac{1}{n \cdot q} \left(\frac{1}{\omega} i \mathcal{P}_{n,\omega_a-\omega}^{\perp} \frac{\vec{n}}{2} \chi_{n,\omega_a} + i R_{n,\omega_a-\omega}^{(1)} \chi_{n,\omega_a} + \left[p_n^{(1)} \chi_{n,\omega} \right] \right), \qquad (3.65)$$

$$\Psi_{n,\hat{\omega}}^{(3)} = \frac{1}{n \cdot q} \left(i R_{n,\omega_a-\omega}^{(2)} \chi_{n,\omega_a} + \left[p_n^{(2)} \chi_{n,\omega} \right] + \frac{1}{\omega_a + \omega_b} i R_{n,\omega_a-\omega_b-\omega}^{(1)} i \mathcal{P}_{n,\omega_b}^{\perp} \frac{\vec{n}}{2} \chi_{n,\omega_a} \right)$$

$$+ \left[p_n^{(1)} \frac{1}{\omega} i \mathcal{P}_{n,\omega_a-\omega}^{\perp} \frac{\vec{n}}{2} \chi_{n,\omega_a} \right] + \left[p_n^{(1)} i R_{n,\omega_a-\omega}^{(1)} \chi_{n,\omega_a} \right] + \frac{1}{2} i R_{n,\omega_a-\omega_b-\omega}^{(1)} i R_{n,\omega_b}^{(1)} \chi_{n,\omega_a} \right)$$

Here there is an implicit integration over the repeated indices ω_a and ω_b . For the gluon superfield first it is useful to expand $W^{\dagger}G_{\mu\nu}W$:

$$W^{\dagger}igG_{\mu\nu}W = \frac{n_{\mu}}{2} \left[i\bar{n}\cdot\mathcal{D},i\mathcal{D}_{\perp\nu}\right] - \frac{n_{\nu}}{2} \left[i\bar{n}\cdot\mathcal{D},i\mathcal{D}_{\perp\mu}\right] + \left[i\mathcal{D}_{\perp\mu},i\mathcal{D}_{\perp\nu}\right] + \frac{\bar{n}_{\mu}}{2} \frac{n_{\nu}}{2} \left[in\cdot\mathcal{D},i\bar{n}\cdot\mathcal{D}\right] \\ + \frac{n_{\mu}}{2} \frac{\bar{n}_{\nu}}{2} \left[i\bar{n}\cdot\mathcal{D},in\cdot\mathcal{D}\right] + \frac{\bar{n}_{\mu}}{2} \left[in\cdot\mathcal{D},i\mathcal{D}_{\perp\nu}\right] - \frac{\bar{n}_{\nu}}{2} \left[in\cdot\mathcal{D},i\mathcal{D}_{\perp\mu}\right] \\ = \frac{n_{\mu}}{2} \left[\bar{\mathcal{P}}g\mathcal{B}_{\perp\nu}\right] - \frac{n_{\nu}}{2} \left[\bar{\mathcal{P}}g\mathcal{B}_{\mu}^{\perp}\right] + \left[\bar{\mathcal{P}}g\mathcal{B}_{\mu\nu}^{\perp\perp}\right] - \frac{\bar{n}_{\mu}n_{\nu}}{4} \left[\bar{\mathcal{P}}gn\cdot\mathcal{B}\right] + \frac{n_{\mu}\bar{n}_{\nu}}{4} \left[\bar{\mathcal{P}}gn\cdot\mathcal{B}\right] \\ - \frac{\bar{n}_{\mu}}{2} \left[\bar{\mathcal{P}}g\mathcal{B}_{\nu}^{\perp2}\right] + \frac{\bar{n}_{\nu}}{2} \left[\bar{\mathcal{P}}g\mathcal{B}_{\mu}^{\perp2}\right] , \qquad (3.66)$$

where $g\mathcal{B}_{\mu\nu}^{\perp\perp}$ and $g\mathcal{B}_{\mu}^{\perp2}$ are given by the combinations of fields in Eq. (3.30). Using this result to determine the first few terms $\mathcal{G}_{n,\hat{\omega}}^{(k)\mu\nu}$ from expanding Eq. (3.20), we find

$$ig\mathcal{G}_{n,\hat{\omega}}^{(1)\mu\nu} = \frac{\omega}{2(n\cdot q)} \Big[n^{\nu} (g\mathcal{B}_{n\perp}^{\mu})_{\omega} - n^{\mu} (g\mathcal{B}_{n\perp}^{\nu})_{\omega} \Big], \qquad (3.67)$$

$$ig\mathcal{G}_{n,\hat{\omega}}^{(2)\mu\nu} = \frac{1}{n\cdot q} \Big\{ \Big[\mathcal{P}_{\perp}^{\mu} (g\mathcal{B}_{n\perp}^{\nu})_{\omega} \Big] - \Big[\mathcal{P}_{\perp}^{\nu} (g\mathcal{B}_{n\perp}^{\mu})_{\omega} \Big] + \Big[(g\mathcal{B}_{n\perp}^{\mu}), (g\mathcal{B}_{n\perp}^{\nu}) \Big]_{\omega} \\
+ \frac{\omega}{4} (\bar{n}^{\mu} n^{\nu} - n^{\mu} \bar{n}^{\nu}) (gn \cdot \mathcal{B}_{n})_{\omega} + \frac{\omega}{2} \Big[iR_{n,\omega-\omega_{a}}^{(1)}, n^{\nu} (g\mathcal{B}_{n\perp}^{\mu})_{\omega_{a}} - n^{\mu} (g\mathcal{B}_{n\perp}^{\nu})_{\omega_{a}} \Big] \\
- \frac{1}{2} \Big[n^{\nu} p_{n}^{(1)} \omega (g\mathcal{B}_{n\perp}^{\mu})_{\omega} - n^{\mu} p_{n}^{(1)} \omega (g\mathcal{B}_{n\perp}^{\nu})_{\omega} \Big] \Big\}, \\
ig\mathcal{G}_{n,\hat{\omega}}^{(3)\mu\nu} = \frac{-1}{2(n\cdot q)} \Big\{ \Big[(\bar{n}^{\mu} \mathcal{P}_{\perp}^{\nu} - \bar{n}^{\nu} \mathcal{P}_{\perp}^{\mu}) (gn \cdot \mathcal{B}_{n})_{\omega} \Big] - \Big[in \cdot \partial_{n} \big(\bar{n}^{\mu} g\mathcal{B}_{\perp,\omega}^{\nu} - \bar{n}^{\nu} g\mathcal{B}_{\perp,\omega}^{\mu} \big) \Big] \\
+ \Big[\bar{n}^{\mu} g\mathcal{B}_{\perp,\omega}^{\nu}, gn \cdot \mathcal{B}_{n} \Big] - \Big[\bar{n}^{\nu} g\mathcal{B}_{\perp,\omega}^{\mu}, gn \cdot \mathcal{B}_{n} \Big] - \Big[n^{\nu} \omega (g\mathcal{B}_{n\perp}^{\mu})_{\omega} p_{n}^{(2)\dagger} - n^{\mu} \omega (g\mathcal{B}_{n\perp}^{\nu})_{\omega} p_{n}^{(2)\dagger} \Big] \\
+ \dots \Big\},$$

where again there is an implicit integration over ω_a in terms where it appears. Here the ellipsis denotes terms in $\mathcal{G}_n^{(3)\mu\nu}$ with an $iR_n^{(1,2)}$ or $p_n^{(1)}$ which were not needed for our analysis. The $(gn\cdot\mathcal{B}_n)$ and $[in\cdot\partial_n g\mathcal{B}_{\perp}^{\mu}]$ terms are further reduced to \mathcal{P}_{\perp} 's, $(g\mathcal{B}_{\perp}^{\mu})$'s, and χ_n 's by using the equation of motion in Eq. (3.32). Finally, recall that the expansion coefficients in Eqs. (3.63,3.65,3.67) do not encode the RPI relations between collinear and soft fields which can be determined using Eq. (2.29).

The above results can be used to expand the RPI basis of operators in terms of operators in the homogeneous basis as in Eq. (3.29).

3.7 Applications

3.7.1 Scalar Current

As a first example to show how the expansion of a RPI current works, we expand the scalar chiral-even bilinear currents (LL+RR), for processes with a hard external vector q^{μ} up to order λ^3 . In the basis built from superfields there is only one current that satisfies these conditions

$$\bar{\Psi}_{n,\hat{\omega}_1} \not q \Psi_{n,\hat{\omega}_2} . \tag{3.68}$$

All the other possible currents (for example $\bar{\Psi}_{n,\hat{\omega}_1}\gamma_{\mu}q_{\nu}\mathcal{G}^{\mu\nu}_{n,\hat{\omega}_3}\Psi_{n,\hat{\omega}_2}$) have expansions that start at $\mathcal{O}(\lambda^4)$ or beyond. To recover the basis with a homogeneous power counting, all we have to do is to expand (3.68) using Eq. (3.65),

Thus all the $\mathcal{O}(\lambda^3)$ terms (the twist-3 terms on the last three lines) are connected. Eq. (3.69) agrees with the original derivation of these constraints given in Eqs. (122-126) of Ref. [58]. The ease at which Eq. (3.69) was derived demonstrates the power of the invariant operator formalism. In this example there is only one supercurrent to $\mathcal{O}(\lambda^3)$, so all Wilson coefficients are connected to the coefficient of the leading operator $\bar{\chi}_{n,\omega_1} \bar{n} \chi_{n,\omega_2}$. Note that here all of the connected operators involve a q_{\perp} , which we have counted as $\mathcal{O}(\lambda^0)$. We will see below that for situations with two collinear directions, where in the end its natural to specialize to a frame where $q_{\perp} = 0$, the connections tend to appear at higher twist. For situations with three or more collinear directions RPI will provide useful constraints on the basis already at lowest order.

3.7.2 General Quark and Gluon Operators

In this Section we enumerate an operator basis for the general set of collinear quark and gluon operators up to $\mathcal{O}(\lambda^4)$. This basis is useful for many applications, and we keep our notation as general as possible. In particular we consider up to 4 distinct collinear directions (which for example could be used for $e^+e^- \rightarrow 4$ jets, or $gg, qg, q\bar{q} \rightarrow$ 2 jets). We also discuss a basis both for the homogeneous operators with a definite power counting, and for the RPI operators.

For processes with a hard q^{μ} , the most general basis of homogeneous quark operators in SCET up to $\mathcal{O}(\lambda^4)$ is

$$O^{(0a)} = \bar{\chi}_{n_1,\omega_1} \Gamma \chi_{n_2,\omega_2}, \qquad O^{(1a)} = \bar{\chi}_{n_1,\omega_1} \Gamma_{\alpha} \mathcal{P}_{n_1\perp}^{\dagger \alpha} \chi_{n_2,\omega_2}, \qquad O^{(1a)} = \bar{\chi}_{n_1,\omega_1} \Gamma_{\alpha} \mathcal{P}_{n_1\perp}^{\dagger \alpha} \chi_{n_2,\omega_2}, \qquad O^{(1c)} = \bar{\chi}_{n_1,\omega_1} \Gamma_{\alpha} (ig\mathcal{B}_{n_3\perp}^{\alpha})_{\omega_3} \chi_{n_2,\omega_2}, O^{(2a)} = \bar{\chi}_{n_1,\omega_1} \Gamma_{\alpha\beta} \mathcal{P}_{n_1\perp}^{\dagger \alpha} \mathcal{P}_{n_2\perp}^{\beta} \chi_{n_2,\omega_2}, \qquad O^{(2b)} = \bar{\chi}_{n_1,\omega_1} \Gamma_{\alpha\beta} \mathcal{P}_{n_1\perp}^{\dagger \alpha} \mathcal{P}_{n_1\perp}^{\beta} \chi_{n_2,\omega_2}, O^{(2c)} = \bar{\chi}_{n_1,\omega_1} \Gamma_{\alpha\beta} \mathcal{P}_{n_2\perp}^{\beta} \chi_{n_2,\omega_2}, \qquad O^{(2d)} = \bar{\chi}_{n_1,\omega_1} \Gamma_{\alpha\beta} \mathcal{P}_{n_1\perp}^{\dagger \alpha} (g\mathcal{B}_{n_3\perp}^{\beta})_{\omega_3} \chi_{n_2,\omega_2}, O^{(2e)} = \bar{\chi}_{n_1,\omega_1} \Gamma_{\beta\alpha} (g\mathcal{B}_{n_3\perp}^{\beta})_{\omega_3} \mathcal{P}_{n_2}^{\perp \alpha} \chi_{n_2,\omega_2}, \qquad O^{(2f)} = \bar{\chi}_{n_1,\omega_1} \Gamma_{\alpha\beta} [\mathcal{P}_{n_3}^{\perp \alpha} (g\mathcal{B}_{n_3}^{\perp \beta})_{\omega_3}] \chi_{n_2,\omega_2}, O^{(2g)} = \bar{\chi}_{n_1,\omega_1} \Gamma_{\alpha\beta} (g\mathcal{B}_{n_3\perp}^{\alpha})_{\omega_3} (g\mathcal{B}_{n_4\perp}^{\beta})_{\omega_4} \chi_{n_2,\omega_2}, \qquad O^{(2h)} = (\bar{\chi}_{n_1,\omega_1} \Gamma_1 \chi_{n_2,\omega_2}) (\bar{\chi}_{n_3,\omega_3} \Gamma_2 \chi_{n_4,\omega_4})$$

If we need to specify the subscripts we write for example $O^{(2g)}(\omega_1, \omega_3, \omega_4, \omega_2)$, with the ω_i listed from left to right. Due to the equations of motion in Eqs. (3.31,3.32) we did not need to consider $in \cdot \partial_n \chi_n$ or $gn \cdot \mathcal{B}_n$. For each operator there may be a set of different Dirac, flavor, and color structures $\Gamma^j_{\alpha_1\cdots\alpha_n}$ which depend on the particular phenomena being studied (including also two choices for color for the Γ_i in the four-quark operators $O^{(2i)}$). In general for each independent $\Gamma^{j}_{\alpha_{1}\cdots\alpha_{n}}$ structure the operator has a Wilson coefficient that must be determined order by order in perturbation theory. We included in Eq. (3.70) the mixed quark and gluon operators. For pure gluon operators up $\mathcal{O}(\lambda^{4})$ we have the homogeneous basis

$$\begin{aligned}
O^{(0b)} &= \mathcal{B}_{n_{1},\omega_{1}}^{\perp\mu} \mathcal{B}_{n_{2},\omega_{2}}^{\perp\nu}, & O^{(1d)} &= \mathcal{B}_{n_{1},\omega_{1}}^{\perp\mu} \mathcal{P}_{n_{1}\perp}^{\dag\alpha} \mathcal{B}_{n_{2},\omega_{2}}^{\perp\nu}, & (3.71) \\
O^{(1e)} &= \mathcal{B}_{n_{1},\omega_{1}}^{\perp\mu} \mathcal{P}_{n_{2}\perp}^{\alpha} \mathcal{B}_{n_{2},\omega_{2}}^{\perp\nu}, & O^{(1f)} &= \mathcal{B}_{n_{1},\omega_{1}}^{\perp\mu} \mathcal{B}_{n_{2},\omega_{2}}^{\perp\nu} \mathcal{B}_{n_{3},\omega_{3}}^{\perp\tau}, \\
O^{(2i)} &= \mathcal{B}_{n_{1},\omega_{1}}^{\perp\mu} \mathcal{P}_{n_{1}\perp}^{\dag\beta} \mathcal{B}_{n_{2},\omega_{2}}^{\perp\nu}, & O^{(2j)} &= \mathcal{B}_{n_{1},\omega_{1}}^{\perp\mu} \mathcal{P}_{n_{2}\perp}^{\beta} \mathcal{B}_{n_{2},\omega_{2}}^{\perp\nu}, \\
O^{(2k)} &= \mathcal{B}_{n_{1},\omega_{1}}^{\perp\mu} \mathcal{P}_{n_{2}\perp}^{\alpha} \mathcal{P}_{n_{2}\perp}^{\beta} \mathcal{B}_{n_{2},\omega_{2}}^{\perp\nu}, & O^{(2l)} &= \left[\mathcal{P}_{n_{1}\perp}^{\alpha} \mathcal{B}_{n_{1},\omega_{1}}^{\perp\mu}\right] \mathcal{B}_{n_{2},\omega_{2}}^{\perp\nu} \mathcal{B}_{n_{3},\omega_{3}}^{\perp\tau}, \\
O^{(2m)} &= \mathcal{B}_{n_{1},\omega_{1}}^{\perp\mu} \left[\mathcal{P}_{n_{2}\perp}^{\alpha} \mathcal{B}_{n_{2},\omega_{2}}^{\perp\nu}\right] \mathcal{B}_{n_{3},\omega_{3}}^{\perp\tau}, & O^{(2n)} &= \mathcal{B}_{n_{1},\omega_{1}}^{\perp\mu} \mathcal{B}_{n_{2},\omega_{2}}^{\perp\nu} \left[\mathcal{P}_{n_{3}\perp}^{\alpha} \mathcal{B}_{n_{3},\omega_{3}}^{\perp\tau}\right], \\
O^{(2o)} &= \mathcal{B}_{n_{1},\omega_{1}}^{\perp\mu} \mathcal{B}_{n_{2},\omega_{2}}^{\perp\nu} \mathcal{B}_{n_{3},\omega_{3}}^{\perp\sigma} \mathcal{B}_{n_{4},\omega_{4}}^{\perp\tau}.
\end{aligned}$$

Here we do not need to consider operators with $gn \cdot \mathcal{B}_n$ and $gn \cdot \partial_n \mathcal{B}_{n\perp}^{\mu}$ because using the equations of motion in Eq. (3.32) they can be written in terms of the operators in Eq. (3.71), and are hence redundant.

To setup the computation of constraints on Wilson coefficients we also need to build an RPI basis of operators using the objects in Eq. (3.13) and $i\partial_n^{\mu}$. Because each operator will be RPI, its Wilson coefficient is truly independent of those for other operators in the basis. The RPI operators can then be expanded in terms of homogeneous operators made out of of gauge invariant objects, and doing so we obtain operators in the homogeneous basis with all the constraints coming from reparametrization invariance. The number of constraints on Wilson coefficients is equal to the number of homogeneous operators minus the number of RPI operators, once we have accounted for linear dependencies [46, 73].

Let's construct the RPI basis of operators which is the analog of those in Eqs. (3.70) and (3.71). The operators with no $i\partial_n^{\mu}$ derivatives are

$$\mathbf{Q}^{(0q)} = \bar{\Psi}_{n_1,\hat{\omega}_1} \Gamma \Psi_{n_2,\hat{\omega}_2} , \qquad \qquad \mathbf{Q}^{(0g)} = \mathcal{G}_{n_1,\hat{\omega}_1}^{\mu\nu} \, \mathcal{G}_{n_2,\hat{\omega}_2}^{\sigma\tau} , \qquad (3.72)$$

where the basis of Dirac structures Γ , and contraction of indices $\mu\nu\sigma\tau$ in $\mathbf{Q}^{(0g)}$ depends on the kind of current we are studying. For cases without a q^{μ} the subscripts $\hat{\omega}_i$ are erased and RPI operators are multiplied by the $\hat{\Delta}_{ij}$ factors shown in Eq. (3.17). Recall that we do not have a good power counting in the RPI basis, this basis makes the RPI properties transparent but the power counting more tricky. When $\mathbf{Q}^{(aq)}$ and $\mathbf{Q}^{(ag)}$ are expanded in terms of operators that are homogeneous in the power counting, they contain a leading order term, so they are relevant operators to consider at LO. Of the RPI objects only $i\partial_n^{\mu}$ starts at leading order, so theoretically we can construct an infinite set of LO operators using $(i\partial_n)^k$ for any k. However, the structure of this operator provides additional constraints. In particular the $\mathcal{O}(\lambda^0)$ term is $i\partial_n^{\mu} = (n^{\mu}/2)\overline{\mathcal{P}}_n + \cdots$, and the collinear momentum $\overline{\mathcal{P}}_n$ acting on a *n*-collinear field such as χ_{n,ω_1} just gives a number, ω_1 , which can be absorbed into the Wilson coefficient $C(\omega_1,\omega_2)$. For cases with a q^{μ} this implies that adding $i\partial_n^{\mu}$'s in a scalar operator (where all vector indices are contracted) most often gives an operator that differs from one we already have only at $\mathcal{O}(\lambda)$. For these scalar operators we can count $i\partial_n^\mu \sim \mathcal{O}(\lambda)$ when determining which RPI operators are required, and for simplicity we follow this counting in the remainder of this Section. If we have an operator with a free vector index μ , then this index can be carried by $i\partial_n^{\mu} = (n^{\mu}/2)\overline{\mathcal{P}}_n + \cdots$, and the partial derivative does count as $\mathcal{O}(\lambda^0)$.

The expansion of the RPI operators in Eq. (3.72) in terms of homogeneous operators up to $\mathcal{O}(\lambda^4)$ is

$$\mathbf{Q}^{(0q)} = \bar{\Psi}_{n_{1},\hat{\omega}_{1}}^{(1)} \Gamma \Psi_{n_{2},\hat{\omega}_{2}}^{(1)} + \bar{\Psi}_{n_{1},\hat{\omega}_{1}}^{(2)} \Gamma \Psi_{n_{2},\hat{\omega}_{2}}^{(1)} + \bar{\Psi}_{n_{1},\hat{\omega}_{1}}^{(1)} \Gamma \Psi_{n_{2},\hat{\omega}_{2}}^{(2)} + \bar{\Psi}_{n_{1},\hat{\omega}_{1}}^{(1)} \Gamma \Psi_{n_{2},\hat{\omega}_{2}}^{(2)} + \bar{\Psi}_{n_{1},\hat{\omega}_{1}}^{(1)} \Gamma \Psi_{n_{2},\hat{\omega}_{2}}^{(3)} + \bar{\Psi}_{n_{1},\hat{\omega}_{1}}^{(3)} \Gamma \Psi_{n_{2},\hat{\omega}_{2}}^{(1)} + O(\lambda^{5}) ,$$

$$\mathbf{Q}^{(0g)} = \mathcal{G}_{n_{1},\hat{\omega}_{1}}^{(1)\mu\nu} \mathcal{G}_{n_{2},\hat{\omega}_{2}}^{(1)\sigma\tau} + \mathcal{G}_{n_{1},\hat{\omega}_{1}}^{(2)\mu\nu} \mathcal{G}_{n_{2},\hat{\omega}_{2}}^{(1)\sigma\tau} + \mathcal{G}_{n_{1},\hat{\omega}_{1}}^{(1)\mu\nu} \mathcal{G}_{n_{2},\hat{\omega}_{2}}^{(2)\sigma\tau} + \mathcal{G}_{n_{1},\hat{\omega}_{1}}^{(2)\mu\nu} \mathcal{G}_{n_{2},\hat{\omega}_{2}}^{(2)\sigma\tau} + \mathcal{G}_{n_{1},\hat{\omega}_{1}}^{(3)\mu\nu} \mathcal{G}_{n_{2},\hat{\omega}_{2}}^{(1)\mu\nu} \mathcal{G}_{n_{2},\hat{\omega}_{2}}^{(3)\sigma\tau} + O(\lambda^{5}) ,$$

$$(3.73)$$

where the $\Psi_{n,\hat{\omega}}^{(k)}$ and $\mathcal{G}_{n,\hat{\omega}}^{(k)}$ are given in Eqs. (3.65) and (3.67). To look for RPI relations the results of this expansion must be compared to power suppressed operators which

also can generate $\mathcal{O}(\lambda^3)$ and $\mathcal{O}(\lambda^4)$ terms. Up to this order the power suppressed operators involving two or more quark fields are

$$\mathbf{Q}^{(1a)} = \bar{\Psi}_{n_1,\hat{\omega}_1}\Gamma_{\alpha}\,i\partial_{n_2}^{\alpha}\,\Psi_{n_2,\hat{\omega}_2}\,, \qquad \mathbf{Q}^{(1b)} = \bar{\Psi}_{n_1,\hat{\omega}_1}\Gamma_{\alpha}\,i\overline{\partial}_{n_1}^{\alpha}\,\Psi_{n_2,\hat{\omega}_2}\,, \qquad (3.74)$$

$$\mathbf{Q}^{(1c)} = \bar{\Psi}_{n_1,\hat{\omega}_1}\Gamma_{\beta\beta'}\,\mathcal{G}_{n_3,\hat{\omega}_3}^{\beta\beta'}\,\Psi_{n_2,\hat{\omega}_2}\,, \qquad \mathbf{Q}^{(2a)} = \bar{\Psi}_{n_1,\hat{\omega}_1}\Gamma_{\alpha\alpha'}\,i\overline{\partial}_{n_2}^{\alpha}\,i\partial_{n_2}^{\alpha'}\,\Psi_{n_2,\hat{\omega}_2}\,, \qquad \mathbf{Q}^{(2a)} = \bar{\Psi}_{n_1,\hat{\omega}_1}\Gamma_{\alpha\alpha'}\,i\overline{\partial}_{n_2}^{\alpha'}\,\Psi_{n_2,\hat{\omega}_2}\,, \qquad \mathbf{Q}^{(2c)} = \bar{\Psi}_{n_1,\hat{\omega}_1}\Gamma_{\alpha\beta\beta'}\,i\overline{\partial}_{n_2}^{\alpha'}\,\Psi_{n_2,\hat{\omega}_2}\,, \qquad \mathbf{Q}^{(2c)} = \bar{\Psi}_{n_1,\hat{\omega}_1}\Gamma_{\alpha\beta\beta'}\,[i\partial_{n_3}^{\alpha}\mathcal{G}_{n_3,\hat{\omega}_3}^{\beta\beta'}]\,\Psi_{n_2,\hat{\omega}_2}\,, \qquad \mathbf{Q}^{(2e)} = \bar{\Psi}_{n_1,\hat{\omega}_1}\Gamma_{\alpha\beta\beta'}\,[i\partial_{n_3}^{\alpha}\mathcal{G}_{n_3,\hat{\omega}_3}^{\beta\beta'}]\,\Psi_{n_2,\hat{\omega}_2}\,, \qquad \mathbf{Q}^{(2e)} = \bar{\Psi}_{n_1,\hat{\omega}_1}\Gamma_{\alpha\beta\beta'}\,\mathcal{G}_{n_3,\hat{\omega}_3}^{\beta\beta'}\,\mathcal{G}_{n_3,\hat{\omega}_3}^{\beta\beta'}\,\mathcal{G}_{n_2,\hat{\omega}_2}^{\beta\beta'}\,, \qquad \mathbf{Q}^{(2e)} = \bar{\Psi}_{n_1,\hat{\omega}_1}\Gamma_{\alpha\beta\gamma\delta}\,\mathcal{G}_{n_3,\hat{\omega}_3}^{\alpha\beta}\,\mathcal{G}_{n_4,\hat{\omega}_4}^{\gamma\delta}\,\Psi_{n_2,\hat{\omega}_2}\,, \qquad \mathbf{Q}^{(2e)} = \bar{\Psi}_{n_1,\hat{\omega}_1}\Gamma_{\alpha\beta\gamma\delta}\,\mathcal{G}_{n_3,\hat{\omega}_3}^{\beta\beta'}\,\mathcal{G}_{n_4,\hat{\omega}_4}^{\gamma\delta}\,\Psi_{n_2,\hat{\omega}_2}\,, \qquad \mathbf{Q}^{(2e)} = \bar{\Psi}_{n_1,\hat{\omega}_1}\Gamma_{\alpha\beta\gamma\delta}\,\mathcal{G}_{n_3,\hat{\omega}_3}^{\gamma\delta}\,\mathcal{G}_{n_4,\hat{\omega}_4}^{\gamma\delta}\,\Psi_{n_2,\hat{\omega}_2}\,, \qquad \mathbf{Q}^{(2e)} = \bar{\Psi}_{n_1,\hat{\omega}_1}\Gamma_{\alpha\beta\gamma\delta}\,\mathcal{G}_{n_3,\hat{\omega}_3}^{\gamma\delta}\,\mathcal{G}_{n_4,\hat{\omega}_4}^{\gamma\delta}\,\mathcal{G}_{n_4,\hat{\omega}_4}^{\gamma\delta'}\,\Psi_{n_4,\hat{\omega}_4}\,, \qquad \mathbf{Q}^{(2e)} = \bar{\Psi}_{n_1,\hat{\omega}_1}\Gamma_{\alpha\beta\gamma\delta}\,\mathcal{G}_{n_3,\hat{\omega}_3}^{\gamma\delta}\,\mathcal{G}_{n_4,\hat{\omega}_4}^{\gamma\delta'}\,\Psi_{n_4,\hat{\omega}_4}\,, \qquad \mathbf{Q}^{(2e)} = \bar{\Psi}_{n_1,\hat{\omega}_1}\Gamma_{\alpha\beta\gamma\delta}\,\mathcal{G}_{n_3,\hat{\omega}_3}^{\gamma\delta'}\,\mathcal{G}_{n_4,\hat{\omega}_4}^{\gamma\delta'}\,\mathcal{G}_{n_4,\hat{\omega}_4}^{\gamma\delta'}\,\mathcal{G}_{n_4,\hat{\omega}_4}^{\gamma\delta'}\,\mathcal{G}_{n_4,\hat{\omega}_4}^{\gamma\delta'}\,\mathcal{G}_{n_4,\hat{\omega}_4}^{\gamma\delta'$$

Again a minimal basis for Dirac structures Γ will depend on the process being studied and may differ between the various $\mathbf{Q}^{(ix)}$ operators. Such a basis will also in general differ from the one for the homogeneous operators in Eq. (3.70). We will adopt notation such as $\mathbf{Q}^{(2g)}(\hat{\omega}_1, \hat{\omega}_3, \hat{\omega}_4, \hat{\omega}_2)$ when we wish to specify these subscripts. For a field basis for the higher order operators with gluon fields (whose expansion starts at $\mathcal{O}(\lambda^3)$ or $\mathcal{O}(\lambda^4)$) we have

$$\mathbf{Q}^{(1d)} = \mathcal{G}_{n_1,\hat{\omega}_1}^{\mu\nu} i\partial_{n_2}^{\alpha} \mathcal{G}_{n_2,\hat{\omega}_2}^{\sigma\tau}, \qquad \mathbf{Q}^{(1e)} = \mathcal{G}_{n_1,\hat{\omega}_1}^{\mu\nu} i\overleftarrow{\partial}_{n_1}^{\alpha} \mathcal{G}_{n_2,\hat{\omega}_2}^{\sigma\tau}, \qquad (3.75)$$

$$\mathbf{Q}^{(1f)} = \mathcal{G}_{n_1,\hat{\omega}_1}^{\mu\nu} \mathcal{G}_{n_2,\hat{\omega}_2}^{\sigma\tau} \mathcal{G}_{n_3,\hat{\omega}_3}^{\alpha\beta}, \qquad \mathbf{Q}^{(2i)} = \mathcal{G}_{n_1,\hat{\omega}_1}^{\mu\nu} i\partial_{n_2}^{\alpha} i\partial_{n_2}^{\beta} \mathcal{G}_{n_2,\hat{\omega}_2}^{\sigma\tau}, \qquad \mathbf{Q}^{(2i)} = \mathcal{G}_{n_1,\hat{\omega}_1}^{\mu\nu} i\overleftarrow{\partial}_{n_1}^{\alpha} i\overleftarrow{\partial}_{n_1}^{\beta} \mathcal{G}_{n_2,\hat{\omega}_2}^{\sigma\tau}, \qquad \mathbf{Q}^{(2k)} = \mathcal{G}_{n_1,\hat{\omega}_1}^{\mu\nu} i\overleftarrow{\partial}_{n_2}^{\alpha} \mathcal{G}_{n_2,\hat{\omega}_2}^{\sigma\tau}, \qquad \mathbf{Q}^{(2m)} = \mathcal{G}_{n_1,\hat{\omega}_1}^{\mu\nu} [i\partial_{n_2}^{\gamma} \mathcal{G}_{n_2,\hat{\omega}_2}^{\sigma\tau}] \mathcal{G}_{n_3,\hat{\omega}_3}^{\alpha\beta}, \qquad \mathbf{Q}^{(2m)} = \mathcal{G}_{n_1,\hat{\omega}_1}^{\mu\nu} [i\partial_{n_2}^{\gamma} \mathcal{G}_{n_2,\hat{\omega}_2}^{\sigma\tau}] \mathcal{G}_{n_3,\hat{\omega}_3}^{\alpha\beta}, \qquad \mathbf{Q}^{(2m)} = \mathcal{G}_{n_1,\hat{\omega}_1}^{\mu\nu} \mathcal{G}_{n_2,\hat{\omega}_2}^{\sigma\tau} \mathcal{G}_{n_3,\hat{\omega}_3}^{\alpha\beta} \mathcal{G}_{n_4,\hat{\omega}_4}^{\alpha\delta}. \qquad \mathbf{Q}^{(2m)} = \mathcal{G}_{n_1,\hat{\omega}_1}^{\mu\nu} \mathcal{G}_{n_2,\hat{\omega}_2}^{\sigma\tau} \mathcal{G}_{n_3,\hat{\omega}_3}^{\alpha\beta} \mathcal{G}_{n_4,\hat{\omega}_4}^{\alpha\delta}. \qquad \mathbf{Q}^{(2m)} = \mathcal{G}_{n_1,\hat{\omega}_1}^{\mu\nu} \mathcal{G}_{n_2,\hat{\omega}_2}^{\alpha\beta} \mathcal{G}_{n_3,\hat{\omega}_3}^{\alpha\beta} \mathcal{G}_{n_4,\hat{\omega}_4}^{\alpha\delta}. \qquad \mathbf{Q}^{(2m)} = \mathcal{G}_{n_1,\hat{\omega}_1}^{\mu\nu} \mathcal{G}_{n_2,\hat{\omega}_2}^{\alpha\beta} \mathcal{G}_{n_4,\hat{\omega}_4}^{\alpha\delta}. \qquad \mathbf{Q}^{(2m)} = \mathcal{G}_{n_1,\hat{\omega}_1}^{\mu\nu} \mathcal{G}_{n_4,\hat{\omega}_4}^{\alpha\beta} \mathcal{G}_{n_4,\hat{\omega}_4}^{\alpha\delta}. \qquad \mathbf{Q}^{(2m)} = \mathcal{G}_{n_1,\hat{\omega}_1}^{\mu\nu} \mathcal{G}_{n_4,\hat{\omega}_4$$

We will include a basis of Dirac structures and expand the RPI operators in Eqs. (3.74) and (3.75) in terms of the homogeneous ones in several of the examples below, and consider whether there are non-trivial RPI relations on a case-by-case basis.

3.7.3 Deep Inelastic Scattering for Quarks at Twist-4

In this Section we consider spin-averaged DIS at twist-4. This provides a test of our technique of constructing a minimal basis, for an example where the basis is already well known [42, 62, 63]. We will see that RPI constrains the Wilson coefficients of the homogeneous collinear operators. Our analysis is really of scalar operators with one collinear direction, $q_{\perp} = 0$, with overall derivatives set to zero. DIS is the most popular application for these operators, so we frame our discussion in that language. For simplicity we consider the QCD electromagnetic current $J^{\mu} = \bar{q}\gamma^{\mu}q$ for one-flavor of quark. (We briefly discuss the generalization to non-singlet operators in a footnote.) The study of higher twist in DIS and related processes is an active field of research, for example [23, 28, 36, 53, 66, 87]. In the language of SCET, DIS was first studied in [9], whose notation we follow. The virtual photon has momentum transfer $q^2 = -Q^2$, and $x = Q^2/(2p \cdot q)$ is the Bjorken variable.

In the Breit frame the momentum of the virtual photon is $q^{\mu} = Q(\bar{n}^{\mu} - n^{\mu})/2$, and the incoming proton momentum is $p^{\mu} = n^{\mu}\bar{n} \cdot p/2 + \bar{n}^{\mu}m_p^2/(2\bar{n} \cdot p)$ where m_p is the mass of the proton. Expanding in m_p/Q we have $\bar{n} \cdot p = Q/x - xm_p^2/Q + \dots$ The energetic proton has a small invariant mass $p^2 = m_p^2 \sim \Lambda_{\rm QCD}^2$, and in the Breit frame it is described by collinear fields in the effective theory with a power counting in $\lambda =$ $\Lambda_{\rm QCD}/Q$. It is convenient to pick this frame in order to be able to assign definite power counting to momentum components. What reparametrization invariance enforces is that all results are invariant to small perturbations about this frame, encoded by changes to the collinear reference vector n^{μ} . Since these changes are small we are free to use the same power counting when studying the RPI relations. There is a larger class of frame independence, which says for example that the same results would be found if we compare an analysis in the Breit-frame with an analysis made about the initial proton rest frame, but this set of "big" frame transformations does not encode non-trivial dynamic information that relates coefficients of operators at higher twist. All final results are of course entirely frame independent. For spin-averaged DIS the hadronic tensor has the structure

$$T_{\mu\nu} = \left(-g_{\mu\nu} + \frac{q_{\mu}q_{\nu}}{q^2}\right)T_1(x,Q^2) + \left(p_{\mu} + \frac{q_{\mu}}{2x}\right)\left(p_{\nu} + \frac{q_{\nu}}{2x}\right)T_2(x,Q^2), \quad (3.76)$$

where

$$T_{\mu\nu}(p,q) = \frac{1}{2} \sum_{\text{spin}} \langle p | \hat{T}_{\mu\nu}(q) | p \rangle, \qquad \hat{T}_{\mu\nu}(q) = i \int d^4 z e^{iq \cdot z} \mathbf{T}[J_{\mu}(z), J_{\nu}(0)].$$
(3.77)

The scalar structure functions T_i can be projected out of $T_{\mu\nu}$ using

$$T_1(Q^2, x) = -\frac{1}{2} \left(g^{\mu\nu} - \frac{4x^2}{Q^2 + 4m_p^2 x^2} p^\mu p^\nu \right) T_{\mu\nu} ,$$

$$T_2(Q^2, x) = -\frac{2x^2}{Q^2 + 4m_p^2 x^2} \left(g^{\mu\nu} - \frac{12x^2}{Q^2 + 4m_p^2 x^2} p^\mu p^\nu \right) T_{\mu\nu} .$$
(3.78)

The expansion of T_1 and T_2 has been carried out up to twist-4 with the Wilson coefficients determined at tree level in Refs. [42, 62, 63]. To simplify our calculations we will make use of the fact that the projections in Eq. (3.78) commute with taking the proton matrix element, and hence can be applied directly to $\hat{T}_{\mu\nu}$ to give \hat{T}_1 and \hat{T}_2 , where $\frac{1}{2} \sum_{\text{spin}} \langle p | \hat{T}_i | p \rangle = T_i(Q^2, x)$. Thus we consider the expansion of \hat{T}_1 and \hat{T}_2 in scalar chiral-even operators, by writing

$$\hat{T}_i = \sum_j \int [d\omega_k] C_j^{[i]}(\omega_k) O_j(\omega_k) .$$
(3.79)

Here $[d\omega_k] = d\omega_1 \cdots d\omega_n$ is the integration measure over the independent parton momenta ω_k carried by the Wilson coefficients $C_j^{[i]}$ and the operators O_j . The superscript [i] indicates that the Wilson coefficients for the two tensor structures will in general differ. We also consider a basis of RPI operators \mathbf{Q}_j by writing

$$\hat{T}_i = \sum_j \int [d\hat{\omega}_k] \, \hat{C}_j^{[i]}(\hat{\omega}_k) \mathbf{Q}_j(\hat{\omega}_k) \,. \tag{3.80}$$

Unlike the O_j 's the \mathbf{Q}_j 's do not contain contributions of a definite order in the power

counting. Using the RPI \mathbf{Q}_j operators we can test if there are relations between the Wilson coefficients $C_j^{[i]}$ of the O_j 's. A connection would mean, for example, that the one-loop coefficient for a twist-4 operator is determined by a coefficient at twist-2 at all orders in α_s .

We first write down a gauge invariant basis of chiral-even quark operators that are homogeneous in the power counting. This can be done using the general basis in Eq. (3.70) with all directions $n_i = n$. Furthermore, since the DIS matrix element is forward, we have $\langle p | [\mathcal{P}^{\mu}O] | p \rangle = 0$ for any operator O. Thus we are free to integrate \perp -label momentum operators by parts, and hence can ignore all terms with $\mathcal{P}_{\perp}^{\dagger}$'s in Eq. (3.70). (If we consider our analysis to be of the general scalar operators with one collinear direction, then this is the only simplification that we make which relies on the form of the final matrix element.) For simplicity we also drop the squarebrackets from inside $O^{(2f)}$ in Eq. (3.70). A minimal basis of chiral-even parity-even Dirac structures between the *n*-collinear quark fields is easily constructed using the properties of the SCET χ_n fields. We have i) just $\{\vec{n}\}$ when there are no vector indices on fields, ii) no elements at all when there is one vector index, and iii) just $\{\vec{\eta}g_{\perp}^{\mu\nu}, i\epsilon_{\perp}^{\mu\nu}\vec{\eta}\gamma_5\}$ or $\{\vec{\eta}g_{\perp}^{\mu\nu}, \vec{\eta}\gamma_{\perp}^{\mu}\gamma_{\perp}^{\nu}\}$ for two vector indices on fields. Here ii) is the standard fact that the spin-averaged case does not have twist-3 terms. (For polarized DIS it does not suffice to only consider the scalar operators.) For the four-quark operators we can have $\Gamma_1 \otimes \Gamma_2 = \{ \vec{n} \otimes \vec{n}, \vec{n} \gamma_5 \otimes \vec{n} \gamma_5 \}$ and color structures $1 \otimes 1$ or $T^A \otimes T^A$. Thus the basis is

$$O_{1} = \bar{\chi}_{n,\omega_{1}} \frac{\vec{\hbar}}{2} \chi_{n,\omega_{2}}, \qquad O_{2} = \bar{\chi}_{n,\omega_{1}} \frac{\vec{\hbar}}{2} \mathcal{P}_{\perp}^{2} \chi_{n,\omega_{2}}, \qquad (3.81)$$

$$O_{3a} = \bar{\chi}_{n,\omega_{1}} \frac{\vec{\hbar}}{2} (g\mathcal{B}_{n\perp}^{\mu})_{\omega_{3}} \mathcal{P}_{\perp}^{\perp} \chi_{n,\omega_{2}}, \qquad O_{3b} = \bar{\chi}_{n,\omega_{1}} \frac{\vec{\hbar}}{2} \mathcal{P}_{\perp}^{\perp} (g\mathcal{B}_{n\perp}^{\mu})_{\omega_{3}} \chi_{n,\omega_{2}}, \qquad O_{3b} = \bar{\chi}_{n,\omega_{1}} \frac{\vec{\hbar}}{2} \mathcal{P}_{\perp}^{\perp} (g\mathcal{B}_{n\perp}^{\mu})_{\omega_{3}} \chi_{n,\omega_{2}}, \qquad O_{4b} = \bar{\chi}_{n,\omega_{1}} \frac{\vec{\hbar}}{2} \mathcal{P}_{\perp} (g\mathcal{B}_{n\perp})_{\omega_{3}} \chi_{n,\omega_{2}}, \qquad O_{5} = \bar{\chi}_{n,\omega_{1}} \frac{\vec{\hbar}}{2} (g\mathcal{B}_{n\perp})_{\omega_{3}} (g\mathcal{B}_{n\perp})_{\omega_{4}} \chi_{n,\omega_{2}}, \qquad O_{6} = \bar{\chi}_{n,\omega_{1}} \frac{\vec{\hbar}}{2} \operatorname{Tr} \left[(g\mathcal{B}_{n\perp})_{\omega_{3}} (g\mathcal{B}_{n\perp})_{\omega_{4}} \right] \chi_{n,\omega_{2}}, \qquad O_{8} = \bar{\chi}_{n,\omega_{1}} \frac{\vec{\hbar}}{2} \operatorname{Tr} \left[(g\mathcal{B}_{n\perp}^{\mu})_{\omega_{3}} (g\mathcal{B}_{\perp}^{\perp n})_{\omega_{4}} \right] \chi_{n,\omega_{2}}$$

$$O_{9} = \left[\bar{\chi}_{n,\omega_{1}}\frac{\vec{n}}{2}\chi_{n,\omega_{2}}\right] \left[\bar{\chi}_{n,\omega_{3}}\frac{\vec{n}}{2}\chi_{n,\omega_{4}}\right], \qquad O_{10} = \left[\bar{\chi}_{n,\omega_{1}}\frac{\vec{n}}{2}\gamma_{5}\chi_{n,\omega_{2}}\right] \left[\bar{\chi}_{n,\omega_{3}}\frac{\vec{n}}{2}\gamma_{5}\chi_{n,\omega_{4}}\right], O_{11} = \left[\bar{\chi}_{n,\omega_{1}}\frac{\vec{n}}{2}\gamma_{5}T^{A}\chi_{n,\omega_{2}}\right] \left[\bar{\chi}_{n,\omega_{3}}\frac{\vec{n}}{2}\gamma_{5}T^{A}\chi_{n,\omega_{4}}\right], \quad O_{12} = \left[\bar{\chi}_{n,\omega_{1}}\frac{\vec{n}}{2}T^{A}\chi_{n,\omega_{2}}\right] \left[\bar{\chi}_{n,\omega_{3}}\frac{\vec{n}}{2}T^{A}\chi_{n,\omega_{4}}\right]$$

Recall that in an operator like O_2 the position space analog of $\mathcal{P}_{\perp}^{\mu}$ is to translate all gluon and quark fields in χ_{n,ω_2} in x_{\perp} , differentiate twice with respect to x_{\perp}^{μ} , and then set $x_{\perp} = 0$. The basis shown in Eq. (3.81) can be used to describe twist-4 effects in DIS at any order in α_s . Note that we have already discussed and taken into account the quark and gluon equations of motion in the general result in Eq. (3.70) and hence already in Eq. (3.81). For $O_{5,7}$ there are two color structures associated with the product of \mathcal{B}_n 's, but these are picked out by consider Wilson coefficients $C_{5,7}$ that are odd or even in the exchange $\omega_3 \leftrightarrow \omega_4$. The forward proton matrix element of these operators will be proportional to an overall δ -function, which is $\delta(\omega_1 - \omega_2)$ for $O_{1,2}$, $\delta(\omega_1 + \omega_3 - \omega_2)$ for $O_{3a,3b,4a,4b}$, $\delta(\omega_1 + \omega_3 + \omega_4 - \omega_2)$ for O_{5-8} , and $\delta(\omega_1 + \omega_3 - \omega_2 - \omega_4)$ for O_{9-12} .

Next we derive the analogous results for the RPI basis of chiral-even operators. From Eq. (3.77) the hadronic tensor operator $\hat{T}_{\mu\nu}$ depends on q^{μ} which we use as our reference vector. To construct this basis we cannot use n^{μ} or \bar{n}^{μ} . Comparing Eqs. (3.77) and Eq. (3.78) we see that it suffices to construct a basis of scalar operators for the expansion of $g^{\mu\nu}\hat{T}_{\mu\nu}$ and $p^{\mu}p^{\nu}\hat{T}_{\mu\nu}$. The forward proton matrix element of the expansion of these operators then yields an expansion for the observables T_1 and T_2 . Thus, for the scalar basis we allow any number of q's to appear, but only zero or two p's. This implies that at twist-2 there is only one RPI bilinear quark operator

$$\mathbf{Q}_{1} = \bar{\Psi}_{n,\hat{\omega}_{1}} \not \!\!\! / \, \Psi_{n,\hat{\omega}_{2}} \,. \tag{3.82}$$

At twist-3 there are no scalar chiral-even RPI operators. The candidate operators $\bar{\Psi}_n i \partial_n \Psi_n$ and $\bar{\Psi}_{n,\hat{\omega}_1} \not/(q \cdot i\partial_n) \Psi_{n,\hat{\omega}_2}$ are ruled out by the equations of motion in Eqs. (3.35) and (3.36). Another possible operator is $\bar{\Psi}_n \not/(p \cdot \partial_n) \Psi_n$ but it starts at twist-6, since $(p \cdot \partial_n) \sim O(\lambda^2)$, being suppressed either by an $n \cdot p$ or $n \cdot \partial_n$, and \not/p adds another factor 2 to the power counting when it is squeezed between the *n*-collinear fermion fields χ_n . All the operators with $\mathcal{G}_n^{\mu\nu}$, like for example $\bar{\Psi}_n \gamma_\mu q_\nu \mathcal{G}_n^{\mu\nu} \Psi_n$, have expansions whose lowest term is twist-4 because the Dirac structure of the twist-3 component of this operator vanishes between the *n*-collinear fermion fields, since $\bar{\chi}_n \not/$ $\chi_n = 0$. Thus the power suppressed terms start at twist-4 in agreement with the homogeneous basis in Eq. (3.81). Writing out the RPI operators different from zero at twist-4 and not connected by operator relations we have

$$\begin{aligned} \mathbf{Q}_{2} &= \bar{\Psi}_{n,\hat{\omega}_{1}} \gamma_{\mu} q_{\nu} \mathcal{G}_{n,\hat{\omega}_{3}}^{\mu\nu} \Psi_{n,\hat{\omega}_{2}} , & \mathbf{Q}_{3} &= \bar{\Psi}_{n,\hat{\omega}_{1}} \not{\!\!\!\!/} \gamma_{\mu} \gamma_{\nu} ig \mathcal{G}_{n,\hat{\omega}_{3}}^{\mu\nu} \Psi_{n,\hat{\omega}_{2}} , & (3.83) \end{aligned}$$
$$\begin{aligned} \mathbf{Q}_{4} &= -g^{2} \bar{\Psi}_{n,\hat{\omega}_{1}} \not{\!\!\!\!/} \gamma_{\mu} q_{\nu} \mathcal{G}_{n,\hat{\omega}_{3}}^{\mu\nu} \gamma_{\alpha} q_{\beta} \mathcal{G}_{n,\hat{\omega}_{4}}^{\alpha\beta} \Psi_{n,\hat{\omega}_{2}} , & \mathbf{Q}_{5} &= -g^{2} \bar{\Psi}_{n,\hat{\omega}_{1}} \not{\!\!\!/} \gamma_{\mu} q_{\nu} \gamma_{\alpha} q_{\beta} \mathrm{Tr}[\mathcal{G}_{n,\hat{\omega}_{3}}^{\mu\nu} \mathcal{G}_{n,\hat{\omega}_{4}}^{\alpha\beta}] \Psi_{n,\hat{\omega}_{2}} \\ \mathbf{Q}_{6} &= -g^{2} \bar{\Psi}_{n,\hat{\omega}_{1}} \not{\!\!/} q_{\mu} q_{\nu} (\mathcal{G}_{n,\hat{\omega}_{3}})^{\mu}_{\alpha} \mathcal{G}_{n,\hat{\omega}_{4}}^{\alpha\nu} \Psi_{n,\hat{\omega}_{2}} , & \mathbf{Q}_{7} &= -g^{2} \bar{\Psi}_{n,\hat{\omega}_{1}} \not{\!\!/} q_{\mu} q_{\nu} \mathrm{Tr}[(\mathcal{G}_{n,\hat{\omega}_{3}})^{\mu}_{\alpha} \mathcal{G}_{n,\hat{\omega}_{4}}^{\alpha\nu}] \Psi_{n,\hat{\omega}_{2}} , \\ \mathbf{Q}_{8} &= \left[\bar{\Psi}_{n,\hat{\omega}_{1}} \not{\!\!/} \Psi_{n,\hat{\omega}_{2}} \right] \left[\bar{\Psi}_{n,\hat{\omega}_{3}} \not{\!/} \Psi_{n,\hat{\omega}_{4}} \right] , & \mathbf{Q}_{9} &= \left[\bar{\Psi}_{n,\hat{\omega}_{1}} \not{\!/} \gamma_{5} \Psi_{n,\hat{\omega}_{2}} \right] \left[\bar{\Psi}_{n,\hat{\omega}_{3}} \not{\!/} T^{A} \Psi_{s,\hat{\omega}_{4}} \right] , \\ \mathbf{Q}_{10} &= \left[\bar{\Psi}_{n,\hat{\omega}_{1}} \not{\!/} T^{A} \Psi_{n,\hat{\omega}_{2}} \right] \left[\bar{\Psi}_{n,\hat{\omega}_{3}} \not{\!/} T^{A} \Psi_{n,\hat{\omega}_{4}} \right] , & \mathbf{Q}_{11} &= \left[\bar{\Psi}_{n,\hat{\omega}_{1}} \not{\!/} T^{A} \gamma_{5} \Psi_{n,\hat{\omega}_{2}} \right] \left[\bar{\Psi}_{n,\hat{\omega}_{3}} \not{\!/} T^{A} \gamma_{5} \Psi_{n,\hat{\omega}_{4}} \right] . \end{aligned}$$

One can think of other possible operators at twist-4, but all of them are either ruled out by the equations of motion and operator relations, or start at higher twist. For example, there are not operators with both p^{μ} and $\mathcal{G}_{\hat{\omega}_3}^{\mu\nu}$, like $\bar{\Psi}_{n,\hat{\omega}_1}\gamma_{\mu}p_{\nu}\mathcal{G}_{n,\hat{\omega}_3}^{\mu\nu}\Psi_{n,\hat{\omega}_2}$, because they all start at higher twist. We have integrated by parts making all derivatives act to the right, since here our interest is in forward matrix elements, and we removed $i\partial_{\nu}\mathcal{G}_{n,\hat{\omega}}^{\mu\nu}$ with the gluon equation of motion in Eq. (3.37). The operator $\bar{\Psi}_{n,\hat{\omega}_1}\notin$ $(i\partial_n \cdot i\partial_n)\Psi_{n,\hat{\omega}_2} = \bar{\Psi}_{n,\hat{\omega}_1}\notin i\partial_n \Psi_{n,\hat{\omega}_2}$, and is removed by the quark equation of motion in Eq. (3.35). For the operators with two $\mathcal{G}'s$ only the structures in \mathbf{Q}_{4-7} have expansions that start at twist-4. For example, $\mathcal{G}_{n,\hat{\omega}_3}^{\mu\alpha}\mathcal{G}_{n,\hat{\omega}_4\alpha\nu}$ at LO is proportional to $(\mathcal{B}_{\perp}^{\alpha})_{\omega_3}(\mathcal{B}_{\alpha}^{\perp})_{\omega_4}n^{\mu}n^{\nu}$ so closing the indexes with γ^{μ} or γ^{ν} generates a \not{m} that next to χ_n gives zero.

It is less obvious that operators with one \mathcal{G}_n and one $i\partial_n$ are redundant and can be eliminated from the RPI basis. Consider the operator

To remove it we use a manipulation discussed by Jaffe and Soldate in Ref. [63]. First we write

$$\mathcal{G}_{n}^{\mu\nu}i\partial_{\nu}^{n} = \mathcal{G}_{n}^{\mu\nu}i\hat{\mathcal{D}}_{\nu}^{n} - \mathcal{G}_{n}^{\mu\nu}[1/(q\cdot i\partial_{n})q_{\alpha}ig\mathcal{G}_{n\nu}^{\alpha}], \qquad (3.85)$$

and note that the term with two \mathcal{G}_n 's can be ignored since it is already in our basis. Next using the definition (3.20) we can write

$$q_{\mu}ig\mathcal{G}_{n}^{\mu\nu}i\hat{\mathcal{D}}_{\nu}^{n} = q^{\mu}[i\hat{\mathcal{D}}_{\mu}^{n},i\hat{\mathcal{D}}_{n}^{\nu}]i\hat{\mathcal{D}}_{\nu}^{n} = \frac{1}{2}\left\{-q^{\mu}[i\hat{\mathcal{D}}_{\nu}^{n},[i\hat{\mathcal{D}}_{\mu}^{n},i\hat{\mathcal{D}}_{n}^{\nu}]] - (i\hat{\mathcal{D}}_{n})^{2}iq\cdot\partial_{n} + iq\cdot\partial_{n}(i\hat{\mathcal{D}}_{n})^{2}\right\}.$$
(3.86)

The double commutator term is turned into a four-quark operator by the gluon equations of motion in Eq. (3.37). For the remaining terms we write $(i\hat{\mathcal{D}}_n)^2 = i\hat{\mathcal{P}}_n$ / $i\hat{\mathcal{D}}_n + \frac{i}{2}\sigma_{\mu\nu}\mathcal{G}_n^{\mu\nu}$, where the $\sigma_{\mu\nu}$ term gives \mathbf{Q}_3 , and terms involving $(i\hat{\mathcal{P}}_n)^2$ are turned into the operators \mathbf{Q}_2 , \mathbf{Q}_3 , \mathbf{Q}_4 , and \mathbf{Q}_6 by the quark equation of motion in Eq. (3.35). (They are not simply set to zero, since $[\hat{\mathcal{P}}_n]$ does not commute with $\delta(\hat{\omega}-2q\cdot i\partial_n)$.) Finally, we can also rule out the only other non-trivial operator $\bar{\Psi}_{n,\hat{\omega}_1} \not\equiv i \partial_n \gamma_{\mu} q_{\nu} \mathcal{G}_{n,\hat{\omega}_3}^{\mu\nu} \Psi_{n,\hat{\omega}_2}$. Using the gluon equation of motion we write

where the ellipsis denotes operators with two \mathcal{G}_n 's or four-quark fields that are part of the basis. The Bianchi identity in Eq. (3.38) gives another relation for the two operators on the LHS of Eq. (3.87) and implies that they can be written in terms of \mathbf{Q}_3 , \mathbf{Q}_* , \mathbf{Q}_4 , and \mathbf{Q}_6 . Thus both the operators $\bar{\Psi}_{n,\hat{\omega}_1} \not q_i \partial_n \gamma_\mu q_\nu \mathcal{G}_{n,\hat{\omega}_3}^{\mu\nu} \Psi_{n,\hat{\omega}_2}$ and $\bar{\Psi}_{n,\hat{\omega}_1} \not q_i \partial_n \mathcal{G}_{n,\hat{\omega}_3}^{\mu\nu} \Psi_{n,\hat{\omega}_2}$ are redundant. Finally we note that the order of the \mathcal{G}_n 's in an operator like \mathbf{Q}_6 is not important, since we can always symmetrize or antisymmetrize its Wilson coefficient in $\hat{\omega}_3$ and $\hat{\omega}_4$. Note that when considering the transformation of the operators under charge conjugation one must consider both the operator and its Wilson coefficient. We discuss an example below in Eq. (3.94).

The number of independent RPI operators in Eq. (3.81) is smaller than in the basis

of homogeneous operators in Eq. (3.83), implying that there exist further constraints on the Wilson coefficients of the homogeneous basis at twist-4. To find the constraints we must expand the operators in Eq. (3.83) in terms of those in Eq. (3.81). We start with \mathbf{Q}_4 through \mathbf{Q}_{11} which are in one-to-one correspondence with operators in the homogeneous basis,

$$\mathbf{Q}_{4} = \frac{\omega_{3}\omega_{4}}{4(n \cdot q)} O_{5}, \qquad \mathbf{Q}_{5} = \frac{\omega_{3}\omega_{4}}{4(n \cdot q)} O_{6}, \qquad \mathbf{Q}_{6} = -\frac{\omega_{3}\omega_{4}}{4(n \cdot q)} O_{7},
\mathbf{Q}_{7} = \frac{-\omega_{3}\omega_{4}}{4(n \cdot q)} O_{8}, \qquad \mathbf{Q}_{8} = \frac{1}{(n \cdot q)^{2}} O_{9}, \qquad \mathbf{Q}_{9} = \frac{1}{(n \cdot q)^{2}} O_{10},
\mathbf{Q}_{10} = \frac{1}{(n \cdot q)^{2}} O_{12}, \qquad \mathbf{Q}_{11} = \frac{1}{(n \cdot q)^{2}} O_{11}. \qquad (3.88)$$

Here the order of the $\hat{\omega}_i$ subscripts in operators on the left exactly matches up with the ω_i subscripts on the right. For the remaining operators whose expansions start at twist-4 and for Q_1 that starts at twist-2, we have

$$\begin{aligned} \mathbf{Q}_{2}(\hat{\omega}_{1},\hat{\omega}_{3},\hat{\omega}_{2}) &= \frac{1}{(n\cdot q)^{2}} \left[\frac{\omega_{3}}{2\omega_{2}} O_{4a}(\omega_{1},\omega_{3},\omega_{2}) + \frac{\omega_{3}}{2\omega_{1}} O_{4b}(\omega_{1},\omega_{3},\omega_{2}) \right] + \dots, \end{aligned}$$
(3.89)
$$& - O_{3a}(\omega_{1},\omega_{3},\omega_{2}) + O_{3b}(\omega_{1},\omega_{3},\omega_{2}) \right] + \dots, \end{aligned} \\ \mathbf{Q}_{3}(\hat{\omega}_{1},\hat{\omega}_{3},\hat{\omega}_{2}) &= \frac{2}{(n\cdot q)^{2}} \left[\frac{\omega_{1}}{\omega_{2}} O_{4a}(\omega_{1},\omega_{3},\omega_{2}) + O_{4b}(\omega_{1},\omega_{3},\omega_{2}) - 2O_{3b}(\omega_{1},\omega_{3},\omega_{2}) \right] + \dots, \end{aligned} \\ \mathbf{Q}_{1}(\hat{\omega}_{1},\hat{\omega}_{2}) &= \frac{1}{n\cdot q} O_{1}(\omega_{1},\omega_{2}) + \frac{\bar{n}\cdot q}{(n\cdot q)^{2}} \left[\left\{ \frac{-1}{\omega_{1}\omega_{2}} + \frac{d}{d\omega_{1}} \frac{1}{\omega_{1}} + \frac{d}{d\omega_{2}} \frac{1}{\omega_{2}} \right\} O_{2}(\omega_{1},\omega_{2}) \right] \\ &+ \left\{ \frac{2}{(\omega_{a}-\omega_{2})^{2}} - \frac{d}{d\omega_{2}} \frac{2}{\omega_{a}-\omega_{2}} \right\} \left\{ O_{3a}(\omega_{1},\omega_{a}-\omega_{2},\omega_{a}) - O_{3b}(\omega_{1},\omega_{a}-\omega_{2},\omega_{a}) \right\} \\ &+ \left\{ \frac{-2}{(\omega_{1}-\omega_{a})^{2}} - \frac{d}{d\omega_{1}} \frac{2}{\omega_{1}-\omega_{a}} \right\} \left\{ O_{3a}(\omega_{a},\omega_{1}-\omega_{a},\omega_{2}) - O_{3b}(\omega_{a},\omega_{1}-\omega_{a},\omega_{2}) \right\} \\ &+ \left\{ \frac{-1}{\omega_{1}\omega_{2}} + \frac{d}{d\omega_{1}} \frac{1}{\omega_{1}} \right\} O_{4a}(\omega_{a},\omega_{1}-\omega_{a},\omega_{2}) + \frac{d}{d\omega_{2}} \frac{1}{\omega_{a}} O_{4a}(\omega_{1},\omega_{a}-\omega_{2},\omega_{a}) \right\} \\ &+ \left\{ \frac{-1}{\omega_{1}\omega_{2}} + \frac{d}{d\omega_{2}} \frac{1}{\omega_{2}} \right\} O_{4b}(\omega_{1},\omega_{a}-\omega_{2},\omega_{a}) + \frac{d}{d\omega_{1}} \frac{1}{\omega_{a}} O_{4b}(\omega_{a},\omega_{1}-\omega_{a},\omega_{2}) \right\} \\ &+ \cdots . \end{aligned}$$

Here the ellipses indicate terms involving operators \mathcal{O}_{5-12} that have already occurred

in \mathbf{Q}_{4-11} and hence they are no longer important for determining the linear independent combinations. It is interesting to note that expanding the operator \mathbf{Q}_* gives the same combination of O_{3a} and O_{3b} that appears in $\mathbf{Q}_2 - 4\omega_1/\omega_3\mathbf{Q}_3$, so even if we had not eliminated \mathbf{Q}_* from the RPI basis, the implications for the homogeneous basis would be the same.

The three RPI operators in Eq. (3.89) have expansions in terms of six homogeneous operators O_1 , O_2 , O_{3a} , O_{3b} , O_{4a} , and O_{4b} , so there are three RPI relations. The Wilson coefficients of these six homogeneous operators are determined by three coefficients, $\hat{C}_{1,2,3}$ in the RPI basis. It is convenient to trade $\hat{C}_{1,2,3}$ for the three coefficients C_1 , C_{3a} , and C_{3b} . The remaining coefficients C_2 , C_{4a} , and C_{4b} are then determined by RPI. We find

$$C_{2}(\omega_{1},\omega_{2}) = -\frac{\bar{n}\cdot q}{n\cdot q} \Big\{ \frac{1}{\omega_{1}\omega_{2}} + \frac{1}{\omega_{1}} \frac{d}{d\omega_{1}} + \frac{1}{\omega_{2}} \frac{d}{d\omega_{2}} \Big\} C_{1}(\omega_{1},\omega_{2}),$$

$$C_{4a}(\omega_{1},\omega_{3},\omega_{2}) = -\frac{1}{2} C_{3a}(\omega_{1},\omega_{3},\omega_{2}) - \frac{\omega_{1}}{2\omega_{2}} C_{3b}(\omega_{1},\omega_{3},\omega_{2}) + \frac{\bar{n}\cdot q}{n\cdot q} \frac{\bar{\omega}_{2}\omega_{3}}{\omega_{2}\omega_{3}} C_{1}(\omega_{1},\omega_{2}-\omega_{3}) - \frac{\bar{n}\cdot q}{n\cdot q} \frac{(\omega_{2}+\omega_{3})}{\omega_{3}(\omega_{2})^{2}} C_{1}(\omega_{1}+\omega_{3},\omega_{2}),$$

$$C_{4b}(\omega_{1},\omega_{3},\omega_{2}) = -\frac{\omega_{2}}{2\omega_{1}} C_{3a}(\omega_{1},\omega_{3},\omega_{2}) - \frac{1}{2} C_{3b}(\omega_{1},\omega_{3},\omega_{2}) + \frac{\bar{n}\cdot q}{n\cdot q} \frac{(\omega_{1}-\omega_{3})}{\omega_{3}(\omega_{1})^{2}} C_{1}(\omega_{1},\omega_{2}-\omega_{3}) - \frac{\bar{n}\cdot q}{n\cdot q} \frac{\bar{n}\cdot q}{\omega_{1}\omega_{3}} C_{1}(\omega_{1}+\omega_{3},\omega_{2}).$$

$$(3.90)$$

We have cross-checked the relation for C_2 with a tree-level matching computation. Note that $C_2(\omega_1, \omega_2)$ multiplies a matrix element that gives $\delta(\omega_1 - \omega_2)$, while $C_{4a,4b}(\omega_1, \omega_3, \omega_2)$ multiplies a $\delta(\omega_1 + \omega_3 - \omega_2)$, and that we have used these δ -functions at various intermediate steps. That is, the result in Eq. (3.90) applies for a basis of operators, whose matrix elements have vanishing total derivatives.

Our operator bases can be compared to the flavor singlet and parity even basis of Jaffe and Soldate in Ref. [63] which has one operator at twist-2, and 12 operators at twist-4.³ There is a simple correspondence between the 11 operators in our RPI

³The notation in Eq. (3.81) suggests that all quark bilinears are flavor singlet contractions if χ_n has multiple flavor components. To incorporate other possibilities for the flavor indices is straightforward [63]. We consider χ_n as a doublet of SU(2) flavor, or a triplet of SU(3) flavor, with elements

basis in Eq. (3.82,3.83) and the QCD operators in their basis. The correspondence is one-to-one for \mathbf{Q}_1 , the four-quark operators \mathbf{Q}_{8-11} , and the operators $\mathbf{Q}_{2,3}$ that have one $\mathcal{G}_n^{\mu\nu}$. For the operators with two $\mathcal{G}_n^{\mu\nu}$'s we have four operators compared to their six, but the difference is accounted for by the way in which the twist towers are enumerated. We used continuous $\hat{\omega}_i$'s where even and odd symmetry under the interchange $\hat{\omega}_3 \leftrightarrow \hat{\omega}_4$ encodes two possible color structures with f^{ABC} and d^{ABC} , while Ref. [63] uses a discrete basis with integer powers of $(i\bar{n} \cdot D_n)$, where the choice of which operators to eliminate by integration by parts implies that the two color structures yield different operators. Our homogeneous basis has 14 operators up to twist-4, and most closely corresponds to an enumeration of an operator basis in terms of the so-called "good" quark and gluon fields. The good quark and gluon fields have been discussed in Refs. [7, 64, 70]. In this basis the power counting is manifest. From the three RPI relations in Eq. (3.90) the number of independent short distance Wilson coefficients is 11, and so encodes the same amount of information as the OPE basis from Ref. [63]. Note that there is no room in the traditional OPE in DIS for a correspondence with higher order operators with soft fields. In our language, the validity of the OPE for DIS with generic x implies that soft degrees of freedom are not needed, and one can consider that fluctuations from that region are reabsorbed into the collinear fields.

When the basis of bilinear quark operators is considered in the forward proton matrix element it can be reduced even further as discussed in detail in Ref. [42]. In this process it is found that the matrix elements of operators like O_2 , O_{4a} , and O_{4b} do not provide independent information. Hence at this level the RPI relations in Eq. (3.90) do not appear to have practical implications.

 $[\]chi_n^f$. For photon currents one has a charge matrix in flavor space in each QCD current, which is $\hat{Q} = \text{diag}(2/3, -1/3, -1/3)$ for SU(3). Thus, at leading order in the electromagnetic interactions one must simply introduce a \hat{Q}^2 in all bilinear-quark operators, O^1 through O^8 in Eq. (3.81). When counting the four-quark operators O^9 to O^{12} induced by photons we double the number of operators because there are two possibilities, $\hat{Q}^2 \otimes 1$ and $\hat{Q} \otimes \hat{Q}$. In this notation the flavor singlet contraction for the four-quark operators is $1 \otimes 1$. For the RPI basis of operators the analysis of flavor structures is identical, and hence flavor does not modify the constraints in Eq. (3.90).
3.7.4 Deep Inelastic Scattering for Gluons at Twist-4

Next let us consider the minimal basis for pure gluon DIS operators up to twist-4. We proceed in a similar manner to our construction for quarks, first writing the homogeneous basis and then the RPI basis to check if reparametrization invariance provides constraints on the homogeneous operators. The homogeneous basis is

$$O_{1} = \operatorname{Tr}\left[(g\mathcal{B}_{n\perp})_{\omega_{1}} \cdot (g\mathcal{B}_{n\perp})_{\omega_{2}}\right], \qquad (3.91)$$

$$O_{2} = \operatorname{Tr}\left[(g\mathcal{B}_{n\perp}^{\mu})_{\omega_{1}}\mathcal{P}_{\perp}^{2}(g\mathcal{B}_{n\mu}^{\perp})_{\omega_{2}}\right], \qquad (3.91)$$

$$O_{3,4} = \operatorname{Tr}\left[(g\mathcal{B}_{n\perp}^{\mu})_{\omega_{1}}(g\mathcal{B}_{n\perp}^{\nu})_{\omega_{2}}\mathcal{P}_{\perp}^{\alpha}(g\mathcal{B}_{n\perp}^{\beta})_{\omega_{3}}\right]\Gamma_{\mu\nu\alpha\beta}^{1,2}, \qquad (3.91)$$

$$O_{5,6} = \operatorname{Tr}\left[(g\mathcal{B}_{n\perp}^{\mu})_{\omega_{1}}(g\mathcal{B}_{n\perp}^{\nu})_{\omega_{2}}(g\mathcal{B}_{n\perp}^{\alpha})_{\omega_{3}}(g\mathcal{B}_{n\perp}^{\beta})_{\omega_{4}}\right]\Gamma_{\mu\nu\alpha\beta}^{1,2}, \qquad (3.91)$$

$$O_{7,8} = \operatorname{Tr}\left[(g\mathcal{B}_{n\perp}^{\mu})_{\omega_{1}}(g\mathcal{B}_{n\perp}^{\nu})_{\omega_{2}}\right]\operatorname{Tr}\left[(g\mathcal{B}_{n\perp}^{\alpha})_{\omega_{3}}(g\mathcal{B}_{n\perp}^{\beta})_{\omega_{4}}\right]\Gamma_{\mu\nu\alpha\beta}^{1,2}, \qquad (3.91)$$

$$O_{9} = \operatorname{Tr}\left[(g\mathcal{B}_{n\perp}^{\mu})_{\omega_{1}}\mathcal{P}_{\mu}^{\perp}\mathcal{P}_{\perp}^{\nu}(g\mathcal{B}_{n\nu}^{\perp})_{\omega_{2}}\right], \qquad (3.91)$$

where $\Gamma^{1,2}_{\mu\nu\alpha\beta} = \{g_{\mu\nu}g_{\alpha\beta}, g_{\mu\alpha}g_{\nu\beta}\}$ and the traces are over color. Recall that the equations of motion (3.32) were used to eliminate the operators $gn \cdot \mathcal{B}_n$ and $in \cdot \partial_n (g\mathcal{B}^{\mu}_{n\perp})$. Again since the basis is designed for taking forward matrix elements we are free to integrate by parts and hence we do not consider $\mathcal{P}^{\mu\dagger}_{\perp}$. There is a third tensor structure, $\Gamma^3_{\mu\nu\alpha\beta} = g_{\mu\beta}g_{\alpha\nu}$, that can also be considered for O_{3-8} , but which can always be eliminated. For $O_{3,4}$ this is done using integration by parts and the cyclic trace, giving

$$\operatorname{Tr}\left[(g\mathcal{B}_{n\perp}^{\mu})_{\omega_{1}}(g\mathcal{B}_{n\perp}^{\nu})_{\omega_{2}}\mathcal{P}_{\perp}^{\alpha}(g\mathcal{B}_{n\perp}^{\beta})_{\omega_{3}}\right]\Gamma_{\mu\nu\alpha\beta}^{3} = -O_{4}(\omega_{2},\omega_{3},\omega_{1}) - O_{3}(\omega_{3},\omega_{1},\omega_{2}). \quad (3.92)$$

For O_{5-8} the cyclic property of the trace suffices to eliminate $\Gamma^3_{\mu\nu\alpha\beta}$ in an analogous manner. The operator $\text{Tr}[(g\mathcal{B}^{\mu}_{n\perp})_{\omega_1}\mathcal{P}^{\alpha}_{\perp}(g\mathcal{B}^{\nu}_{n\perp})_{\omega_2}(g\mathcal{B}^{\beta}_{n\perp})_{\omega_3}]$ is also not needed in the basis because it can be put into the form of the operators O_3 and O_4 . This is done by acting with the $\mathcal{P}^{\alpha}_{\perp}$ on the two \mathcal{B}_{\perp} 's to the right, using the cyclic property of the trace, and again noting that $O_{3,4}$ encode all orderings for the ω_i subscripts.

For forward spin averaged matrix elements the RPI basis of gluon operators up

to twist-4 is

$$\begin{aligned} \mathbf{Q}_{1} &= q_{\nu}q_{\beta}\mathrm{Tr}\left[ig\mathcal{G}_{n,\hat{\omega}_{1}}^{\mu\nu}ig\mathcal{G}_{n,\hat{\omega}_{2}}^{\alpha\beta}\right]g_{\mu\alpha}\,, \end{aligned} \tag{3.93} \\ \mathbf{Q}_{2} &= \mathrm{Tr}\left[ig\mathcal{G}_{n,\hat{\omega}_{1}}^{\mu\nu}ig\mathcal{G}_{n,\hat{\omega}_{2}}^{\alpha\beta}\right]g_{\mu\alpha}g_{\nu\beta}\,, \\ \mathbf{Q}_{3} &= q_{\nu}q_{\sigma}\mathrm{Tr}\left[ig\mathcal{G}_{n,\hat{\omega}_{1}}^{\mu\nu}ig\mathcal{G}_{n,\hat{\omega}_{2}}^{\alpha\beta}ig\mathcal{G}_{n,\hat{\omega}_{3}}^{\rho\sigma}\right]g_{\mu\alpha}g_{\beta\rho}\,, \\ \mathbf{Q}_{4} &= q_{\nu}q_{\beta}q_{\sigma}\mathrm{Tr}\left[ig\mathcal{G}_{n,\hat{\omega}_{1}}^{\mu\nu}ig\mathcal{G}_{n,\hat{\omega}_{2}}^{\alpha\beta}i\partial_{\mu}ig\mathcal{G}_{n,\hat{\omega}_{3}}^{\rho\sigma}\right]g_{\alpha\rho}\,, \\ \mathbf{Q}_{5,6} &= q_{\nu}q_{\beta}q_{\sigma}q_{\lambda}\mathrm{Tr}\left[ig\mathcal{G}_{n,\hat{\omega}_{1}}^{\mu\nu}ig\mathcal{G}_{n,\hat{\omega}_{2}}^{\alpha\beta}ig\mathcal{G}_{n,\hat{\omega}_{3}}^{\rho\sigma}ig\mathcal{G}_{n,\hat{\omega}_{4}}^{\tau\lambda}\right]\Gamma_{\mu\alpha\rho\tau}^{1,2}\,, \\ \mathbf{Q}_{7,8} &= q_{\nu}q_{\beta}q_{\sigma}q_{\lambda}\mathrm{Tr}\left[ig\mathcal{G}_{n,\hat{\omega}_{1}}^{\mu\nu}ig\mathcal{G}_{n,\hat{\omega}_{2}}^{\alpha\beta}\right]\mathrm{Tr}\left[ig\mathcal{G}_{n,\hat{\omega}_{3}}^{\rho\sigma}ig\mathcal{G}_{n,\hat{\omega}_{4}}^{\tau\lambda}\right]\Gamma_{\mu\alpha\rho\tau}^{1,2}\,. \end{aligned}$$

Here we remove a possible operator $q_{\nu}q_{\beta} \operatorname{Tr}\left[(ig\mathcal{G}_{n,\hat{\omega}_{1}})^{\nu}{}_{\alpha}(i\partial)^{2}ig\mathcal{G}_{n,\hat{\omega}_{2}}^{\alpha\beta}\right]$ by writing $(i\partial)^{2}\mathcal{G}_{n,\hat{\omega}_{2}}^{\alpha\beta} = (i\partial_{\mu})i\partial^{\mu}\mathcal{G}_{n,\hat{\omega}_{2}}^{\alpha\beta}$, and then using the Bianchi identity in Eq. (3.38) to rewrite this operator in terms of operator with two \mathcal{G}_{n} 's, plus $(i\partial_{\mu})i\partial^{\alpha}\mathcal{G}_{n,\hat{\omega}_{2}}^{\beta\mu}$ and $(i\partial_{\mu})i\partial^{\beta}\mathcal{G}_{n,\hat{\omega}_{2}}^{\mu\alpha}$. The last two terms are removed by the gluon equation of motion. There is no need to include the analog of \mathbf{Q}_{4} with the $i\partial_{\mu}$ acting on $ig\mathcal{G}_{n,\hat{\omega}_{2}}^{\alpha\beta}$, because it is related to \mathbf{Q}_{4} by integration by parts up to a term, $i\partial_{\mu}ig\mathcal{G}_{n,\hat{\omega}_{1}}^{\mu\nu}$ that reduces to other operators through the gluon equation of motion. Again the cyclic nature of the trace allows one to remove $\Gamma_{\mu\alpha\rho\tau}^{3}$ for \mathbf{Q}_{5-8} .

In order to consider the effect of charge conjugation on these basis one must consider the transformation of

$$\int [d\hat{\omega}_j] \hat{C}_i(\hat{\omega}_j) \mathbf{Q}_i(\hat{\omega}_j), \quad \text{or} \quad \int [d\omega_j] C_i(\omega_j) O_i(\omega_j), \quad (3.94)$$

where \hat{C}_i is the Wilson coefficient associated with \mathbf{Q}_i , and C_i the Wilson coefficient associated with O_i . We can impose constraints on $\hat{C}_i(\hat{\omega}_j)$ and $C_i(\omega_j)$ such that (3.94) is *C*-invariant. For example, note that under charge conjugation \mathbf{Q}_3 transforms into

$$-q_{\nu}q_{\sigma}\mathrm{Tr}\left[ig\mathcal{G}_{n,\hat{\omega}_{3}}^{\mu\nu}ig\mathcal{G}_{n,\hat{\omega}_{2}}^{\alpha\beta}ig\mathcal{G}_{n,\hat{\omega}_{1}}^{\rho\sigma}\right]g_{\mu\alpha}g_{\beta\rho},\qquad(3.95)$$

so to make it C-invariant we impose that $\hat{C}_3(\hat{\omega}_1,\hat{\omega}_2,\hat{\omega}_3) = -\hat{C}_3(\hat{\omega}_3,\hat{\omega}_2,\hat{\omega}_1)$. Simi-

lar considerations apply to the homogeneous basis. For example, the combinations $O_3(\omega_1, \omega_2, \omega_3) - O_3(\omega_2, \omega_1, \omega_3)$ and $O_4(\omega_1, \omega_2, \omega_3) + O_4(\omega_1, \omega_3, \omega_2) + O_3(\omega_3, \omega_2, \omega_1)$ are even under charge conjugation.

Next we must expand the RPI basis in Eq. (3.93) in terms of the homogeneous basis in Eq. (3.91) to find possible constraints. We first expand \mathbf{Q}_{5-8} , they have only operators with four $g\mathcal{B}^{\mu}_{\perp}$'s, that is O_{5-8} ,

$$\mathbf{Q}_{5,6} = \frac{\omega_1 \omega_2 \omega_3 \omega_4}{16} O_{5,6} , \qquad \qquad \mathbf{Q}_{7,8} = \frac{\omega_1 \omega_2 \omega_3 \omega_4}{16} O_{7,8} . \qquad (3.96)$$

Next we expand $\mathbf{Q}_{3,4}$ to find

$$\mathbf{Q}_{3} = \frac{\omega_{1}\omega_{3}}{4(n \cdot q)} \Big[-O_{4}(\omega_{1}, \omega_{2}, \omega_{3}) - O_{4}(\omega_{3}, \omega_{1}, \omega_{2}) - O_{3}(\omega_{2}, \omega_{3}, \omega_{1}) \Big] + \cdots ,$$

$$\mathbf{Q}_{4} = \frac{\omega_{1}\omega_{2}\omega_{3}}{8} \Big[O_{4}(\omega_{1}, \omega_{2}, \omega_{3}) - \frac{\omega_{3}}{\omega_{1}} O_{3}(\omega_{2}, \omega_{3}, \omega_{1}) \Big] + \cdots ,$$
(3.97)

where we integrate over the repeated ω_a variable. The ellipses in Eq. (3.97) indicate terms involving operators O_{5-8} that have already occurred in \mathbf{Q}_{5-8} and hence are no longer important for determining the linear independent combinations. Eq. (3.97) implies that O_3 and O_4 have Wilson coefficients that are independent of other operators in the basis. When we expand the remaining RPI operators $\mathbf{Q}_{1,2}$, we may also have terms with $O_{1,2,9}$ which have two $g\mathcal{B}^{\mu}_{\perp}$'s. We find

$$\mathbf{Q}_{1}(\hat{\omega}_{1},\hat{\omega}_{2}) = \frac{\omega_{1}\omega_{2}}{4}O^{1}(\omega_{1},\omega_{2}) + \frac{(\bar{n}\cdot q)}{4(n\cdot q)} \left(2 - \frac{d}{d\omega_{1}}\omega_{1} - \frac{d}{d\omega_{2}}\omega_{2}\right)O^{2}(\omega_{1},\omega_{2}) + \dots,$$

$$\mathbf{Q}_{2}(\hat{\omega}_{1},\hat{\omega}_{2}) = \dots,$$
 (3.98)

where the ellipsis indicates terms involving operators O_{3-8} that have already occurred in \mathbf{Q}_{3-8} . The fact that O^9 does not occur in the expansion of any of the RPI operators indicates that it is ruled out by RPI (explaining why we listed it last in the basis). Furthermore, the operators O^1 and O^2 only enter in the combination obtained from expanding \mathbf{Q}^1 , and so their Wilson coefficients are related by

$$C_{2}(\omega_{1},\omega_{2}) = \frac{\bar{n} \cdot q}{n \cdot q} \left(2 + \omega_{1} \frac{d}{d\omega_{1}} + \omega_{2} \frac{d}{d\omega_{2}} \right) \frac{C_{1}(\omega_{1},\omega_{2})}{\omega_{1}\omega_{2}}$$
$$= \frac{\bar{n} \cdot q}{\omega_{1}\omega_{2} n \cdot q} \left(\omega_{1} \frac{d}{d\omega_{1}} + \omega_{2} \frac{d}{d\omega_{2}} \right) C_{1}(\omega_{1},\omega_{2}) .$$
(3.99)

For the gluon DIS operators the RPI relations are similar to that for the quark basis, namely it is the collinear operators with \mathcal{P}_{\perp} 's that are constrained. This was also observed in Ref. [4] for the heavy-to-light currents at second order in the power counting. Overall there are eight homogeneous operators for spin-averaged gluon DIS up to twist-4, and seven independent Wilson coefficients.

An analysis of twist-4 gluon matrix elements was done in Ref. [6] using leadingorder Feynman diagram, based on the methods of Ref. [42]. To the best of our knowledge, the complete linear independent bases of twist-4 pure glue operators given in Eq. (3.91) and (3.93) have not been given earlier in the literature.

3.7.5 Two Jet production: n-n' operators

An important application for operators with two-collinear directions, n-n', is the study of two jet phenomena and event shapes. The effective theory SCET has been used to study jets at leading order in the power expansion and various orders in the α_s expansion in Refs. [12, 13, 22, 47, 48, 59, 65, 71, 72, 90, 96]. Another interesting application is to describing parton showers with SCET [16, 17], where both leading and subleading operators with two-collinear directions play some role. In this Section we study the leading and first power suppressed quark operators with two-collinear directions. For two jet processes it is convenient to use the center-of-momentum (CM) frame where the two jets are back to back. In this frame we can take $n' = \bar{n}$ so that $n' \cdot n = 2$. Our main interest will be in the operators that do not vanish in this frame, however part of our discussion touches on the additional operators that do.

To be concrete we consider operators that appear in two jet production from a virtual photon of momentum q^{μ} in $e^+e^- \rightarrow J_n J_{n'}$. In QCD the fundamental hadronic

operator is the current $J^{\mu} = \bar{\psi}\gamma^{\mu}\psi$, which is conserved $\partial_{\mu}J^{\mu} = 0$ or $q_{\mu}J^{\mu} = 0$, is odd under charge-conjugation, and transforms as a vector under parity and time-reversal. To describe high-energy jet production this current is matched onto a series of SCET currents $J_{\ell}^{(k)}(\omega_i) \sim \lambda^k$ with Wilson coefficients $C_{\ell}(\omega_i)$,

$$J^{\mu} = \sum_{n,n'} \sum_{k=0}^{\infty} \sum_{\ell} \int [\prod_{i} d\omega_{i}] C_{\ell}(\omega_{i}) \left[J_{\ell}^{(k)}(\omega_{i}) \right]_{2\text{-jet}}^{\mu}.$$
 (3.100)

Here k denotes the power in λ , the subscript ℓ denotes members of the basis at a given order, and the ω_i are the set of gauge invariant momentum fractions upon which the operator depends. We also sum over all collinear directions n and n', and the appropriate ones for a given computation are picked out by the jet-momenta in the states. Because of this sum we are free to swap $n \leftrightarrow n'$ when considering symmetry implications. The C, P, and T symmetry properties of $C_{\ell}(\omega_i)J_{\ell}^{(k)}(\omega_i)$ are the same as J^{μ} , and they also satisfy current conservation, $q_{\mu}[J_{\ell}^{(k)}(\omega_i)]^{\mu} = 0$. Finally, since the matching takes place at a hard scale where perturbation theory is valid, the SCET operators should have the same LL + RR chirality as J^{μ} .

We first construct a basis of SCET operators that is homogeneous in the power counting and with even chirality. For the construction of this basis it is convenient to define

$$g_T^{\mu\nu} = g^{\mu\nu} - \frac{q^{\mu}q^{\nu}}{q^2}, \qquad \qquad \gamma_T^{\mu} = \gamma^{\mu} - \frac{q^{\mu}\not{q}}{q^2}, \qquad (3.101)$$
$$r_{-}^{\mu} = \frac{n \cdot q}{2} n'^{\mu} - \frac{n' \cdot q}{2} n^{\mu}, \qquad \qquad r_{+}^{\mu} = \frac{n \cdot q}{2} n'^{\mu} + \frac{n' \cdot q}{2} n^{\mu},$$

where r_{-}^{μ} is odd under $n \leftrightarrow n'$ and r_{+} is even. We also define s_{\pm}^{μ} as r_{\pm}^{μ} with $n \to \bar{n}$ and $n' \to \bar{n}'$. Four of these objects are transverse to q^{μ} , $q_{\mu}g_{T}^{\mu\nu} = 0$, $q_{\mu}\gamma_{T}^{\mu} = 0$, and $q \cdot r_{-} = q \cdot s_{-} = 0$, which is helpful for satisfying current conservation. For constructing the homogeneous basis it suffices to consider the vectors $\{r_{-}, q, s_{-}, s_{+}\}$ in place of $\{n, \bar{n}, n', \bar{n}'\}$. When we specialize to the CM frame, $q_{n\perp}^{\mu} = q_{n'\perp}^{\mu} = 0$, $s_{\pm}^{\mu} = \mp r_{\pm}^{\mu}$, and the vector $r_{+}^{\mu} = q^{\mu}$, and hence r_{+}^{μ} and s_{\pm}^{μ} do not need to be considered.

In a general frame the LO operator is $\bar{\chi}_{\bar{n},\omega_1}\Gamma^{\mu}\bar{\chi}_{n,\omega_2}$ with $\Gamma^{\mu} = \{\gamma^{\mu}_T, r^{\mu}_- \not q, r^{\mu}_- /$

 $r_{-}, g_{T}^{\mu\nu}r_{\nu}^{+}\not{q}, r_{-}^{\mu}\not{r}_{+}, g_{T}^{\mu\nu}r_{\nu}^{+}\not{r}_{-}, g_{T}^{\mu\nu}r_{\nu}^{+}\not{r}_{+}\}$ plus terms where r_{+} or r_{-} are replaced by s_{\pm} . No terms with q^{μ} are allowed by current conservation. Things become much simpler if we focus on operators that are non-zero in the CM frame. In the CM frame $\bar{\chi}_{n',\omega_1}\not{q}$ $\bar{\chi}_{n,\omega_2} = 0, \ \bar{\chi}_{n',\omega_1}\not{r}_{-}\bar{\chi}_{n,\omega_2} = 0$, and the vectors r_{+} and s_{\pm} become redundant, so there is only one operator at lowest order

$$J_0^{(0)} = \bar{\chi}_{n',\omega_1} \gamma_T^{\mu} \chi_{n,\omega_2} \,. \tag{3.102}$$

Here $\omega_i = {\omega_1, \omega_2}$ and for brevity we suppress the index μ on the LHS.

To construct a homogeneous basis at NLO we again consider only operators which are non-vanishing in the CM frame. In the CM frame we can take the total transverse momentum of the jet equal to zero, so we have the relations $\bar{\chi}_{n',\omega_1}\Gamma_{\mu}\mathcal{P}_{\perp}^{\mu}\chi_{n,\omega_2} =$ $\bar{\chi}_{n',\omega_1}\Gamma_{\mu}\mathcal{P}_{\perp}^{\dagger \mu}\chi_{n,\omega_2} = 0$, with Γ_{μ} any gamma structure, and hence do not need to consider operators with a single \mathcal{P}_{\perp} . Again all operators with a \not{q} or \not{r}_{\perp} vanish, as do those with $q \cdot (g\mathcal{B}_{n\perp})$ and $r_{\perp} \cdot (g\mathcal{B}_{n\perp})$, and the analogs with $n \to n'$. Operators with three γ 's can all reduce to operators with a single γ plus terms that are zero in the CM frame. This implies that at NLO there are only two operators

$$J_{1}^{(1)} = r_{-}^{\mu} \bar{\chi}_{n',\omega_{1}} \gamma_{\nu} (g \mathcal{B}_{n\perp}^{\nu})_{\omega_{3}} \chi_{n,\omega_{2}},$$

$$J_{2}^{(1)} = r_{-}^{\mu} \bar{\chi}_{n',\omega_{1}} \gamma_{\nu} (g \mathcal{B}_{n'\perp}^{\nu})_{\omega_{3}} \chi_{n,\omega_{2}}.$$
 (3.103)

Linear combinations of these two SCET currents can both be made odd under charge conjugation by imposing appropriate conditions on their coefficients under $\omega_1 \leftrightarrow -\omega_2$.

To see if there are constraints on the Wilson coefficients we write down a basis of RPI operators up to NLO. The objects γ_T^{μ} and $g_T^{\mu\nu}$ are invariant under RPI and can be used for this construction, but the object r_{\pm}^{μ} cannot. We find the basis

$$\mathbf{J}_{0}^{(0)} = \bar{\Psi}_{n',\hat{\omega}_{1}}\gamma_{T}^{\mu}\Psi_{n,\hat{\omega}_{2}},$$

$$\mathbf{J}_{1}^{(1)} = \bar{\Psi}_{n',\hat{\omega}_{1}}g_{\mu\alpha}^{T}\gamma_{\beta}ig\mathcal{G}_{n,\hat{\omega}_{3}}^{\alpha\beta}\Psi_{n,\hat{\omega}_{2}},$$

$$\mathbf{J}_{2}^{(1)} = \bar{\Psi}_{n',\hat{\omega}_{1}}g_{\mu\alpha}^{T}\gamma_{\beta}ig\mathcal{G}_{n',\hat{\omega}_{3}}^{\alpha\beta}\Psi_{n,\hat{\omega}_{2}},$$

$$\mathbf{J}_{3}^{(1)} = g_{\mu\lambda}^{T}\bar{\Psi}_{n',\hat{\omega}_{1}}\gamma_{\alpha}q_{\beta}\left[i\partial_{n}^{\lambda}ig\mathcal{G}_{n,\hat{\omega}_{3}}^{\alpha\beta}\right]\Psi_{n,\hat{\omega}_{2}},$$

$$\mathbf{J}_{4}^{(1)} = g_{\mu\lambda}^{T}\bar{\Psi}_{n',\hat{\omega}_{1}}\gamma_{\alpha}q_{\beta}\left[i\partial_{n'}^{\lambda}ig\mathcal{G}_{n,\hat{\omega}_{3}}^{\alpha\beta}\right]\Psi_{n,\hat{\omega}_{2}},$$

$$\mathbf{J}_{4}^{(1)} = g_{\mu\lambda}^{T}\bar{\Psi}_{n',\hat{\omega}_{1}}\gamma_{\alpha}q_{\beta}\left[i\partial_{n'}^{\lambda}ig\mathcal{G}_{n',\hat{\omega}_{3}}^{\alpha\beta}\right]\Psi_{n,\hat{\omega}_{2}},$$

$$\mathbf{J}_{4}^{(1)} = g_{\mu\lambda}^{T}\bar{\Psi}_{n',\hat{\omega}_{1}}\gamma_{\alpha}q_{\beta}\left[i\partial_{n'}^{\lambda}ig\mathcal{G}_{n',\hat{\omega}_{3}}^{\alpha\beta}\right]\Psi_{n,\hat{\omega}_{2}},$$

$$\mathbf{J}_{5}^{(1)} = g_{\mu\lambda}^{T} \bar{\Psi}_{n',\hat{\omega}_{1}} \gamma_{\alpha} q_{\beta} i g \mathcal{G}_{n,\hat{\omega}_{3}}^{\alpha\beta} i \partial_{n}^{\lambda} \Psi_{n,\hat{\omega}_{2}}, \qquad \mathbf{J}_{6}^{(1)} = g_{\mu\lambda}^{T} \bar{\Psi}_{n',\hat{\omega}_{1}} (-i\overleftarrow{\partial}_{n'}^{\lambda}) \gamma_{\alpha} q_{\beta} i g \mathcal{G}_{n',\hat{\omega}_{3}}^{\alpha\beta} \Psi_{n,\hat{\omega}_{2}}.$$

Here we do not write down RPI operators which vanish in the CM frame when expanded, such as $\bar{\Psi}_{n'}g_{\mu\nu}^T i\partial_n^{\nu}\Psi_n$ or operators with only the Dirac structure \not{q} . This set also includes three γ operators since in $\mathbf{J}_1^{(1)}$ replacing $\Gamma_{\mu\alpha\beta} = g_{\mu\alpha}^T\gamma_\beta$ by $\Gamma_{\mu\alpha\beta} = \gamma_T^{\mu}\gamma_{\alpha}\gamma_\beta$ gives an operator that vanishes in the CM frame, and any other order for the γ 's is then redundant. The same is true for $\Gamma_{\mu\alpha\beta} = \gamma_{\mu}^T \not{q} \gamma_\alpha q_\beta - q^2 g_{\mu\alpha}^T \gamma_\beta$. Analogous arguments rule out three γ terms replacing the tensor in $\mathbf{J}_2^{(1)}$. There are no others operators with $i\partial_n^{\lambda}$ or $i\partial_{n'}^{\lambda}$ besides $\mathbf{J}_{3-6}^{(1)}$ at LO. To see why, notice that for the operators with $i\partial_n^{\lambda}$ and $i\partial_{n'}^{\lambda}$, only the contraction with $g_{\mu\lambda}^T$ has the potential to give a LO term. Momentum conservation requires $q^{\lambda} = i\partial_n^{\lambda} + i\partial_{n'}^{\lambda}$ and because $q^{\lambda}g_{\mu\lambda}^T = 0$, we can exchange $i\partial_{n'}^{\lambda}$ and $i\partial_n^{\lambda}$. The operators $\mathbf{J}_{3-6}^{(1)}$ correspond to keeping $i\partial_n^{\lambda}$ when we have a $\mathcal{G}_{n,\hat{\omega}}^{\alpha\beta}$.

The number of operators in Eq. (3.104) is greater than that in the homogeneous basis, and when expanded $\mathbf{J}_{0,1,2} \rightarrow J_{0,1,2}$. Thus the operators in Eqs. (3.102) and (3.103) are not connected by RPI. For two jet production the constraints imposed by considering the CM frame are strong enough that RPI provides no further information. (RPI could still constrain the homogeneous basis of operators in a general frame, but does not have practical implications for determining the basis of operators for an analysis to be carried out with homogeneous operators in the CM frame.)

3.7.6 Three Jet Production: n_1 - n_2 - n_3 operators

Here we analyze operators for three jet production. As in the two jet case, we consider production of jets from e^+e^- scattering through a virtual photon. To construct the minimal SCET basis needed for a matching we could proceed like in the previous cases, by writing down both the most general homogeneous basis and RPI basis consistent with the symmetry of the process, and expanding the RPI basis to find possible connections. In the two jet processes the interesting terms in the homogeneous basis are made of only two operators up to NLO. However, for three jets the homogeneous basis has many operators at LO since we have three distinct directions n_1 , n_2 and n_3 . With only two directions we could greatly reduced the number of operators by focusing on the ones that do not vanish in the center of momentum frame, meaning those that do not vanish when $n_1 \to n$, $n_2 \to \bar{n}$, and $q_{\perp} \to 0$. This choice rules out many operators because of the relations $\bar{\chi}_{\bar{n}}\Gamma_{\perp}\not{n}\chi_n = \bar{\chi}_{\bar{n}}\Gamma_{\perp}\not{n}\chi_n = 0$, where Γ_{\perp} is a Dirac structure without \vec{n} or \not{n} factors. With three directions there is more freedom, for example the perpendicular direction of $n_1 - \bar{n}_1$ is not the same as the one of $n_2 - \bar{n}_2$, or of $n_3 - \bar{n}_3$. Now to construct the homogeneous basis, we can use n_1 , \bar{n}_1 , n_2 , \bar{n}_2 , n_3 , \bar{n}_3 , γ^{μ} , so in the three jet case we have a bigger set of objects available.

On the other hand the RPI basis still has a reasonable number of objects, namely Ψ_{n_i} , $\mathcal{G}_{n_i}^{\mu\nu}$, $\partial_{n_i}^{\mu}$, γ^{μ} , and q^{μ} . The $\partial_{n_i}^{\mu}$ operators and q^{μ} are connected by momentum conservation, $i\partial_{n_1}^{\mu} + i\partial_{n_2}^{\mu} + i\partial_{n_3}^{\mu} = q^{\mu}$, so one of them can be eliminated. Hence we expect that reparametrization invariance will give a large number of connections on the homogeneous basis, so many in fact that it is not even convenient to write down the homogeneous basis. It is much quicker to just write only the RPI basis and expand it to determine a basis of allowed homogeneous operators.

The RPI basis for three jets at LO is made of two quarks fields and a gluon field (we do not consider here the case with pure gluon jets). n_1 and n_2 will be the directions of the quark and antiquark jets, and n_3 will be the direction of the gluon jet. As for the two jet case, because of current conservation, the only objects that can carry the vector index and are RPI invariant are $g_T^{\mu\nu}$ and γ_T^{μ} . The RPI basis is

$$\mathbf{J}_{1} = \bar{\Psi}_{n_{1},\hat{\omega}_{1}}\gamma_{\nu} \not\notin \gamma_{\mu}^{T}q_{\sigma}ig\mathcal{G}_{n_{3},\hat{\omega}_{3}}^{\nu\sigma}\Psi_{n_{2},\hat{\omega}_{2}}, \qquad \mathbf{J}_{2} = \bar{\Psi}_{n_{1},\hat{\omega}_{1}}\gamma_{\mu}^{T}\not\notin \gamma_{\nu}q_{\sigma}ig\mathcal{G}_{n_{3},\hat{\omega}_{3}}^{\nu\sigma}\Psi_{n_{2},\hat{\omega}_{2}},
\mathbf{J}_{3} = \bar{\Psi}_{n_{1},\hat{\omega}_{1}}\gamma_{\sigma}\gamma_{\nu}\gamma_{\mu}^{T}ig\mathcal{G}_{n_{3},\hat{\omega}_{3}}^{\nu\sigma}\Psi_{n_{2},\hat{\omega}_{2}} \qquad \mathbf{J}_{4} = \bar{\Psi}_{n_{1},\hat{\omega}_{1}}\gamma_{\mu}^{T}\gamma_{\nu}\gamma_{\sigma}ig\mathcal{G}_{n_{3},\hat{\omega}_{3}}^{\nu\sigma}\Psi_{n_{2},\hat{\omega}_{2}},
\mathbf{J}_{5} = g_{\mu\alpha}^{T}\bar{\Psi}_{n_{1},\hat{\omega}_{1}}\gamma_{\nu}q_{\sigma}i\overleftarrow{\partial}_{n_{1}}^{\alpha}ig\mathcal{G}_{n_{3},\hat{\omega}_{3}}^{\nu\sigma}\Psi_{n_{2},\hat{\omega}_{2}}, \qquad \mathbf{J}_{6} = g_{\mu\alpha}^{T}\bar{\Psi}_{n_{1},\hat{\omega}_{1}}\gamma_{\nu}q_{\sigma}ig\mathcal{G}_{n_{3},\hat{\omega}_{3}}^{\nu\sigma}i\partial_{n_{2}}^{\alpha}\Psi_{n_{2},\hat{\omega}_{2}},
\mathbf{J}_{7} = g_{\mu\alpha}^{T}\bar{\Psi}_{n_{1},\hat{\omega}_{1}}\not\notin \gamma_{\nu}\gamma_{\sigma}i\overleftarrow{\partial}_{n_{1}}^{\alpha}ig\mathcal{G}_{n_{3},\hat{\omega}_{3}}^{\nu\sigma}\Psi_{n_{2},\hat{\omega}_{2}}, \qquad \mathbf{J}_{8} = g_{\mu\alpha}^{T}\bar{\Psi}_{n_{1},\hat{\omega}_{1}}\not\# \gamma_{\nu}\gamma_{\sigma}ig\mathcal{G}_{n_{3},\hat{\omega}_{3}}^{\nu\sigma}i\partial_{n_{2}}^{\alpha}\Psi_{n_{2},\hat{\omega}_{2}},
\mathbf{J}_{9} = g_{\mu\nu}^{T}\bar{\Psi}_{n_{1},\hat{\omega}_{1}}\not\# ig\mathcal{G}_{n_{3},\hat{\omega}_{3}}^{\nu\sigma}i\overleftarrow{\partial}_{12\sigma}\Psi_{n_{2},\hat{\omega}_{2}}, \qquad \mathbf{J}_{10} = \bar{\Psi}_{n_{1},\hat{\omega}_{1}}\gamma_{\mu}^{T}\not\# \gamma_{\nu}ig\mathcal{G}_{n_{3},\hat{\omega}_{3}}^{\nu\sigma}i\overleftarrow{\partial}_{12\sigma}\Psi_{n_{2},\hat{\omega}_{2}},
\mathbf{J}_{11} = \bar{\Psi}_{n_{1},\hat{\omega}_{1}}\gamma_{\mu}^{T}q_{\nu}ig\mathcal{G}_{n_{3},\hat{\omega}_{3}}^{\nu\sigma}i\overleftarrow{\partial}_{12\sigma}\Psi_{n_{2},\hat{\omega}_{2}}. \qquad (3.105)$$

For the first four operators we chose the Dirac structures $\{\gamma_{\nu} \not\in \gamma_{\mu}^{T} q_{\sigma}, \gamma_{\mu}^{T} \not\in \gamma_{\nu} q_{\sigma}, \gamma_{\sigma} \gamma_{\nu} \gamma_{\mu}^{T}, \gamma_{\sigma} \gamma_{\nu} \gamma_{\mu}^{T}, \gamma_{\sigma} \gamma_{\nu} \gamma_{\mu}^{T}, \gamma_{\sigma} \gamma_{\nu} \gamma_{\mu}^{T}, \gamma_{\sigma} \gamma_{\nu} \gamma_$

 $\gamma_{\mu}^{T}\gamma_{\nu}\gamma_{\sigma}\}$ in order to simplify the transformation of the basis under charge conjugation. Since $\{\gamma_{\mu}^{T}, \not{q}\} = 0$ the sum of the first two structures gives $-g_{\mu\nu}^{T} \not{q}q_{\sigma}$, and using the antisymmetry of $\mathcal{G}_{n_{3}}^{\nu\sigma}$ the sum of the last two gives $4g_{\mu\nu}^{T}\gamma_{\sigma}$, so structures with a $g_{\mu\nu}^{T}$ are redundant. Other three γ operators are also redundant. We have used the equations of motion and Bianchi identity in Eqs. (3.35,3.38) to eliminate $i\partial_{n_{i}}$, and momentum conservation to eliminate $i\partial_{n_{3}}^{\mu} = q^{\mu} - i\partial_{n_{1}}^{\mu} - i\partial_{n_{2}}^{\mu}$. For the operators \mathbf{J}_{9-11} we have a derivative contracted with $ig\mathcal{G}_{n_{3}}^{\nu\sigma}$, and we can use the gluon equation of motion, $i\partial_{n_{3}\sigma}\mathcal{G}_{n_{3}}^{\sigma\nu} = (q_{\sigma} - i\partial_{n_{1}\sigma} - i\partial_{n_{2}\sigma})\mathcal{G}_{n_{3}}^{\sigma\nu} = \ldots$, where the ellipsis denotes higher twist terms, to eliminate $(i\partial_{n_{1}}^{\sigma} + i\partial_{n_{2}}^{\sigma})$ and leave only $i\partial_{12}^{\sigma} \equiv (i\partial_{n_{1}}^{\sigma} - i\partial_{n_{2}}^{\sigma})$. Note that we cannot use the trick used in DIS for \mathbf{Q}_{*} , to eliminate \mathbf{J}_{9-11} , because here $i\partial_{n_{1,2}}^{\alpha}$ and $\mathcal{G}_{n_{3}}^{\nu\sigma}$ have different collinear directions. Operators with two or more derivatives are redundant for the construction of the LO basis of RPI currents with one-vector index μ , and hence do not need to be considered.

We can match the three jet RPI basis of currents with the basis of homogeneous SCET currents by writing

$$\sum_{n_1,n_2,n_3} \sum_{\ell} \int [\prod_i d\hat{\omega}_i] \, \hat{C}_{\ell}(\hat{\omega}_i) \, [\mathbf{J}_{\ell}(\hat{\omega}_i)]^{\mu} = \sum_{n_1,n_2,n_3} \sum_{\ell} \int [\prod_i d\omega_i] \, C_{\ell}(\omega_i) \, [\mathcal{J}_{\ell}(\omega_i)]_{3-\text{jet}}^{\mu} + \dots \, .$$

$$(3.106)$$

On the RHS the integration variable was changed using $\hat{\omega}_i = n \cdot q \, \omega_i$ and any additional $n \cdot q$ factors were absorbed into the Wilson coefficients $C_{\ell}(\omega_i)$. We can determine the currents $[\mathcal{J}_l(\omega_i)]_{3\text{-jet}}^{\mu}$ of the homogeneous basis, whose form is as in Eq. (3.70), by just expanding the currents (3.105) using Eqs. (3.65) and (3.67). This yields the homogeneous operator basis

$$\mathcal{J}_{1} = \bar{\chi}_{n_{1},\omega_{1}} (g \mathcal{B}_{n_{3}}^{\perp})_{\omega_{3}} \not q \gamma_{T}^{\mu} \chi_{n_{2},\omega_{2}}, \qquad (3.107)$$

$$\mathcal{J}_{2} = \bar{\chi}_{n_{1},\omega_{1}} \gamma_{T}^{\mu} \not q (g \mathcal{B}_{n_{3}}^{\perp})_{\omega_{3}} \chi_{n_{2},\omega_{2}}, \qquad (3.107)$$

$$\mathcal{J}_{3} = \omega_{3} \bar{\chi}_{n_{1},\omega_{1}} (g \mathcal{B}_{n_{3}}^{\perp})_{\omega_{3}} \not q_{3} \gamma_{T}^{\mu} \chi_{n_{2},\omega_{2}}, \qquad (3.107)$$

$$\mathcal{J}_{4} = \omega_{3} \bar{\chi}_{n_{1},\omega_{1}} \gamma_{T}^{\mu} \not q_{3} (g \mathcal{B}_{n_{3}}^{\perp})_{\omega_{3}} \chi_{n_{2},\omega_{2}}, \qquad (3.107)$$

$$\begin{split} \mathcal{J}_{5} &= \omega_{1} \, \bar{\chi}_{n_{1},\omega_{1}} n_{1T}^{\mu} (g \mathcal{B}_{n_{3}}^{\perp})_{\omega_{3}} \chi_{n_{2},\omega_{2}} \,, \\ \mathcal{J}_{6} &= \omega_{2} \, \bar{\chi}_{n_{1},\omega_{1}} n_{2T}^{\mu} (g \mathcal{B}_{n_{3}}^{\perp})_{\omega_{3}} \chi_{n_{2},\omega_{2}} \,, \\ \mathcal{J}_{7} &= \omega_{1} \, \bar{\chi}_{n_{1},\omega_{1}} n_{1T}^{\mu} \, \not{\!\!\!\!/} \, \not{\!\!\!\!/} \, \eta_{3} (g \mathcal{B}_{n_{3}}^{\perp})_{\omega_{3}} \chi_{n_{2},\omega_{2}} \,, \\ \mathcal{J}_{8} &= \omega_{2} \, \bar{\chi}_{n_{1},\omega_{1}} n_{2T}^{\mu} \, \not{\!\!\!/} \, \not{\!\!\!/} \, \eta_{3} (g \mathcal{B}_{n_{3}}^{\perp})_{\omega_{3}} \chi_{n_{2},\omega_{2}} \,, \\ \mathcal{J}_{9} &= \omega_{3} \, \bar{\chi}_{n_{1},\omega_{1}} \, \not{\!\!\!/} \, \left[n_{3}^{\nu} (g \mathcal{B}_{n_{3}\perp}^{\mu})_{\omega_{3}} - n_{3T}^{\mu} (g \mathcal{B}_{n_{3}\perp}^{\nu})_{\omega_{3}} \right] (n_{2\nu}\omega_{2} + n_{1\nu}\omega_{1}) \chi_{n_{2},\omega_{2}} \,, \\ \mathcal{J}_{10} &= \omega_{3} \, \bar{\chi}_{n_{1},\omega_{1}} \gamma_{T}^{\mu} \, \not{\!\!/} \, \left[n_{3}^{\nu} (g \mathcal{B}_{n_{3}}^{\perp})_{\omega_{3}} - \not{\!\!\!/} \, \eta_{3} (g \mathcal{B}_{n_{3}\perp}^{\nu})_{\omega_{3}} \right] (n_{2\nu}\omega_{2} + n_{1\nu}\omega_{1}) \chi_{n_{2},\omega_{2}} \,, \\ \mathcal{J}_{11} &= \bar{\chi}_{n_{1},\omega_{1}} \gamma_{T}^{\mu} (g \mathcal{B}_{n_{3}\perp}^{\nu})_{\omega_{3}} (n_{2\nu}\omega_{2} + n_{1\nu}\omega_{1}) \chi_{n_{2},\omega_{2}} \,, \end{split}$$

where $n_{1T}^{\mu} = n_1^{\mu} - q^{\mu}(n_1 \cdot q)/q^2$, $n_{2T}^{\mu} = n_2^{\mu} - q^{\mu}(n_2 \cdot q)/q^2$ and $\not q = (\bar{n}_3 \cdot q)\not n_3^{\mu}/2 + (n_3 \cdot q)\not n_3^{\mu}/2$. To simplify the results we did not bother to write out the terms with $q_{\mu}(g\mathcal{B}_{n_3\perp}^{\mu})$ in Eq. (3.107), which are terms that vanish in a frame where $q_{\perp n_3} = 0$. In some cases we have absorbed RPI factors in the Wilson coefficients $C_{\ell}(\omega_i)$ when carrying out the expansion.

The tree level matching from QCD to SCET for three jets comes from matching two Feynman diagrams in QCD onto the operator basis in Eq. (3.107), and is done at the hard scale $\mu = Q$. This gives

$$C_1 = C_6 = \frac{-2}{n_1 \cdot n_3 \,\omega_1 \omega_3}, \quad C_2 = -C_5 = \frac{-2}{n_2 \cdot n_3 \,\omega_2 \omega_3}, \quad C_{3,4} = C_{7-11} = 0.$$
(3.108)

The results for these Wilson coefficients are invariant under type-III RPI as expected.

The above matching computation can be compared with the tree level SCET computations for parton showers in Ref. [17], where three final state jets are considered. To compare the calculations we take the two stages of matching of Ref. [17] both at $\mu = Q$, and we split the operators in Eqs. (27,28) of Ref. [17] into two parts, $\mathcal{O}_3 = \mathcal{O}_{3a} + \mathcal{O}_{3b}$ and $\mathcal{O}_3^{(2)} = \mathcal{O}_{3a}^{(2)} + \mathcal{O}_{3b}^{(2)}$. The matching computation of Ref. [17] used a frame $q_{n_2\perp} = 0$ for \mathcal{O}_{3a} and $\mathcal{O}_{3a}^{(2)}$ and a frame $q_{n_1\perp} = 0$ for \mathcal{O}_{3b} and $\mathcal{O}_{3b}^{(2)}$. With these frame choices, we confirm that $C_1\mathcal{J}_1 + C_6\mathcal{J}_6 = \mathcal{O}_{3a} + \mathcal{O}_{3a}^{(2)}$ and $C_2\mathcal{J}_2 + C_5\mathcal{J}_5 = \mathcal{O}_{3b} + \mathcal{O}_{3b}^{(2)}$, providing a cross-check on the results in Eq. (3.108).

3.7.7 Two Jets from Gluon Fusion: $gg \rightarrow q\bar{q}$ operators

Next we consider the example of the production of two quark jets from gluon fusion, which is relevant for the LHC. In this application we will see that RPI substantially constrains the number and structure of operators. This basis of operators have not yet been constructed. The factorization theorem for $pp \rightarrow 2$ jets has been discussed in Ref. [68], and were also considered recently in Ref. [11] using SCET. SCET has also been used to resum electroweak Sudakov logarithms by solving RGE equations for four quark collinear operators in Refs. [35, 98, 99], and to consider Higgs production from pp collisions [1].

We consider the incoming gluons to be collinear in different directions, which is appropriate for the high energy collision of energetic protons at the LHC, and we assume that the final state jets have a large perpendicular momentum relative to the beam axis. Hence the final jets are described by two additional collinear directions, making four in total. Unlike our previous examples, here there is not an external q^{μ} vector, the hard interaction takes place entirely between strongly interacting particles. Hence this is an example of the case ii) discussed above Eq. (3.7).

Similarly to the three jets case, it is convenient to directly write the RPI basis without first writing the homogeneous basis, because the presence of four collinear directions imply that there are a large number of homogeneous operators, many of which are restricted by RPI. Due to the absence of an external hard vector q^{μ} in this process, in the definition of the currents we make use the RPI delta function factors of Eq. (3.15), $\hat{\Delta}_{km}$. The general formula for matching the RPI operators onto homogeneous operators is

$$i\sum_{n_1,n_2,n_3,n_4}\sum_{\ell}\int[\prod_{i,j}d\hat{\omega}_{ij}]\,\hat{C}_{\ell}(\hat{\omega}_i)\,[\prod_{km}\hat{\Delta}_{km}]\mathbf{Q}_{\ell} = i\sum_{n_1,n_2,n_3,n_4}\sum_{\ell}\int[\prod_i d\omega_i]\,C_{\ell}(\omega_i)\,[O_{\ell}(\omega_i)]_{gg\to q\bar{q}}$$
$$+\dots, \qquad (3.109)$$

where we use the same manipulations needed to get Eq. (3.11). Note that here we have divided the RPI operators into the δ -functions in $\hat{\Delta}_{km}$ which depend on $\hat{\omega}_{km}$, and the remainder of the operator \mathbf{Q}_{ℓ} that does not. The starting point for building a basis for \mathbf{Q}_{ℓ} is the object $\bar{\Psi}_{n_1} \mathcal{G}_{n_3}^{\mu\nu} \mathcal{G}_{n_4}^{\alpha\beta} \Psi_{n_2}$. We assume a LL + RR chirality for the quarks which is suitable when strong interactions produce massless quarks, and hence include either γ^{λ} or $\gamma^{\lambda} \gamma^{\sigma} \gamma^{\tau}$. Since the overall operator is a scalar, all the vector indices on the field strengths and on the Dirac structure must be contracted with $g_{\mu\nu}$'s or $i\partial_{n_i}^{\mu}$'s. We can use the equations of motion and Bianchi identity in Eqs. (3.35,3.37,3.38) to eliminate terms with $i\partial_{n_i}$ in any operator, and terms with $\partial_{n_3\mu}\mathcal{G}_{n_3}^{\mu\nu}$ or $\partial_{n_4\mu}\mathcal{G}_{n_4}^{\mu\nu}$. In addition, momentum conservation implies $i\partial_{n_1}^{\mu} + i\partial_{n_2}^{\mu} + i\partial_{n_3}^{\mu} + i\partial_{n_4}^{\mu} = 0$, and we will use this to eliminate all operators with an $i\partial_{n_1}$. This leaves twenty operators for the RPI basis

$$\begin{aligned} \mathbf{Q}_{1} &= \bar{\Psi}_{n_{1}} \gamma_{\beta} g_{\nu \alpha} \, ig \mathcal{G}_{n_{3}}^{\mu \nu} i \partial_{n_{4} \mu} ig \mathcal{G}_{n_{4}}^{\alpha \beta} \Psi_{n_{2}} \,, & \mathbf{Q}_{2} &= \bar{\Psi}_{n_{1}} \gamma_{\mu} g_{\nu \alpha} i \partial_{n_{3} \beta} \, ig \mathcal{G}_{n_{3}}^{\mu \nu} ig \mathcal{G}_{n_{4}}^{\alpha \beta} \Psi_{n_{2}} \,, \\ \mathbf{Q}_{3} &= \bar{\Psi}_{n_{1}} \gamma_{\mu} g_{\nu \alpha} \, ig \mathcal{G}_{n_{3}}^{\mu \nu} ig \mathcal{G}_{n_{4}}^{\alpha \beta} i \partial_{n_{2} \beta} \Psi_{n_{2}} \,, & \mathbf{Q}_{4} &= \bar{\Psi}_{n_{1}} \gamma_{\beta} g_{\nu \alpha} \, ig \mathcal{G}_{n_{4}}^{\mu \nu} ig \mathcal{G}_{n_{4}}^{\alpha \beta} i \partial_{n_{2} \mu} \Psi_{n_{2}} \,, \\ \mathbf{Q}_{5} &= \bar{\Psi}_{n_{1}} \gamma_{\nu} \gamma_{\alpha} \gamma_{\beta} \, ig \mathcal{G}_{n_{3}}^{\mu \nu} i \partial_{n_{4} \mu} ig \mathcal{G}_{n_{4}}^{\alpha \beta} \Psi_{n_{2}} \,, & \mathbf{Q}_{6} &= \bar{\Psi}_{n_{1}} \gamma_{\mu} \gamma_{\nu} \gamma_{\alpha} i \partial_{n_{3} \beta} \, ig \mathcal{G}_{n_{3}}^{\alpha \beta} ig \mathcal{G}_{n_{4}}^{\alpha \beta} \psi_{n_{2}} \,, \\ \mathbf{Q}_{7} &= \bar{\Psi}_{n_{1}} \gamma_{\mu} \gamma_{\nu} \gamma_{\alpha} \, ig \mathcal{G}_{n_{3}}^{\mu \nu} ig \mathcal{G}_{n_{4}}^{\alpha \beta} i \partial_{n_{2} \mu} \Psi_{n_{2}} \,, & \mathbf{Q}_{8} &= \bar{\Psi}_{n_{1}} \gamma_{\nu} \gamma_{\alpha} \gamma_{\beta} \, ig \mathcal{G}_{n_{3}}^{\alpha \beta} ig \mathcal{G}_{n_{4}}^{\alpha \beta} i \partial_{n_{2} \mu} \Psi_{n_{2}} \,, \\ \mathbf{Q}_{9} &= \bar{\Psi}_{n_{1}} \gamma_{\alpha} \, ig \mathcal{G}_{n_{3}}^{\mu \nu} i \partial_{n_{4} \mu} ig \mathcal{G}_{n_{4}}^{\alpha \beta} i \partial_{n_{2} \nu} i \partial_{n_{2} \beta} \Psi_{n_{2}} \,, \\ \mathbf{Q}_{10} &= \bar{\Psi}_{n_{1}} \gamma_{\mu} i \partial_{n_{3} \alpha} \, ig \mathcal{G}_{n_{3}}^{\mu \nu} ig \mathcal{G}_{n_{4}}^{\alpha \beta} i \partial_{n_{2} \nu} i \partial_{n_{2} \beta} \Psi_{n_{2}} \,. \end{aligned} \tag{3.110}$$

The other ten operators \mathbf{Q}_{11-20} have the same structure as Eq. (3.110) but with a trace over color for the gluon operators, for example $\mathbf{Q}_{11} = \bar{\Psi}_{n_1} \gamma_\beta g_{\nu\alpha} \operatorname{Tr}[ig\mathcal{G}_{n_3}^{\mu\nu}i\partial_{n_4\mu}ig\mathcal{G}_{n_4}^{\alpha\beta}]\Psi_{n_2}$. Note that \mathbf{Q}_{1-10} have \mathcal{G}_{n_3} to the left of \mathcal{G}_{n_4} , so one might think that there are ten more operators with the \mathcal{G} 's in the other order. However, in Eq. (3.109) we sum over $n_{3,4}$ and integrate over $d\hat{\omega}_3 d\hat{\omega}_4$, and hence include operators obtained from the interchange $n_3 \leftrightarrow n_4$, $\hat{\omega}_3 \leftrightarrow \hat{\omega}_4$. Recall that the directions n_i are only determined by the matrix elements. So if we consider a matrix element with gluons in the n and n' direction then there is a contribution from $n_3 = n$, $n_4 = n'$, and from $n_3 = n'$, $n_4 = n$. Other possible operators might be $\bar{\Psi}_{n_1}\gamma_{\alpha} i\partial_{n_3\beta} ig\mathcal{G}_{n_3}^{\mu\nu} i\partial_{n_4\mu} ig\mathcal{G}_{n_4}^{\alpha\beta} i\partial_{n_2\nu}\Psi_{n_2}$, $\bar{\Psi}_{n_1}\gamma_{\mu}i\partial_{n_3\alpha} ig\mathcal{G}_{n_3}^{\mu\nu}i\partial_{n_4\nu}ig\mathcal{G}_{n_4}^{\alpha\beta}i\partial_{n_2\beta}\Psi_{n_2}$ and similarly with the trace. We can use the Bianchi identity (3.38) to rule them out. For example, in the first operator we have implicitly already used the Bianchi identity for the $i\partial_{n_4\mu}ig\mathcal{G}_{n_4}^{\alpha\beta}$ term because we did not write operators with $i\partial_{n_4}ig\mathcal{G}_{n_4}^{\alpha\beta}$. But we can apply the Bianchi identity to $i\partial_{n_3\beta}ig\mathcal{G}_{n_3}^{\mu\nu}$, that is not connected with γ 's. In this way we can write this operator in terms of \mathbf{Q}_1 , \mathbf{Q}_4 and operators with three gluon fields. Note that we do not need to consider operators with $i\partial_n \cdot i\partial_{n'}$ since all these contracted derivatives are contained in the $\hat{\Delta}_{km}$'s.

A natural frame for analyzing $gg \to q\bar{q}$ is the CM frame with the choices $\bar{n}_1 = n_2$, $\bar{n}_2 = n_1$, $\bar{n}_3 = n_4$, $\bar{n}_4 = n_3$. We expand the currents (3.110) with an eye towards using them in this frame. Actually, only the condition $\bar{n}_3 = n_4$, $\bar{n}_4 = n_3$ is necessary to find the following operators

$$O_{1} = \omega_{4} \bar{\chi}_{n_{1},\omega_{1}} \#_{4} (g\mathcal{B}_{n_{3}\perp}^{\mu}) \omega_{3} (g\mathcal{B}_{n_{4}\mu}^{\perp}) \omega_{4} \chi_{n_{2},\omega_{2}}, \qquad (3.111)$$

$$O_{2} = \omega_{3} \bar{\chi}_{n_{1},\omega_{1}} \#_{3} (g\mathcal{B}_{n_{3}\perp}^{\mu}) \omega_{3} (g\mathcal{B}_{n_{4}\mu}^{\perp}) \omega_{4} \chi_{n_{2},\omega_{2}}, \qquad (3.111)$$

$$O_{3} = \omega_{2} \bar{\chi}_{n_{1},\omega_{1}} (g\mathcal{B}_{n_{3}\perp}) \omega_{3} (g\mathcal{B}_{n_{4}\perp}^{\perp}) \omega_{4} \chi_{n_{2},\omega_{2}}, \qquad (3.111)$$

$$O_{4} = \omega_{2} \bar{\chi}_{n_{1},\omega_{1}} (gn_{2} \cdot \mathcal{B}_{n_{3}}^{\perp}) \omega_{3} (g\mathcal{B}_{n_{4}\perp}) \omega_{4} \chi_{n_{2},\omega_{2}}, \qquad (3.111)$$

$$O_{5} = \omega_{4} \bar{\chi}_{n_{1},\omega_{1}} \#_{4} (g\mathcal{B}_{n_{3}\perp}) \omega_{3} (g\mathcal{B}_{n_{4}\perp}) \omega_{4} \chi_{n_{2},\omega_{2}}, \qquad (3.111)$$

$$O_{6} = \omega_{3} \bar{\chi}_{n_{1},\omega_{1}} \#_{4} (g\mathcal{B}_{n_{3}\perp}) \omega_{3} (g\mathcal{B}_{n_{4}\perp}) \omega_{4} \chi_{n_{2},\omega_{2}}, \qquad (3.111)$$

$$O_{7} = \omega_{2} \omega_{3} \omega_{4} \bar{\chi}_{n_{1},\omega_{1}} \#_{3} \#_{4} (g\mathcal{B}_{n_{3}\perp}) \omega_{3} (gn_{2} \cdot \mathcal{B}_{n_{4}}^{\perp}) \omega_{4} \chi_{n_{2},\omega_{2}}, \qquad (3.111)$$

$$O_{8} = \omega_{2} \omega_{3} \omega_{4} \bar{\chi}_{n_{1},\omega_{1}} \#_{3} \#_{4} (gn_{2} \cdot \mathcal{B}_{n_{3}}^{\perp}) \omega_{3} (g\mathcal{B}_{n_{4}\perp}) \omega_{4} \chi_{n_{2},\omega_{2}}, \qquad (3.111)$$

 O_{11-20} have the same structure of (3.111) but with a trace over color of the two gluon operators. O_i is given by the expansion of \mathbf{Q}_i for i = 1, 2, 5, 6, by the expansion of a suitable linear combination of \mathbf{Q}_i and \mathbf{Q}_{i-1} for i = 3, 6, and of \mathbf{Q}_i and \mathbf{Q}_{i-3} for i = 4, 8. $O_{9/10}$ are given by the expansion of a suitable linear combination of $\mathbf{Q}_{9/10}$, $\mathbf{Q}_{1/2}$ and $\mathbf{Q}_{4/3}$. In some cases we have absorbed reparametrization invariant prefactors that appear in the expansion into the Wilson coefficients $C_l(\omega_i)$. By using momentum conservation it is possible to reduce these ten operators to just four independent operators at leading order in SCET.⁴

It is straightforward to carry out the matching from QCD onto the SCET operators in Eq. (3.111). At tree level there are three Feynman diagrams. The amplitude squared is also known analytically at one-loop [43], and a full matching computation at this order involves regulating infrared singularities in the same way for the loops in QCD and SCET before subtracting. The only point to be careful about is the sum over the n_i 's in Eq. (3.109), since definite values for these n_i 's should be determined by the states. For example, if we consider the tree level $gg \rightarrow q\bar{q}$ matrix element of O_1 with perpendicular polarization for the gluons then

$$\left\langle q_{n_{1}'}(p_{1}')\bar{q}_{n_{2}'}(p_{2}) \right| \sum_{n_{i}} \int [d\omega_{j}]iC_{1}(\omega_{1},\omega_{3},\omega_{4},\omega_{2})O_{1}(\omega_{j}) \left| g_{n_{3}'}^{\perp}(p_{3}')g_{n_{4}'}^{\perp}(p_{4}') \right\rangle$$

$$= ig^{2}C_{1}(\omega_{1}',\omega_{3}',\omega_{4}',\omega_{2}')\omega_{4}' \left[\epsilon_{n_{3}\perp}^{\mu A} \epsilon_{n_{4}\perp\mu}^{B} \right] \left[\bar{u}_{n_{1}'}\eta_{4}'T^{A}T^{B}u_{n_{2}'} \right]$$

$$+ ig^{2}C_{1}(\omega_{1}',\omega_{4}',\omega_{3}',\omega_{2}')\omega_{3}' \left[\epsilon_{n_{3}\perp}^{\mu A} \epsilon_{n_{4}\perp\mu}^{B} \right] \left[\bar{u}_{n_{1}'}\eta_{3}'T^{B}T^{A}u_{n_{2}'} \right].$$

$$(3.112)$$

The two terms come from the cases $n_{3,4} = n'_{3,4}$ and $n_{3,4} = n'_{4,3}$ respectively. Therefore to determine the C_{ℓ} 's it suffices to compute terms contributing to the color structure $T^A T^B$ in QCD, which at tree level gives

$$C_{1} = \frac{-1}{(n_{3} \cdot n_{4})\omega_{3}\omega_{4}}, \quad C_{2} = \frac{1}{(n_{3} \cdot n_{4})\omega_{3}\omega_{4}}, \quad C_{3} = \frac{2}{(n_{2} \cdot n_{4})\omega_{2}\omega_{4}}, \quad C_{5} = \frac{1}{(n_{2} \cdot n_{4})\omega_{2}\omega_{4}},$$
$$C_{4} = C_{6-20} = 0. \quad (3.113)$$

Note that the results for the C_{ℓ} 's are invariant under type-III RPI transformations as expected, and that in the frame used for our computation $n_3 \cdot n_4 = 2$. We have confirmed that a consistent result is obtained by considering the $T^B T^A$ terms. Eq. (3.112) expresses the interesting fact that with distinct collinear directions for all final state particles, only the color ordered QCD amplitudes are needed for the matching which

⁴We thank W. Waalewijn for his explicit derivation of this point.

determines the SCET Wilson coefficients.

3.8 Conclusion

In SCET the momenta of collinear particles are decomposed with light-like vectors n^{μ} and \bar{n}^{μ} , where \vec{n} is close to the direction of motion. The vectors n^{μ} and \bar{n}^{μ} are required to define collinear operators that have a definite order in the power counting. However, there is a freedom in defining n and \bar{n} , which leads to reparametrization constraints. The decomposition of operators in the theory must satisfy these constraints in order to be consistent. This reparametrization invariance gives nontrivial relations among the Wilson coefficients of collinear operators occurring at different orders in the power counting, and for situations with multiple collinear directions gives constraints on the form of operators making up a complete basis.

In this chapter we have constructed objects that are invariant under both collinear gauge transformations and reparametrization transformations, a superfields Ψ_{n_i} for fermions and a superfield $\mathcal{G}_{n_i}^{\mu\nu}$ for gluons. Here the subscript n_i denotes an equivalence class of light-like vectors under RPI. The superfields are invariant under collinear gauge transformations through a reparametrization invariant Wilson line \mathcal{W}_{n_i} that is the generalization of the usual W_{n_i} . We constructed RPI operators out of these superfields by introducing reparametrization invariant δ -functions. The δ -functions act on the RPI operators to pick out large momenta, and are convoluted with hard Wilson coefficients that must be computed by matching computations. The power of the RPI operators is that they encode information about the minimal basis of Wilson coefficients. However, they do not have a definite power counting order. By expanding them in λ one obtains a minimal basis of operators with a good power counting, where all constraints on the Wilson coefficients are made explicit. The final basis of operators with a good power counting involves a two-component field χ_{n_i} for quarks, a field $\mathcal{B}_{n_i\perp}^{\mu}$ for the two physical gluon polarizations, derivatives $\mathcal{P}_{n_i\perp}^{\mu}$, and delta functions $\delta(\omega - \overline{\mathcal{P}}_{n_i})$ that pick out the large momenta of these collinear fields. That is it. Other field components such as $n_i \cdot \mathcal{B}_{n_i}$, and other derivatives such as

 $in_i \cdot \partial_{n_i}$, are eliminated from the purely collinear operator basis using the equations of motion.

This procedure was applied to several processes. We studied spin-averaged DIS for quarks at twist-4, as a means of testing our setup in a framework where the power suppressed basis of operators is well understood. We then constructed a minimum basis of pure glue operators for DIS at twist-4. These applications involve a single collinear direction. For processes with multiple collinear directions we considered operator bases for jet production. Useful constraints from RPI were not found for the first power suppressed operators in $e^+e^- \rightarrow 2$ jets. On the other hand, already at leading order in the power counting, RPI provided important constraints on the complete basis of operators for $e^+e^- \rightarrow 3$ jets with three distinct collinear directions. RPI was also very useful in constructing a complete basis of operators for gluon fusion producing two quark initiated jets, where there are four collinear directions. In this case the process of interest is pp \rightarrow 2 jets, which will be studied at the LHC. We expect the complete bases of operators constructed here will be a useful ingredient in the study of factorization theorems for this process. The steps we used to construct complete basis will also be useful when considering factorization for processes with more jets in the final state. In general we found that RPI becomes more powerful for processes involving more jets, essentially because the number of vectors n_i and \bar{n}_i proliferates faster than the number of objects that must be considered to build the RPI basis.

An interesting observation discussed in Section 3.7.7 is that when matching from QCD onto SCET operators describing multiple collinear directions n_i , the Wilson coefficient is determined by the color ordered QCD amplitude. Since results for multi-leg QCD amplitudes are often expressed in a color ordered form, this should simplify the matching of QCD amplitudes onto SCET.

Chapter 4

Parton Showers to NLO

4.1 Introduction

A final state shower MC is based on the "strongly-ordered limit" which describes the leading log contribution (accounting for soft emission by angular ordering or other approximations). In this kinematic configuration, each radiated particle comes off much more collinear to its parent than the previous one, a situation that can be formulated in terms of perpendicular momenta or invariant masses, *i.e.*

$$q_{0\perp} \gg q_{1\perp} \gg q_{2\perp} \gg \dots, \qquad \text{or} \qquad q_0^2 \gg q_1^2 \gg q_2^2 \gg \dots.$$

$$(4.1)$$

Furthermore, and important for practical computation, each emission is independent of the previous one to leading log order. Thus, if we have calculated the differential cross section for (i - 1)-parton emission, $d\sigma_{i-1}$, then we can obtain the *i*-parton case as

$$d\sigma_i = \frac{P_{i \to jk}^{(0)}}{q_{i-1}^2} d\sigma_{i-1}, \qquad (4.2)$$

where $P^{(0)}$ is the leading order (LO) "splitting function" that captures the probability of the $(i-1)^{th}$ emitted parton to split into two others, jk, and q_{i-1}^2 is its virtuality. Thus, we can formulate the process in terms of a probabilistic Markov chain of (i-1) $1 \rightarrow 2$ particle splittings. The probabilities are determined by the functions $P_{i\rightarrow jk}^{(0)}$, which are related to Altarelli-Parisi kernels. As an example, for $q \rightarrow qg$ in QCD, after averaging and summing over spins,

$$P_{q \to qg}^{(0)} = \frac{\alpha_s}{2\pi} C_F \frac{1+z^2}{1-z},$$
(4.3)

where z is the momentum fraction of the daughter with respect to the parent. This classical, probabilistic process gives rise to the parton shower algorithms used by event generators to model radiation through Monte Carlo, such as Pythia [91, 92], Herwig [5, 38], and Sherpa [54] (although by now more sophisticated generalizations such as the dipole shower [32, 33] are becoming more popular). Given some initial virtuality, q_0^2 , and an initial momentum fraction, x_0 , MCs generate the virtuality and the momentum fraction of the daughter particle after the spitting. The virtuality is determined by a Sudakov factor, $\Delta(q^2, q_0^2)$, which gives the probability of a parton to evolve from q_0^2 to q^2 without branching,

$$\Delta(q^2, q_0^2) = \exp\left[-\int_{q_0^2}^{q^2} \frac{dq'^2}{q'^2} \int dx \, \frac{\alpha_s}{2\pi} P_{jk}^{(0)}(x)\right].$$
(4.4)

The traditional LL parton shower makes a difficult problem tractable, but has some shortcomings related to its dependence on the leading log approximation. Even though the splitting functions only dominate in soft and collinear limits, the shower is used everywhere in order to generate events that cover the full phase space. In addition, since each emission is independent from the previous one in the shower, the LL approximation does not include any spin or color correlations. Furthermore, the procedure is classical and the interference between different amplitudes is only probed beyond leading log.

The hierarchy of scales in the parton shower makes it amenable to an effective field theory treatment. Since the shower regime occurs for fields in the soft and collinear regions, we can describe it with Soft-Collinear Effective Theory (SCET) [8, 10, 14, 18]. SCET is the appropriate effective theory for studying parton showers because it is designed to reproduce exactly the limit of soft and collinear particles. Moreover, SCET is organized in an expansion of power counting parameter that makes it convenient for classifying all corrections. The first work on parton showers using SCET was Refs. [16, 17], where the authors showed how the splitting function and the Sudakov factor emerge naturally in SCET. They reproduced the LL parton showers using SCET, and showed how higher order virtual corrections can be encoded by matching onto Wilson coefficients in the effective theory. Unfortunately they introduced choices and approximations at several points along the way, which makes our task of identifying the full set of corrections beyond LL difficult, so some modification of their setup will be required.

Before discussing our approach, we give here a brief discussion of literature on improvements to the basic shower Monte Carlo picture. To begin with, consider a simple setup where one declares that an artificial scale μ_0 divides collinear from hard radiation. We describe emissions above μ_0 through fixed-order tree-level calculations, and those beneath by running shower Monte Carlo. Each regime would get an accurate treatment, but interfacing the two results in leading-log introduces sensitivity to the scale μ_0 . This is because the LO (in α_s) result contains no LL-resummation. The CKKW algorithm [31] defines a method to include this information. One distributes the particles in an event according to the probabilities given by the exact tree-level matrix element, with μ_0^2 as a lower cutoff on the virutality between any two particles. One then clusters the event using the k_T algorithm [29] to determine the splitting virtualities, q_i^2 . With these scales in hand, one reweights the event by multiplication by appropriate Sudakov factors, as well as factors of $\alpha_s(q_i)/\alpha_s(Q)$, where Q is some hard scale. We can then run a parton shower algorithm on these amplitudes, vetoing any splitting virtuality harder than μ_0^2 to avoid double counting. The resulting distribution depends on μ_0 only at the subleading order $\alpha_s(\alpha_s \ln^2(Q/\mu_0))^m$.

Another important effect concerns soft gluons which are also kinematically enhanced. Collinear emissions reinforce the picture of partonic radiation as an isolated jet as they get distributed within some narrow cone about the original hard parton. Apriori soft emissions have no preferred direction and can communicate between elements of the shower. Fortunately, azimuthal averaging confines these emissions to lie within the cone containing the nearest collinear gluon as well as the showering quark or gluon. This effect is called angular-ordering and can be accomodated by methods such as evolving the shower by decreasing angle monotonically, as is done in Herwig [81], or by enforcing it with a veto in a mass-ordered shower (rightmost expression in Eq. (4.1)), which is an option in Pythia [91]. Additional considerations treated in shower programs include putting α_s at the scale of each splitting, and encoding momentum conservation at each vertex, which give the parton shower information beyond an analytic LO/LL calculation. All of these are treated in different fashions by different MC codes.

There are of course further corrections to include to go to NLO in α_s and NLL in kinematic logs. The most effort to date has gone to working out the NLO/LL contribution to incorporate one-loop-corrected amplitudes. Adding α_s corrections involves the numerical challenge of combining real and virtual results which separately diverge. The basic resolution is to extract the pole-portion of the real emission of *i*-partons and include it along with the virtual contributions to the i - 1 case. Unfortunately this does not sum leading logs. One cannot blindly extend the CKKW procedure to NLO/LL, as it leads to double-counting problems; the Sudakov factors in the reweighting contain a portion of the one-loop contributions. Thus seperately adding on the full one-loop result would clearly double count.

There are two main solutions to the NLO/LL merging problem. The older of the two, MC@NLO [52], works by means of subtraction, finding the places where the Sudakovs will contribute at NLO, and removing the splitting function contribution. While conceptually clear, this has numerical and theoretical complications. The full amplitude and splitting function portions are calculated separately before subtraction, which is time-consuming. Furthermore, since the subtractions occur for the amplitude squared, one cannot guarantee positivity of the result. Hence one deal with negatively weighted events (and in some pathological cases negative cross-sections, see eg. [20]). To avoid the computational difficulties of process-by-process subtraction and the problem of negative weights, an alternative is the POWHEG algorithm [82]. It keeps the IR-safe NLO cross-section manifest, and defines a Sudakov factor based on a modified splitting function to handle LL resummation. In this way, it makes use of quantities already obtained in the fixed order NLO calculation, requiring fewer additional steps for its implementation for each known process. The conservation of probability obeyed by the splittings and related Sudakov factor avoid double countings and give back $\sigma_{\rm NLO}$ upon integration.

Another approach to go beyond LO/LL is to incorporate subleading logs by including in the Sudakov the contribution of the $\mathcal{O}(\alpha_s^2)$ corrections to the Altarelli-Parisi splitting kernels, $P_{qq}^{(1)}$. Unfortunately, doing this alone would fail to conserve probability if one did not also correct the probability for real emission in Eq. (4.3). Some of the subleading contributions take the form of $1 \rightarrow 3$ splittings, requiring a modification of the usual $1 \rightarrow 2$ algorithm. The KRKMC group aims to incorporate these subleading contributions in shower MC [60, 61, 93]. Similar to CKKW, their corrections take the form of a multiplicative reweighting. For a particular configuration of partons in phase space, they reweight by a factor that includes the insertion of $1 \rightarrow 3$ "defects" as loop-corrected $1 \rightarrow 2$ splittings that account for the affects of $P_{qq}^{(1)}$. If ρ is the fully differential cross-section, they define a corrected weight for n partons, w_n as:

$$w_n = \frac{\rho_{\rm LO}(k_1, \dots, k_n) + \sum_{r=1}^{n/2} \rho_{\rm N^r LO}(k_1, \dots, k_n)}{\rho_{\rm LO}(k_1, \dots, k_n)},$$
(4.5)

where r determines the number of defect insertions in any configuration.

In this chapter we set up an EFT framework to classify and study perturbative and power corrections to parton showers. The results of this work will be presented in Ref. [21]. While the ultimate goal is to facilitate the implementation of a complete NLL/NLO parton shower algorithm, our goals here are much more modest. The main objective of our work is to explain a computational framework where we can enumerate and classify all the needed theoretical ingredients in a NLO/NLL shower at a theoretical rather than algorithmic level. When carrying out explicit computations within this framewrok we focus primarily on shower power corrections in the fully differential cross section for an arbitrary number of emissions, that is

$$\frac{d\sigma}{d\vec{p}_1^3\cdots d\vec{p}_n^3},\tag{4.6}$$

for *n*-partons. Similarly to [17], we use an operator approach based on SCET. A main issue to resolve is taking into account different possibilities for the kinematic configuration of subsequent emissions, to go beyond the strong ordering described in Eq. (4.1). We overcome this issue by setting up a tower of related soft collinear effective fields theory called $SCET_i$, which also helps us deal with several technical obstacles. We formulate the shower description as a standard matching procedure between operators in different $SCET_i$. Power corrections are encoded by performing matching computations at subleading order in the kinematic power expansion between different regions. The hierarchy between regions is expressed by a power counting parameter $\lambda \ll 1$. These power corrections modify the structures that initiate the shower, branching probabilities, as well as opening up the $1 \rightarrow 3$ splitting channel. Virtual perturbative α_s corrections are encoded by performing matching calculations beyond tree level between $SCET_i$ theories. Finally, corrections to the Sudakov nobranching probabilities are encoded through anomalous dimensions of leading and subleading operators at the appropriate order within the different $SCET_i$ s. We will carry out the necessary computations for the power corrections, and a subset of the required calculations for anomalous dimensions occuring for operators beyond the leading shower. This analysis includes the leading corrections to the shower from interference and from spin correlations. As much as possible we attempt to give pointers for additional computations that are needed in places where our analysis is incomplete. For example, to simplify things we have not treated color correlations since doing so increases the basis of operators and the number of computations, but does not change the conceptual setup.

The outline of this chapter is as follows. We review the Bauer-Schwartz SCET shower method in subsection 4.2.1 and discuss the technical obstructions to extending it to the level needed to determine the desired power corrections. We present our SCET_i framework in subsection 4.2.2 to resolve these issues. In subsection 4.3 we analyze the LL shower in the SCET_i framework, and show that the transition between SCETs, SCET_i \rightarrow SCET_{i+1}, can be encoded by operator replacement rules on single parton collinear fields. Soft emissions in SCET_i are discussed, and we summarize the correspondence between SCET_i objects and LL shower ingredients to show how the mapping works. In subsection 4.4, we use the SCET_i formulation to classify and compute various corrections to the shower to $\mathcal{O}(\lambda^2)$ in the cross-section. Two main categories of branching corrections emerge, which we refer to as hard scattering and jet-substructure corrections. We also discuss ingredients needed for renormalization group evolution, and derive all the LL anomalous dimensions for our subleading operators. A summary of the NLL/NLO corrections is presented as a table mapping ingredients needed for the subleading shower to those in SCET. Conclusions are given in subsection 4.6.

Many details are relegated to the appendices B and C. We describe finite reparametrization transformations in appendix B.1, we use RPI in our matching computations to disentangle kinematic choices from kinematic power corrections. Details on the matching of QCD \rightarrow SCET₁, SCET₁ \rightarrow SCET₂, and SCET₂ \rightarrow SCET₃ with generalization to SCET_i to SCET_{i+1} can be found in appendices B.2, B.3, and B.4, respectively. Appendix C contains a cross-check on our results, where we integrate a subset of our power suppressed terms to rederive the abelian terms in $P_{q \rightarrow qg}^{(1)}$, namely the $\mathcal{O}(\alpha_s)$ correction to the $q \rightarrow qg$ splitting function [39].

Before we start analyzing parton showers in SCET, we introduce the operator notation we will use for this chapter. Most of our discussion will involve interactions with collinear fields, and we use the notation χ_n for quarks and $\mathcal{B}_{n\perp}^{\mu}$ for gluons defined in subsection 2.1.1. We proved in subsection 3.4 that we need only three key building blocks to construct operators in SCET, these are: χ_n , $\mathcal{B}_{n\perp}^{\mu}$, and $\mathcal{P}_{n\perp}^{\mu}$, each of which carry a power counting scaling of $\mathcal{O}(\lambda)$. A general notation for the *i*-parton operators we will consider in this chapter is

$$\mathcal{O}^{(j,k,\ell)}(n_1^{[\ell_1]},\dots,n_{j+k}^{[\ell_{j+k}]}) = \left[\prod_{a=1}^{j/2} (\mathcal{P}_{n_a\perp})^{\ell_a} \chi_{n_a}\right] \left[\prod_{b=j/2+1}^{j} (\mathcal{P}_{n_b\perp})^{\ell_b} \bar{\chi}_{n_b}\right] \left[\prod_{c=1}^{k} (\mathcal{P}_{n_c\perp})^{\ell_c} g \mathcal{B}_{n_c\perp}\right],$$
(4.7)

where the number of partons is the sum of quarks and gluons, j + k = i, and the total number of \perp derivatives is $\ell = \sum_{m=1}^{j+k} \ell_m$. The arguments in brackets denote the collinear direction for each parton field, with the superscript denoting the number of \perp derivatives acting upon it. The power counting scaling of these operators is $\mathcal{O}^{(j,k,\ell)} \sim \lambda^{j+k+\ell}$. These operators are tensors in the space of spinors and Lorentz vectors, and the indices get contracted with structures contained in the Wilson coefficient C for the operator. If $C\mathcal{O}$ is a Lorentz scalar then j is even. Since the collinear fields carry a label referring to a specific light-cone vector, these operators describe particles in a specific region of phase space. We have seen in chapter 3 that the Wilson coefficients and the operators are in general convoluted. In this chapter, because we only treat tree-level quantities there are now convolutions between Wilson coefficients and operators and we do not need to use the notation in Eq.(3.5).

4.2 Obtaining the Parton Shower with SCET

4.2.1 Bauer-Schwartz Method

The original application of SCET to study and improve the parton shower was carried out in Refs. [16, 17] by Bauer & Schwartz. The main reasons why SCET is useful here are:

- The SCET fields, soft and collinear quarks and gluons, encode the infrared contributions which are exactly where the parton shower amplitudes have their dominant contributions in phase space.
- Since SCET is improvable order-by-order in the kinematic expansion parameter,
 λ, one has the potential to systematically correct the shower.

We will give a short overview of the Bauer-Schwartz approach, and then discuss the complications that arise when trying to extend the analysis to NLO in the λ expansion. In this subsection we will use notation that is not found elsewhere in this chapter to retain consistency with Refs. [16, 17].

The procedure of [16, 17] starts by constructing *i*-parton operators, \mathcal{O}_i , through matching SCET to QCD at a hard scale. For example, their \mathcal{O}_2 will equal $\mathcal{O}^{(2,0,0)}(n_1, n_2)$ in the notation of Eq. (4.7), and \mathcal{O}_3 will be $\mathcal{O}^{(2,1,0)}(n_1, n_2, n_3)$. As we run $\mathcal{O}_i(\mu)$ down, the leading log renormalization group evolution (LL RGE) does not mix operators and the exponential evolution operator encodes the no-branching probability. The evolution continues until another parton becomes apparent at a scale $\mu = p_T$.

If we have an *i*-parton operator, $\mathcal{O}_i = \mathcal{O}^{(j,i-j,0)}(n_1,\ldots,n_i)$ with all *n*'s distinct, then it has the RG solution $\mathcal{O}_i(Q) = U^{(j,i-j,0)}(Q,\mu)\mathcal{O}_i(\mu)$ with

$$U^{(j,i-j,0)}(Q,\,\mu) \,=\, \exp\left[-\int_{\mu}^{Q} \frac{d\mu'}{\mu'} \gamma^{(j,i-j,0)}(\mu')\right],\tag{4.8}$$

where $\gamma^{(j,i-j,0)}$ is the operator's anomalous dimension. The leading-log resummation effects of the Sudakov factor in the PS enter through one-loop operator running in SCET, as dictated by the cusp anomalous dimension. The one-loop cusp portion is especially easy to calculate in SCET as it depends solely on the number of collinear fields, even though the loop calculations do generically involve soft loops as well [16, 17],

$$\gamma_{\rm LL}^{(n_q, n_g, 0)}(\mu) = -\frac{\alpha_s}{\pi} \left[\frac{n_q}{2} C_F + \frac{n_g}{2} C_A \right] \log \frac{\mu^2}{Q^2} \,. \tag{4.9}$$

This form of the kernel yields a product of Sudakov factors which are the no-branching probabilities for each parton in the operator:

$$U_{\rm LL}^{(j,i-j,0)}(Q,\,\mu) \,=\, \Delta_q^{\frac{j}{2}}(Q,\,\mu) \Delta_g^{\frac{i-j}{2}}(Q,\,\mu). \tag{4.10}$$

Thus, in agreement with Ref. [31], we can account for leading-log effects for any particle multipleity by simply multiplying matrix elements by appropriate Sudakov factors.

As we run $\mathcal{O}_i(\mu)$ down, another parton becomes apparent at a scale $\mu = p_T$. To account for this, Bauer & Schwartz devised a "threshold matching" of \mathcal{O}_i to a new, higher multiplicity operator, $\mathcal{O}_{i+1}^{(i)}$, where the subscript still denotes the number of partons in the operator and the superscript tracks the parent operator. The general threshold matching equation is [16, 17]

$$\left[C_n^{(j)}\langle \mathcal{O}_n^{(j)}\rangle\right]_{\mu=p_T+\epsilon} = \left[C_{n+1}^{(j)}\langle \mathcal{O}_{n+1}^{(j)}\rangle\right]_{\mu=p_T-\epsilon}.$$
(4.11)

After more running and threshold matching, we eventually wind up with $\mathcal{O}_n^{(i)}$ for various n > i. The n - i particles emitted at increasingly lower scales by this process correspond to the parton showering of the original fields created at the hard scale by \mathcal{O}_i . Refs. [16, 17] also showed that an appropriate list of SCET operators (\mathcal{O}_i 's and $\mathcal{O}_i^{(n)}$'s) can interpolate between fixed-order QCD and parton shower (PS) calculations of IR-safe observables. Furthermore, Ref. [16, 17] carried out the important task of including $\mathcal{O}(\alpha_s)$ effects from matching QCD to SCET at one-loop.

That these subsequent emissions reproduce the usual parton shower splitting function emerges easily from SCET. Consider an operator $\mathcal{O}_i = \bar{\chi}_{n_0}\Omega$, where Ω is arbitrary and we have made explicit a single collinear quark field, $\bar{\chi}_{n_0}$. If we emit a collinear gluon from the quark, $q(q_0^{\mu}) \rightarrow q(q_1^{\mu})g(k_1^{\mu})$, the amplitude for the process is:

$$A_{LO}^{X+qg} = \bar{u}_n(q_1)\rho^{\alpha} \frac{\bar{q}_0}{q_0^2} \Omega, \qquad (4.12)$$

where u_{n_0} is the collinear quark spinor, and ρ^{α} is the combination of the SCET gluon emission Feynman rule plus the $\bar{\chi}_{n_0}$ Wilson line emission (the quark $\mathcal{L}_n^{(0)}$ can be found in Eq. (2.16)),

Note that ρ^{α} in SCET comes entirely from $\bar{\chi}_{n_0}$ without reference to anything residing in Ω . The subscript $(n_0 \perp)$ refers to components perpendicular to n_0^{μ} and \bar{n}^{μ} , which we denote by \perp for the remainder of this computation. The amplitude in Eq. (4.12) is gauge invariant and $k_1^{\alpha}\rho_{\alpha} = 0$. Squaring A_{LO}^{X+qg} and summing over spins we have $\sum_{spin} \bar{u}_n(q_1)u_n(q_1) = \bar{q}_1 n_0/2$, and the gluon polarization sum denoted $\sum_{spin} \epsilon_{\alpha} \epsilon_{\beta}^* = d_{\alpha\beta}$. Since ρ^{α} commutes with n_0 , we get an answer proportional to $\rho^{\alpha}\rho^{\dagger\beta} d_{\alpha\beta}$, where without loss of generality we can use a light-cone gauge, $d_{\alpha\beta} = -g_{\alpha\beta} + (\bar{n}_{\alpha}k_{1\beta} + k_{1\alpha}\bar{n}_{\beta})/\bar{k}_1$. Crucially, this is a Dirac scalar:

$$\rho^{\alpha} \rho^{\dagger \beta} d_{\alpha \beta} \equiv |\rho|^2 = 2 \left(\frac{2 q_0^2}{\bar{k}_1 \bar{q}_0} - \frac{q_{1\perp}^2}{\bar{q}_1^2} + \frac{2 q_{0\perp} \cdot q_{1\perp}}{\bar{q}_0 \bar{q}_1} - \frac{q_{0\perp}^2}{\bar{q}_0^2} \right) \times \mathbb{I}_4, \tag{4.14}$$

where we have used the on-shell conditions $q_1^2 = 0$ and $k_1^2 = 0$.

In a frame where $q_0^{\perp} = 0$ we have $q_{1\perp} = -k_{1\perp}$ and $\bar{q}_0/q_0^2 = 1/(n_0 \cdot q_0)$. Here $n_0 \cdot q_0 = n_0 \cdot k_1 + n_0 \cdot q_1 = -k_{1\perp}^2/[\bar{q}_0 z(1-z)]$, where $z \equiv \bar{q}_1/\bar{q}_0$. Thus we have the simpler expression

and

$$\rho^{\alpha} \rho^{\dagger \beta} d_{\alpha \beta} = 2 \left(\frac{2 n \cdot k_1}{\bar{k}_1} - \frac{2 q_{1\perp}^2}{\bar{q}_1 \bar{k}_1} - \frac{q_{1\perp}^2}{\bar{q}_1^2} \right) \times \mathbb{I}_4 = -\frac{2 k_{1\perp}^2}{\bar{q}_0^2} \frac{(1+z^2)}{z^2 (1-z)^2} \,. \tag{4.16}$$

Putting these properties together in the full amplitude squared we get

$$|A_{LO}^{X+qg}|^{2} = \frac{g^{2}C_{F}}{(n_{0}\cdot q_{0})^{2}} \frac{\bar{q}_{1}}{2} \operatorname{Tr} \left[\not{\!\!\!/}_{0} \rho^{\alpha} \Omega \Omega^{\dagger} \rho^{\dagger\beta} \right] d_{\alpha\beta} = \frac{g^{2}C_{F} \bar{q}_{1}}{(n_{0}\cdot q_{0})^{2}} |\rho|^{2} \operatorname{Tr} \left[\frac{\not{\!\!\!/}_{0}}{2} \Omega \Omega^{\dagger} \right]$$

$$= g^{2}C_{F} 2z \frac{(1+z^{2})}{|k_{1\perp}|^{2}} \operatorname{Tr} \left[\bar{q}_{0} \frac{\not{\!\!\!/}_{0}}{2} \Omega \Omega^{\dagger} \right].$$
(4.17)

Thus, all information about the emission factors out to the front of the amplitude squared and is independent of the rest of the process encoded by Ω . Unlike the analogous computation in full QCD there was no need to expand the amplitude to obtain this result. In order to recover Eq. (4.3), we still need to include the z-dependence from phase space, since $P_{jk}^{(0)}(z)$ operates at the level of the cross section. Using $d^3k/(2E_k) = dk^- d^2k_\perp/(2k^-)$, for q_1 and k_1 we have

$$\frac{dq_1^- d^2 q_{1\perp}}{2\bar{q}_1} \frac{dk_1^- d^2 k_{1\perp}}{2\bar{k}_1} = \frac{dq_0^- d^2 q_{0\perp}}{2\bar{q}_0} \frac{dz \, d^2 k_{1\perp}}{2z(1-z)} \,. \tag{4.18}$$

Thus we recover the expected 1/(1-z) dependence from the measure. Combining pieces and performing the trivial azimuthal integral $d\phi_{k_1}$, we get the expected expression:

$$d\sigma_{X+qg} = dz \, \frac{dk_{1\perp}^2}{k_{1\perp}^2} P_{q\to qg}^{(0)}(z) \, d\sigma_{X+q}, \qquad (4.19)$$

where $P_{q \to qg}^{(0)}(z)$ is the quark splitting function in Eq. (4.3). Here $d\sigma_{X+q}$ is the crosssection for the rest of the process with emission of a momentum q_0 quark, and the corresponding amplitude squared is $\operatorname{Tr}\left[\frac{\bar{q}_0}{2}\not{\pi}_0\Omega\Omega^{\dagger}\right]$. Whether Ω represents a simple hard current or an entire chain of collinear splittings, we see that the $q \to qg$ emission factors out with the expected logarithmic singularities, as in Eq. (4.2).

In order to obtain their results, Bauer and Schwartz introduced choices and approximations at several points which obscure the path toward systematically computing NLO corrections in the SCET power-counting parameter, λ . Indeed, they concluded that obtaining these corrections "... may be prohibitively difficult" [16]. Some of the issues one encounters trying to work at higher orders are:

1. At NLO, it becomes crucial to distinguish which simplifications correspond to approximations with power corrections, and which involve a choice of coordinates where the final answer is coordinate independent. For example, in general a collinear state has nonzero momentum components perpendicular to the index n of the field that annihilates it. In Refs. [16, 17] a choice was made to have collinear SCET fields in the operators only create particles whose momenta perfectly aligns with their index direction, n:

$$\chi_n |q\rangle = \delta_{n,n_q}$$
, where $n_q^\mu = q^\mu / E_q$. (4.20)

Eq. (4.20) enforces certain kinematical restrictions on final state particles, and required that fermions fields be rotated to an appropriate $n_q^{\mu} \operatorname{via} \xi_n \to (\#/4)\xi_{n_q}$.

2. At LO it was possible to avoid a potential double counting between collinear and soft fields by dropping soft emission and Wilson line emission, and taking only collinear emissions with transverse polarization. The threshold matching procedure is designed to avoid double counting of collinear operators, such as a Lagrangian emission from \mathcal{O}_2 and direct emission from $\mathcal{O}_3^{(2)}$, since only one of these is allowed to operate at a time. However, the threshold matching in Eq. (4.11) makes the technical procedure for incorporating power corrections unclear.

3. Threshold matching contains another impediment to systematic improvement. The basic idea is that the initial operator \mathcal{O}_2 has nonzero projection onto Fock states of any multiplicity, but the number of particles created by an operator is a scale-dependent question. The matching scales are determined by the strong ordering kinematics, $p_{1\perp} \gg \ldots \gg p_{m\perp}$. At the scale of an emission, say $p_{1\perp}$, one matches to the operator $\mathcal{O}_2^{(3)}$. However, going to higher orders in the shower necessitates encoding departures from the strong-ordering condition needed for this procedure.

There is also an interplay between some of these. For example, if we wish to construct a $q\bar{q}g$ operator that is obtained from a time-ordered product of the SCET lagrangian with \mathcal{O}_2 , then there is a kinematic conflict since the mother quark has nonzero invariant mass, but the quark in \mathcal{O}_2 is suppose to have null momentum to enforce Eq. (4.20). Encoding the emission in $\mathcal{O}_3^{(2)}$ via threshold matching avoids the conflict with the choice in 1. In carrying out their method, Bauer & Schwartz carefully enumerated the above approximations. They affect the ability to include corrections in λ , but do not impact the LO shower.

Building on the work of Refs. [16, 17], the main goal of the framework we develop in the next subsection is to overcome this list of issues so that we can determine the NLO corrections to the shower using SCET.

4.2.2 Using $SCET_i$

The main feature of the parton shower is the ability to capture the dominant physics of particles emitted in kinematically hierarchical regions of phase space. Our goal is to reformulate the SCET interface with the shower using a standard sequence of matching and running steps in different versions of SCET,

$$QCD \to SCET_1 \to SCET_2 \to \dots \to SCET_N.$$
 (4.21)

We refer to this as the SCET_i procedure. The key distinction between a SCET at one stage and the next is the definition of the corresponding resolution parameters $1 \gg \lambda_1 \gg \lambda_2 \gg \cdots \gg \lambda_N$, where λ^i is the power counting parameter of SCET_i. As we move down the chain, the corresponding SCET resolves smaller invariant masses $\sim (Q\lambda_j)^2$, and has a different meaning for its collinear sectors $\{[n_i]\}_{SCET_j}$. To keep track of this we will attach a subscript to the operators to denote the SCET_i in which its fields live,

$$\mathcal{O}_i^{(j,k,\ell)}(n_1,\ldots,n_{j+k}). \tag{4.22}$$

Effectively with Eq. (4.21), we partiton the momenta of particles in an event into classes,

$$\Omega_0 \supset \Omega_1 \supset \ldots \supset \Omega_N \,, \tag{4.23}$$

where Ω_j is defined to contain the momenta of all propogators or particles with invariant mass $p^2 \sim (Q\lambda_j)^2$ or smaller, or an equivalent condition on relative perpendicular momenta.¹ The allowed momenta in Ω_i correspond to the IR particles of SCET_i. The sequence of SCET_j's is truncated when we resolve a scale of order the parton shower cutoff, $Q\lambda^N = p_T^{\text{cut}} \simeq 1 \text{ GeV}$, that is in SCET_N.

The strongly ordered configuration of partons in Eq. (4.1) corresponds with removing a single q_j^2 in Ω_j as we pass from $\Omega_j \to \Omega_{j+1}$. However with Eq. (4.23) nothing stops us from having multiple emissions at a single scale. If two mother particles, with q_j^2 and q_{j+1}^2 , are associated to the same Ω_k , then when we integrate out the scale

¹Note that we do not associate a strong hiearchy to the hard scales \bar{p}_i in each SCET_i, and we consider $\bar{p}_i \sim Q$ despite the fact that the partons loose energy as they split and evolve. This does not imply that the hard scale for each SCET_j is fixed, but mearly that parametrically it does not tend to decrease as rapidly as the invariant masses encoded in the power counting parameter λ_j .

in SCET_k this configuration just contributes to an operator with a different parton multiplicity from the strongly ordered case. Thus, with the Eq. (4.21) setup there is no obstacle to considering corrections from an arbitrary assignment of q_j^2 's to Ω_k 's. This resolves issue 3. of Subsection 4.2.1.

To carry out calculations in the SCET_i framework it is convenient and sufficient to take a specific definition of the power counting parameters, $\lambda_i = (\lambda)^i$. The motivation for this is that we want the hierarchy between neighboring splittings to stay the same throughout the shower so as not to privilege any portion of it. We will see in Subsection 4.4.4 that this democratic setup allows us to interpret our $\mathcal{O}(\lambda)$ corrections to *i*-parton amplitudes as universal corrections to the splitting probability, given at LO by Eq. (4.3). Our definition of collinearity will change as we go to lower scales, and from Eq. (2.2) fields collinear to *n* within Ω_i have:

$$(n \cdot q_i, \bar{q}_i, q_{i\perp}) \sim (\lambda^{2i}, 1, \lambda^i) \bar{q}_i, \qquad (4.24)$$

and virtuality $\sim (\bar{q}_i)^2 \lambda^{2i}$. For convenience we use the same auxillary vector \bar{n}^{μ} defined for *n*-collinear fields in SCET₁ for all subsequent collinear fields in SCET_i's that descend from these *n*-collinear mothers. In SCET_i $\mathcal{L}_n^{(0)}$ again only couples collinear fields in the same direction *n*. Since different SCET_i's have different definitions of collinearity, our description of identical physical processes changes when we switch to a theory with a lower scale.

We depict this in Fig. 4-1, where the left panel is in SCET_i and the right panel is in SCET_{i+1} . In SCET_i , the quark $(\vec{q_1})$ and gluon $(\vec{k_1})$ are n_0 -collinear. This means that at LO they are emitted from a qqg vertex in the LO SCET_i Lagrangian (or a Wilson line). Schematically, the amplitude for a \perp -polarized gluon looks like²

$$A^{q\bar{q}g} = C_{q\bar{q}} \int dx \langle 0|T\{\mathcal{L}_{\text{SCET}_i}(x)\mathcal{O}^{q\bar{q}}\}|q\bar{q}g\rangle, \qquad (4.25)$$

namely like the first term in Eq. (2.22). The right-hand panel of Fig. 4-1 denotes the same configuration as seen by $SCET_{i+1}$. The scale of this theory is lower and

² From here on we will drop the superscript (0) and the subscript n from the collinear Lagrangian.



Figure 4-1: The same three-parton process as seen in two different SCET's, SCET_i and SCET_{i+1}. Above: Kinematic configuration of the quarks and gluon. The solid cones represent the regions considered collinear to the vectors drawn. Below: Feynman diagrams for the corresponding amplitude. Note that in SCET_{i+1} we have removed a propagating degree of freedom. The amplitude thus comes from a higher dimension operator $\mathcal{O}^{q\bar{q}g}$, rather than time-ordered product of $\mathcal{L}_{\text{SCET}_i}$ with the current $J^{q\bar{q}}$, as seen for SCET_i.

the definition of collinearity stricter, so the quark and gluon are not collinear here. Therefore, the amplitude now comes from an operator with three partons,

$$A^{q\bar{q}g} = C_{q\bar{q}q} \langle 0 | \mathcal{O}^{q\bar{q}g} | q\bar{q}g \rangle, \tag{4.26}$$

as in Eq. (2.23). As usual in EFT, continuity of the S-matrix is maintained by matching $SCET_i$ to $SCET_{i+1}$ to calculate $C_{q\bar{q}g}$.

Before moving on, we list some technical advantages to working in the SCET_i framework:

1. Collinear fields in SCET with different *n*-labels, as well as soft fields, do not overlap in Hilbert space. SCET provides built in mechanisms to avoid double counting that will be inherited by terms in the $SCET_i$ chain, such as zerobin subtractions [78]. This allows us to separate an *i*-jet process with *i* distinguished partons, from an (i - 1)-jet process with *i* partons, where two are collinear and unresolved. Lower-scale SCET_{*i*}'s distinguish processes more finely based on their stricter definition of collinearity. This resolves issue 2. from Subsection 4.2.1. This SCET property also illuminates simplified structures in the power corrections, such as the form of the amplitude interference (cf. subsection 4.4.4).

- 2. Soft modes communicate between collinear sectors and threaten the factorization of different jets. Fortunately, SCET constrains the interactions they have with collinear fields. In fact, one can decouple them with soft Wilson lines in the LO SCET Lagrangian. At LO we maintain factorization and derive the angular ordering and coherent branching of soft emissions (cf. subsection 4.3.2). Soft interactions which are power suppressed can also be systematically studied in SCET with Lagrangians available in the literature [15, 24, 25].
- 3. In SCET_i we have a symmetry group RPI_i which corresponds to coordinate choices. In SCET_{i+1} only a subset of this RPI_{i+1} \subset RPI_i remains a symmetry of the new theory. The kinematics in the coset portion RPI_i/RPI_{i+1} within SCET_i become a set of higher-dimension operators in SCET_{i+1}, and describe configurations which would not otherwise be contained in the SCET_{i+1} Lagrangian (cf. subsection 4.3 and appendix B.1). This resolves issue 1. from Subsection 4.2.1.
- 4. In matching between SCET_i and SCET_{i+1}, higher order operators in the lower-scale theory are needed to reproduce the physics of the higher one. We proved that all higher order purely collinear operators can be built from quark fields (χ_n), perpendicular gluon fields (B_{⊥n}), and the perpendicular momentum operators (P_{⊥n}) (cf. subsection 2.1.1). Thus the symmetries and equations of motion of SCET greatly simplify the operator basis one needs to consider at each order in λ (cf. subsection 4.4 and appendices B.2 and B.3).

The final SCET_N corresponds to the scale where the shower stops, *i.e.* where $Q\lambda^N \sim p_T^{\text{cut}}$. In SCET_N, we only require the coefficients of the operators where all

collinear partons belong to distinct sectors and which have no $\mathcal{P}_{n\perp}$'s, $C_N^{(j,k,0)}\mathcal{O}_N^{(j,k,0)}$. Here we have $n_j^{\mu} = p_j^{\mu}/\bar{p}_j$ for each parton that is each perticle is collinear to a different direction, this is as in Eq. (4.20), but we only do this for operators which have just one parton in an equivalence class. The more general operators with more than one parton in a single collinear direction and \perp -momenta are required for the steps through intermediate SCETs, to calculate NLO corrections. The coefficients $C_N^{(j,k,0)}$ encode the history of the shower, including branching and evolution, and can be written entirely in terms of: dot products $n_i \cdot n_{i'}$ (that is equivalent to products of physical parton momenta), hard momenta \bar{p}_i , and the renormalization scale of dimensional regularization, μ . The dot products of ns carry the power counting in λ . It is also worth mentioning that SCET maintains gauge invariance for power corrections, so the Wilson coefficients encoding the shower information are gauge invariant.

Our goal is to carry out the transition $SCET_i \rightarrow SCET_{i+1}$ with a standard matching procedure, and to find a correspondence between the $SCET_i$ setup and the information needed for carrying out the probabilistic parton shower. A traditional LO parton shower needs four basic ingredients: an evolution variable, no-branching Sudakov factors, the splitting function, and a notion of factorization yielding an iterative algorithm. Given a desired evolution variable, a correspondence must be set up with the $n_i \cdot n_{i'}$'s in the SCET_i procedure. At LO this is straightforward, but at NLO this will effect the manner in which the power corrections enter a probabilistic shower code. The no-branching Sudakov's are obtained by renormalization of operators in each $SCET_i$, with calculations that are along the same lines as those covered in detail in Refs. [16, 17]. At LO, only operators with distinct collinear sectors are required, while at NLO the renormalization becomes more complicated since operators may now contain two or more partons that are collinear in the same direction. The splitting functions appear in the matching between $SCET_i$ and $SCET_{i+1}$. At NLO, this includes $1 \rightarrow 3$ splitting functions as well as corrections where more partons appear at the start of the shower. The majority of the splittings in a tree still factorize at NLO, and the properties of SCET control complications in the interference.

In the above discussion we did not identify the parameter λ with an outcome in

the probabilistic shower. As far as the shower is concerned λ is merely a bookkeeping device which helps determine what pieces are needed beyond LO, and those pieces depend only on physical momentum and not on λ . One could try defining $\lambda_1 = k_{1\perp}/Q$, $\lambda_2 = k_{2\perp}/Q$, etc., but this is not ideal since probabilistically there is a chance for events where $k_{1\perp} \sim Q$ or $k_{1\perp} \sim p_T^{\text{cut}}$. The organization in Eq. (4.23) instead exploits the fact that on average showers are strongly-ordered. Our expansion in λ will then on average give a description of the most likely first deviations from strong-ordering. The correspondence between shower ingredients and ingredients in SCET_i implies a probabilistic interpretation, rather than fixed unique values for the scales $Q\lambda_i$. In the end, we pass to the shower the information obtained along the way to determining the operators of SCET_N to NNLO in λ at the amplitude-squared level. Again, what is important is to obtain the ingredients in the shower, and in this sense SCET_i is a crutch to derive the NLO shower ingredients and not a physical theory that outputs precisely the desired NLO shower.

Now that we have defined the basics of the $SCET_i$ setup and its advantages, we are in a position to give an operational overview of our results. Building on item 4 above, the bulk of this work is devoted to studying how one derives $SCET_{i+1}$ from $SCET_i$. We will find that many simplifications occur. We can write down the operator which reproduces the strongly-ordered parton shower in a closed form expression for an arbitrary number of partons (subsection 4.3.1). Improving the parton shower then simply corresponds to matching at higher orders. We will find one set of corrections that only involve fields near the hard-scattering process. One can determine them from matching to QCD amplitudes with a small number of partons (subsection 4.4.1). They represent a set of non-resummable contributions, which we can interpret as improving the initial condition from the hard-scale matrix element. We will also find a generalizable set of corrections for an arbitrary number of partons that admits a closed form expression (subsection 4.4.2). This leads to a correction of the substructure of the jet, and we can use it to cross-check it by known $\mathcal{O}(\alpha_s)$ corrections to the usual parton splitting functions. Lastly, when we construct the squared amplitude the interference between all of these operators in the $SCET_i$ picture is straightforward

(subsection 4.4.4).

It is also worth commenting explicitly on shower ingredients that we do not compute in this work. We only treat the case of a showering quark $q \rightarrow qg$ and for the abelian portion ($\propto C_F^2$). We have left gluon splittings, $g \rightarrow q\bar{q}$ and $g \rightarrow gg$ out of this analysis, though the extension to these cases should be straightforward. We have not computed the evolution factors at NLL required for a full NLL shower with a proper resummation of double logs. We have also not determined the full effect of NLO power corrections from subleading soft interactions, although we briefly examine the factorized structure of such corrections in subsection (4.3.2). Finally, and most importantly, we do not attempt here to develop a realistic numerical algorithm for implementing an NLL shower. These items are all left to future investigations.

4.3 Parton Shower in SCET via Operator Replacement

In the previous subsection, we presented our approach of using a series of EFTs, the $SCET_i$, to handle processes with a hierarchy of many scales. We will now use this technique to calculate the leading contribution to a series of collinear emissions, as occurs in the parton shower. Our ultimate goal is to incorporate corrections, but as a starting point we want to easily reproduce the strongly-ordered configuration Eq. (4.1). We can do this if we declare that in a shower, the i^{th} particle decomposes as:

$$(n \cdot q_i, \, \bar{q}_i, \, q_{i\perp}) \sim (\lambda^{2i}, \, 1, \, \lambda^i) \, Q, \tag{4.27}$$

and therefore has virtuality $q_i^2 \sim Q^2 \lambda^{2i}$. strongly-ordered *i*-gluon radiation in Fig. 4-6. This is exactly the same condition as Eq. (4.24), which we used to define the EFT, SCET_{*i*}. Thus, it is natural to treat the *i*th emission in the theory SCET_{*i*}.

To calculate the operators that describe *i* emissions in the strongly-ordered limit, we will perform a series of matchings $SCET_i \rightarrow SCET_{i+1}$. We will find that the most efficient way to describe the process at LO is to be in $SCET_i$ for *i*-parton radiation.


Figure 4-2: We construct the operators that reproduce strongly-ordered gluons through a series of matchings and emissions. *Horizontal* dashed arrows refer to the radiation of a gluon from a time-ordered product of the SCET_i lagrangian with the operator creating fields at the point marked by \otimes . *Diagonal* solid arrows denote the matching the process to a higher multiplicity operator in SCET_{i+1}.

Thus, we emit and match *i*-times in series, as shown by Fig. (4-2). At LO, we will show that one can implement this using an operator replacement rule. In the case of $q \rightarrow qg$ emission, it takes the form:

$$\chi_{n_1} \to c \, g \mathcal{B}^{\alpha}_{n_3 \perp} \, \chi_{n_2} \,, \tag{4.28}$$

where χ_n and $\mathcal{B}_{n\perp}^{\alpha}$ are the SCET fields associated with collinear quarks and gluons respectively, and C is the Wilson coefficient whose indices are suppressed. Though we do not compute them, there should be similar $g\mathcal{B}_{n_1\perp}^{\alpha} \to c' \bar{\chi}_{n_2}\chi_{n_3} + c'' g\mathcal{B}_{n_2\perp}^{\beta}g\mathcal{B}_{n_3\perp}^{\gamma}$ rules as well. In SCET, each collinear field carries the label n, which gives its direction of collinearity. Note that the quark field on the LHS of (4.28) has a different one from those on the RHS. This relates to the stricter definition of collinearity in



Figure 4-3: The opening angle of the light grey (blue) cone is λ^{2i} , and the aperture of the dark grey (red) one is $\lambda^{2(i+1)}$. The particle with momentum p is collinear to both n and n' in SCET_i, but only to n' in SCET_{i+1}. RPI in SCET_i allows us to move the index direction, n, anywhere inside the appropriate collinear cone while keeping the theory invariant.

 $SCET_{i+1}$ shown in Fig. 4-3. In order to perform the matching we will make use of the reparametrization invariance (RPI) to change fields' index labels, n.

4.3.1 LO Shower Revisited

We first want to reproduce the strongly-ordered contribution to *i*-gluon radiation from the quark in $q\bar{q}$ pair production. Our iterative matching procedure for multiple EFTs takes a particularly simple form at LO (LO, NLO, *etc.* refer to the expansion in λ). For our standard example, we take the process $e^+e^- \rightarrow \text{jets}$. Starting in QCD, we couple the quarks to another sector via the operator, $J^{\mu}_{\text{QCD}} = \bar{q} \Gamma^{\mu} q$. This allows us to avoid complications that come from the initial state such a backward evolution. In SCET₁ (which is equivalent to the usual SCET), matching to QCD at tree-level converts the quark coupling to the following operator at LO: $\bar{\chi}_{n_0}\Gamma^{\mu}\chi_{\bar{n}}$, which produces q and \bar{q} in different collinear directions, for details on the matching QCD to SCET₁ see appendix B.2. Using the notation in Eq. (4.7) we write the SCET₁ operator in the following way:

$$\bar{\chi}_{n_0} \Gamma^{\mu} \chi_{\bar{n}} = \left(C_{1, \text{LO}}^{(2,0,0)} \right)_{ij} \left(\mathcal{O}_1^{(2,0,0)}(n_0, \bar{n}) \right)_{ij}$$
(4.29)

where

$$\left(\mathcal{O}_{1}^{(2,0,0)}(n_{0},\bar{n}) \right)_{ij} = (\bar{\chi}_{n_{0}})_{i} (\chi_{\bar{n}})_{j},$$

$$\left(C_{1,\,\text{LO}}^{(2,0,0)} \right)_{ij} = (\Gamma^{\mu})_{ij}.$$

$$(4.30)$$

and *i* and *j* are spinor indices. The subscripts 1 in Eq. (4.30) means that the fields are defined in SCET₁. Our focus is on gluon emissions from the quark, and we always take the antiquark in the same direction, \bar{n} . We can therefore use the following shorthand notation:

$$\mathcal{O}_i^{(2,k,0)}(n_1, n_1', \dots, n_k', \bar{n}) \equiv \mathcal{O}_i^{(k)}(n_1, n_1', \dots, n_k'), \qquad (4.31)$$

where the subscript indicates that the operators are defined in SCET_i . In the rest of this chapter we will drop the spinor indices to make the notation more readable. Using the above convention we write the operator in Eq. (4.29) as

$$\bar{\chi}_{n_0} \Gamma^{\mu} \chi_{\bar{n}} = C_{1, \, \mathrm{LO}}^{(0)} \mathcal{O}_1^{(0)}(n_0) \,. \tag{4.32}$$

The LO derivations are independent of the exact structure of Γ^{μ} . In fact, even the antiquark is a spectator, and we could just as easily use $\mathcal{O}^{(q)} = \bar{\chi}_{n_0}\Omega$, where Ω is arbitrary. However, as we will discuss in sec. 4.4, matching QCD to SCET₁ at higher orders requires us to specify Ω .

We start with single gluon radiation. In this case, shown in Fig. 4-4, the emission amplitude is³:

$$A_{\rm LO}^{q\bar{q}g} = C_{1,\rm LO}^{(0)} \langle 0| \int dx \, T\{\mathcal{L}^{\rm SCET_1}(x)\mathcal{O}_1^{(0)}(n_0)\} | q_{n_0} g_{n_0} \bar{q}_{\bar{n}} \rangle$$
(4.33)

where we have now labelled the collinear directions of the particles in the state

³All the amplitudes we write in this work refer only to the hadronic part of $e^+e^- \rightarrow \text{jets}$, thus $A_{\text{LO}}^{q\bar{q}g}$ is the amplitude of $\gamma^* \rightarrow q\bar{q}g$.



Figure 4-4: Vector labels for single (A) and double (B) gluon emission.

 $|q_{n_0}g_{n_0}\bar{q}_{\bar{n}}\rangle$ for later convenience. The SCET₁ Lagrangian is given in Eq. (2.16). Here we study the process in the center of mass frame with $p_{\gamma} = (Q, 0, 0, 0)$ and the quark (q_0) and and antiquark $(p_{\bar{q}})$ along the directions $n_0 = (1, 0, 0, 1)$ and $\bar{n} = (1, 0, 0, -1)$, respectively:

$$p_{\gamma}^{\mu} = \frac{Q}{2} n_{0}^{\mu}, + \frac{Q}{2} \bar{n}^{\mu}$$

$$p_{\bar{q}}^{\mu} = \frac{n_{0} \cdot p_{\bar{q}}}{2} \bar{n}^{\mu},$$

$$q_{0}^{\mu} = \frac{\bar{q}_{0}}{2} n_{0}^{\mu} + \frac{n_{0} \cdot q_{0}}{2} \bar{n}^{\mu}.$$
(4.35)

We decompose the emitted quark (q_1) and gluon (k_1) along the directions (n_0, \bar{n})

$$q_{1}^{\mu} = \frac{\bar{q}_{1}}{2} n_{0}^{\mu} + (q_{1})_{n_{0}\perp}^{\mu} + \frac{n_{0} \cdot q_{1}}{2} \bar{n}^{\mu}, \qquad (4.36)$$
$$k_{1}^{\mu} = \frac{\bar{k}_{1}}{2} n_{0}^{\mu} + (k_{1})_{n_{0}\perp}^{\mu} + \frac{n_{0} \cdot k_{1}}{2} \bar{n}^{\mu}.$$

The variables are defined in Fig. 4-4. By momentum conservation we have $(k_1)_{n_0\perp} = -(q_1)_{n_0\perp}$, $Q = \bar{q}_0 = \bar{k}_1 + \bar{q}_1$ and $n_0 \cdot p_{\bar{q}} = Q - n_0 \cdot q_1 - n_0 \cdot k_1$. We take all the external particle on -shell, thus $n_0 \cdot q_1 = -(q_1)_{n_0\perp}^2/\bar{q}_1$ and similarly for $n_0 \cdot k_1$. As we discussed in subsection 4.2.1, [16, 17] showed that single gluon emission in SCET reproduces the splitting function (Eq. (4.3)) and factorization behavior (Eq. (4.17)) of the standard parton shower. Take this simple behavior of a single radiation will reproduce the shower for an arbitrary number of gluons.

We proceed to two-gluon emission as this demonstrates the benefit of going to a



Figure 4-5: Left panel: Single gluon emission in SCET₁ comes from the time-ordered product of the lagrangian with a quark-creating operator, $A = \langle 0|T\{\mathcal{L}_{\text{SCET}_1}\mathcal{O}_1^{(0)}\}|q\bar{q}\rangle$. Right panel: For parent quarks which are too virtual for SCET₂, the gluon comes from the central vertex via a higher-dimensional operator, $A = \langle 0|\mathcal{O}_2^{(1)}|qg\bar{q}\rangle$.

lower-scale EFT as more gluons are radiated. If $k_{1\perp}$ and $k_{2\perp}$ are the perpendicular momenta with respect to their parents of the gluons emitted first and second, then strong ordering dictates that $k_{2\perp} \ll k_{1\perp}$. SCET₁ makes no parametric difference between the two, as all perpendicular momenta are assigned a factor λ . Since SCET₂ has a stricter definition of collinearity (a cone size of $\mathcal{O}(\lambda^2)$), it can distinguish that of the two fields. This allows us to integrate out the parent of k_1 , as its offshellness is too hard $\mathcal{O}(Q^2\lambda^2)$ for SCET₂ and match to a higher-dimension operator. In this way, we remove information unneeded to reproduce strong-ordering. By systematically putting it back (subsection 4.4.2), we can improve upon the standard strongly-ordered parton shower.

We want to describe the less collinear emission (k_1) as coming from a higherdimension SCET₂ operator (Fig. 4-5), in this case $C_{2,LO}^{(1)}\mathcal{O}_2^{(1)}(n_1,n_1')$. To find it, we just need to find the SCET₂ operator that reproduces the amplitude for single gluon emission in Eq. (4.34). As one can see, Eq. (4.34) makes explicit reference to the label direction n_0 . In SCET₁, this labels the virtual quark (q_0) , the real quark (q_1) , and the gluon (k_1) directions. However, as we have discussed (cf. Fig. 4-1), we have a tighter definition of collinearity in SCET₂, such that q_1 and k_1 cannot share the same label direction with each other or q_0 , which is not even a propagating degree of freedom in the theory. Spinors in SCET have a label dependence, and so we can have different spinors for q_1 according to whether we are in SCET₁ or SCET₂. This makes it difficult to match at the level of an unintegrated quantity such as an amplitude. The momentum p decomposed along n' and \bar{n} is This leads to a technical complication in the computation, but one that is easy to surmount, making use of finite reparametrization invariance transformations that allow us to rotate the label vector while keeping the theory invariant, Fig 4-3. We define reparametrization invariant in subsection 2.2 and we construct finite RPI transformation in appendix B.1. We define new labels n_1 and n'_1 as the basis where the final quark, and gluon respectively have zero perpendicular momentum,

$$q_{1} = \bar{q}_{1} \frac{n_{1}}{2},$$

$$k_{1} = \bar{k}_{1} \frac{n_{1}'}{2}.$$
(4.37)

The momenta are especially simple in this basis, q_1 and k_1 are on-shell, which means $n_1 \cdot q_1 = n'_1 \cdot k_1 = 0$. In SCET₁, we are free to use n_0 or n_1 to describe the q_1 quark and k_1 gluon because we can use the RPI symmetry to transform between this choice and other equivalent choices. Whatever choice we make the results for the Wilson coefficients in the matching will be the same. Since n_1 is also a valid index for the field in SCET₂, we can do the matching computation using the same spinor, $u_{n_1}(q_1)$, in both theories. The following results show how to express a spinor in the n_1 direction in terms of one in the n_0 direction and how to rotate the vector n_0 to $n_1(n'_1)$ such that the quark(gluon) field with perpendicular momentum $(q_1)_{n_0\perp}(k_1)_{n_0\perp}$ in the n_0 -basis has 0 perpendicular momentum in the $n_1(n'_1)$ -basis:

$$u_{n_{0}} = \frac{\not{n}_{0} \vec{\pi}}{4} u_{n_{1}},$$

$$n_{1}^{\alpha} = n_{0}^{\alpha} + \frac{2(q_{1})_{n_{0}\perp}}{\overline{q}_{1}} - \frac{(q_{1})_{n_{0}\perp}^{2}}{\overline{q}_{1}^{2}} \bar{n}^{\alpha},$$

$$n_{1}^{\prime \alpha} = n_{0}^{\alpha} + \frac{2(k_{1})_{n_{0}\perp}}{\overline{k}_{1}} - \frac{(k_{1})_{n_{0}\perp}^{2}}{\overline{k}_{1}^{2}} \bar{n}^{\alpha}.$$

$$(4.38)$$

The two different n_i -vectors' directions lie within cones of size λ , and so we apply the transformation in SCET₁. It is simple to check that in the new basis, $(q_1)_{n_1\perp} =$ $q_1 - (n_1 \cdot p)\bar{n}/2 - \bar{q}_1 n_1/2 = 0$ and similarly for $(k_1)_{n'_1 \perp}$. In appendix B.1 we derive Eq. (4.38) and other rotation formulas. Acting on Eq. (4.34), we get:

where $q_0 = q_1 + k_1$. Terms with subscript n_0 , n_1 , n'_1 are in the frame defined by q_0 , q_1 , k_1 , respectively. Having changed bases, we can easily write the SCET₂ operator that reproduce the amplitude (4.39) $C_{2,\text{LO}}^{(1)}(\mu)\mathcal{O}_2^{(1)}(n_1,n'_1)$, where:⁴

$$\mathcal{O}_{2}^{(1)}(n_{1},n_{1}') = (\bar{\chi}_{n_{1}})_{j} g \mathcal{B}_{n_{1}'\perp}^{\alpha}(\chi_{\bar{n}})_{k} , \qquad (4.40)$$

$$C_{2,LO}^{(1)} = U_{LL}^{(0)}(n_{0})(Q,\mu_{1}) \left[\frac{\bar{q}_{0}}{q_{0}^{2}} \left(n_{0}^{\alpha} + \frac{(\not{q}_{1})_{n_{0}\perp}\gamma_{n_{1}'\perp}^{\alpha}}{\bar{q}_{1}} \right) \frac{\not{n}\not{n}_{0}}{4} \Gamma^{\mu} \right]_{jk} \Theta_{\lambda^{2}}[n_{1} \cdot n_{1}'] . \qquad (4.41)$$

We have included in the Wilson coefficient the leading log RG-kernel $U_{LL}^{(0)}$ that comes from running down the SCET₁ operator $\mathcal{O}_1^{(0)}$ from the scale Q to the scale $\mu_1 \sim \lambda Q$ where we have the emission. From Eq. (4.10) we have

$$U_{\rm LL}^{(0)}(n_0)(Q,\mu_1) = \Delta_q(Q,\mu_1), \qquad (4.42)$$

where we have simplified the superscript as in Eq. (4.31), that is

$$U_{\rm LL}^{(2,k,0)}(n_1, n'_1, \dots, n'_k, \bar{n}) \equiv U_{\rm LL}^{(k)}(n_1, n'_1, \dots, n'_k).$$
(4.43)

When $U_{\rm LL}$ refers to an operator where all the collinear directions are distinct, we will drop ns from the notation. In the interpretation of this factor as a Sudakov, we see that it accounts for the non-branching from the scale Q to the μ_1 of the emission. We can write the coefficient $C_{2,\rm LO}^{(1)}$ only in terms of scalar product of ns, this is shown in Eq. (B.53) in appendix B.3, for the running coefficient we can take $\mu_1 = \sqrt{|n_1 \cdot n_1'|}Q$.

The function $\Theta_{\lambda^2}[n_1 \cdot n'_1]$ in Eq. (4.41) is a function that encodes the information

⁴ See appendix B.3 for more detail on this matching.

that $(n_1 \cdot n'_1) \leq \lambda^2$. The SCET₂ operator $\mathcal{O}_2^{(1)}(n_1, n'_1)$ knows that the gluon and the quark are not collinear in SCET₂, that is $q_1 \cdot k_1 > \lambda^4 Q^2$ (or $n_1 \cdot n'_1 > \lambda^4$), but it does not know that they were collinear in SCET₁. We calculated the coefficient $C_{2,\text{LO}}^{(1)}$ in a phase space region where $q_1 \cdot k_1 \leq \lambda^2 Q^2$ (or $n_1 \cdot n'_1 \leq \lambda^2$) in the matching to the SCET₁ $\mathcal{O}_1^{(0)}(n_0)$ and this information must be encoded in the Wilson coefficient. We encode that $n_1 \cdot n'_1 \leq \lambda^2$ using the Θ function in $C_{2,\text{LO}}^{(1)}$. For a LL shower we can take $\Theta = 1$, because there is not problem of overlap between different regions. However, once we start considering power corrections, we will need Θ to turn on and off particular regions of phase space. For later convenience we define the complement to the Θ function as

$$\tilde{\Theta}_{\lambda^k}[n_i \cdot n_j] = 1 - \Theta_{\lambda^k}[n_i \cdot n_j].$$
(4.44)

Schematically, we have

$$\Theta_{\lambda^{k}}[n_{i} \cdot n_{j}] = \begin{cases} 1 & n_{i} \cdot n_{j} \lesssim \lambda^{k} \\ 0 & n_{i} \cdot n_{j} \gtrsim \lambda^{k} \end{cases}$$

$$(4.45)$$

In principle, we could choose Θ to be the usual step-function, θ , but for practical integration, it is better to define a smoothed step. We give an example of such a function in Eq. (B.61), and plot it in Fig. 4-9.

The coefficient $C_{2,\text{LO}}^{(1)}$ has a power counting in λ because of the presence of multiple SCETs. This is different from what happens in a usual SCET matching where all the coefficients are of order λ^0 . $C_{2,\text{LO}}^{(1)}$ has an overall weight λ^{-1} . We get λ^{-2} from the SCET₁ propagator, $1/q_0^2$. The numerator is proportional to λ and comes from the vertex: $\left(n_{q_0}^{\alpha} + (\not{q}_1)_{n_0\perp}\gamma_{n_1'\perp}^{\alpha}/\bar{q}_1\right)$. The second term is $\mathcal{O}(\lambda)$ because of the power counting of $(\not{q}_1)_{n_0\perp} \sim \lambda$. Since n_0^{α} also gets contracted with $\mathcal{B}_{n_1'\perp}^{\alpha}$, it only contributes its perpindicular component in the n_1' frame. From Eq. (4.38), we see that $(n_0)_{n_1'\perp} \sim$ $n_0 - n_1' \sim (k_1)_{n_0\perp}/\bar{k}_1 \sim \lambda$. The vertex thus contributes the power of λ corresponding to the perpendicular momentum of the daughter with respect to its mother.

We can obtain $C_{2,\text{LO}}^{(1)}\mathcal{O}_2^{(1)}(n_1,n_1')$ from the original two-parton operator, $C_{1,\text{LO}}^{(0)}(n_0)\mathcal{O}_1^{(0)}(n_0)$,

in two steps: first we multiply it by the running factor $U_{LL}^{(0)}(n_0)(Q,\mu_1) = \Delta_q(Q,\mu_1)$, where the Δ_i are given in relation to the RG kernel in Eqs. (4.9-4.10), and second we apply the replacement rule

$$(\bar{\chi}_{n_0})_i \to (c^{\alpha}_{\rm LO}(n_0))_{ji} (\bar{\chi}_{n_1})_j g \mathcal{B}^{n'_1 \perp}_{\alpha},$$
 (4.46)

where $c_{\rm LO}^{\alpha}$ is:

$$c_{\rm LO}^{\alpha}(n_0) = \frac{\bar{q}_0}{q_0^2} \left(n_0^{\alpha} + \frac{(q_1)_{n_0 \perp}^{\mu} \gamma_{n_1' \perp}^{\alpha}}{\bar{q}_1} \right) \frac{\not{n} \not{n}_0}{4} \Theta_{\lambda^2}[n_1 \cdot n_1'] \,. \tag{4.47}$$

The relation (4.46) is the operator statement of splitting in the parton shower. The scale μ_1 defines the endpoint of running in the UV theory. As we evolve down, more partons become apparent. We can see this here by the presence of two fields where there had been one. It makes the basic aspects of the shower manifest. The replacement rule affects the quark alone, and so we see that the amplitude for splitting factorizes off from the rest of the process. The RG kernel gets modified to reflect the changing no-branching probabilities. We can interpret the vertex portion of $c_{\rm LO}^{\alpha}$ as the "square root" of the splitting function. The spinor projector ($\# \eta_0/4$) in Eq. (4.47) rotates the spin-sum from $\#_1$ to $\#_0$ in accordance with Eq. (4.2). The remaining part of $c_{\rm LO}^{\alpha}$ after stripping off the Sudakovs is:

$$P_{\alpha} \equiv \frac{\bar{q}_0}{q_0^2} \left(n_0^{\alpha} + \frac{(\not{q}_1)_{n_0 \perp} \gamma_{n'_1 \perp}^{\alpha}}{\bar{q}_1} \right), \qquad (4.48)$$

which squares to a trivial Dirac structure. Furthermore, even though $\rho_{\alpha}(\bar{q}/q_0^2) \neq P_{\alpha}$ because of the RPI rotations we performed (where ρ is defined in Eq. (4.15)), where we square $|\rho|^2(\bar{q}_0/q_0^2)^2 = |P|^2$ with respect the gauge polarization sum, $d_{\alpha\beta}$, so

$$|P|^2 = \frac{1+z^2}{q_{1\perp}^2}.$$
(4.49)

Just as before, including the z-dependence from the measure and spin-sum, we recover the the standard splitting function $\propto (1 + z^2)/(1 - z)$. Thus, c_{LO}^{α} weights the probability assigned to the expectation value of $C_{2,\text{LO}}^{(1)}(\mu)\mathcal{O}_2^{(1)}(n_1,n_1')$ appropriately.

Having computed the LO result for a single gluon, it is straightforward to proceed to an arbitrary number of emissions. In SCET₂, we know that two-gluon emission comes from the *T*-product of the lagrangian with $C_{2,LO}^{(1)}(n_1, n'_1)\mathcal{O}_2^{(1)}(n_1, n'_1)$. We can proceed by emitting a third with another SCET₂ lagrangian insertion. The situation is now exactly parallel to when we were considering two-gluon emission in SCET₁. Once again, we are carrying around more information than we need to reproduce the strongly-ordered contribution, so we should integrate out the scale $\sim Q\lambda^2$ to get a new theory, SCET₃, that has a two-gluon operator, $C_{3,LO}^{(2)}(n_2, n'_1, n'_2)\mathcal{O}_3^{(2)}(n_2, n'_1, n'_2)$ at LO. Similarly to before, the amplitude has the contribution,

$$A_{\rm LO}^{q\bar{q}gg} = C_{2,\rm LO}^{(1)}(\mu) \langle 0| \int dx \, T\{\mathcal{L}_{\rm SCET_2}(x) \mathcal{O}_2^{(1)}(n_2, n_1')\} |q_{n_2} g_{n_2} g_{n_1'} \bar{q}_{\bar{n}} \rangle \,. \tag{4.50}$$

The vertex for gluon emission in the SCET₂ lagrangian is identical to that in SCET₁. Thus, integrating out the parent of the Lagrangian-emitted, second-most collinear gluon, we obtain a two-gluon SCET₃ operator exactly as before. Similar to the matching SCET₁ to SCET₂, we can obtain the SCET₃ operator $C_{3,LO}^{(2)}\mathcal{O}_3^{(2)}(n_2, n'_1, n'_2)$ from the SCET₂ $C_{2,LO}^{(1)}(\mu)\mathcal{O}_2^{(1)}(n_1, n'_1)$, just multiplying it by the running factor for $\mathcal{O}_1^{(1)}$,

$$U_{\rm LL}^{(1)} = \Delta_q(\mu_0, \mu_1) \,\Delta_g^{1/2}(\mu_0, \mu_1) \,, \tag{4.51}$$

and applying the replacement rule

$$(\bar{\chi}_{n_2})_i \to (c_{\rm LO}^{\alpha}(n_1))_{ji} (\bar{\chi}_{n_3})_j g \mathcal{B}_{\alpha}^{n'_2 \perp}, \qquad (4.52)$$

$$c_{\rm LO}^{\alpha}(n_1) = \frac{\overline{q}_1}{q_1^2} \left(n_1^{\alpha} + \frac{(\not{q}_2)_{n_1 \perp} \gamma_{n'_2 \perp}^{\alpha}}{\overline{q}_2} \right) \frac{\not{n} \not{n}_1}{4} \Theta_{\lambda^4} [n_2 \cdot n'_2],$$

where n_2 and n'_2 are the direction where the quark and the second emitted gluon are collinear, they are defined in appendix B.1. One can iterate this procedure to obtain the LO result for N-gluon emission. If we use the replacement rule N-1 times we go



Figure 4-6: Left panel: our kinematical convention for a strongly ordered process. Quark momenta are denoted by q_i and gluon momenta by k_i . Right panel: power counting of the LO coefficient in SCET_N. The powers of λ with negative exponents refer to the propagator contribution to the amplitude. Those with positive exponents refer to the perpendicular momentum of the gluon with respect to its parent, since SCET vertices are proportional to it.

down to SCET_N operator $C_{N,LO}^{(N-1)}\mathcal{O}_N^{(N+1)}(n_N, n'_1, \ldots, n'_{N+1})$, after which Lagrangian emissions are no longer distinguished as separate particles. We have⁵

$$\mathcal{O}_{N}^{(N-1)}(n_{N}, n_{1}', \dots, n_{N-1}') = \bar{\chi}_{n_{N}} \left(\prod_{k=1}^{N-1} g \mathcal{B}_{\alpha_{k}}^{n_{k}' \perp} \right) \chi_{\bar{n}} , \qquad (4.53)$$

$$C_{N, \text{LO}}^{(N-1)} = \left(\prod_{k=1}^{N-1} U_{\text{LL}}^{(k-1)}(\mu_{k-1}, \mu_{k}) c_{\text{LO}}^{\alpha_{k}}(n_{k-1}) \right) \Gamma^{\mu} , \qquad (4.53)$$

$$c_{\text{LO}}^{\alpha_{k}}(n_{k-1}) = \frac{\bar{q}_{k-1}}{q_{k-1}^{2}} \left(n_{k-1}^{\alpha_{k}} + \frac{(q_{k\perp})_{n_{k-1}} \gamma_{n_{k}'\perp}^{\alpha_{k}}}{\bar{q}_{k}} \right) \frac{\bar{n} m_{k-1}}{4} \Theta_{\lambda^{2k}} [(n_{k} \cdot n_{k}')] , \qquad U_{\text{LL}}^{(k-1)}(\mu_{k-1}, \mu_{k}) = \Delta_{q}(\mu_{k-1}, \mu_{k}) (\Delta_{g}(\mu_{k-1}, \mu_{k}))^{(k-1)/2} ,$$

Where the variables for *i* emission are defined in Fig. 4-6, $\mu_0 = Q$, $\mu_k = \sqrt{|n_k \cdot n'_k|}Q$ and $q_k = \left(q_i + \sum_{j=k-i}^n k_j\right)^2$. n_k and n'_k are the directions where the quark q_k and the external quark k_k have not perpendicular component, they can be related to n_{k-1} through reparametrization invariant (*cf.* appendix B.1). We can extend the argument we used to calculated the power counting of $C_{2,\text{LO}}^{(1)}$ to the SCET_N coefficient in Eq. (4.53), just counting the contributions form vertices and the propagators,

⁵ Because we only analyze gluon emissions from a quark, we have at SCET_N exactly N-1 gluon emissions, so the SCET_N operator is $\mathcal{O}_{N-1}^{(N)}$.

 $C_{N,\text{LO}}^{(N-1)}$ has a power counting $\sum_{i}^{N-1} 1/\lambda^{-i} = \lambda^{-N(N-1)/2}$, see Fig. (4-6).

Similarly to the discussion above Eq. (4.49), we can extract the vertex part of $c_{\text{LO}}^{\alpha_k}$ to define P^{α_k} . We get that

$$|P^{\alpha_k}|^2 = \frac{1+z_k^2}{q_{k\perp n_{k-1}}^2},\tag{4.54}$$

where $z_k \equiv \bar{q}_k/\bar{q}_{k-1}$. Thus, we get that the amplitude squared goes like the factorized product of the appropriate $1 \to 2$ splitting functions. Since $\mathcal{O}_i^{(i-1)}(n_{i-1}, n'_1, \ldots, n'_{i-1})$ is just built up from the repeated use of Eq. (4.46), we see that it requires no added information after we compute the first $q \to qg$ splitting. Thus, what we need to pass to a shower algorithm comes just from single real and single virtual gluon computations, as we list below in Subsection 4.3.3 in Table 4.1. Collinear splitting is entirely handled by the splitting rule in Eq. (4.46) at LL order.⁶

We have seen that to describe LO parton shower we only need operators with χ_n and \mathcal{B}_n fields, but when we go to NLO, we will see that we can also have operator with $\mathcal{P}_{n\perp}$. In the final SCET_N where all χ_n and \mathcal{B}_n fields have their own n_i directions we do not need operators with factors of $\mathcal{P}_{\perp n_i}$ in the operator. Such operators encode redundant information that is specified by RPI. That is, in SCET_N we only have the operator $\mathcal{O}_N^{(N-1)}(n_N, n'_1, \ldots, n'_{N-1})$ and all the corrections are encoded in the Wison coefficients.

4.3.2 Soft emissions

SCET describes soft degrees of freedom using soft quark and gluon field: $q_s(x)$ and $A_s(x)$. In this work we focus on fully differential cross section where we can always distinguish collinear and soft modes, but in an integrated cross section we have to

⁶It is straightforward to see that we do not have additional contributions at LO. Firstly, consider the possibility of operators that do not take the form of a single-field replacement rule. These would depend on the details of the hard process that produced the quark in the first place and could threaten the factorization of the shower. In fact, we will get such terms when we match QCD \rightarrow SCET₁, but they are always suppressed, as we discuss in sec. 4.4. Returning to single-field replacement, let us consider matching SCET₁ \rightarrow SCET₂, as results in this case will generalize to all SCET_i. Rule (4.46) sends $\chi_{n_1} \rightarrow C\mathcal{B}^{\mu}_{n_3 \perp} \chi_{n_2}$. At LO, we cannot get such a replacement involving multiple gluon fields, $\mathcal{B}_{n_j \perp}$, as this implies that we have integrated out multiple, hard ($\sim Q\lambda^2$) propagators. Such a contribution would not be strongly ordered, and is suppressed. We will see in sec. 4.4 that we do have such contributions at higher orders.

implement soft emissions with some form of zero-bin subtractions [78] to avoid interference between soft and collinear radiation. As we have seen in subsection 2.1, the collinear sector and the soft sector couple through the covariant derivative λ^2 , $iD_s^{\mu} = i\partial^{\mu} + gA_s^{\mu}$, (Eq. (2.13)) acting on the collinear fields. At LO the collinear particles only couple to the $n \cdot A_s$ component of the soft gluons and the soft-collinear factorization guarantees that we can absorb all this couplings into Wilson line Y(x)along the direction of the collinear particle, Eq. (2.15). In SCET this is accomplished by making the field redefinitions in Eqs. (2.14), so the new collinear fields no longer couple to soft gluons through their kinetic term. The outcome for the composite fields considered here is that

$$\chi_n \to Y_n \chi_n , \qquad \mathcal{B}_n^\mu \to Y_n \mathcal{B}_n^\mu Y_n^\dagger .$$

$$(4.55)$$

Note that here we consider nonabelian soft interactions which is why the soft Wilson lines do not cancel for the \mathcal{B}^{μ}_{n} field.

In matching SCET_i to SCET_{i+1} we will only consider external soft modes in SCET_{i+1} with momenta $k^{\mu} \sim Q\lambda^{2(i+1)}$. These modes are contained within the soft modes in SCET_i. We do not consider particles with soft momenta $k^{\mu} \sim Q\lambda^{2i}$ that could not be encoded by onshell modes in SCET_{i+1}. Since such modes are forced to have larger momenta than the soft fields in SCET_{i+1} they are not responsible for IR divergences, and any contributions from momenta of this type can be encoded in the Wilson coefficients of our SCET_{i+1} operators with its field content.

In a given $SCET_i$ after making the field redefinition the effect of soft gluons is encoded by Wilson lines Y_n in the operators, with the form

$$\bar{\chi}_{n_N}^{(0)} Y_{n_N}^{\dagger} \prod_{k=1}^N Y_{n'_k} \mathcal{B}_{n'_k \perp}^{(0)\alpha_k} Y_{n'_k}^{\dagger} \Gamma_{\mu} Y_{\bar{n}} \chi_{\bar{n}} .$$
(4.56)

In the context of $SCET_i$, the angular ordering property of the soft emission and the coherent parton branching formalism for soft emission with multiple hard partons

emerge naturally. If we take the Fourier transform of $Y_n(x)$ we get

$$Y = 1 + \sum_{m=1}^{\infty} \sum_{\text{perms}} \frac{(-g)^m}{m!} \frac{n \cdot A_s^{a_1} \cdots n \cdot A_s^{a_m}}{n \cdot k_1 n \cdot (k_1 + k_2) \cdots n \cdot (\sum_{i=1}^m k_i)} T^{a_m} \cdots T^{a_1}$$
(4.57)

where $k_1, k_2, ..., k_n$ are the momenta of the gluon fields. The eikonal structure of (4.57) leads to angular ordering. If a collinear particle with momentum q_i in the n_i direction emits a soft gluon of momentum k_s , the amplitude acquires a term proportional to

$$F_{\text{soft}} = \frac{n_i \cdot \varepsilon_s}{n \cdot k_s} = \frac{q_i \cdot \varepsilon_s}{q_i \cdot k_s} + O(\lambda), \qquad (4.58)$$

where ε_s is the polarization vector of the soft radiation and $q_i^{\mu} = \bar{q} n_i^{\mu}/2$ up to power corrections. If $A_n(q_1, q_2, \dots, q_n)$ is the amplitude to emit *n* collinear particles with momenta q_1, q_2, \dots, q_n and A_{n+1} the amplitude with one more emission k_s in the soft region, we get $A_{n+1}(q_1, q_2, \dots, q_n, k_s) \sim A_n(q_1, q_2, \dots, q_n) \sum_{i=1}^n C_i q_i \cdot \varepsilon_s/q_i \cdot k$, where C_i is a color factor. For the cross section this implies

$$d\sigma_{n+1} = d\sigma_n \frac{dE_s}{E_s} \frac{d\Omega_s}{2\pi} \frac{\alpha_s}{2\pi} \sum_{i,j} C_{i,j} W_{i,j}, \qquad (4.59)$$

where $d\Omega_s$ and E_s are the element of solid angle and the energy of the emitted soft gluon, $C_{i,i}$ is a color factor. Here

$$W_{i,j} = \frac{E_s^2 q_i \cdot q_j}{q_i \cdot k_s q_j \cdot k_s} \tag{4.60}$$

is called the radiation function. After integration of $W_{i,j}$ over azimuthal angular variables the soft gluons only contribute when the gluon is confined to the cones centered in the direction of the particle *i* or *j*, and are hence angular ordered.

To see how coherent branching emerges we consider the effect of soft gluons in our leading order matching between SCET_i and SCET_{i+1} . If we consider effects encoded by operators with exactly the same collinear field content in SCET_i and SCET_{i+1} then graphs involving soft gluons agree and there is no contribution to the matching. If we consider instead the calculations that lead to the LO replacement rule $\bar{\chi}_{n_0} \to c_{\text{LO}} \bar{\chi}_{n_1} \mathcal{B}_{n'_1}^{\perp}$, then the soft gluons are encoded by

SCET_i:
$$\bar{\chi}_{n_0} Y_{n_0}^{\dagger}$$
, SCET_{i+1}: $c_{\text{LO}} \bar{\chi}_{n_1} Y_{n_1}^{\dagger} Y_{n_1'} \mathcal{B}_{n_1'}^{\perp} Y_{n_1'}^{\dagger}$. (4.61)

For soft gluons at wide angles relative to n_0 , n_1 , and n'_1 the effect of attachments to $Y_{n_1}^{\dagger}Y_{n'_1}$ are power suppressed because soft attachments to these two lines cancel up to terms that are power suppressed by $n_1 \cdot n'_1 \sim \lambda^{2i}$. The remaining attachment to $Y_{n'_1}^{\dagger}$ looks the same as attachments to $Y_{n_0}^{\dagger}$ at leading power since $n_0 \cdot n'_1 \sim \lambda^{2i}$. Thus wide angle soft gluons do not resolve the substructure revealed by matching to SCET_{i+1} and effectively only couple to the overall color charge of the parent quark $\bar{\chi}_{n_0}$. Soft radiation that is close in angle to n_1 and n'_1 resolves the split quark $\bar{\chi}_{n_1}$ and gluon $\mathcal{B}_{n'_1}^{\perp}$, compensating for the $n_1 \cdot n'_1$ suppression by additional collinear singularities in its propagator factors. Thus the coherent branching formalism for soft gluons emerges naturally for amplitudes in our SCET_i picture.

From the SCET point of view it would be natural to distinguish soft and collinear radiation in the shower and treat them independently, being careful not to double count. For simplicity all available shower codes treat them in a coherent fashion. The effect of accounting for soft coherent branching in the shower typically leads to modifications of the Sudakov probability factors (see for example Ref. [44]), and effects the choice of evolution variable or by adding additional vetoes. In the context of SCET the implications of this were discussed recently in Ref. [19].

4.3.3 Summary LO Parton Showers

In table 4.1 we summarize results for the mapping between the LO/LL parton shower at and our SCET_i picture. In the first column we put the elements needed for showering, and in the central column the translation to elements in the SCET_i setup. The usual splitting function is related to our replacement rule $\bar{\chi}_{n_0} \rightarrow c^{\text{LO}} \bar{\chi}_{n_1} \mathcal{B}_{n'_1}^{\perp}$, that in turn is related to the SCET₂ coefficient of the operator $\mathcal{O}_2^{(1)}$. The LL Sudakov comes from LL running factors related to the one-loop cusp anomoulous dimension as in Ref. [17]. At leading order soft emission in SCET_i is taken into account adding

Shower Ingredients	Quantity in SCET_i	Found In:
Splitting function	Replacement rule	Eq.(4.46)
LL Sudakov factor	One-loop cusp	Eq. (4.10)
	anomalous dimension	
Soft emission	Soft amplitude	Eq. (4.59)
Evolution variables	$ar{q}, \ n_i \cdot n_j$	

Table 4.1: Mapping between parton shower and $SCET_i$ at LO/LL.

soft Wilson lines Y_n into our operators. This leads to angular ordering and coherent branching, which must be accounted for with modifications to the shower to account for the soft singular regions. Finally, showers are constructed with different choices of evolution variables and the choice effects the structure of power corrections. In SCET_i we have seen that we can write all coefficients in terms of the large momenta (\bar{q}) and dot product of ns vectors $(n_i \cdot n_j)$, which are natural variables in the SCET_i picture. At LO/LL the translation to any appropriate set of evolution variables is straightforward. Beyond this order one must systematically account for additional power corrections induced by changing variable from \bar{q}_i and $n_i \cdot n_j$ to the desired evolution variables for the shower.

4.4 SCET Power Corrections to the Shower

As we have seen in the previous section, we reproduce the usual parton shower by matching collinear gluon emissions to increasingly lower-scale EFTs, the SCET_i. Our goal is to catalog the leading corrections (in λ) to the cross section for the emission of an arbitrary number of collinear gluons to a quark. By this we mean all amplitude terms to LO and NLO, as well as those at NNLO that can interfere with LO. As we will argue in subsection. 4.4.4, in many cases of interest, there is no LO/NLO interference, and so we focus on NLO/NLO and LO/NNLO. Just as in the strongly-ordered case, it is convenient to integrate down to SCET_{i+1} when describing the emission of *i*-gluons. We obtain these corrections by doing our matching computations at higher order. We will show that there are two distinct types of subleading matching, and they have a different physical interpretation:

- One set originates in matching QCD → SCET₁ at higher orders. This generates a set of subleading terms that remain suppressed as we move down to lower-scale SCET_j's. We call them *hard-scattering* corrections as they involve the details of the hard-scale process that created our initial hadrons. Also, they are most important for partons radiated closest to the original vertex.
- The other set comes from the subleading matching $\text{SCET}_i \to \text{SCET}_{i+1}$. They involve processes described by the SCET_i lagrangian, but ones that get integrated out into higher dimension operators at lower scales. These corrections are ubiquitous. They do not depend on the hard-scattering details, and we can determine them for arbitrary $\text{SCET}_i \to \text{SCET}_{i+1}$ once we have found them in $\text{SCET}_1 \to \text{SCET}_2$. Furthermore, they relate to known $\mathcal{O}(\alpha_s)$ corrections to the $q \to qg$ splitting function. For this reason we call them *jet-structure* corrections.

Determining the above to NLO in the cross-section will only involve single and double gluon emission. Thus, we will never need to compute in a lower-scale theory than $SCET_3$. We perform all the QCD \rightarrow SCET₁ \rightarrow SCET₂ \rightarrow SCET₃ matchings for these amplitudes necessary to calculate the corrections in appendices B.2-B.4. Below, we discuss the final results for the corrections, with Subsection 4.4.1 focusing on the hard-scattering corrections and Subsection 4.4.2 on the jet-corrections ones. For these portions of the work, the matching is only done at tree level, though formulas in the appendices include one-loop RG kernels. We discuss the effects of LL running on correction terms in Subsection 4.4.3. Lastly, in Subsection 4.4.4, we will study the amplitude squared and we will see there is a great simplification of the interference structure in SCET_N. We will also describe steps going from our subleading operators toward the systematic improvement of the parton shower to NLL.

4.4.1 Hard-Scattering Corrections

Let us begin by examining the matching $QCD \rightarrow SCET_1$ for single gluon emission collinear to the quark. For this case, all corrections are of the hard-scattering type. Unlike LO, corrections can have dependence on the process that creates the $\bar{q}q$ pair. For concreteness, we will consider the coupling of QCD quarks to the vector current $J^{\mu}_{\text{QCD}} = \bar{q} \dot{\gamma}^{\mu} q$. The matching is performed in the centre of mass frame with the initial virtual photon with momentum $p_{\gamma} = (Q, 0, 0, 0)$. The full details of matching calculation for QCD to SCET₁ is performed in appendix B.2. To reproduce the full QCD current J^{μ}_{QCD} , we need an infinite tower of SCET₁ operators increasingly higher order in λ . However, to get the required amplitude to NNLO, we only need four:

$$A_{\text{to NNLO}}^{q\bar{q}g} = C_{1,\text{LO}}^{(0)} \int dx \langle 0|T\{\mathcal{L}_{\text{SCET}_{1}}(x)\mathcal{O}_{1}^{(0)}(n_{0})\}|q_{n_{0}}g_{n_{0}}\bar{q}_{\bar{n}}\rangle + C_{1}^{(1)}(n_{0},n_{0})\langle 0|\mathcal{O}_{1}^{(1)}(n_{0},n_{0})|q_{n_{0}}g_{n_{0}}\bar{q}_{\bar{n}}\rangle + C_{1,\mathcal{T}}^{(1)}(n_{0},n_{0})\langle 0|\mathcal{T}_{1}^{(1)}(n_{0},n_{0})|q_{n_{0}}g_{n_{0}}\bar{q}_{\bar{n}}\rangle + C_{1}^{(1)}(n_{1},n_{1}')\langle 0|\mathcal{O}_{1}^{(1)}(n_{1},n_{1}')|q_{n_{1}}g_{n_{1}'}\bar{q}_{\bar{n}}\rangle,$$
(4.62)

where

$$\mathcal{O}_{1}^{(0)}(n_{0}) = \bar{\chi}_{n_{0}}\chi_{\bar{n}}, \qquad (4.63)$$

$$\mathcal{O}_{1}^{(1)}(n_{0}, n_{0}) = \bar{\chi}_{n_{0}} g\mathcal{B}_{n_{0}\perp}^{\alpha}\chi_{\bar{n}}, \qquad (4.63)$$

$$\mathcal{T}_{1}^{(1)}(n_{0}, n_{0}) = \bar{\chi}_{n_{0}} \left[\mathcal{P}_{n_{0}\perp}^{\beta} g\mathcal{B}_{n_{0}\perp}^{\alpha} \right] \chi_{\bar{n}}, \qquad (4.63)$$

$$\mathcal{O}_{1}^{(1)}(n_{1}, n_{1}') = \bar{\chi}_{n_{1}} g\mathcal{B}_{n_{1}\perp}^{\alpha}\chi_{\bar{n}}.$$

We gave the expression for $C_{1,LO}^{(0)}$ in Eq. (4.41). The amplitude from the operator $\mathcal{O}_{1}^{(0)}(n_{0})$ is shown in the first diagram in the SCET₁ column of Fig. 4-7, those from $\mathcal{O}_{1}^{(1)}(n_{0}, n_{0})$ and $\mathcal{T}_{1}^{(1)}(n_{0}, n_{0})$ in the second, and that for $\mathcal{O}_{1}^{(1)}(n_{1}, n_{1}')$ in the third. We call $\mathcal{O}_{1}^{(1)}(n_{0}, n_{0})$ and $\mathcal{T}_{1}^{(1)}(n_{0}, n_{0})$ "two-jet" operators as they are labeled only with two distinct collinear directions $(n_{0} \text{ and } \bar{n})$ (for the direction of the antiquark we do not put a symbol following the convention in Eq. (4.31)). They describe a gluon collinear to the quark. We obtain the coefficients $C_{1}^{(1)}(n_{0}, n_{0})$ and $C_{1,\mathcal{T}}^{(1)}(n_{0}, n_{0})$ by expanding the QCD amplitude in the limit of small gluon momentum transverse to the quark's direction with the usual SCET parametrics: $(n_{0}, \bar{k}_{1}, k_{1n_{0}\perp}) \sim (\lambda^{2}, 1, \lambda)Q$. There are an additional set of two-jet configurations corresponding to the gluon being collinear to the antiquark. These are trivial to obtain by charge conjugation and we do not discuss them separately here. The operator $\mathcal{O}_{1}^{(1)}(n_{1}, n_{1}')$ is a three-jet configuration,



Figure 4-7: Matching QCD to $SCET_1$ to $SCET_2$ for one emission collinear to the quark direction. We depict the operator structures that lead to this process in each of the three theories. The QCD contribution is standard. In $SCET_1$, we either emit a collinear gluon through the time-ordered product of the Lagrangian with an two-parton operator, or from three-parton operators. In $SCET_2$, this configuration only arises from a higher-dimension three-parton operators.

as it describes three distinct directions where the quark, the gluon and the antiquark are far apart according to the definition of collinearity in SCET₁. Whenever we have an operator where each field has its own index label, we can chose for the purpose of matching the n_i such that they are exactly aligned with the external particle momenta. The coefficients $C_1^{(1)}(n_0, n_0)$ and $C_1^{(1)}(n_1, n'_1)$, are discussed in Eqs. (B.36) and give:

$$C_{1}^{(1)}(n_{0}, n_{0}) = \frac{1}{Q} (n_{0}^{\mu} - \bar{n}^{\mu}) \gamma_{n_{0} \perp}^{\alpha},$$

$$C_{1,\mathcal{T}}^{(1)}(n_{0}, n_{0}) = \frac{1}{\bar{q}_{1}\bar{k}_{1}} \gamma_{n_{0} \perp}^{\mu} \gamma_{n_{0} \perp}^{\beta} \gamma_{n_{0} \perp}^{\alpha} - \frac{2}{\bar{q}_{1}Q} g^{\beta\mu} \gamma_{n_{0} \perp}^{\alpha},$$
(4.64)

We use the same kinematic variables defined in Fig. 4-4. For $C_1^{(1)}(n_0, n_0)$ and $C_{1,\mathcal{T}}^{(1)}(n_0, n_0)$ the initial current is not a spectator, so neither term is simply proportional to the γ^{μ} that we started with. This dependence on the details of the rest of process is a characteristic feature of these hard-scattering corrections. We give the coefficient $C_1^{(1)}(n_1, n_1')$ of the operator with quark, gluon, and antiquark in distinct collinear directions in Eq. (B.39).

Going to SCET₂ for single gluon emission is straightforward. The basis of operators needed to reproduce the amplitude (4.62) is equal to (4.63), but where they are defined in SCET₂ instead of SCET₁: $\mathcal{O}_2^{(0)}(n_0)$, $\mathcal{O}_2^{(1)}(n_0, n_0)$, $\mathcal{T}_2^{(1)}(n_0, n_0)$, and $\mathcal{O}_2^{(1)}(n_1, n'_1)$. As the computations get more complicated with subsequent emissions, we wish to minimize our effort by only including those terms necessary to give the corrections to a shower algorithm. This means we are only interested in the following:

- Since we construct observables by squaring and integrating amplitudes, we will need to keep those NNLO contributions that can interfere with LO. These give terms at the same order as an NLO operator squared with itself. We do not compute NNLO amplitude terms which have zero interference with the LO amplitude. A list of the necessary computations is found in Appendix B.3.
- 2. Our ultimate goal is not a complete SCET_i theory from which one can do computations, but an improved shower algorithm. In Table 4.1, we give a list of those ingredients needed to construct a map between SCET_i and a LL parton shower. We will augment the map with items needed for corrections (Table 4.2 in subsection 4.5), but will not calculate contributions which only contain redundant information for the shower.

The latter point has important implications for the sorts of operator structures we need to consider. If we wanted to do computations in SCET₂, then of course we would need all operators and Wilson coefficients to the order we are working. As discussed above, single gluon contributions in SCET₂ where the gluon and the quark are collinear at $\mathcal{O}(\lambda^2)$ (*i.e.* $\mathcal{O}_2^{(1)}(n_0, n_0)$ or $\mathcal{T}_2^{(1)}(n_1, n_1)$) correspond to a quark which does not split until after the scale of matching SCET₁ \rightarrow SCET₂. However the corresponding no-branching probability is already determined in SCET₁, so the coefficients of these operators in SCET₂ are not required. Thus, we only need to calculate those single gluon contributions where each field has its own index label in SCET₂, meaning determine the Wilson coefficient $C_2^{(1)}(n_1, n_1')$ of $\mathcal{O}_2^{(1)}(n_1, n_1')$. The matching equation for $C_2^{(1)}(n_1, n_1')$ in SCET₂ is:

$$C_{1,LO}^{(0)}(n_0) \int dx^4 \langle 0|T\{\mathcal{L}_{SCET_1}(x)\mathcal{O}_1^{(0)}(n_0)\}|q_{n_1} g_{n'_1} \bar{q}_{\bar{n}}\rangle + C_1^{(1)}(n_0, n_0) \langle 0|\mathcal{O}_1^{(1)}(n_0, n_0)|q_{n_1} g_{n'_1} \bar{q}_{\bar{n}}\rangle + C_1^{(1)}(n_1, n'_1) \langle 0|\mathcal{O}_1^{(1)}(n_1, n'_1)|q_{n_1} g_{n'_1} \bar{q}_{\bar{n}}\rangle + C_{1,\mathcal{T}}^{(1)}(n_0, n_0) \langle 0|\mathcal{T}_1^{(1)}(n_0, n_0)|q_{n_1} g_{n'_1} \bar{q}_{\bar{n}}\rangle = C_2^{(1)}(n_1, n'_1) \langle 0|\mathcal{O}_2^{(1)}(n_1, n'_1)|q_{n_1} g_{n'_1} \bar{q}_{\bar{n}}\rangle.$$
(4.65)

It is convenient to decompose $C_2^{(1)}(n_1, n_1')$ as

$$C_2^{(1)} = C_{2, \text{LO}}^{(1)} + C_{2, \text{NLO}}^{(1)H,a} + C_{2, \text{NLO}}^{(1)H,b} + C_{2, \text{NNLO}}^{(1)H}, \qquad (4.66)$$

where $C_{2,LO}^{(1)}$ is the coefficient multiplying $\langle 0|\mathcal{O}_{2}^{(1)}(n_{1},n_{1}')|q_{n_{1}}g_{n_{1}'}\bar{q}_{\bar{n}}\rangle$ that reproduces the second line in Eq. (4.65), $C_{2,NLO}^{(1)H,a}$ the third line, *etc.* $C_{2,LO}^{(1)}$ was calculated in (4.41) using RPI to rotate objects in the SCET₁ amplitude such that they can come from SCET₂ operators. The third and fourth terms give NLO coefficients, which we can calculate in a similar manner to $C_{2,LO}^{(1)}$. Their values are derived in Eqs. (B.57) and (B.59):

$$C_{2,\text{NLO}}^{(1)H,a} = \frac{1}{Q} \Big(\frac{\bar{k}_{1} n_{1}^{\prime \mu} + \bar{q}_{1} n_{1}^{\mu}}{\bar{q}_{0}} - \Big(1 + \frac{\bar{q}_{1} \bar{k}_{1}}{2 \bar{q}_{0}^{2}} (n_{1} \cdot n_{1}^{\prime}) \Big) \bar{n}^{\mu} \Big) \gamma_{n_{1}^{\prime} \perp}^{\alpha} \Theta_{\lambda^{2}} [n_{1} \cdot n_{1}^{\prime}] , \qquad (4.67)$$

$$C_{2,\text{NLO}}^{(1)H,b} = -\frac{2}{(n_{1} \cdot n_{1}^{\prime}) \bar{q}_{1} \bar{k}_{1}} \gamma^{\alpha} \not p_{\gamma} \gamma_{T}^{\mu}$$

$$+ \Big[\frac{1}{(n \cdot p_{\bar{q}}) \bar{k}_{1}} \Big(\gamma_{T}^{\mu} \not p_{\gamma} - \bar{q}_{1} n_{1T}^{\mu} \Big) + \frac{2(n \cdot p_{\bar{q}})}{(n_{1} \cdot n_{1}^{\prime}) \bar{q}_{1} \bar{k}_{1}} \bar{n}_{T}^{\mu} \Big] \gamma^{\alpha} \tilde{\Theta}_{\lambda^{2}} [n_{1} \cdot n_{1}^{\prime}] ,$$

$$C_{2,\text{NNLO}}^{(1)H} = \Big(\frac{1}{2 Q} \Big(\gamma_{n_{1}^{\prime} \perp}^{\mu} \sqrt{|n_{1} \cdot n_{1}^{\prime}|} \not p_{1}^{\mu} + \bar{n}^{\mu} \frac{\bar{q}_{1}}{Q} (n_{1} \cdot n_{1}^{\prime}) \Big) \gamma_{n_{1}^{\prime} \perp}^{\alpha}$$

$$+ \frac{\bar{k}_{1}}{Q^{2}} \Big(\sqrt{|n_{1} \cdot n_{1}^{\prime}|} v_{1}^{\mu} - \bar{n}^{\mu} (n_{1} \cdot n_{1}^{\prime}) \frac{(\bar{k}_{1}^{2} - \bar{q}_{1}^{2})}{2 Q^{2}} \Big) \gamma_{n_{1}^{\prime} \perp}^{\alpha} \Big) \Theta_{\lambda^{2}} [n_{1} \cdot n_{1}^{\prime}] .$$

Here n_1 and n'_1 are aligned with the direction of the quark and the gluon for the matching, as in Eq. (4.37), and v_1 is defined in Eq. (B.14). In Eq. (4.67) we have not displayed the running factors that come from evolution of the SCET₁ operators. The SCET₁ operators in Eqs. (4.63) can have different running factors, in particular the two-jet and three-jet operators have different LL evolution. Therefore it is important



Figure 4-8: Plot of the ratios of the amplitudes squared for $\gamma^* \to q\bar{q}g$ for $R_{\rm LO} = |A^{q\bar{q}g}|_{\rm LO}^2/|A^{q\bar{q}g}|_{\rm QCD}^2$ (blue) and $R_{\rm NLO} = |A^{q\bar{q}g}|_{\rm NLO,2-jet}^2/|A^{q\bar{q}g}|_{\rm QCD}^2$ (red) versus $|k_1|_{n_0\perp}$, for $\bar{k}_1/\bar{q}_0 = 0.4$. The amplitudes are evaluated without running factors.

to decompose the coefficient $C_2^{(1)}$ as in Eq. (4.66) so that we can keep track of which SCET₁ evolution factor to include for each one. The running of these operators is discussed further in subsection 4.4.3.

To illustrate the effects of including hard-scattering corrections, in Fig. 4-8 we plot the ratios $R_{\rm LO} = |A^{q\bar{q}g}|_{\rm LO}^2/|A^{q\bar{q}g}|_{\rm QCD}^2$ and $R_{\rm NLO} = (|A^{q\bar{q}g}|_{\rm LO}^2 + |A^{q\bar{q}g}|_{\rm to NNLO,2-jet}^2)/|A^{q\bar{q}g}|_{\rm QCD}^2$ versus the gluon perp momentum. Here $|A^{q\bar{q}g}|_{\rm QCD}^2$ is the QCD amplitude squared for one-gluon emission, $|A^{q\bar{q}g}|_{\rm LO}^2$ is the SCET₂ amplitude squared for one-gluon emission from the LO coefficient $C_{2\rm LO}^{(1)}\mathcal{O}_2^{(1)}$ (from Eq. (B.62)), and $|A^{q\bar{q}g}|_{\rm to NLO,2-jet}^2$ is the NLO amplitude squared for one-gluon emission in the two-jet region that comes from the coefficients $C_{2,\rm NLO}^{(1)H_a}$ and $C_{2,\rm NNLO}^{(1)H}$ (given in Eq. (B.65)). Notice that including corrections up to NNLO in the amplitudes extends the region where tree-level SCET₁ and QCD agree. The advantage of using the one-gluon SCET amplitude over QCD is that it is incorporated into shower language of branching and no-branching, and also automatically avoids double counting with LL emissions.

In the above computation we have seen that the SCET₂ coefficient $C_2^{(1)}$ and corresponding operator $\mathcal{O}_2^{(1)}$ carry results that occured in several different operators in SCET₁. This operator carries the information that its fields are in distinct equivalence



Figure 4-9: Plot of $\Theta(x)_{\Lambda,a}$ defined in Eq. (B.61) for $\Lambda = 5$ and a = 0.8. The parameter Λ tells where the function switches from 0 to 1 and the parameter a how fast it does it.

classes, $\{[n_j]\}$, with separations $\gg \lambda^4$. However it does not know just how inequivalent they are, whether they are separated by $\sim \lambda^2$ or by $\sim \lambda^0$, which is information that was stored in the operators in SCET₁. In SCET₂ this information must be stored in the Wilson coefficient in the matching procedure. For the case of $C_1^{(1)}(n_0, n_0)$ and $C_1^{(1)}(n_1, n'_1)$, we need to keep their contributions separate as they correspond to parametrically distinct regions of phase space in SCET₁. To enforce this distinction, we use a Wilsonian cutoff on the dot product of collinear directions for different fields with the smoothed θ -function, $\Theta_{\lambda j}[n_i, n'_i]$ described in Subsection 4.3.1. The notation means that $\Theta \to 0$ once $n_i \cdot n'_i \gtrsim \lambda^j$, and $\tilde{\Theta} = 1 - \Theta$ behaves in the opposite fashion.

Therefore, $C_{2,\text{NLO}}^{(1)H,a} \propto \Theta_{\lambda^2}[n_1,n_1']$ and $C_{2,\text{NLO}}^{(1)H,b} \propto \tilde{\Theta}_{\lambda^2}[n_1,n_1']$, as these enforce where in the SCET₁ phase space these contributions originated. In appendix B.3, we define a smooth Θ function in Eq.(B.61), which is plotted in Fig. 4-9. It depends on two parameter, $\Theta(x)_{\Lambda,a}$, The parameter Λ tells where the function switches from 0 to 1 and the parameter *a* how fast it does it. By virtue of its appearance in the Wilson coefficients this Θ function merges the two-jet amplitude squared with a quark and gluon in one-jet, $(|A^{q\bar{q}g}|_{\text{LO}}^2 + |A^{q\bar{q}g}|_{\text{to NNLO}, 2-\text{jet}}^2)$, with the three-jet amplitude squared $|A^{q\bar{q}g}|_{\text{NNLO}, 3-\text{jet}}^2$, that is the amplitude squared that come from using the coefficient $C_{2,\text{NNLO}}^{(1)H,b}$. The full expression is given in Eq. (B.64). As shown in Fig. B-3, the theta



Figure 4-10: Merging of the two-jet and three-jet amplitudes squared for $\gamma^* \rightarrow q\bar{q}g$ process using a smooth theta function. Plots of the two-jet amplitude square, $|A^{q\bar{q}g}|_{\rm LO}^2 + |A^{q\bar{q}g}|_{\rm to NNLO, 2-jet}^2$, (green), three-jet, $|A^{q\bar{q}g}|_{\rm NLO, 3-jet}^2$, (blue) and sum (red) versus $|k_1|_{n_0\perp}$. The amplitudes are evaluated without running coefficients for $\bar{k}_1/\bar{q}_0 = 0.4$.

function smoothly merges the two squared amplitudes.

With two-gluon emission, the SCET₁ graphs will include jet-structure corrections in addition to hard-scattering ones. It is straightforward to distinguish the types as the former result from taking time-ordered products of the SCET₁ Lagrangian with operators generated by the LO replacement rule, Eq. (4.46), while the latter will come only from terms involving a power suppressed SCET₁ operator. To fully identify the subleading contributions to two-gluon emission, we must match down to SCET₂ where the LL contribution is first uniquely identified. We already know that the LO result comes from two applications of Eq. (4.46).

In Fig. (4-11), we show the contributions to two-gluon emission in QCD, SCET₁, SCET₂, and SCET₃. The first column in the SCET₁ category correspond to the jet-structure corrections to be considered in the next section. In the second column we have a set of hard-scattering corrections from taking the *T*-product of the SCET₁ Lagrangian with the suppressed single gluon operators we calculated above in Eqs. (4.64), $C_1^{(1)}\mathcal{O}_1^{(1)}$ and $C_{1,\mathcal{T}}^{(1)}\mathcal{T}_1^{(1)}$.

In considering the basis of operators in $SCET_2$ we do not need operators such as



Figure 4-11: Matching QCD to $SCET_1$ to $SCET_2$ to $SCET_3$ for two-gluon emission collinear to the quark direction. Once again, we depict the operator structures that lead to this process in each of the theories. Gluons drawn away from the central vertex are emitted by the renormalizable lagrangian in that theory, while those coming from the vertex are due to higher-dimension operators.

 $\mathcal{T}_2^{(1)}(n_1, n_1')$ since, $\mathcal{P}_{n_1'\perp} \mathcal{B}_{n_1'\perp} = 0$, with n_1' lying along the gluon momentum. The coefficient of such operators are determined by RPI, and we can use RPI in SCET₂ to make a coordinate choice where they are not necessary. As mentioned above in the single gluon matching section, our interest is only in calculating those terms needed to improve a shower algorithms, which precludes us from considering operators such as $\mathcal{T}_2^{(1)}(n_0, n_0)$ or $\mathcal{O}_2^{(1)}(n_0, n_0)$. Therefore, for double gluon emission we only need to calculate the coefficients of the following operators

$$\mathcal{O}_{2}^{(1)}(n_{1}, n_{1}') = \bar{\chi}_{n_{1}}g\mathcal{B}_{n_{1}'\perp}^{\alpha}\chi_{\bar{n}}$$

$$\mathcal{O}_{2}^{(2)}(n_{2}, n_{2}, n_{1}') = \bar{\chi}_{n_{2}}g\mathcal{B}_{n_{2}\perp}^{\alpha}g\mathcal{B}_{n_{1}'\perp}^{\beta}\chi_{\bar{n}},$$

$$\mathcal{O}_{2}^{(2)}(n_{2}, n_{1}', n_{1}') = \bar{\chi}_{n_{2}}g\mathcal{B}_{n_{1}'\perp}^{\alpha}g\mathcal{B}_{n_{1}'\perp}^{\beta}\chi_{\bar{n}},$$

$$\mathcal{O}_{2}^{(2)}(n_{2}, n_{1}', n_{2}') = \bar{\chi}_{n_{2}}g\mathcal{B}_{n_{1}'\perp}^{\alpha}g\mathcal{B}_{n_{2}'\perp}^{\beta}\chi_{\bar{n}}.$$

$$(4.68)$$

Thus in SCET₂, we are interested in *two* gluon operators where two fields can have the same label. When we pass to SCET₃ we can restrict our interest to only $\mathcal{O}_3^{(2)}(n_2, n'_1, n'_2)$, the operator whose collinear fields are all in distinct light-cone directions.

The coefficients of $\mathcal{O}_2^{(1)}(n_1, n_1')$ needed to compute the leading power corrections

are given in Eqs. (B.59) and (B.57). We get an NLO contribution to the two gluon amplitude by computing the matrix element, $C_{2\text{NLO}}^{(1)}\langle 0|T\{\mathcal{L}_{\text{SCET}_2}\mathcal{O}_2^{(1)}\}|q\bar{q}gg\rangle$ (fifth column in Fig. 4-11). The power counting of the contribution does not change as the gluon from $\mathcal{L}_{\text{SCET}_2}$ gives a vertex×propagator factor of λ^{-3} , which is the same as LO. There are also coefficients we need from two-gluon matching calculations for the operator $\mathcal{O}_2^{(2)}$ (sixth column in Fig. 4-11). Putting in the index structures on the fields, these include $C_{2,\text{NLO}}^{(2)J}(n_2, n'_1, n'_2)$ for $\mathcal{O}_2^{(2)}(n_2, n_2, n'_1), C_{2,\text{NLO}}^{(2)J}(n_2, n'_1, n'_1)$ for $\mathcal{O}_2^{(2)}(n_2, n'_1, n'_1)$ and $C_{2,\text{NLO}}^{(2)J}(n_2, n_2, n'_2) + C_{2,\text{NNLO}}^{(2)H}(n_2, n_2, n'_2)$ for $\mathcal{O}_2^{(2)}(n_2, n'_1, n'_2)$. Since only the last one multiplies an operator that interferes with LO, it is the only one we will need to keep track of to NNLO for the hard scattering. (We will see in the next subsection that there are jet-structure corrections to all three terms in the sixth column at NLO.) All other hard-scattering contributions are beyond the order we need. Thus we only need to compute $C_2^{(2)H}(n_2, n_2, n'_2)$ given by the matching equation

$$C_{1}^{(1)} \int dx \langle 0 | T \{ \mathcal{L}_{\text{SCET}_{1}}(x) \mathcal{O}_{1}^{(1)}(n_{0}, n_{0}) \} | q_{n_{2}} g_{n_{2}} g_{n_{1}'} \bar{q}_{\bar{n}} \rangle$$

$$- C_{2,\text{NLO}}^{(1)}(n_{2}, n_{1}') \int dx \langle 0 | T \{ \mathcal{L}_{\text{SCET}_{2}}(x) \mathcal{O}_{2}^{(1)}(n_{2}, n_{1}') \} | q_{n_{2}} g_{n_{2}} g_{n_{1}'} \bar{q}_{\bar{n}} \rangle$$

$$= C_{2}^{(2)H}(n_{2}, n_{2}, n_{1}') \langle 0 | \mathcal{O}_{2}^{(2)}(n_{2}, n_{2}, n_{1}') | q_{n_{2}} g_{n_{2}} g_{n_{1}'} \bar{q}_{\bar{n}} \rangle ,$$

$$(4.69)$$

where we subtract the time-ordered product graph in SCET₂ from the time-ordered product graph in SCET₁. The result for $C_2^{(2)H}(n_2, n_2, n'_2)$ is given in Eq. (B.104). It is easy to see why $\mathcal{O}_2^{(2)}$ only gets hard-scattering contributions up to this order. By definition, hard-scattering has to involve a suppressed operator from the QCD \rightarrow SCET₁ matching, and so we begin at NLO at the lowest order. Including a second gluon, but demanding that we cannot write it as coming from a SCET₂ lagrangian emission takes us to one order higher, namely NNLO.

All the contributions we have discussed so far have come from the hard-scattering, single-gluon, suppressed operators in $SCET_1$. There are also those with two gluons. That is to say a process where neither gluon comes from the $SCET_1$ Lagrangian,

represented by the diagram in the third $SCET_1$ column in Fig. 4-11. One example is double emission from the antiquark, as shown in the third QCD graph of Fig. 4-11. We know from applying Eq. (4.46) twice, that LO for this process is at $\mathcal{O}(\lambda^{-3})$ counting only vertex \times propagator factors, as these are all we need to compare different $q\bar{q}gg$ processes. We readily see that double antiquark emission is $\sim \lambda^0$. This follows from the same arguments we give above for single emission from the antiquark. Thus, they are N³LO. This is higher order than we are analyzing. Besides antiquark vertices, we also have subleading emissions from the quark in QCD that arise from the suppressed SCET-spinor portion of the QCD quark propagator (cf. appendix B.2). In fact, we have already such contributions for single emission. However, if both emissions come from the suppressed propagator, once again, this is $\sim \lambda^0$ at lowest order, and so we can neglect it. Mixed antiquark/suppressed spinor contributions are also N³LO. Thus, we do not need corrections to double emission collinear to the quark if they do not involve at least one $SCET_1$ Lagrangian insertion. We can extend this argument to further emissions without using the $SCET_1$ Lagrangian. Once again, these will go as $\mathcal{O}(\lambda^0)$, while LO goes like $\mathcal{O}(\lambda^{-\frac{i(i+1)}{2}})$. Thus, to the order we are working, we only need the single gluon hard-scattering correction (plus Lagrangian insertions). Furthermore, we only needed the NLO Wilson coefficient $C_1^{(1)}$, given in Eq. (4.64).

4.4.2 Jet-Structure Corrections

The *jet-structure* corrections only involve contributions from the SCET₁ Lagrangian. These arise from the graphs in the first SCET₁ column in Fig. 4-11. We specifically designed our leading order replacement rule in Eq. (4.46), so when it is used twice it only contains that part of double emission corresponding to the leading stronglyordered limit. This occurs for the gluons having collinearities $\sim \lambda$, λ^2 , respectively. However SCET₁ also describes other kinematic situations and in this section we compute the corrections from these regions.

The prescription for obtaining two-gluon jet-structure corrections is to compute the double gluon emission amplitude in SCET₁ coming from two lagrangian inserstions and take different limits on the relative collinearities of n_2 , n'_2 , and n'_1 , where these labels refer to the null vectors exactly proportional the corresponding particle momenta. We can define:

$$A_{\rm NLO}^{q\bar{q}gg} = C_{1,\rm LO}^{(0)} \int dx_1 dx_2 \langle 0|T\{\mathcal{L}_{\rm SCET_1}(x_1)\mathcal{L}_{\rm SCET_1}(x_2)\mathcal{O}_1^{(0)}\}|q_{n_2}g_{n_1'}g_{n_2'}\bar{q}_{\bar{n}}\rangle, \quad (4.70)$$

and then calculate,

n

$$\lim_{n_2 \cdot n'_2 \sim \lambda^2} A_{\text{NLO}}^{q\bar{q}gg} = C_{2,\text{NLO}}^{(2)J}(n_2, n'_1, n'_2) \langle 0 | \mathcal{O}_2^{(2)} | q_{n_2} g_{n'_1} g_{n'_2} \bar{q}_{\bar{n}} \rangle$$
(4.71)

$$\lim_{\substack{1'n_2'\sim\lambda^4}} A_{\rm NLO}^{q\bar{q}gg} = C_{2,\,\rm NLO}^{(2)J}(n_2,\,n_1',\,n_1')\langle 0|\mathcal{O}_2^{(2)}|q_{n_2}g_{n_1'}g_{n_2'}\bar{q}_{\bar{n}}\rangle \tag{4.72}$$

$$\lim_{n_2 \cdot n'_2 \sim \lambda^4} A_{\text{NLO}}^{q\bar{q}gg} = C_{2,\text{LO}}^{(1)} \langle 0 | T \{ \mathcal{L}_{\text{SCET}_2} \mathcal{O}_2^{(1)} \} | q_{n_2} g_{n'_1} g_{n'_2} \bar{q}_{\bar{n}} \rangle + C_{2,\text{NLO}}^{(2)J} (n_2, n_2, n'_1) \langle 0 | \mathcal{O}_2^{(2)} | q_{n_2} g_{n'_1} g_{n'_2} \bar{q}_{\bar{n}} \rangle$$

$$(4.73)$$

We make a few things about the above equations. Firstly, there is a four-parton correction operator that has the same index structure as LO, (n_2, n_2, n'_1) . We cannot obtain it as a pure limit of $A_{\text{NLO}}^{q\bar{q}gg}$, and we will need to subtract off the LO contributions. Secondly, the limit in Eq. (4.71) does not lead to an expansion of any part of $A_{\text{NLO}}^{q\bar{q}gg}$, as the scaling of the *n*-indices' dot products is exactly that from SCET₁. Even though it just gives back the same expression as the SCET₁ amplitude $A_{\text{NLO}}^{q\bar{q}gg}$, the SCET₂ result for $C_{2,\text{NLO}}^{(2)J}(n_2, n'_1, n'_2)\mathcal{O}_2^{(2)}$ tells us something more. This Wilson coefficient is proportional to $\tilde{\Theta}_{\lambda^2}[n'_1 \cdot n'_2]\tilde{\Theta}_{\lambda^2}[n_2 \cdot n'_2]$, where the $\tilde{\Theta}$'s only have support outside the phase space region of Eq. (4.72), as well as the strongly-ordered limit of Eq. (4.73), (see Eqs. (4.45) and (B.61) for the definition of Θ , $\tilde{\Theta}$). The full results for the Wilson coefficients shown in Eq. (4.73) can be found in Eqs. (B.80), (B.94), and (B.102). At the amplitude level, given a particular phase space configuration for an external state, we will only ever need one of these terms for double gluon emission in SCET₂. Squaring the result is straightforward as there will be no interference between them.

We will now examine how to improve the matching of SCET_i to SCET_{i+1} , and show that the jet-structure corrections computed here generalize to a matching result for this case. We first notice that the first two operators above do not interfere with the one giving LO, as they have different index structures. The subleading term in Eq. (4.73) does inhabit the same region of phase space as $C_{2,LO}^{(1)} \mathcal{O}_2^{(1)}$, but as we will argue in Subsection 4.4.4 LO/NLO interference cancels out of most observables of interest. Before proceeding, we note that our description of corrections to two-gluon emission gets even simpler when we match to SCET₃. In SCET₃ the only operator we need has distinct collinear directions for all fields. Thus, we can write all hard and jet corrections to two-gluon emission we have found in the coefficient, $C_3^{(2)}$ for the operator $\mathcal{O}_3^{(2)}(n_2, n'_1, n'_2) = \bar{\chi}_{n_2} g \mathcal{B}_{n'_1 \perp}^{\alpha} g \mathcal{B}_{n'_2 \perp}^{\beta} \chi_{\bar{n}}$, as we do in Eqs. (B.110). The same will hold for *i*-gluon emission in SCET_{i+1}. Our NLO jet-structure operators therefore have the following form:

$$C_{3,\text{NLO}}^{(2)J,I}(n_2, n_1, n_1')\mathcal{O}_3^{(2)} = h_I^{\alpha\beta} \bar{\chi}_{n_2} g \mathcal{B}_{n_1'\perp}^{\alpha} g \mathcal{B}_{n_2'\perp}^{\beta} \Gamma^{\mu} \chi_{\bar{n}} , \qquad (4.74)$$

where $h_I^{\alpha\beta}$ is given by Eq. (B.118). Here $I = \{1, 2, 3\}$ and we distinguish the coefficients $C_{3,\text{NLO}}^{(2)J,I}$ depending from which SCET₂ operators they come from in order to properly account for their renormalization group evolution factors in SCET₂. For I = 1, it comes form the SCET₂ operator $\mathcal{O}_2^{(2)}(n_2, n'_2, n'_1)$, for I = 2 from $\mathcal{O}_2^{(2)}(n_2, n'_1, n'_1)$ and for I = 3 from $\mathcal{O}_2^{(2)}(n_2, n_2, n'_1)$.

When doing the LO matching for SCET_i to SCET_{i+1} , we found that the replacement rule to go from SCET_1 to SCET_2 generalized to the case of *i*-gluon stronglyordered emission. Similarly, we can take the above operator, Eq. (4.74), and recast it as a replacement rule for our original current insertion, $C_{1,\text{LO}}^{(0)}\mathcal{O}_1^{(0)}$. It takes the form of a $1 \rightarrow 3$ replacement rule:

$$\bar{\chi}_{n_0} \to h_I^{\alpha\beta} \bar{\chi}_{n_2} g \mathcal{B}^{\alpha}_{n_1 \perp} g \mathcal{B}^{\beta}_{n_2 \perp}, \qquad (4.75)$$

with contributions from I = 1, 2, 3.

If we want to consider the NLO radiation of i+1 gluons, we can perform a very similar matching between SCET_i and SCET_{i+2} to the one above for $\text{SCET}_1 \to \text{SCET}_3$ to obtain an operator $C_{i+2,\text{NLO}}^{(i+1)J} \mathcal{O}_{i+2}^{(i+1)}$. Since the first (i-1) emissions are strongly ordered, they completely factor out. Thus, the amplitude for the emission of the final two gluons will be identical to that for simple two-gluon emission. We can therefore take the (i - 1) gluon LO operator, $C_{i,\text{LO}}^{(i-1)}\mathcal{O}_i^{(i-1)}$, and use the replacement rule in Eq. 4.75), to obtain $C_{i+2,\text{NLO}I}^{(i+1)J}\mathcal{O}_{i+2}^{(i+1)}$. Our NLO replacement rule corresponds to violating strong ordering at any location in the shower, either by taking the j^{th} and $(j + 1)^{\text{th}}$ gluons to have the same parametric collinearity with respect to their parents, $k_{j+1\perp} \sim k_{j\perp}$, or by including the region of phase space where the propagator between them is hard, and so we get no collinear divergence as the quark and second gluon become collinear.⁷

It is not difficult to see that this gives an NLO contribution for any j. If we have i-gluon strongly-ordered emission, the propagators and vertices will go as $\lambda^{-i(i+1)/2}$, where the j^{th} gluon contributes λ^{-j} . If we violate strong ordering as we mention above for any two gluons, the product of their vertices times propagators goes like λ^{-2j} instead of $\lambda^{-(2j+1)}$. Thus, we can insert $\bar{\chi}_{n_0} \to h_I \bar{\chi}_{n_2} g \mathcal{B}_{n'_1 \perp} g \mathcal{B}_{n'_2 \perp}$ instead of two successive $\bar{\chi}_{n_0} \to c_{\text{LO}} \bar{\chi}_{n_1} \mathcal{B}_{n'_1 \perp}$'s in operator matching as a "defect" in strong ordering at any stage and obtain an NLO jet-structure correction. The Θ -functions contained in the Wilson coefficients, $C_{i,\text{NLO}}^{(i+1)J}$, allow us to read off at which step in the shower we violated strong-ordering. In appendix C, we show how an integrated version of $h_I^{\alpha\beta}$ is related to the subleading splitting function which serves as a cross-check on our computations.

4.4.3 Operator Running

Up until now, our discussion of matching has taken place mostly at tree-level. Connecting to the leading-log parton shower, however, also requires that we include the effects of full one-loop and two-loop cusp needed for NLL running. For this reason, our final expressions for Wilson coefficients in Apps. B.2-B.4 all include the neces-

⁷At this point, one may ask why we do not go farther and consider the case $k_{j+1\perp} \gg k_{j\perp}$. In fact, we do not have to. Since the amplitude for *i*-gluon emission has an underlying Bose symmetry, we are free to partition phase space into *i*! regions, each of which gives an identical contribution to the cross section. Thus, to get the final answer, we only need to integrate over one of them. While we can choose this region such that $k_{j+1\perp} \gg k_{j\perp}$ never occurs, we are forced to include $k_{j+1\perp} \sim k_{j\perp}$. If we do not wish to partition phase space in this manner then the Bose symmetry implies that the result for $k_{j+1\perp} \gg k_{j\perp}$ can be obtained from the configurations already discussed.

sary notation for evolution kernels. Identifying the power suppressed amplitudes as corresponding to perturbative corrections to more inclusive observables it is natural to take only LL evolution for power suppressed or α_s suppressed corrections, and include NLL evolution only for the leading shower terms. In this section we give arguments that determine the LL evolution for the subleading operators, and discuss what precisely is missing to obtain the full NLL evolution kernel.

To set the stage we consider as an example the running in SCET₁. We can consider our matching to QCD to have taken place at a scale Q, but we can run down to a lower scale μ , as is necessary before we further match to SCET₂. The zero and single gluon operators in SCET₁ acquire the following factors (*cf.* the tree-level version in Eq. (4.64)):

$$C_{0}^{(0)}(n_{0}) = U^{(2,0,0)}(n_{0})(Q,\mu)\gamma_{n_{0}\perp}^{\mu}$$

$$C_{1}^{(1)}(n_{0},n_{0}) = U^{(2,1,0)}(n_{0},n_{0})(Q,\mu) \otimes \frac{n_{0}^{\mu} - \bar{n}^{\mu}}{Q}\gamma_{n_{0}\perp}^{\alpha},$$

$$C_{1,\mathcal{T}}^{(1)}(n_{0},n_{0}) = U_{\mathcal{T}}^{(2,1,1)}(n_{0},n_{0})(Q,\mu) \otimes \frac{1}{\bar{q}_{1}\bar{k}_{1}} \Big(\gamma_{n_{0}\perp}^{\mu}\gamma_{n_{0}\perp}^{\beta}\gamma_{n_{0}\perp}^{\alpha} - \frac{2}{\bar{q}_{1}Q}g^{\mu\beta}\gamma_{n_{0}\perp}^{\alpha}\Big),$$

$$C_{1}^{(1)}(n_{1},n_{1}') = -U^{(2,1,0)}(n_{1},n_{1}')(Q,\mu) \Big(\frac{2}{(n_{1}\cdot n_{1}')\bar{q}_{1}\bar{k}_{1}}\gamma_{n_{0}\perp}^{\alpha}\not{p}_{\gamma}\gamma_{T}^{\mu}$$

$$+ \Big[\frac{1}{(n\cdot p_{\bar{q}})\bar{k}_{1}}\Big(\gamma_{T}^{\mu}\not{p}_{\gamma} - \bar{q}_{1}n_{1T}^{\mu}\Big) + \frac{2(n\cdot p_{\bar{q}})}{(n_{1}\cdot n_{1}')\bar{q}_{1}\bar{k}_{1}}\bar{n}_{T}^{\mu}\Big]\gamma_{n_{0}\perp}^{\alpha}\Big).$$
(4.76)

 $U^{(j,k,l)}(n_1, n_2, \ldots, n_{j+k})$ is the RG-kernel of the operator $\mathcal{O}^{(j,k,l)}(n_1, n_2, \ldots, n_{j+k})$ defined in Eq. (4.7) and $U_{\mathcal{T}}^{(2,1,1)}$ is the RG-kernel for the operator $\mathcal{T}_1^{(1)}$. Since the antiquark is always collinear to \bar{n} , we avoid writing \bar{n} in the RG-kernel's argument. We inserted the symbol \otimes in the second and third line of Eq. (B.40), because when we have an operator where two or more fields share the same collinear direction, there can be a convolution in the fraction momentum \bar{p} between the running factor and the coefficient. If the field are collinear in the same direction, they can talk each other in SCET and they can exchange momentum in running down from scale Q to μ . The anomalous dimension of an operator is independent of which SCET_i it is defined, but does depend on the field content and in particular how many different collinear directions are in the operator. Thus, the RG-kernel of the operator $\bar{\chi}_{n_0} \mathcal{B}^{\alpha}_{n_0\perp} \chi_{\bar{n}}$ is

different from that of $\bar{\chi}_{n_1} g \mathcal{B}^{\alpha}_{n'_1 \perp} \chi_{\bar{n}}$.

In Ref. [17] the LL part of $U^{(j,i-j,0)}(Q, \mu)$ was related to the Sudakov form factor Eq. (4.10) (up to accounting for the soft effects of angular ordering). The cusp term in the anomalous dimension gives the LL part, and comes from soft and collinear one-loop diagrams. The result from the soft diagrams is constrained by that of the collinear diagrams in order to cancel out infrared sensitivity that cannot be absorbed in local counterterms at the hard scale. Here we will use this same argument, but in reverse, in order to determine the LL anomalous dimension of various subleading operators.

Due to the soft-collinear factorization, the soft structure only depends on the number of collinear directions. After making the field redefinition operators like $\bar{\chi}_{n_0}\chi_{\bar{n}}$ and $\bar{\chi}_{n_0}g\mathcal{B}^{\alpha}_{n_0\perp}\chi_{\bar{n}}$ both have $Y^{\dagger}_{n_0}Y_{\bar{n}}$, and so both have the same soft divergences, and hence have the same one-loop cusp term. Thus the leading-log resummation only depends on the number of collinear index directions in the operator. We therefore have

$$U_{\rm LL}^{(2,0,0)}(n_0) = U_{\rm LL}^{(2,1,1)}(n_0, n_0) = U_{\rm LL,\mathcal{T}}^{(2,1,1)}(n_0, n_0) \,. \tag{4.77}$$

Thus at LL order we have the full set of evolution kernels for subleading collinear operators and we account for these factors in the appendix expressions. Since this is a LL effect we expect the effects of soft radiation and angular ordering to be incorporated in a manner identical to the evolution factor in the LL shower.

An important consequence of this result for the LL evolution is that it justifies treating our hard-scattering corrections as improvements to the fixed-order, matrixelement calculation that goes into a shower algorithm. Correcting the two-jet amplitude with either $C_{1,\text{NLO}}^{(1)}$ or $C_{1,\mathcal{T}}^{(1)}$, we see that the LL resummation is the same as that in the standard shower except that there is an extra parton already inside the leading jet. We thus get a shower correction just by using a matrix element improved by including our hard-scattering terms. This is different from simply running a LL shower on higher order matrix elements, as different anomalous dimensions control each operator's evolution. Some, operators, like those just mentioned with only n_0 and \bar{n} collinear directions, run like two-jet configurations, that is with a quark-antiquark Sudakov. Others, (e.g. $C_1^{(1)}(n'_1, n_1)\mathcal{O}_1^{(1)}$) have three-parton running since they have three distinct collinear directions. This latter set corresponds to the usual implementation of fixed order corrections in parton showers, but the former is a novel type of shower improvement whose implementation will require further study.

On the other hand, the effect of jet-structure corrections is not to modify the initial scattering process, but to go hand in hand with the NLL change to the leading operators' running. The complication we must face for this calculation is that this correction to the evolution kernel must in principal be carried out in the same scheme used to distinguish the phase space regions for the jet-structure corrections, and hence may depend on the choice for the Θ functions. Furthermore it is likely that power suppressed soft effects will also have implications for the subleading evolution kernel. Our lack of an appropriate NLL evolution factor for the shower is due to these two issues. In addition at this level of accuracy one must take into account the appropriate scale choice for $\alpha_s(\mu)$ in the LL expressions.

To setup the distinction between kinematic regions we used Wilsonian type Θ functions, but from the point of view of evolution $\overline{\text{MS}}$ would be simpler. Although it is not directly relevant to our setup it is nevertheless still interesting to consider how the NLL evolution kernel would arise in $\overline{\text{MS}}$. As we will discuss in Appendix C, when integrated over phase space in dimensional regularization the jet-structure corrections give the real emission portion of $P_{qq}^{(1)}$, which is the $\mathcal{O}(\alpha_s)$ correction to the Altarelli-Parisi splitting kernel. Combined with known SCET results for single-emission at one-loop, we can recover all of $P_{qq}^{(1)}$. Obtaining this expression is important both conceptually and practically. It validates our formal expansion in λ , showing that corrections to $\mathcal{O}(\lambda^2)$, along with a set of previously calculated SCET one-loop diagrams, capture the contributions needed for NLL resummation. On the practical side it provides a cross check on the computations.

With $P_{qq}^{(1)}$ in hand we can extend the argument of [16, 17] that the Sudakov factor gives the LL part of the the RG kernel $U^{(i)}(Q, \mu)$ (Eq. (4.10)) to the NLL level, looking at $U^{(0)}(Q,\mu)$ for running of the operator $C_{1,LO}^{(0)}\mathcal{O}_1^{(0)}$. Using the Sudakov factor of [31] for quarks, we have:

$$\Delta_q(Q,\mu) = \exp\left\{-\frac{C_F}{2\pi} \int_{\mu}^{Q} \frac{d\mu'}{\mu'} \alpha_s(\mu') \int_{\frac{\sqrt{\mu'}}{Q}}^{1-\frac{\sqrt{\mu'}}{Q}} dz \frac{1+z^2}{1-z}\right\},\tag{4.78}$$

where we recognize $P_{qq}^{(0)}$, Eq. (4.3). Performing the z integral and expanding in the limit of large Q gives:

$$\Delta_q(Q,\mu) \approx \exp\left\{\frac{C_F}{\pi} \int_{\mu}^{Q} \frac{d\mu'}{\mu'} \alpha_s(\mu') \left[\log\left(\frac{\mu'^2}{Q^2}\right) + \frac{3}{2}\right]\right\},\tag{4.79}$$

which is identical to $U^{(0)}(Q,\mu)$ at one-loop. The term in the exponent proportional to $\log(\mu'^2/Q^2)$ sums the leading logs in the parton shower. Interpreting Eq. (4.79) as an RG kernel, this log piece is coming from the one-loop cusp anomalous dimension, C_F . The factor of 3/2 is the remaining part of the one-loop anomalous dimension, and it sums part of the collinear NLL.⁸ In order to get the full NLL summation, one also needs corrections corresponding to the two-loop cusp anomalous dimension. This is a known result in SCET for the operator $\bar{\chi}_n\chi_{\bar{n}}$, which we can relate to $P_{qq}^{(1)}$, by adding the subleading splitting function to the exponent of $\Delta_q(Q,\mu)$. Unlike Eq. (4.79), this new expression actually sums next-to-leading logs in the $\overline{\text{MS}}$ scheme. We wish to stress, however, that the ultimate goal to improve parton showers through resummation is to include all next-to-leading-logs. In this work, we have not considered the effects of soft NLL, nor those related to the two-loop running of α_s . Furthermore, one needs the NLL running for all of our operators. While our formulas in Apps. B.2-B.4 do include all LL running, we only have NLL for $C_i^{(0)}\mathcal{O}_i^{(0)}$. The collinear-NLL-improved Sudakov corresponding to it is:

$$\Delta_q^{\text{NLL}}(Q,\mu) = \exp\left\{-\int_{\mu}^{Q} \frac{d\mu'}{\mu'} \int_{\frac{\sqrt{\mu'}}{Q}}^{1-\frac{\sqrt{\mu'}}{Q}} dz \left[P_{qq}^{(0)}(z,\mu') + P_{qq}^{(1)}(z,\mu')\right]\right\}, \quad (4.80)$$

⁸ Since Eq. (4.79) resums some NLL contributions [31] calls it the NLL Sudakov factor.

where $P_{qq}^{(0)}$ given in Eq. (4.3) and $P_{qq}^{(1)}$ in [39]. Once again, we integrate in z, expanding in large Q to get:

$$\Delta_{q}^{\text{NLL}}(Q,\mu) = \exp\left\{\int_{\mu}^{Q} \frac{d\mu'}{\mu'} \left[\frac{\alpha_{s}(\mu')}{\pi}C_{F}\left(\log\left(\frac{\mu'^{2}}{Q^{2}}\right) + \frac{3}{2}\right) + \frac{\alpha_{s}^{2}(\mu')}{4\pi^{2}}C_{F}\left(C_{g}\left(\frac{67}{9} - \frac{\pi^{2}}{3}\right) - \frac{20}{9}C_{F}T_{F}n_{F}\right)\log\left(\frac{\mu'^{2}}{Q^{2}}\right)\right]\right\}, \quad (4.81)$$

where the term $\propto \alpha_s^2$ reproduces the known result for the two-loop cusp anomalous dimension.

4.4.4 From Operators Toward a Corrected Shower

As discussed previously, our end goal is to match down to an EFT, $SCET_N$, where each field has its own index direction. Further Lagrangian emission from these operators is physically meaningless, as the resolution scale is set $\sim \mathcal{O}(\text{GeV})$, below which we stop computing in perturbation theory and pass to a hadronization routine. Thus in SCET_N we match everything to the single operator $\mathcal{O}_N^{(N-1)}(n_N, n'_1, \ldots, n'_{N-1})$ and all the information Lo and NLO is encoded in the Wilson coefficients. We have calculated the coefficients From this $SCET_N$, we define a map to different shower algorithm ingredients. We gave those needed for a LL shower Monte Carlo in Table 4.1, which only required single gluon considerations for collinear effects. To include the leading SCET power corrections and consistently resum collinear NLL, we will need operators with an arbitrary of partons in the mapping. However, as we saw in Secs. 4.4.1 and 4.4.2, we only needed to do two-gluon emission computations to obtain the amplitudes to NNLO that contribute the leading corrections in λ to observables. For the hard-scattering corrections, this was because modifying the emissions of particles further away from the initial hard-scale production led to increasingly suppressed terms. For jet-structure, while corrections can occur anywhere, the leading ones only involve the most minimal deviation from strong-ordering, that affecting nearest-neighbor The LO coefficient for $SCET_N$ is given in Eq. (4.53), and for NLO are given in appendix B.4. As we demonstrate below, upon squaring these amplitudes, the simplicity of

the picture remains, and we only need to consider the non-trivial phase space of two particles at a time.

Interference for LO² and Jet-Structure Corrections

It is a general statement about SCET fields with different n index labels that they have no overlap in Hilbert space. As an example, we can take two different operators, \mathcal{O}_{n_1} and \mathcal{O}_{n_2} . All the fields in \mathcal{O}_{n_1} , except that labeled by n_1 (e.g. χ_{n_1}), have exact analogs in \mathcal{O}_{n_2} , creating the same type of particle and having the same index label. Instead, \mathcal{O}_{n_2} contains a field labeled by n_2 , which may or may not be the same as n_1 . We thus have:⁹

$$\langle q_1, q_2, \dots, q_m | \mathcal{O}_{n_1}^{\dagger} | 0 \rangle \langle 0 | \mathcal{O}_{n_2} | q_1, q_2, \dots, q_m \rangle = \delta_{n_1, n_2} \langle q_1, q_2, \dots, q_m | \mathcal{O}_{n_1}^{\dagger} | 0 \rangle \langle 0 | \mathcal{O}_{n_2} | q_1, q_2, \dots, q_m \rangle.$$

$$(4.82)$$

It is to guarantee a relation like Eq. (4.82) that our Wilson coefficients contain Θ -functions (*cf.* Eqs. (4.45) and (B.61)), which will cutoff the overlap regions in phase space once we begin integrating. The amplitude squared is particularly easy in $SCET_N$, where we have only the operator $\mathcal{O}_N^{(N-1)}(n_N, n'_1, \ldots, n'_{N-1})$, and where each particle is defined in a different collinear direction.

SCET_i simplifies the amplitude squared by distinguishing at the operator level which configurations are strongly-ordered and which are not. This means that we have no interference between $C_{N,LO}^{(N-1)}\mathcal{O}_N^{(N-1)}$ and $C_{N,NLO}^{(N-1)J}\mathcal{O}_N^{(N-1)}$ where $C_{N,LO}^{(N-1)}$ is the LO SCET_N coefficient defined in Eq. (4.53), and $C_{N,NLO}^{(N-1)J}$ is the NLO SCET_N coefficient defined in Eq.(B.119). Even though the \mathcal{O} 's are the same, the Θ -functions in the C's enforce different conditions, where the former is strongly ordered, while the latter is not. Thus, in the analog of Eq. (4.82), the Kronecker delta will give zero.

We get a further simplification when we square the NLO contributions. Looking

⁹By RPI, n_1 and n_2 do not have to be exactly equal, but must concur up to an angle of $\mathcal{O}(\lambda^i)$ in SCET_i .


Figure 4-12: Amplitude squared for the LO operator $C_{NLO}^{(i)} \mathcal{O}_N^{(i)}$. Though we do the amplitude computation in SCET_N , we illustrate the process here with a cut SCET_1 Feynman diagram in order to emphasize the simple ladder structure.

at $C_{N, \text{NLO}}^{(N-1)J}$ in detail, we have:

$$C_{N,\text{NLO}}^{(N-1)J} = \sum_{l=1}^{N-2} C_{N,\text{NLO}}^{(N-1)J}(l) , \qquad (4.83)$$

where

$$C_{N,\text{NLO}}^{(N-1)J}(l) = \sum_{I=1}^{3} \left[\left(\prod_{k=1}^{l-1} U_{\text{LL}}^{(k-1)}(\mu_{k-1}, \mu_{k}) c_{\text{LO}}^{\alpha_{k}}(n_{k-1}) \right) U_{\text{LL}}^{(l-1)}(\mu_{k-1}, \mu_{k}) \otimes h_{I}^{\alpha,\beta}(n_{l+1}, n_{l}', n_{l+1}') \right. \\ \left. \times \left(\prod_{k=l+1}^{N-1} U_{\text{LL}}^{(k-1)}(\mu_{k-1}, \mu_{k}) c_{\text{LO}}^{\alpha_{k}}(n_{k-1}) \right) \right] \Gamma^{\mu} .$$

$$(4.84)$$

In $C_{N, \text{NLO}}^{(N-1)J}(l)$ we have made explicit that the $l, (l+1)^{\text{th}}$ gluons violate strong-ordering and come with the factor $h^{\alpha\beta}$ of the subleading splitting rule, Eq. (4.75). The sum in the last term over I counts the different types of NLO jet-structure terms given in Eq. (4.74). The $c_{\text{LO}}^{\alpha_k}$ are defined in Eqs. (B.115), and the U's are running factors. The complete explanation of the symbols in Eq. (4.84) can be found in the discussion around Eq. (B.120). Since different l correspond to a violation of strong-ordering at different points in the shower, each of the $C_{i,\text{NLO}}^{(i+1)J}(l, l+1)$ codes a different index structure. Therefore, there is no interference for different values of l, and we have that the amplitude squared to NNLO for jet-structure corrections is just the sum of squares of the individual operators:

$$|A^{q(N-1)g\bar{q}J}|^2_{\text{to NNLO}} = |A^{q(N-1)g\bar{q}}|^2_{\text{LO}} + |A^{q(N-1)g\bar{q}J}|^2_{\text{NNLO}}, \qquad (4.85)$$

where

$$|A^{q(N-1)g\bar{q}}|_{\rm LO}^2 = |C_{N\,\rm LO}^{(N-1)}|^2 |\langle 0|\mathcal{O}_N^{(N-1)}|q(N-1)g\bar{q}\rangle|^2, \qquad (4.86)$$
$$|A^{q(N-1)g\bar{q}J}|_{\rm NNLO}^2 = \sum_{l=1}^{N-2} |C_{N,\rm NLO}^{(N-1)J}(l)|^2 |\langle 0|\mathcal{O}_N^{(N-1)}|q(N-1)g\bar{q}\rangle|^2,$$

where with $|q(N-1)g\bar{q}\rangle$ we indicate a state with N-1 gluon emission. The simplification even extends inside each of the terms, since the j^{th} gluon only gets contracted with itself. Diagramatically, this means there are zero nearest-neighbor crossings in the $|\text{LO}|^2$ diagram, as we see in Fig. 4-12 and a maximum of *one* in the $|\text{NLO}|^2$ case shown in Fig. 4-13. We thus only slightly modify the factorized emission formula, Eq. (4.2).

We have proved that at NLO for the jet-structure corrections the only non trivial part on the amplitude squared involves at most two near gluons. We can see why terms that have interference with more than two gluons are suppressed, just looking at the internal propagators in the amplitude. The amplitude for i + 1 emissions has a factor $1/q_1^2 \times 1/q_2^2 \times \cdots \times 1/q_i^2$ due to the propagators. The LO term comes form the strong-ordered region where each new emission is more collinear then the previous one, that means where $q_1^2 \gg q_2^2 \gg \cdots \gg q_i^2$, Eq. (4.1). The jet-structure NLO is given when $q_j^2 \sim q_{j+1}^2$ and this allows the two gluons k_{j+1} and k_{j+2} to share the same region of the phase space and so to interference. Let us suppose that we instead want to interference the gluon k_{j+1} with k_{j+3} . When these two gluons have the same momentum we have $q_j^2 \sim q_{j+1}^2 \sim q_{j+2}^2$, and therefore the associated amplitude is clearly suppressed with respect to the LO and jet-structure NLO amplitudes.



Figure 4-13: Contribution to the amplitude squared of the jet-structure piece at NLO. We are using a cut SCET₁ Feynman diagram to show that the SCET_N operator, $C_{N,\text{NLO}}^{(N)J}\mathcal{O}_N^{(N)}$, contains only a single deviation from the simple ladder structure apparent at LO (Fig. 4-12).

Interference for Hard-Scattering Corrections

The corrections to the cross-section to $\mathcal{O}(\lambda^2)$ involve hard-scattering ones as well. Unlike the jet-structure case, these involve NNLO amplitude terms, as well. As we argued above, they only involve the gluons closest to the hard interaction. Thus, we will not need to sum over many terms as we do in Eq. (4.86). In fact, for hardscattering corrections, we only need to worry about interfering SCET_i operators that arise from acting with the LO replacement rule Eq. (4.46) on $C_{1,\text{LO}}^{(0)}$, $C_{1,\text{NLO}}^{(1)}$, and $C_{1,\tau}^{(1)}$, which are given in Eqs. (4.30) and (4.64). Since the 2nd through *i*th gluons arise from the LO rule for all three coefficients, they proceed as in the LO/LO case. The interference to look at in detail is that of the first two gluons. We only need to compute single emission graphs, as we can add the other emissions to the amplitude squared by multiplying the appropriate splitting functions and propagators. In SCET_N, we have:

$$|A^{q(N-1)g\bar{q}H}|^{2}_{\text{to NNLO}} = \left(|C_{N,\text{LO}}^{(N-1)\dagger}C_{N,\text{NLO}}^{(N+1)H} + C_{N,\text{NLO}}^{(N-1)H\dagger}C_{N,\text{LO}}^{(N-1)}| + |C_{N,\text{NLO}}^{(N-1)H}|^{2} + |C_{N,\text{NLO}}^{(N-1)H}|^{2} + |C_{N,\text{NLO}}^{(N-1)\dagger}C_{N,\text{NLO}}^{(N-1)H\dagger}C_{N,\text{LO}}^{(N-1)}| \right) |\langle 0|\mathcal{O}_{N}^{(N-1)}|q(N-1)g\bar{q}\rangle|^{2}$$

$$(4.87)$$

The Wilson coefficients are found in Eqs. (B.114), (B.116), and (B.122), respectively. The only nontrivial interference in Eq. (4.87) occurs between the first two-gluon emissions.

The interference between LO and NLO simplifies in many cases of interest. For example for one-gluon emission

$$|A_{\rm LO/NLO}^{qg\bar{q}}|_{\mu\nu}^{2} = \frac{4\bar{q}_{1}\bar{p}_{\bar{q}}}{q_{0}^{2}}k_{1\perp\nu}(n_{\mu}-\bar{n}_{\nu}).$$
(4.88)

If we can cleanly separate the initial and final states (e.g. $e^+e^- \rightarrow \text{jets}$), then by a classic proof (*i.e.* that in [84])) involving the Ward identity, once we have integrated over final state vector quantities (we are allowed to keep scalars such as z_i unintegrated), the resulting differential observable depends on $g^{\mu\nu}|A_{\text{LO/NLO}}|^2_{\mu\nu}$, which for Eq. (4.88) is zero. While this is quite straightforward for leptonic initial states, one may be able to extend it to certain hadronic ones as well. Once again, as we said in sec. 4.4.1, one can account for these corrections by modifying the hard-scale matrix element and then running a parton shower modified to include the different no-branching probabilities for different phase space configurations of the same particle content. Also unlike standard shower corrections, we can avoid double counting issues, in principle, because all contributions, whether LO, hard-scattering, or jet-structure corrections are kept separately in distinct operators.

4.5 Correction Map at NLO/NLL

We summarize our results for parton shower in Table 4.2 including ingredients necessary for NLO/NLL accuracy in the cross-section. Since it is easier, in the table we use the language of $SCET_1$ to discuss the corrections, rather than referring to terms in the

Category	Shower Ingredients	Quantity in $SCET_i$	Found In:
Hard Scattering	Hard matrix elements	Wilson coeff. of $\bar{\chi}_{n_1} \mathcal{B}_{n'_1} \chi_{\bar{n}}$	Eq. (4.67)
	with more partons	in SCET ₁	
	Power correction	Wilson coeff. of $\bar{\chi}_{n_0} \mathcal{B}_{n_0} \chi_{\bar{n}}$	Eq. (4.67)
	to initial branching	and $\bar{\chi}_{n_0}[\mathcal{P}_{\perp}\mathcal{B}_{n_0}]\chi_{\bar{n}}$	
	within the leading jet		
	$\mathcal{O}(\alpha_s)$ hard virtual	One-loop matching for	See [16, 17]
	correction	$ar{\chi}_{n_1}\chi_{ar{n}}$	
Jet Structure	$1 \rightarrow 3$	Double gluon real	Eq. (4.74)
	Splitting functions	emission in $SCET_1$	
	$\mathcal{O}(\alpha_s)$ virtual correction	One-loop correction	Left for
	for LO $1 \rightarrow 2$ splittings	to $1 \rightarrow 2$ replacement rule	future work
	Probability	Compare SCET	Left for
	of $1 \rightarrow 2$ vs. $1 \rightarrow 3$	amplitudes	future work
No Branching	NLL Sudakov factor for	NLL anomalous dimension	Left for
Probabilities	leading branching	for leading operators	future work
	LL Sudakovs for	LL anomalous dimensions	Eqs.(4.10)
	subleading branching	for subleading operators	and (4.77)
Soft Emission	Subleading corrections	Include effects of soft	Left for
	from soft gluons	emission from subleading	future work
		SCET soft Lagrangians	

Table 4.2: Mapping between ingredients for a NLO parton shower algorithm and computations in SCET_i . For exclusive cross-sections these ingredients would together yield results accurate to NLO in α_s , NLO in the power expansion (λ), and with corresponding NLL resummation.

final SCET_N. In SCET_N the features of SCET operators that avoid double counting and allow the various contributions to be distinguished are encoded by Θ functions in the Wilson coefficients. In considering the totally differential cross-section we found at NLO two kinds of power corrections. This includes a set of matrix-element corrections that we called hard-scattering corrections (Section 4.4.4), and a set of contributions that improve double real emissions that we called jet-structure corrections (Section 4.4.4).

In the hard category we have overall three different kinds of corrections. The first is due the the SCET₁ operator $\bar{\chi}_{n_1} \mathcal{B}_{n'_1} \chi_{\bar{n}}$ that gives the SCET₂ coefficient $C_{2,\text{NLO}}^{(1)H,b}$ in Eq. (4.67). It is an improvement of the hard matrix element that takes into account an extra parton. The second is due to the SCET₁ operators $\bar{\chi}_{n_0} \mathcal{B}_{n_0} \chi_{\bar{n}}$ and $\bar{\chi}_{n_0} [\mathcal{P}_{\perp} \mathcal{B}_{n_0}] \chi_{\bar{n}}$ that give the SCET₂ coefficients $C_{2,\text{NLO}}^{(1)H,a}$ and $C_{2,\text{NNLO}}^{(1)H}$ in Eq. (4.67). This correction also accounts for more partons, but it describes a situation where they are initially emitted close to the collinear quark rather than widely separated, a correction to the initial branching within the leading jet. It is important to note that the two kinds of corrections have different renormalization group evolution and thus different Sudakov no-branching factors. For a full NLL resummation we also need a third type of hard scattering correction, the one-loop virtual corrections to the leading shower operator. For the required operator $\bar{\chi}_{n_0}\chi_n$ these were discussed in Refs. [16, 17].

For the category of jet-structure corrections there are again several ingredients to consider. We derived a replacement rule for two emissions $1 \rightarrow 3$, which should be added on top of two leading order $1 \rightarrow 2$ splittings. This correction takes into account emissions in a region of the phase space that is not strongly-ordered. In addition at NLO/NLL we require the $\mathcal{O}(\alpha_s)$ virtual correction to the LO splitting rule. This would be derived from a one loop matching computation that should be straightforward, but was not considered here. In addition the shower requires a new type of probability for when to do a $1 \rightarrow 3$ splitting versus a standard $1 \rightarrow 2$ splitting. The shower we are discussing has positive weights, since the square of amplitudes is positive, and it is this additional probability function that accounts for situations where the original LL $1 \rightarrow 2$ shower yields a result that is larger than the physical observable of interest. A full investigation of this is left to the future.

We also saw that the Sudakov factors, that give the no branching probability, are associated with the running factors of the operator and in turn with their anomalous dimension (Section 4.4.3). To NLL we need the NLL Sudakovs factor for leading branching and the LL Sudakovs for subleading branching that are associated repsectively to the NLL anomalous dimensions for the leading operators and LL anomalous dimensions for the subleading operators. At LL we have the Sudakovs for subleading branching (Eqs. (4.10) and (4.77)), but we have not yet calculated the NLL Sudakov for leading branching in the scheme appropriate for our setup, as described in Subsection 4.4.3. The last item in the table is the treatment of soft radiation at NLO. This can be achieved by considering time-ordered products for the matching of QCD to SCET_1 and SCET_i to SCET_{i+1} that involve soft gluons and subleading soft Lagrangians that are known in SCET [15, 24, 25]. One must then work out the effect that these NLO soft amplitudes have on interference. The associated soft calculations and investigations have been left for future work.

Not included in table 4.2, but worth recalling, is the fact that our NLO results have been expressed in terms of a particular set of variables, namely dot products of lightlike vectors $n_i \cdot n_j$ and large light-cone momenta \bar{q}_i . As mentioned in Subsection 4.3.3 there will be corrections at NLO/NLL that arise because of the choice of shower evolution variables and the translation of the SCET_i results to this choice, and these must be considered on a case by case basis. They are not accounted for elsewhere in our NLO corrections.

4.6 Conclusion

In this work we developed a rigorous framework based on a tower of independent but related EFTs, the SCET_i, to study corrections to the parton shower. The work of [16, 17] showed how to formulate the LL parton shower in terms of SCET, and how virtual corrections are straightforward to incorporate by one-loop matching. The SCET_i framework extends these ideas in a manner that makes it easy to deal with double counting issues, the issue of disentangling coordinate choices from kinematic power corrections, and the construction of a complete set of operators for corrections at a desired order. The interference structure and hence the leading corrections to spin correlations and color correlations are also straightforward to work out in the SCET_i setup.

The SCET_i are iteratively used to integrate out the characteristic scale, $Q\lambda^i$ for increasing *i*. This approach allows one to perform a systematic expansion which can correct both the hard-scale process that produce partons and the parton shower itself. We described the parton shower through the operators $\mathcal{O}_j^{(i)}$ in SCET_j and used standard matching procedures to make the transition from SCET_j to SCET_{j+1} where more partons become apparent. Performing the matching relied crucially on the RPI symmetry of SCET, and we extended the usual infinitesimal version to carry out the finite rotations in each $SCET_i$ that we needed. A summary of ingredients required for the NLO/NLL shower are given in Table 4.2, including both calculations carried out here as well as those left for future work. The main result of our work are:

- 1. An easy replacement rule for the LL shower, $\chi_{n_0} \to c_{\text{LO}}^{\alpha} \chi_{n_1} g \mathcal{B}_{n_1'\perp}^{\alpha}$, where c_{LO} is related to the standard LO splitting-function. Also a demonstration that angular ordering and coherent branching for soft emissions emerge naturally in the SCET_i framework.
- 2. At NLO we found two kinds of corrections: hard-scattering corrections and jet-structure corrections. The hard-scattering corrections depend on the hard process and appear near the top of the shower tree. They came from matching QCD to $SCET_1$ at higher order. Since they only occur at the top of the shower one can treat these as a modified form of matrix-element correction. A subset of these corrections correspond to the usual implementation of fixed-order matrix elements, while the remaining ones give power corrections to the initial branching in the LL shower.
- 3. The jet-structure corrections are independent from what happens at the hard scale, hence they are universal for any process we want to study. They come from matching SCET_i to SCET_{i+1} at higher order for any *i*. They can appear anywhere in the shower tree and they take into account emissions in regions of the phase space that are not strongly-ordered. For these corrections we found that the NLO operator are related to the LO operator via a replacement rule for two emissions: $\chi_{n_0} \to h_I^{\alpha\beta} \chi_{n_2} g \mathcal{B}^{\alpha}_{n'_1 \perp} g \mathcal{B}^{\beta}_{n'_2 \perp}$.
- 4. The SCET_i picture allows us to easily take into account interference. Once we reach the final SCET_N theory all the fields are labeled in a different collinear directions. Because in SCET we can only can contract collinear fields that share the same collinear direction, in SCET_N calculating the amplitude squared becomes very easy. Kinematic information that is encoded by the shower history

from passing through earlier SCET_i s is encoded by Θ functions in the final SCET_N Wilson coefficients. We proved that when emitting an arbitrary number of partons, the non-trivial part of the amplitude squared involves at most two fields.

The framework developed here allows for systematic improvement to arbitrary orders in the kinematic expansion. There are still several important steps to take, though, before this picture can lead to a practical implementation, including additional computations that we have outlined. We list here three topics whose development would be particularly useful, and which we believe should be straightforward to approach:

- This work has only considered q → qg splittings and an abelian theory. One should include the full nonabelian results and compute the coefficients required for gluon splitting as well. This is required to properly treat color correlation corrections in a manner determined by the NLO interference pattern. For collinear particles we expect that one can include the dominant part of these effects by considering nearest-neighbor interference, leaving the rest of the shower as before.
- 2. Only a subset of the terms required for a full NLL resummation were considered. We computed LL evolution for subleading operators, but did not carry out the computation of the NLL evolution of the leading operator in a scheme that is consistent with our power corrections (we only considered it in $\overline{\text{MS}}$). In order for a consistent treatment as a probabilistic process, the real emission probabilities and Sudakov no-branching corrections must go hand in hand. We also did not consider how to include the effects of our NLO jet-structure $1 \rightarrow 3$ replacements in an algorithm. The non-trivial task is how to simultaneously implement $1 \rightarrow 2$ and $1 \rightarrow 3$ splittings in a consistent manner.
- 3. Since soft modes in SCET can communicate between different collinear jets, they carry the ability to spoil their factorization. Fortunately, this did not

happen when we included their LO couplings. In fact, we were able to derive the condition of angular ordering and coherent branching in SCET_i . It is open question as to what extent NLO soft couplings can be factorized in the shower tree and the necessary SCET computations were not carried out here. The treatment of soft NLO interactions in SCET in the past has always led to factorized structures, so we remain optimistic that such effects will be tractable for the shower.

We also briefly comment on how the corrections in Table 4.2 relate to corrections already implemented in parton shower codes in the literature. In most cases the goal of these codes differed from the strict NLO/NLL shower corrections considered here. This makes a strict association impossible, but there is still a general correspondence that can be made. CKKW [31] is a LO(α_s)/LL procedure whose goal is to merge matrix elements involving multiple partons with a parton shower in a manner that avoids double counting. In our language this corresponds to the real emission hardscattering corrections in the first row of Table 4.2. The $\bar{\chi}_{n_0}\chi_{\bar{n}}$ and $\bar{\chi}_{n_1}\mathcal{B}_{n'_1}^{\perp}\chi_{\bar{n}}$ operators describe processes with different numbers of initial well-separated jets. In CKKW a parameter y_{cut} is used to separate the extra emission in the matrix element from emissions in the shower. In our analysis the contributions from showering $\bar{\chi}_{n_0}\chi_{\bar{n}}$ does not interfere with the direct contribution from $\bar{\chi}_{n_1}\mathcal{B}_{n'_1}^{\perp}\chi_{\bar{n}}$, and this is encoded by Θ functions in the Wilson coefficient of SCET_N.

In MC@NLO [52] and POWHEG [82], virtual and real matrix element corrections at NLO in α_s are incorporated into the shower, with the goal of ensuring that it reproduces an associated cross-section completely at NLO in α_s . The implementation includes careful handling of the cancellation of real and virtual IR divergences. In Table 4.2 our goal was to implement corrections at NLO in powers and NLL in logs/ α_s , but for all jets from the shower rather than just the first jet needed for the NLO crosssection. At NLL we have only terms up to $\mathcal{O}(\alpha_s \log)$ in the total cross-section, and hence does not encode the entire NLO result (in α_s). In our language the corrections that contribute to the NLO cross section correspond to the hard scattering corrections in the first plus second row, and third rows of Table 4.2. In order to compute the NLO cross section it is not necessary to distinguish between the terms in the first and second rows of the table, and these terms are indeed considered simultaneously in MC@NLO and POWHEG.

The hard scattering corrections in our second row correspond to how to treat real radiation at higher precision that has the same strongly ordered kinematic configuration as the LO shower. The work of KRKMC group [60, 61, 93], on the other side, aims to improve the shower algorithm taking into account the Altarelli-Parisi splitting function at NLO: $P_{qq}^{(1)}$. In our language this corresponds to our jet-structure corrections and we have seen in Appendix C how the replacement rule in Eq. (4.75) is related to $P_{qq}^{(1)}$. In our method we can consistently take into account both correction at order α_s in the $1 \rightarrow 2$ splitting function, as well as the subleading contribution that comes from the $1 \rightarrow 3$ splitting function.

156

.

Chapter 5

Conclusions

At hadron colliders we compare data to theoretical predictions for exclusive quantities in order to take into account experimental cuts and detector effects that are necessary to isolate the signal from the the background. Monte Carlo event generators have been proved indispensable for making exclusive theoretical prediction and they are heavily used by experimentalists to simulate events. They are based on a leading log (LL) parton shower algorithm that is defined in the soft-collinear limit and uses a probabilistic Markov chain of $1 \rightarrow 2$ particle splittings to recursively generate partons. There have been several improvements to LL parton showers, but a systematic way to resum next-to-leading log (NLL) is missing in the literature and there is not even a clear method to catalog all the necessary corrections. An improvement in the traditional event generators up to NLL, would allow us to better distribute particles in phase space, and to have a better normalization of the shower. This will in turn improve the shape of prediction for cross-section at the LHC and thus yield more precise theoretical predictions.

In this thesis we developed a rigorous framework to study corrections to partons showers using soft collinear effective field theory (SCET) to pave the way for an implementation of a NLL parton shower algorithm. Our work is based on a tower of independent but related EFTs, the SCET_i. We described the parton shower through the operators $\mathcal{O}_i^{(j)}$ in SCET_i, where j is the number of parton in the shower, and we used standard matching procedures to make the transition from SCET_i to SCET_{i+1} where more partons become apparent. Even if we have many EFTs, we will use only a single power counting parameter and we calculated corrections to next-to-leading order (NLO) in the power counting for the parton shower.

At LO in the power counting we recover the usual LL shower. We also proved that angular ordering and coherent branching for soft emissions emerge naturally in the SCET_i framework. At NLO we found two kinds of corrections: hard-scattering corrections and jet sub-structure corrections. The hard-scattering corrections depend on the initial hard process and one can treat these as a modified form of matrix-element correction. A subset of these corrections correspond to the usual implementation of fixed-order matrix elements, while the remaining ones give power corrections to the initial branching in the LL shower. The jet-structure corrections are the same for any process we want to study. They can appear anywhere in the shower tree and they take into account emissions in regions of the phase space that are not strongly-ordered. For these corrections we found that determing the NLO operator can be encoding by a $1 \rightarrow 3$ particle splitting replacement rule. The SCET_i picture allows us to easily take into account interference. We proved that when emitting an arbitrary number of partons, the non-trivial part of the amplitude squared involves at most two fields.

Thus, our work represents an improvement of parton shower both at the interface with the matrix element and in the parton splitting. In Table 4.2 we enlisted the corrections we calculated as well as all the missing ingredients for a full NLL shower. The next things to calculate are: virtual collection for the LO $1 \rightarrow 2$ splittings, the NLL Sudakov factor for the leading branching and the subleading correction from soft gluons. The last step towards a NLL shower is to implement these corrections in a NLL shower algorithm that is able to disentangle the probability of $1 \rightarrow 3$ splittings versus a standard $1 \rightarrow 2$ splittings.

Performing the matching between different $SCET_i$ relied on the reparametrization invariant (RPI) symmetry of SCET. We extensively studied RPI in Chapter 3. We constructed operators that are invariant under reparametrization transformations and we use them to reduce the number of operators in SCET. We constructed a minimum basis of pure glue operators for DIS at twist-4, for production of two and three jets from e^+e^- , and for production of two jets form gluon fusion.

.

Appendix A

Invariance to the choice of hard-vector q^{μ}

From the construction in section 3.3, a natural question arises about the special role of q^{μ} in Eq. (3.15). What happens if there is more than one possible choice for q^{μ} in a given process? Say we have a q^{μ} and a q'^{μ} with Wilson coefficients that can depend on q^2 , q'^2 , and $q \cdot q'$, where $q_{\perp} \sim q'_{\perp} \sim \lambda$. It turns out that in this case any linear combination of q^{μ} and q'^{μ} in Eq. (3.15) is equally good, and is equivalent to any other choice. Hence, one choice suffices. To prove this we consider the expansion of the reparametrization invariant variable

$$\xi \equiv \frac{2q \cdot q'}{q^2} \pm \sqrt{\left(\frac{2q \cdot q'}{q^2}\right)^2 - \frac{4q'^2}{q^2}} = \frac{n \cdot q'}{n \cdot q} + \mathcal{O}(q_{\perp}^2), \qquad (A.1)$$

where we take the plus sign if the expansion is done with $n \cdot q'/n \cdot q > \bar{n} \cdot q'/\bar{n} \cdot q$ and the minus sign otherwise. Now use this variable to define

$$q' \cdot i\partial - \xi \left(q \cdot i\partial \right) \equiv \hat{Q}_{\text{INV}}^{[2]}, \qquad (A.2)$$

where the operator $\hat{Q}_{\text{INV}}^{[2]}$ is RPI and its expansion starts at order λ^2 . Thus

$$\int d\omega C(\omega) \,\delta(\omega - q' \cdot i\partial) = \int d\omega C(\omega) \,\delta(\omega - \xi \, q \cdot i\partial - \hat{Q}_{\rm INV}^{[2]})$$

$$= \int d\omega' C(\xi\omega') \,\delta(\omega' - q \cdot i\partial - \hat{Q}_{\rm INV}^{[2]}/\xi)$$

=
$$\int d\omega' \left[\tilde{C}(\omega') \,\delta(\omega' - q \cdot i\partial) + \tilde{B}(\omega') \,\hat{Q}_{\rm INV}^{[2]} \,\delta'(\omega' - q \cdot i\partial) + \dots \right], \qquad (A.3)$$

where in the second line we changed the dummy variable to $\omega' = \xi \omega$, In the last line both terms are RPI, and the ellipsis denotes higher order terms which are also RPI order by order in λ . Eq. (A.3) demonstrates that we can swap the parameter $q' \rightarrow q$ in the δ -function, since the change is compensated by a change of notation in the leading order Wilson coefficient $C \rightarrow \tilde{C}$. Given that we imagine starting with a complete basis of RPI operators built with $\delta(\omega - q' \cdot i\partial)$ or with $\delta(\omega' - q \cdot i\partial)$, the higher terms in the series in Eq. (A.3), like \tilde{B} , also simply change a Wilson coefficient in our basis. Thus, the choice of q or q' in the δ -function just corresponds to a different choice of the basis for the invariant operators, and one choice suffices.

Appendix B

Matching $SCET_i$ to $SCET_{i+1}$

In this appendix we perform the matching QCD to SCET_1 in subsection B.2, SCET_1 to SCET_2 in subsection B.3, and SCET_2 to SCET_3 to SCET_N in subsection B.4. A key element that we use in the SCET_i to SCET_{i+1} matching is the finite RPI transformations that are defined in subsection B.1.

B.1 Finite RPI

We have seen in chapter 2 that we can always decompose the momentum p of a particle in two light-cone directions n and \bar{n}

$$p^{\mu} = n \cdot p \frac{\bar{n}^{\mu}}{2} + \bar{p} \frac{n^{\mu}}{2} + p_{\perp}^{\mu} , \qquad (B.1)$$

with $n^2 = 0$, $\bar{n}^2 = 0$ and $n \cdot \bar{n} = 2$. We define p collinear to the direction n if the components of (B.1) scale as $(n \cdot p, \bar{p}, p_{\perp}) \sim (\lambda^2, 1, \lambda) Q$, where Q is the hard scale and $\lambda \ll 1$. The vector n has physical meaning, \vec{n} is the direction where most of the momentum is allocated, that is the direction \vec{p} is inside a cone of opening angle λ around \vec{n} , see Fig (2-1). \bar{n} does not carry any physical meaning and it is only needed to decompose the momentum in (B.1). The power counting λ defines the level of the collinearity. The decomposition (B.1) is not unique, we can shift n by an amount λ and the particle we still be collinear to the direction n. Pictorially this means that

if we move the vector n inside the cone in Fig. (2-1), p is always collinear to it. The transformations that move the n inside the collinear cone are the reparametrization invariant (RPI) transformation of type-I, Eq. (2.24). Thus if a particle is collinear to a direction n, it is also collinear to any directions n' that is related to n with an RPI transformation.

Two collinear sectors in SCET, n_1 and n_2 , are distinct if [9]

$$n_1 \cdot n_2 \gg \lambda^2 \,, \tag{B.2}$$

We will label the external state with the direction n where the particles are collinear to and with a subscript the magnitude of collinearity, so for example $|q_{n_1}q_{n_2}\rangle_1$ is a state with two quarks, one collinear to n_1 and one to n_2 where $n_1 \cdot n_2 \gg (\lambda)^2$. A collinear field in SCET labeled with n_i can create a particle collinear to the n_i direction, but this means that state is also collinear to any $n_j \in [n_i]$, where $[n_i]$ is the equivalence class defined in Eq. (2.25), so we have

$$\xi_{n_i} |q_{n_j}\rangle = \begin{cases} u_{n_i} & \text{if } n_j \in [n_i] \\ 0 & \text{otherwise} \end{cases},$$
(B.3)

The type-I RPI transformation connects vectors that belong to the same class $[n_j]$.

For each $\{n, \bar{n}\}$, the type-I RPI infinitesimal transformations are given in Eq. (2.24) These transformations preserve the relations $n^2 = 0$, $\bar{n}^2 = 0$ and $n \cdot \bar{n} = 2$. In the matching from SCET_i to SCET_{i+1} we need to rotate the direction n to a direction n' that is closer to the momentum of the particle, such that p is collinear to n' in SCET_{i+1}, The RPI transformations allow us to do it. Thus the finite RPI transformations in SCET_i is crucial to match the two theories and hence is not simply a convention. On the other hand the choice within SCET_{i+1} is purely a convention. We have some freedom in choosing n', if λ^{i+1} is the power counting of SCET_{i+1} any n' such that \vec{p} is inside a cone of aperture λ^{i+1} around $\vec{n'}$ is fine, see Fig. (4-3). The momentum p decomposed along n' and \bar{n} is

$$p^{\mu} = n' \cdot p \frac{\bar{n}^{\mu}}{2} + \bar{p} \frac{n'^{\mu}}{2} + p^{\mu}_{n'\perp} .$$
 (B.4)

For this convention we will pick n' as that direction n_p where p as zero perpendicular momentum, that is

$$p = \bar{p}\frac{n_p^{\mu}}{2} + n_p \cdot p\frac{\bar{n}^{\mu}}{2} \,. \tag{B.5}$$

The relation between n and n_p is

$$n_{p}^{\mu} = n^{\mu} + 2 \frac{p_{\perp}^{\mu}}{\bar{p}} - \bar{n}^{\mu} \frac{(p_{\perp})^{2}}{\bar{p}^{2}}.$$
 (B.6)

This is a RPI finite transformation (instead Eq. (2.24) is infinitesimal). It is easy to check that $n_p^2 = 0$, $n_p \cdot \bar{n} = 2$ and that $p_{n_p\perp}^{\mu} = p^{\mu} - n_p \cdot p \, \bar{n}^{\mu}/2 - \bar{p} \, n_p^{\mu}/2 = 0$.

We will derive similar relations for other quantities. To see how the quark field transform we use the RPI invariant fermion field [80]

Because (B.7) is invariant under RPI, $\psi_n = \psi_{n_p}$ and we can write

Multiplying (B.8) by the projector $\frac{\hbar \pi}{4}$ we get the finite RPI relation

$$\xi_n = \frac{\eta \bar{\eta}}{4} \xi_{n_p} \,. \tag{B.9}$$

The relation (B.9) is in agreement with the spinor equation (A7) in [17] once you choose $\bar{n}_1 = \bar{n}_2$. Object with a full Lorentz index like p^{μ} or γ^{μ} , are RPI invariant because there is no reference to the light-cone vectors n and \bar{n} , but on object in the perpendicular direction, like p^{μ}_{\perp} or γ^{μ}_{\perp} are not, because \perp is defines relative to n and \bar{n} . Using the relation $\gamma_{\perp}^{\mu} = \gamma^{\mu} - \bar{n}^{\mu} \, n/2 - n^{\mu} \, n/2$, we derive the expression

Now we focus on one gluon emission. We consider the case of a virtual quark with momentum q_0 emitting an external gluon and quark with momentum k_1 and q_1 respectively. In Fig. (B-1-A) we portrayed one gluon emission where the initial quark q_0 comes from a QCD current $\bar{q}\gamma^{\mu}q$. We call n_0 , n'_1 and n_1 the directions where q_0 ,



Figure B-1: Kinematic variable for one gluon emission (A), two-gluon emission (B) and *i* gluon emission (C). The *n*s vector are defined such that the momentum has not perpendicular component along that directions $n - \bar{n}$.

 k_1 and q_1 zero have perpendicular direction, that is

$$q_{0} = \bar{q}_{0} \frac{n_{0}^{\mu}}{2} + n_{0} \cdot q_{0} \frac{\bar{n}^{\mu}}{2} ,$$

$$k_{1} = \bar{k}_{1} \frac{n_{1}^{\prime \mu}}{2} ,$$

$$q_{1} = \bar{q}_{1} \frac{n_{q_{1}}^{\mu}}{2} ,$$
(B.11)

where $\bar{q}_0 = \bar{q}_1 + \bar{k}_1$ by momentum conservation in the \bar{n} direction, and q_0 has a \bar{n} component because it is off-shell. Using the Eq. (B.6) we can relate n'_1 and n_1 to n_0

$$n_{1}^{\prime \mu} = n_{0}^{\mu} - 2 \frac{(q_{1})_{n_{0}\perp}^{\mu}}{\bar{k}_{1}} - \bar{n}^{\mu} \frac{(q_{1})_{n_{0}\perp}^{2}}{\bar{k}_{1}^{2}},$$

$$n_{1}^{\mu} = n_{0}^{\mu} + 2 \frac{(q_{1})_{n_{0}\perp}^{\mu}}{\bar{q}_{1}} - \bar{n}^{\mu} \frac{(q_{1})_{n_{0}\perp}^{2}}{\bar{q}_{1}^{2}},$$
(B.12)

where we have used the equality $(k_1)_{n_0\perp}^{\mu} = -(q_1)_{n_0\perp}^{\mu}$. Some useful relations are

$$n_{1} \cdot n_{1}' = n_{0} \cdot n_{1}' \frac{\bar{q}_{0}^{2}}{\bar{k}_{1}^{2}} = n_{0} \cdot n_{1} \frac{\bar{q}_{0}^{2}}{\bar{q}_{1}^{2}} = -2 \frac{(q_{1})_{n_{0}\perp}^{2} \bar{q}_{0}^{2}}{\bar{q}_{1}^{2} \bar{k}_{1}^{2}},$$

$$n_{0}^{\mu} = \frac{\bar{k}_{1} n_{1}'^{\mu} + \bar{q}_{1} n_{1}^{\mu}}{\bar{q}_{0}} - \bar{n}^{\mu} (n_{1} \cdot n_{1}') \frac{\bar{q}_{1} \bar{k}_{1}}{2 \bar{q}_{0}^{2}},$$

$$(q_{1})_{n_{0}\perp}^{\mu} = \frac{1}{2} \frac{\bar{q}_{1} \bar{k}_{1}}{\bar{q}_{1} + \bar{k}_{1}} \sqrt{|n_{1} \cdot n_{1}'|} v_{1}^{\mu} - \bar{n}^{\mu} (n_{1} \cdot n_{1}') \frac{\bar{q}_{1} \bar{k}_{1} (\bar{k}_{1}^{2} - \bar{q}_{1}^{2})}{4 \bar{q}_{0}^{3}},$$

$$\gamma_{n_{0}\perp}^{\mu} = \gamma_{n_{1}'\perp}^{\mu} - \bar{n}^{\mu} \frac{(q_{1})_{n_{0}\perp}}{\bar{k}_{1}} - (q_{1})_{n_{0}\perp}^{\mu} \frac{\bar{q}_{1}}{\bar{k}_{1}} - \bar{n}^{\mu} (n_{1} \cdot n_{1}') \frac{\bar{q}_{1}^{2}}{2 \bar{q}_{0}^{2}} \bar{\eta}, \qquad (B.13)$$

where

$$v_1^{\mu} = \frac{n_1^{\mu} - n_1'^{\mu}}{\sqrt{|n_1 \cdot n_1'|}}, \qquad (B.14)$$

is a vector with $|v_1^2| = 2$. Another useful relation is

$$q_0^2 = (q_1 + k_1)^2 = n'_1 \cdot n_1 \frac{\bar{q}_1 \bar{k}_1}{2}$$
 (B.15)

Notice that we can express all quantities in terms of the vectors n'_1 , n_1 and the momenta \bar{q}_1 and \bar{k}_1 .

In two-gluon emissions the kinematic variables are defined in Fig. (B-1-B). We called n_0 , n'_1 , n'_2 and n_1 the directions where q_0 , k_1 , k_2 and q_1 have perpendicular direction zero, that is:

$$q_{0} = \bar{q}_{0} \frac{n_{0}^{\mu}}{2} + n_{0} \cdot q_{0} \frac{\bar{n}^{\mu}}{2} ,$$

$$k_{1} = \bar{k}_{1} \frac{n_{1}^{\prime \mu}}{2} ,$$

$$q_{1} = \bar{q}_{1} \frac{n_{1}^{\mu}}{2} + n_{1} \cdot q_{1} \frac{\bar{n}^{\mu}}{2} ,$$

$$k_{2} = \bar{k}_{2} \frac{n_{2}^{\prime \mu}}{2} ,$$

$$q_{2} = \bar{q}_{2} \frac{n_{2}^{\mu}}{2} ,$$
(B.16)

now q_1 is also off-shell and have a component along \bar{n} . The definition (B.12) is still

valid, and we can define n'_2 and n_2 as

$$n_{2}^{\prime \mu} = n_{1}^{\mu} - 2 \frac{(q_{2})_{n_{1}\perp}^{\mu}}{\bar{k}_{2}} - \bar{n}^{\mu} \frac{(q_{2})_{n_{1}\perp}^{2}}{\bar{k}_{2}^{2}},$$

$$n_{2}^{\mu} = n_{1}^{\mu} + 2 \frac{(q_{2})_{n_{1}\perp}^{\mu}}{\bar{q}_{2}} - \bar{n}^{\mu} \frac{(q_{2})_{n_{1}\perp}^{2}}{\bar{q}_{2}^{2}},$$
(B.17)

where $(k_2)_{n_1\perp} = -(q_2)_{n_1\perp}$. The relations (B.13) are still valid, and also the relations with $0 \to 1$ and $1 \to 2$, these are

$$n_{2} \cdot n_{2}' = n_{1} \cdot n_{2}' \frac{\bar{q}_{0}^{2}}{\bar{k}_{2}^{2}} = n_{0} \cdot n_{1} \frac{\bar{q}_{0}^{2}}{\bar{q}_{2}^{2}} = -2 \frac{(q_{2})_{n_{1}\perp}^{2} \bar{q}_{1}^{2}}{\bar{q}_{1}^{2} \bar{k}_{2}^{2}},$$

$$n_{1}^{\mu} = \frac{\bar{k}_{2} n_{2}'^{\mu} + \bar{q}_{2} n_{2}^{\mu}}{\bar{q}_{1}} - \bar{n}^{\mu} (n_{2} \cdot n_{2}') \frac{\bar{q}_{2} \bar{k}_{2}}{2 \bar{q}_{1}^{2}},$$

$$(q_{2})_{n_{1}\perp}^{\mu} = \frac{1}{2} \frac{\bar{q}_{2} \bar{k}_{2}}{\bar{q}_{2} + \bar{k}_{2}} \sqrt{|n_{2} \cdot n_{2}'|} v_{2}^{\mu} - \bar{n}^{\mu} (n_{2} \cdot n_{2}') \frac{\bar{q}_{2} \bar{k}_{2} (\bar{k}_{2}^{2} - \bar{q}_{2}^{2})}{4 \bar{q}_{1}^{3}},$$

$$\gamma_{n_{1}\perp}^{\mu} = \gamma_{n_{2}\perp}^{\mu} - \bar{n}^{\mu} \frac{(q_{2})_{n_{1}\perp}}{\bar{k}_{2}} - (q_{2})_{n_{1}\perp}^{\mu} \frac{\bar{\eta}}{\bar{k}_{2}} - \bar{n}^{\mu} (n_{2} \cdot n_{2}') \frac{\bar{q}_{2}^{2}}{2 \bar{q}_{1}^{2}} \bar{\eta},$$
(B.18)

where

$$v_2^{\mu} = \frac{n_2^{\mu} - n_2^{\prime \mu}}{\sqrt{|n_2 \cdot n_2^{\prime}|}}, \qquad (B.19)$$

is a vector with $|v_2^2| = 2$. We can write $n_1 \cdot n_1'$ and v_1 in terms of n_2 , n_1' and n_2' :

$$n_1 \cdot n_1' = \frac{\bar{k}_2(n_2' \cdot n_1') + \bar{q}_2(n_2 \cdot n_1')}{\bar{q}_1} - (n_2 \cdot n_2') \frac{\bar{q}_1 \bar{k}_1}{\bar{q}_0^2}, \qquad (B.20)$$

$$v_1^{\mu} = \frac{2\,\bar{k}_2\bar{q}_1\,n_2^{\prime\mu} + 2\,\bar{q}_1\bar{q}_2n_1^{\mu} - \bar{k}_2\bar{q}_2\,(n_2\cdot n_2^{\prime})\,\bar{n}^{\mu} - 2\,\bar{q}_1n_1^{\prime\mu}}{\sqrt{|n_1\cdot n_1^{\prime}|}}\,.\tag{B.21}$$

(B.15) for two emissions is modified to

$$q_0^2 = (q_2 + k_1 + k_2)^2 = n_1' \cdot n_2 \frac{\bar{q}_2 \bar{k}_1}{2} + n_2' \cdot n_2 \frac{\bar{q}_2 \bar{k}_2}{2} + n_1 \cdot n_2' \frac{\bar{k}_1 \bar{k}_2}{2}.$$
 (B.22)

Others useful relations are

$$q_1^2 = (k_1 + q_2)^2 = (n'_1 \cdot n_2) \frac{k_1 \bar{q}_2}{2},$$

$$(k_2 + q_2)^2 = (n'_2 \cdot n_2) \frac{\bar{k}_2 \bar{q}_2}{2}.$$
(B.23)

For *i* gluon emissions Fig. (B-1-C), we call n'_k the light-cone vector parallel to the *k*-gluon, n_i the light-cone vector parallel to the external quark, and n_k the light cone vector such that the *k* internal virtual quark has zero perpendicular momentum with respect to (n_k, \bar{n}) . To calculate n_i, n'_i we can iterate the formulas above up to *i* emissions. That is we can calculate n_i, n'_i from n_{i-1} using Eq. (B.12) with $0 \to (i-1)$, $1 \to i$.

B.2 Matching QCD to $SCET_1$

To study the process of $q \rightarrow qg$ emission, we match the QCD current,

$$J^{\mu}_{\rm QCD} = \bar{q}\gamma^{\mu}q\,,\tag{B.24}$$

to SCET₁ operators for a final state with a quark, antiquark, and gluon. The particle momenta are q_1 for the quark, $p_{\bar{q}}$ for the antiquark, and k_1 for the gluon, (*cf.* Fig. B-2A). We do the matching in the center of mass frame with

$$p_{\gamma} = q_1 + p_{\bar{q}} + k_1 = (Q, 0, 0, 0).$$
 (B.25)

SCET₁, being equivalent to the usual SCET, is formulated as an expansion in the parameter λ . The current in Eq. (B.24) matches onto an infinite series of SCET₁ operators. We will perform the matching up to NNLO for one gluons emission, and focus only on the cases when the gluon is either collinear to the quark or far form both the quark and the antiquark. Obtaining the limit of gluon-antiquark collinearity from our work is a simple exercise in charge conjugation. We can construct the SCET₁ operators out of a few building blocks: the quark field χ_n , the gluon field $\mathcal{B}_{n\perp}^{\alpha}$ and the perpendicular momentum operator $\mathcal{P}_{n\perp}^{\alpha}$, plus Dirac structures. χ_n , $\mathcal{B}_{n\perp}^{\alpha}$ and $\mathcal{P}_{n\perp}^{\alpha}$ all scale as λ . The basis of SCET₁ operators for one emission up to NNLO is [80]. ¹.

$$\mathcal{O}_{1}^{(0)}(n_{0}) = \bar{\chi}_{n_{0}} \chi_{\bar{n}} ,$$

$$\mathcal{O}_{1}^{(1)}(n_{0}, n_{0}) = \bar{\chi}_{n_{0}} g \mathcal{B}_{n_{0} \perp}^{\alpha} \chi_{\bar{n}} ,$$

$$\mathcal{T}_{1}^{(1)}(n_{0}, n_{0}) = \bar{\chi}_{n_{0}} \left[\mathcal{P}_{n_{0} \perp}^{\beta} g \mathcal{B}_{n_{0} \perp}^{\alpha} \right] \chi_{\bar{n}} ,$$

$$\mathcal{O}_{1}^{(1)}(n_{1}, n_{1}') = \bar{\chi}_{n_{1}} g \mathcal{B}_{n_{1}' \perp}^{\alpha} \chi_{\bar{n}} ,$$
(B.26)

 $\mathcal{O}_1^{(0)}(n_0)$ stands for $\mathcal{O}_1^{(0)}(n_0, \bar{n})$ and similarly for the other operators, we do not write the antiquark direction as it is always \bar{n} . $\mathcal{O}_1^{(0)}$ is the LO operator and scales as λ^2 ,

 $[\]overline{\mathcal{T}_{1}^{(1)}(n_{1},n_{1}')}$ is zero because we can choose the directions n_{1} and n_{1}' to align perfectly with particle momenta such that *e.g.* $\mathcal{P}_{n_{1}\perp}\mathcal{B}_{n_{1}\perp}=0$

$$\mathcal{O}_1^{(1)}(n_0, n_0)$$
 and $\mathcal{O}_1^{(1)}(n_1, n_1')$ are the NLO operators, scaling like λ^3 , and $\mathcal{T}_1^{(1)} \propto \lambda^4$.

In SCET₁, two particles are collinear if they are inside a cone of aperture λ , or equivalently if $p_1 \cdot p_2 \lesssim (Q\lambda)^2$. Usually, we formulate this condition with dimensionless quantities, $n_{p_1} \cdot n_{p_2} \leq \lambda^2$, where n_{p_i} is exactly proportional to the particle momentum. To distinguish a two-jet from a three-jet state, we label the external states with the direction to which the particles are collinear. A state $|q_{n_0}\rangle_1$ indicates a state where a quark with momentum q_1 is collinear to the direction n_0 , that is $(\bar{q}_1, n_0 \cdot q_1, (q_1)_{n_0 \perp}) \sim$ $(1, \lambda^2, \lambda)Q$, and this state can be annihilated by any operator, χ_n , where n and n_0 are in the same SCET₁ equivalence class, $\{[n]\}$. The subscript, 1, tells us that we are using the $SCET_1$ classes here. As we will see when we match to lower-scale $SCET_i$, we will change this number appropriately. A two-jet state with a collinear quark and gluon, and an antiquark along \bar{n} is given by $|q_{n_0} g_{n_0} \bar{q}_{\bar{n}}\rangle_1$. The fact that the quark and gluon share an index label implies that $q_1 \cdot k_1 \lesssim (\lambda)^2$. A three-jet state is indicated by $|q_{n_1} g_{n'_1} \bar{q}_{\bar{n}}\rangle_1$, where each particle is collinear to a different direction. The operators $\mathcal{O}_1^{(0)}(n_0), \ \mathcal{O}_1^{(1)}(n_0,n_0) \ \text{and} \ \mathcal{T}_1^{(1)}(n_0,n_0) \ \text{are two-jet operators, that is they can only}$ create a two-jet state, whereas $\mathcal{O}_1^{(1)}(n_1, n_1')$ is for three-jets. Multiplying the operator in (B.26) by the Wilson coefficients, we have

$$J_{\text{QCD}}^{\mu} = C_{1,\text{LO}}^{(0)}(n_0)\mathcal{O}_1^{(0)}(n_0) + C_1^{(1)}(n_0, n_0)\mathcal{O}_1^{(1)}(n_0, n_0) + C_{1,\mathcal{T}}^{(1)}(n_0, n_0)\mathcal{T}_1^{(1)}(n_0, n_0) + C_1^{(1)}(n_1, n_1')\mathcal{O}_1^{(1)}(n_1, n_1') + \dots, \qquad (B.27)$$

where the ellipses indicate higher order terms in λ . We begin by looking at two-jet operators in detail. For this region, because we are in the center of mass frame, the two jets are back to back. We define the kinematics as follows, the antiquark is exactly parallel to $\bar{n} = (1, 0, 0, -1)$, while the quark and the gluon are collinear to $n_0 = (1, 0, 0, 1)$, such that $q_0 = q_1 + k_1$ has no component perpendicular to n_0 and \bar{n} :

$$p_{\bar{q}}^{\mu} = n_0 \cdot p_{\bar{q}} \frac{\bar{n}^{\mu}}{2} ,$$

$$q_1^{\mu} = \bar{q}_1 \frac{n_0^{\mu}}{2} + n_0 \cdot q_1 \frac{\bar{n}^{\mu}}{2} + (q_1)_{n_0 \perp}^{\mu} ,$$



Figure B-2: Matching QCD to SCET₁ for the two-jet configuration: In the first column there are the two Feynman graphs for one-gluon emission in QCD, labelled by the 4-momenta. In the second column there are the two Feynman graphs in SCET₁ that reproduce the same amplitude in the case the quark and gluon are collinear along the direction n_0 . The first graph come from the operator $\mathcal{O}_1^{(0)}$ with the insertion of the SCET₁ Lagrangian, the second graph comes from the operators $\mathcal{O}_1^{(1)}$ and $\mathcal{T}_1^{(1)}$.

$$k_1^{\mu} = \bar{q}_1 \frac{n_0^{\mu}}{2} + n_0 \cdot k_1 \frac{\bar{n}^{\mu}}{2} + (k_1)_{n_0 \perp}^{\mu}, \qquad (B.28)$$

where $(n_0 \cdot q_1, \bar{q}_1, q_{1\perp})$ and $(n_0 \cdot k_1, \bar{k}_1, k_{1\perp})$ scale as $(\lambda^2, 1, \lambda)$, and $(q_1)_{n_0\perp}^{\mu} = -(k_1)_{n_0\perp}^{\mu}$ by momentum conservation. The Wilson coefficients are defined through the equation

$$\langle 0|J_{\text{QCD}}^{\mu}|q_{n_0} g_{n_0} \bar{q}_{\bar{n}} \rangle_1 = C_{1,\text{LO}}^{(0)}(n_0, n_0) \int dx^4 \langle 0|T\{\mathcal{L}_{\text{SCET}_1}(x)\mathcal{O}_1^{(0)}\}|q_{n_0} g_{n_0} \bar{q}_{\bar{n}} \rangle_1$$

$$+ C_1^{(1)}(n_0, n_0) \langle 0|\mathcal{O}_1^{(1)}|q_{n_0} g_{n_0} \bar{q}_{\bar{n}} \rangle_1 + C_{1,\mathcal{T}}^{(1)}(n_0, n_0) \langle 0|\mathcal{T}_1^{(1)}|q_{n_0} g_{n_0} \bar{q}_{\bar{n}} \rangle_1$$

$$(B.29)$$

Calculating the C's for this two-jet process goes as follows. We decompose the QCD amplitude along n_0 and \bar{n} using Eq. (B.28) and we write the QCD spinor in terms of the SCET₁ spinor Eq. (B.31). Expanding in λ up to NNLO, on the RHS, we compute the amplitudes for the three SCET₁ terms. The coefficient $C_{1,LO}^{(0)}$ was already determined from matching QCD to $SCET_1$ for zero gluon emission, it is

$$C_{1,\rm LO}^{(0)} = \gamma^{\mu} \,. \tag{B.30}$$

The coefficients $C_1^{(1)}$ and $C_{1,\mathcal{T}}^{(1)}$ come from solving Eq. (B.29) at NLO and NNLO, respectively. Since $\mathcal{O}_1^{(1)}$ and $\mathcal{T}_1^{(1)}$ are at different orders in λ , there are no ambiguities in solving Eq. (B.29) for both $C_1^{(1)}$ and $C_{1,\mathcal{T}}^{(1)}$.

In order to do the matching, we need the relation between the QCD and SCET spinors. Using Eq. (B.7), we can write:

$$u(p) = \left(1 + \frac{\not p_\perp \vec{n}}{2\,\bar{p}}\right) u_n(p), \qquad (B.31)$$

where u(p) is the QCD spinor and $u_n(p)$ is the SCET₁ one. It easy to see that the SCET spinor satisfies

$$\frac{\vec{n}\vec{n}}{4}u_n = 0,$$

$$\frac{\vec{n}\vec{n}}{4}u_n = 0,$$

$$\sum_s \bar{u}_n^s u_n^s = \bar{p}\frac{\vec{n}}{2}.$$
(B.32)

Note that the normalization in the spin sum does not introduce NLO terms.

The QCD amplitude for $\gamma^* \to q\bar{q}g$ is shown in Fig. B-2:

$$A_{\rm QCD}^{q\bar{q}g} = \bar{u}(q_1)ig\gamma^{\alpha}\frac{i\not{q}_0}{q_0^2}\gamma^{\mu}v(p_{\bar{q}}) - \bar{u}(q_1)ig\gamma^{\mu}\frac{i\not{p}_0}{p_0^2}\gamma^{\alpha}v(p_{\bar{q}}).$$
(B.33)

The amplitude (B.33) is intended to be multiplied by the polarization vector for the gluon, that in general we do not explicitly write. Using Eqs. (B.28) and (B.31) in (B.33) and expanding to NNLO in λ we get

$$A_{\rm QCD}^{q\bar{q}g} = A_{\rm LO}^{q\bar{q}g} + A_{\rm NLO}^{q\bar{q}g} + A_{\rm NNLO}^{q\bar{q}g} , \qquad (B.34)$$

where

From (B.35), and knowing $C_{1,\text{LO}}^{(0)}$, it is easy to determine the other two Wilson coefficients to reproduce $A_{\text{QCD}}^{q\bar{q}g}$, they are

$$C_{1}^{(1)}(n_{0}, n_{0}) = \frac{1}{Q} (n_{0}^{\mu} - \bar{n}^{\mu}) \gamma_{n_{0} \perp}^{\alpha},$$

$$C_{1,\mathcal{T}}^{(1)}(n_{0}, n_{0}) = \frac{1}{\bar{q}_{1} \bar{k}_{1}} \gamma_{n_{0} \perp}^{\mu} \gamma_{n_{0} \perp}^{\beta} \gamma_{n_{0} \perp}^{\alpha} - \frac{2}{\bar{q}_{1} Q} g^{\beta \mu} \gamma_{n_{0} \perp}^{\alpha},$$
(B.36)

where we have used the relation $\bar{q}_1 + \bar{k}_1 = Q$.

For the three-jet operator $O_1^{(1)}(n_1, n'_1)$, the matching was done in subsection 3.7.6, but we will translate to the notation used here. In this case we need three distinct directions in SCET to describe the three external particles, and there is no small parameter to expand in. This means that the amplitude for this operator is exactly equal to the tree-level QCD amplitude for a $q\bar{q}g$ process. One may wonder then, why we simply do not apply this everywhere instead of just the three-jet region. The answer has to do with running effects. The RG kernels of our two-jet operators, \mathcal{O}_1^0 , \mathcal{O}_1^1 , and \mathcal{T}_1^1 , will resum the large collinear logarithms of those configurations (*cf.* Subsection 4.4.3). It is for this reason that we gain by keeping track of these as separate contributions.

Since the three external particles are collinear to three different directions, in principle we have to decompose each particle using its own pair of light-cone vectors, see Eq. (2.1). However, we can show that we need only four independent ones. In the center of mass frame, $q_0 = q_1 + k_1$ is back to back with the antiquark, $p_{\bar{q}} \propto \bar{n}$. We decompose q_0 along (n_0, \bar{n}) such that it has no component perpendicular to them: $q_0 = n \,\bar{n} \cdot q_0/2 + \bar{n} \, n_0 \cdot q_0/2$. Using Eqs. (B.12), we define the light-cone vectors n_1 and n'_1 such that they are parallel to q_1 and k_1 , respectively. We can actually decompose the quark along (n_1, \bar{n}) and the gluon into (n'_1, \bar{n}) . Unlike the two-jet case, where $(q_1)_{n_0\perp} \leq \lambda$, since the quark was collinear to n_0 , here $(q_1)_{n_0\perp} > \lambda$. Summarizing, we have:

$$q_{1}^{\mu} = \bar{n} \cdot q_{1} \frac{n_{1}^{\mu}}{2}, \qquad (B.37)$$

$$p_{\bar{q}}^{\mu} = n \cdot p_{\bar{q}} \frac{\bar{n}^{\mu}}{2}, \qquad (B.37)$$

$$k_{1}^{\mu} = \bar{n} \cdot k_{1} \frac{n_{1}^{\prime \mu}}{2}, \qquad (B.37)$$

where $\bar{n} \cdot q_1$, $n \cdot p_{\bar{q}}$ and $\bar{n} \cdot k_1$ are $\mathcal{O}(1)$. Since a three-jet configuration has $q_1 \cdot k_1 > \lambda^2$, we have that $n_1 \cdot n'_1 > \lambda^2$. Choosing the directions n_1 , n'_1 and \bar{n} exactly parallel to the external momenta means that we do not need $\mathcal{T}_1^{(1)}(n_1, n'_1) = \bar{\chi}_{n_1} \mathcal{P}_{n'_1} \mathcal{B}_{n'_1 \perp}^{\alpha} \chi_{\bar{n}}$, as $\mathcal{P}_{n'_1}$ acting on the gluon field gives zero.

The matching is therefore given by

$$\langle 0|J_{\text{QCD}}^{\mu}|q_{n_1}q_{\bar{n}}\,g_{n_1'}\rangle_1 = C_1^{(1)}(n_1,n_1')\langle 0|O_1^{(1)}(n_1,n_1')|q_{n_1}q_{\bar{n}}\,g_{n_1'}\rangle_1\,,\tag{B.38}$$

and the Wilson coefficient is

where the subscript T applied to a generic four vector f^{μ} means: $f_T^{\mu} \equiv f^{\mu} - p_{\gamma}^{\mu} (f \cdot p_{\gamma})/p_{\gamma}^2$ and p_{γ} is defined in Eq. (B.25). We notice that all the Wilson coefficients in SCET₁ are of order λ^0 . We described the effects of adding running to them in Subsection 4.4.3. As we discuss further in Appendix B.3, we do not need to compute any suppressed two-gluon operators in SCET₁ to the order at which we are working. In the next Appendix we will mach SCET₁ to SCET₂. Before doing it we have to run down the SCET₁ operators form the scale Q to the scale μ_1 where we have the first emission:

$$C_{0}^{(0)}(n_{0}) = U^{(2,0,0)}(n_{0})(Q,\mu)\gamma_{n_{0}\perp}^{\mu}$$

$$C_{1}^{(1)}(n_{0},n_{0}) = U^{(2,1,0)}(n_{0},n_{0})(Q,\mu) \otimes \frac{n_{0}^{\mu} - \bar{n}^{\mu}}{Q}\gamma_{n_{0}\perp}^{\alpha},$$

$$C_{1,\mathcal{T}}^{(1)}(n_{0},n_{0}) = U_{\mathcal{T}}^{(2,1,1)}(n_{0},n_{0})(Q,\mu) \otimes \frac{1}{\bar{q}_{1}\bar{k}_{1}} \Big(\gamma_{n_{0}\perp}^{\mu}\gamma_{n_{0}\perp}^{\beta}\gamma_{n_{0}\perp}^{\alpha} - \frac{2}{\bar{q}_{1}Q}g^{\mu\beta}\gamma_{n_{0}\perp}^{\alpha}\Big),$$

$$C_{1}^{(1)}(n_{1},n_{1}') = -U^{(1)}(n_{1},n_{1}')(Q,\mu) \Big(\frac{2}{(n_{1}\cdot n_{1}')\bar{q}_{1}\bar{k}_{1}}\gamma_{n_{0}\perp}^{\alpha}\not{p}_{\gamma}\gamma_{T}^{\mu}$$

$$+ \Big[\frac{1}{(n\cdot p_{\bar{q}})\bar{k}_{1}}\Big(\gamma_{T}^{\mu}\not{p}_{\gamma} - \bar{q}_{1}n_{1T}^{\mu}\Big) + \frac{2(n\cdot p_{\bar{q}})}{(n_{1}\cdot n_{1}')\bar{q}_{1}\bar{k}_{1}}\bar{n}_{T}^{\mu}\Big]\gamma_{n_{0}\perp}^{\alpha}\Big).$$
(B.40)

For the definition of the running factors $U^{(i,j,k)}(Q,\mu)$ see Section 4.4.3.

B.3 Matching $SCET_1$ to $SCET_2$

B.3.1 One-Gluon Emission

We now match SCET₁ to SCET₂ for one-gluon emission and two-gluon emissions. In this section we only deal with one emission. The basis of SCET₂ operators necessary for the matching up to NNLO is equal to Eq. (B.26) but with the operators defined in SCET₂ instead of SCET₁: $\mathcal{O}_2^{(0)}(n_0)$, $\mathcal{O}_2^{(1)}(n_0, n_0)$, $\mathcal{T}_2^{(1)}(n_0, n_0)$, $\mathcal{O}_2^{(1)}(n_1, n'_1)$.² In the previous section we have matched QCD to SCET₁ for one emission and we saw that in SCET₁ we can have either a two-jet $(|q_{n_0} g_{n_0} \bar{q}_{\bar{n}}\rangle_1)$ or three-jet configuration $(|q_{n_1} g_{n'_1} \bar{q}_{\bar{n}}\rangle_1)$, depending on the collinearity of the external particles. When we go to SCET₂ we reduce the magnitude of collinearity, now particles with momenta p_1 and p_2 can be define collinear only if $p_1 \cdot p_2 \leq (\lambda^2)^2$, and the SCET₂ theory distinguishes between the states $|q_{n_0} g_{n_0} \bar{q}_{\bar{n}}\rangle_2$ and $|q_{n_1} g_{n'_1} \bar{q}_{\bar{n}}\rangle_2$. Because of the difference of collinearity between SCET₁ and SCET₂, a two-jet configuration in SCET₁ can be matched both onto $|q_{n_0} g_{n_0} \bar{q}_{\bar{n}}\rangle_2$ and $|q_{n_1} g_{n'_1} \bar{q}_{\bar{n}}\rangle_2$ in SCET₂. The matching

² As before we do not consider operators like $\mathcal{O}_2^{(1)}(n_0, \bar{n})$ that describe a gluon collinear to the antiquark.

is given by

$$J_{\text{QCD}}^{\mu} = C_{1,\text{LO}}^{(0)}(n_0)\mathcal{O}_1^{(0)}(n_0) + C_1^{(1)}(n_0, n_0)\mathcal{O}_1^{(1)}(n_0, n_0) + C_1^{(1)}(n_1, n_1')\mathcal{O}_1^{(1)}(n_1, n_1') + C_{1,\mathcal{T}}^{(1)}(n_0, n_0)\mathcal{T}_1^{(1)}(n_0, n_0) + \cdots$$
(B.41)
$$= C_2^{(0)}(n_0)\mathcal{O}_2^{(0)}(n_0) + C_2^{(1)}(n_0, n_0)\mathcal{O}_2^{(1)}(n_0, n_0) + C_2^{(1)}(n_1, n_1')\mathcal{O}_2^{(1)}(n_1, n_1') + C_{2,\mathcal{T}}^{(1)}(n_0, n_0)\mathcal{T}_2^{(1)}(n_0, n_0) + \cdots ,$$
(B.42)

where in the first equation we have written the original QCD current in terms of $SCET_1$ operators and in the second equation in terms of $SCET_2$ operators. The ellipses indicate higher order operators. If we close Eq. (B.42) with the state $|q_{n_0} q_{n_0} \bar{q}_{\bar{n}}\rangle_2$ we get

$$C_{1,LO}^{(0)}(n_{0}) \int dx^{4} \langle 0|T \{ \mathcal{L}_{SCET_{1}}(x) \mathcal{O}_{1}^{(0)}(n_{0}) \} |q_{n_{0}} g_{n_{0}} \bar{q}_{\bar{n}} \rangle_{2} + C_{1}^{(1)}(n_{0}, n_{0}) \langle 0| \mathcal{O}_{1}^{(1)}(n_{0}, n_{0}) |q_{n_{0}} g_{n_{0}} \bar{q}_{\bar{n}} \rangle_{2} + C_{1,\mathcal{T}}^{(1)}(n_{0}, n_{0}) \langle 0| \mathcal{T}_{1}^{(1)}(n_{0}, n_{0}) |q_{n_{0}} g_{n_{0}} \bar{q}_{\bar{n}} \rangle_{2} = C_{2}^{(0)}(n_{0}) \int dx^{4} \langle 0|T \{ \mathcal{L}_{SCET_{2}}(x) \mathcal{O}_{2}^{(0)}(n_{0}) \} |q_{n_{0}} g_{n_{0}} \bar{q}_{\bar{n}} \rangle_{2} + C_{2}^{(1)}(n_{0}, n_{0}) \langle 0| \mathcal{O}_{2}^{(1)}(n_{0}, n_{0}) |q_{n_{0}} g_{n_{0}} \bar{q}_{\bar{n}} \rangle_{2} + C_{2,\mathcal{T}}^{(1)}(n_{0}, n_{0}) \langle 0| \mathcal{T}_{2}^{(1)}(n_{0}, n_{0}) |q_{n_{0}} g_{n_{0}} \bar{q}_{\bar{n}} \rangle_{2} ,$$
(B.43)

Because the structure of the operators in Eq. (B.43) is the same in the LHS and RHS, we simply get

$$C_{2}^{(0)}(n_{0}) = C_{1, \text{LO}}^{(0)}(n_{0}),$$

$$C_{2}^{(1)}(n_{0}, n_{0}) = C_{1}^{(1)}(n_{0}, n_{0}),$$

$$C_{2, \mathcal{T}}^{(1)}(n_{0}, n_{0}) = C_{1, \mathcal{T}}^{(1)}(n_{0}, n_{0}).$$
(B.44)

If we close Eq. (B.42) with the state $|q_{n_1} g_{n'_1} \bar{q}_{\bar{n}}\rangle_2$ we have

$$C_{1,\text{LO}}^{(0)}(n_0) \int dx^4 \langle 0 | T \{ \mathcal{L}_{\text{SCET}_1}(x) \mathcal{O}_1^{(0)}(n_0) \} | q_{n_1} g_{n_1'} \bar{q}_{\bar{n}} \rangle_2 + C_1^{(1)}(n_0, n_0) \langle 0 | \mathcal{O}_1^{(1)}(n_0, n_0) | q_{n_1} g_{n_1'} \bar{q}_{\bar{n}} \rangle_2 \\ + C_1^{(1)}(n_1, n_1') \langle 0 | \mathcal{O}_1^{(1)}(n_1, n_1') | q_{n_1} g_{n_1'} \bar{q}_{\bar{n}} \rangle_2 + C_{1,\mathcal{T}}^{(1)}(n_0, n_0) \langle 0 | \mathcal{T}_1^{(1)}(n_0, n_0) | q_{n_1} g_{n_1'} \bar{q}_{\bar{n}} \rangle_2$$

$$= C_2^{(1)}(n_1, n_1') \langle 0 | \mathcal{O}_2^{(1)}(n_1, n_1') | q_{n_1} g_{n_1'} \bar{q}_{\bar{n}} \rangle_2.$$
(B.45)

We decompose $C_2^{(1)}(n_1, n_1')$ as

$$C_2^{(1)} = C_{2,\text{LO}}^{(1)} + C_{2,\text{NLO}}^{(1)H,a} + C_{2,\text{NLO}}^{(1)H,b} + C_{2,\text{NNLO}}^{(1)H}, \qquad (B.46)$$

where $C_{2,LO}^{(1)}$ is the coefficient that multiplied by $\langle 0|\mathcal{O}_2^{(1)}(n_1,n_1')|q_{n_1}g_{n_1'}\bar{q}_{\bar{n}}\rangle_2$ reproduces the second line in Eq. (B.45), $C_{2,NLO}^{(1)H,a}$ the third line and so on. All the SCET₂ coefficients in Eq. (B.44) scale as λ^0 like in SCET₁, but we will see that the SCET₂ coefficients in Eq. (B.46) scale with different powers of λ , thus what is meaningful is not the power counting of the operator by itself but the power counting of the Wilson coefficient×operator. We will prove that

$$C_{2}^{(0)}(n_{0})\mathcal{O}_{2}^{(0)}(n_{0}) \sim \lambda^{4}, \qquad C_{2,\text{LO}}^{(1)}(n_{1},n_{1}')\mathcal{O}_{2}^{(1)}(n_{1},n_{1}') \sim \lambda^{5}, \qquad (B.47)$$

$$C_{2}^{(1)}(n_{0},n_{0})\mathcal{O}_{2}^{(1)}(n_{0},n_{0}) \sim \lambda^{6}, \qquad C_{2,\text{NLO}}^{(1)H,a}(n_{1},n_{1}')\mathcal{O}_{2}^{(1)}(n_{1},n_{1}') \sim \lambda^{6}, \qquad C_{2,\text{NLO}}^{(1)H,b}(n_{1},n_{1}')\mathcal{O}_{2}^{(1)}(n_{1},n_{1}') \sim \lambda^{6}, \qquad C_{2,\text{NLO}}^{(1)H,b}(n_{1},n_{1}')\mathcal{O}_{2}^{(1)}(n_{1},n_{1}') \sim \lambda^{6}, \qquad C_{2,\text{NLO}}^{(1)H}(n_{1},n_{1}')\mathcal{O}_{2}^{(1)}(n_{1},n_{1}') \sim \lambda^{7}.$$

In the second column we have only one operator $\mathcal{O}_2^{(1)}(n_1, n_1')$ and we have decomposed its coefficient according to Eq. (B.46). It is important to notice that the matching constrains but does not conserve the power counting, this is because in SCET₁ the fields scale as λ , but in SCET₂ they scale as λ^2 , for example we have that the LO operator in SCET₁ is $C_1^{(0)}\mathcal{O}_1^{(0)} \sim \lambda^2$, but for the LO operator in SCET₂ we have $C_2^{(0)} \mathcal{O}_2^{(0)} \sim \lambda^5$.³

If we have to calculate cross section with a fixed number of external particles we have to take into account all the SCET₂ operators in Eq. (B.42). What we are interested in this work is to reproduce parton shower. To reproduce the LL parton shower we only need $C_{2,LO}^{(1)}(n_1, n'_1)\mathcal{O}_2^{(1)}(n_1, n'_1)$. To see why we can emit a second gluon, k_2 from it. Before doing it we have to run it down to the scale of the second

 $^{^{3}}$ A consequence of it is that for two emissions we will have NLO operator in SCET₁ that contributes only to NNLO operators in SCET₂

emission using the factor $U^{(1)}(k_{1\perp}, k_{2\perp})$. Because the first gluon comes form the operator $\mathcal{O}_2^{(1)}(n_1, n_1')$ whose coefficient was defined in the matching form SCET₁, and the second gluons comes form inserting the SCET₂ Lagrangian, we know that $k_1^2 \gg k_2^2$, as the parton showering condition requires. The operators $\mathcal{O}_2^{(0)}(n_0)$ and $\mathcal{O}_2^{(1)}(n_0, n_0)$ carry information that we do not need to describe parton shower. For example $\mathcal{O}_2^{(0)}(n_0)$ describes a quark which had not emitted until after the scale of matching $k_{1\perp}$. However, the RG kernels already give us the no-branching probability so, even if they have lower order in λ , we do not use them. For this reason we call $C_{2,\text{LO}}^{(1)}(n_1, n_1')\mathcal{O}_2^{(1)}(n_1, n_1')$ our LO operator in λ and we say that LO is at order λ^5 . The coefficients $C_{2,\text{NLO}}^{(1),H}(n_1, n_1')$ and $C_{2,\text{NNLO}}^{(1),H}(n_1, n_1')$ give corrections to the parton shower for one emission, and they describe a three-parton process. This means that if we want to find correction to a parton shower process that start from a matrix element with two partons, we have to take into account also parton shower starting from a matrix element of three partons. Therefore in the rest of the section we will focus only on the operator we need: $\mathcal{O}_2^{(1)}(n_1, n_1')$.

We now turn to calculate the Wilson coefficient (B.46). We do it in three steps: first we calculate the amplitudes in SCET₁ on the RHS of (B.45), second we rotate it using the finite RPI transformation define in Appendix B.1, so that they overlap with the SCET₂ states, and third we calculate the SCET₂ amplitudes in the LHS of (B.45) and calculate the Wilson coefficients necessary to make the two side of Eq. (B.45) equal. We do it order by order and we start calculating the coefficient $C_{2,LO}^{(1)}$. The SCET₁ amplitude in the first line of the LHS (B.45) is

 $U^{(0)}(n_0)$ is the running factor (see Section 4.4.3), and $\mu_1 \sim \lambda Q$ is at the scale of the emission. In (B.48) we have omitted the term proportional to \bar{n}^{α} because we work in the light-cone gauge where the Wilson lines are equal to the identity. The amplitude (B.48) is written in terms of objects projected in the *n* and \bar{n} directions. As discussed in Appendix B.1, these directions are not suitable for a SCET₂ states but we can use

the formulas (B.9) and write (B.48) in terms of the directions n_1 and n'_1 that are the directions where the quark and gluon have zero perpendicular component, this gives

In (B.49) we have rotated the spinor in the n_1 direction, $\gamma_{n_0\perp}$ in the n'_1 direction and we have written n_0 in terms of n_1 , n'_1 and $q_{n_0\perp}$. We have dropped all the terms proportional to \bar{n}^{α} and we made use of relations ## = 0 and $\bar{\xi}_{n_1}\#_1 = 0$. Because the momentum of the gluon is parallel to n'^{μ}_1 , only the polarizations in the perpendicular direction with respect to n'^{μ}_1 are physical, thus we can neglect the term proportional to n'^{μ}_1 in Eq. (B.49). The SCET₂ amplitude $\langle 0|\bar{\chi}_{n_1}g\mathcal{B}^{\alpha}_{n'_1\perp}\chi_{\bar{n}}|q_{n_1}g_{n'_1}\bar{q}_{\bar{n}}\rangle$ is

$$\langle 0|\bar{\chi}_{n_1}g\mathcal{B}^{\alpha}_{n'_1\perp}\chi_{\bar{n}}|q_{n_1}g^{\alpha}_{n'_1}\bar{q}_{\bar{n}}\rangle = g\bar{\xi}_{n_1}\epsilon^{\alpha}_{n'_1\perp}\xi_{\bar{n}}\,,\tag{B.50}$$

where in Eq. (B.50) we have explicitly written the polarization vector for the gluon. From Eq. (B.49) and Eq. (B.50), we can see that the LO Wilson coefficient is

$$C_{2,\text{LO}}^{(1)} = U^{(0)}(n_0)(Q, (k_1)_{n_0\perp})c_{\text{LO}}^{\alpha}(n_0)\frac{q_0}{q_0^2}\gamma_{n_0\perp}^{\mu}, \qquad (B.51)$$

where

 $\Theta_{\lambda^2}[n_1 \cdot n'_1]$ is a function that encodes the information that $(n_1 \cdot n'_1) \lesssim \lambda^2$, we will say more about it below. Because the matching come from a SCET₁ operator, $(q_1)_{n_0 \perp}$ can go up to order λ and q_0^2 up to λ^2 , thus $C_{2, \text{LO}}^{(1)}$ has power counting λ^{-1} . Using formulas (B.13), we can write (B.51) only in terms of n_1 and n'_1 , this gives

$$C_{2,\text{LO}}^{(1)} = U^{(0)}(n_0)(Q, \mu_1) \left(\frac{\bar{q}_1}{Q} \sqrt{|n_1 \cdot n_1'|} v_1^{\alpha} + \frac{\bar{k}_1}{2Q} \sqrt{|n_1 \cdot n_1'|} \psi_1 \gamma_{n_1'\perp}^{\alpha} \right) \frac{2\bar{q}_0}{(n_1 \cdot n_1') \bar{q}_1 \bar{k}_1} \times \left(\gamma_{n_1'}^{\mu} - \bar{n}^{\mu} \frac{1}{2} \frac{\bar{q}_1}{\bar{q}_0} \sqrt{|n_1 \cdot n_1'|} \psi_1 \right).$$
(B.53)
where v_1^{μ} is defined in Eq. (B.14), $\bar{q}_1 + \bar{k}_1 = Q$. For μ_1 we have to take a quantity that scale as λQ , we choose $\mu_1 = \sqrt{|n_1 \cdot n'_1|}Q$. Because $|v_1^2| = 2$, the power counting of (B.53) is only from the scalar product $n_1 \cdot n'_1$ that here is of order λ^2 . In a similar way we can calculate $C_{2,\text{NLO}}^{(1)H,a}$ and $C_{2,\text{NNLO}}^{(1)H}$.

We have done the matching starting from the QCD current $J^{\mu}_{\text{QCD}} = \bar{q} \gamma^{\mu} q$. If we had started from a general current, $\bar{q} \Gamma^{\mu} q$, the results (B.51) for $C^{(1)}_{2,\text{LO}}$ would have the same upon the substitution

$$\gamma^{\mu}_{n_0 \perp} \to \frac{\not{n} \not{n}_0}{4} \Gamma^{\mu} \,. \tag{B.54}$$

We can obtain $C_{2,LO}^{(1)}\mathcal{O}_2^{(2)}(n_1,n_1')$ from the SCET₁ operator $\bar{\chi}_{n_0}\Gamma^{\mu}\chi_{\bar{n}}$ running down from scale Q to scale μ_1 using the factor $U^{(0)}(Q, \mu_1)(n_0)$, and subsequently using the replacement rule

$$(\bar{\chi}_{n_0})_i \to (c_{\rm LO}^{\alpha})_{ji}(n_0)(\bar{\chi}_{n_1})_j g \mathcal{B}_{\alpha}^{n_1' \perp}$$
 (B.55)

On the other side we will see that the coefficients $C_{2,\text{NLO}}^{(1)H,a}$, $C_{2,\text{NLO}}^{(1)H,b}$, $C_{2,\text{NLO}}^{(1)H}$, are much different in the presence of a different QCD current. We labeled the coefficients whose structures depend on the hard scattering with the upper-script H, that means hardcorrection. Starting from two-gluon emissions, there are corrections whose structures are independent from the initial currents and we will call them jet correction.

We now calculate $C_{2,\text{NLO}}^{(1)H,a}$ and $C_{2,\text{NNLO}}^{(1)H}$. For the NLO and NNLO amplitudes in the second and third line of the LHS of Eq. (B.45) we have

$$\begin{aligned} A_{\rm NLO}^{q\bar{q}g} &= U^{(2,1,0)}(n_0,n_0)(Q,\,\mu_1) \frac{n_0^{\mu} - \bar{n}^{\mu}}{Q} g \bar{u}_{n_0} \gamma_{n_0\perp}^{\alpha} v_{\bar{n}} = \frac{n_0^{\mu} - \bar{n}^{\mu}}{Q} g \bar{u}_{n_1} \gamma_{n_1'\perp}^{\alpha} v_{\bar{n}_{\cdot}}, \\ A_{\rm NNLO}^{q\bar{q}g} &= U_{\mathcal{T}}^{(2,1,1)}(n_0,n_0)(Q,\,\mu_1) \\ &\times \left(\frac{1}{\bar{q}_1} + \frac{1}{\bar{k}_1}\right) \frac{1}{Q} g \bar{u}_{n_0} \gamma_{n_0\perp}^{\mu} (k_1)_{n_0\perp} \gamma_{n_0\perp}^{\alpha} v_{\bar{n}} - \frac{2}{\bar{q}_1 Q} g \bar{u}_{n_0} (k_1)_{n_0\perp}^{\mu} \gamma_{n_0\perp}^{\alpha} v_{\bar{n}} \\ &= U_{\mathcal{T}}^{(2,1,1)}(n_0,n_0)(Q,\,\mu_1) \\ &\times \left(\frac{1}{\bar{q}_1} + \frac{1}{\bar{k}_1}\right) \frac{1}{Q} g \bar{u}_{n_1} \gamma_{n_0\perp}^{\mu} (k_1)_{n_0\perp} \gamma_{n_1'\perp}^{\alpha} v_{\bar{n}} - \frac{2}{\bar{q}_1 Q} g \bar{u}_{n_1} (k_1)_{n_0\perp}^{\mu} \gamma_{n_1'\perp}^{\alpha} v_{\bar{n}}, \end{aligned} \tag{B.56}$$

where in the last equation we have performed the rotation to the directions n_1 and n'_1 . The SCET₂ coefficients needed to reproduce the amplitudes in Eq. (B.56) are

$$C_{2,\,\rm NLO}^{(1)H,a} = U^{(2,1,0)}(n_0, n_0)(Q, \,\mu_1)c_{2,\,\rm NLO}^{H,a}$$

$$C_{2,\,\rm NNLO}^{(1)H} = U_{\mathcal{T}}^{(2,1,1)}(n_0, n_0)(Q, \,\mu_1)c_{2,\,\rm NNLO}^{H}$$
(B.57)

where

$$c_{2,\text{NLO}}^{H,a} = U^{(2,1,0)}(n_0, n_0)(Q, \mu_1) \frac{n_0^{\mu} - \bar{n}^{\mu}}{Q} \gamma_{n_0 \perp}^{\alpha} \Theta_{\lambda^2}[n_1 \cdot n_1']$$

$$= U^{(2,1,0)}(n_0, n_0)(Q, \mu_1) \frac{1}{Q} \Big(\frac{\bar{k}_1 n_1'^{\mu} + \bar{q}_1 n_1^{\mu}}{\bar{q}_0} - \Big(1 + \frac{\bar{q}_1 \bar{k}_1}{2 \bar{q}_0^2} (n_1 \cdot n_1') \Big) \bar{n}^{\mu} \Big) \gamma_{n_1' \perp}^{\alpha} \Theta_{\lambda^2}[n_1 \cdot n_1'] ,$$

$$c_{2,\text{NNLO}}^{H} = U_{\mathcal{T}}^{(2,1,1)}(n_0, n_0)(Q, \mu_1) \Big(\Big(\frac{1}{\bar{q}_1} + \frac{1}{\bar{k}_1} \Big) \frac{1}{Q} \gamma_{n_0 \perp}^{\mu} (k_1)_{n_0 \perp} \gamma_{n_1' \perp}^{\alpha} - \frac{2}{\bar{q}_1 Q} (k_1)_{n_0 \perp}^{\mu} \gamma_{n_1' \perp}^{\alpha} \Big) \Theta_{\lambda^2}[n_1 \cdot n_1']$$

$$= -U_{\mathcal{T}}^{(2,1,1)}(n_0, n_0)(Q, \mu_1) \Big(\frac{1}{2Q} \Big(\gamma_{n_1' \perp}^{\mu} \sqrt{|n_1 \cdot n_1'|} \not_1 + \bar{n}^{\mu} \frac{\bar{q}_1}{Q} (n_1 \cdot n_1') \Big) \gamma_{n_1' \perp}^{\alpha}$$

$$+ \frac{\bar{k}_1}{Q^2} \Big(\sqrt{|n_1 \cdot n_1'|} v_1^{\mu} - \bar{n}^{\mu} (n_1 \cdot n_1') \frac{(\bar{k}_1^2 - \bar{q}_1^2)}{2Q^2} \Big) \gamma_{n_1' \perp}^{\alpha} \Big) \Theta_{\lambda^2}[n_1 \cdot n_1'] .$$
(B.58)

 $C_{2,\,\rm NLO}^{(1)H,a}$ has power counting λ^0 and $C_{2,\,\rm NNLO}^{(1)H} \sim \lambda^1$.

For the coefficient $C_{2,\text{NLO}}^{(1)H,b}$, the matching comes from the SCET₁ three-jet operators where $n_1 \cdot n'_1$ is now of order λ^0 . Because $C_1^{(1)}(n_1, n'_1)$ defined in Eq. (B.39) is already labeled in terms of the directions n_1 , n'_1 and \bar{n} exactly parallel to the external particles, we can simply write

$$C_{2,\text{NLO}}^{(1)H,b}(n_1,n_1') = C_1^{(1)}(n_1,n_1')\tilde{\Theta}_{\lambda^2}[n_1 \cdot n_1'], \qquad (B.59)$$

where $\tilde{\Theta}_{\lambda^2}[n_1 \cdot n'_1]$ carries the information that $(n_1 \cdot n'_1) > \lambda$. Knowing that now $n_1 \cdot n'_1 \sim \lambda^0$, $C_{2,\text{NLO}}^{(1)H,b}(n_1,n'_1)$ scale as λ^0 and

$$C_{2,\text{NLO}}^{(1)H,b}(n_1,n_1')\mathcal{O}_2^{(1)}(n_1,n_1') \sim \lambda^6$$
 (B.60)

The SCET₂ coefficients (B.53) and (B.57) comes from matching SCET₁ to SCET₂ for the two-jet SCET₁ phase space region where $q_1 \cdot k_1 \lesssim \lambda^2 Q^2$ that implies $n_1 \cdot n'_1 \lesssim \lambda^2$, instead the coefficient $C_{2,\text{NLO}}^{(1)H,b}$ comes for the three-jet SCET₁ region where $n_1 \cdot n'_1 > \lambda^2$. Because the operator $\mathcal{O}_2^{(1)}(n_1,n'_1)$ only knows that $n_1 \cdot n'_1 > \lambda^4$, it is not able to distinguish the two regions, and this information has to be encoded in the Wilson coefficients, and we have done it inserting the function Θ and $\tilde{\Theta}$. We can think of $\tilde{\Theta}_{\lambda^2}[x]$ as usual theta function: $\Theta_{\lambda^2}[x] = \theta[\lambda^2 - x]$ and $\tilde{\Theta}_{\lambda^2}[x] = 1 - \Theta_{\lambda^2}[x]$, but if we have to calculate inclusive quantities we need to integrate the phase space and having hard theta functions in the integrand could be problematic. We can define a smoother theta function, an example is in Fig. 4-9

$$\Theta_{\Lambda,a}(x) = \begin{cases} 0 & \text{if } x < \Lambda - a \\ -\operatorname{Sign}(x - \Lambda)e^{\frac{2a\operatorname{Sign}(x - \Lambda)}{(x - \Lambda) - a\operatorname{Sign}(x - \Lambda)}} + (\operatorname{Sign}(x - \Lambda) + 1)e^{-2} & \text{if } \Lambda - a < x < \Lambda + a \\ 1 & \text{if } x > \Lambda + a \\ (B.61) \end{cases}$$

and $\tilde{\Theta}_{\Lambda,a}(x) = 1 - \Theta_{\Lambda,a}(x)$. The parameter Λ tells where the function switches from 0 to 1 and the parameter *a* how fast it does it. To see how this theta function works, we integrate the amplitude squared up to NLO. The LO amplitude squared is

$$|A^{q\bar{q}g}|_{\rm LO}^2 = |\hat{C}_{2,\rm LO}^{(1)}(n_1, n_1')|^2 G(q_1, k_1, k_2, p_{\bar{q}}), \qquad (B.62)$$

where

$$G(q_1, k_1, k_2, p_{\bar{q}}) = {}_2 \langle q_{n_1} g_{n'_1} \bar{q}_{\bar{n}} | \mathcal{O}_1^{(1)\dagger}(n_1, n'_1) | 0 \rangle \langle 0 | \mathcal{O}_1^{(1)}(n_1, n'_1) | q_{n_1} g_{n'_1} \bar{q}_{\bar{n}} \rangle_2.$$
(B.63)

The NLO and NNLO amplitude squared is

$$|A^{q\bar{q}g}|_{\rm NLO}^2 + |A^{q\bar{q}g}|_{\rm NNLO}^2 = |A^{q\bar{q}g}|_{\rm to\,NNLO,\,2-jet}^2 + |A^{q\bar{q}g}|_{\rm NNLO,\,3-jet}^2, \qquad (B.64)$$

where

$$|A^{q\bar{q}g}|_{\text{to NNLO}, 2-\text{jet}}^{2} = (C_{2,\text{LO}}^{(1)\dagger}(n_{1}, n_{1}')(C_{2,\text{NNLO}}^{(1),H}(n_{1}, n_{1}') + C_{2,\text{NLO}}^{(1)H,a\dagger}(n_{1}, n_{1}')) + (C_{2,\text{NLO}}^{(1)H,a\dagger}(n_{1}, n_{1}') + C_{2,\text{NNLO}}^{(1),H\dagger}(n_{1}, n_{1}'))C_{2,\text{NLO}}^{(1)}(n_{1}, n_{1}')$$



Figure B-3: Merging of the two-jet and three-jet amplitudes squared for $q\bar{q}g$ process using a smooth theta function. Plots of the two-jet amplitude square, $|A^{q\bar{q}g}|_{\rm LO}^2 + |A^{q\bar{q}g}|_{\rm NLO, 2-jet}^2$, (green), three-jet, $|A^{q\bar{q}g}|_{\rm NLO, 3-jet}^2$, (blue) and sum (red) versus $k_{1\perp n_0}$. The amplitudes are evaluated two-jet and three-jeted without running coefficients for $\bar{k}_1/\bar{q}_0 = 0.4$.

$$+ |\hat{C}_{2,\text{NLO}}^{(1),Ha}(n_1,n_1')|^2) G(q_1,k_1,k_2,p_{\bar{q}}), |A^{q\bar{q}g}|_{\text{NNLO},3-\text{jet}}^2 = |C_{2,\text{NLO}}^{(1),Hb}(n_1,n_1')|^2 G(q_1,k_1,k_2,p_{\bar{q}}).$$
(B.65)

 $|A^{q\bar{q}g}|^2_{\mathrm{NLO}, 2-\mathrm{jet}}$ is the contribution to the amplitude squared from the two-jet region and $|A^{q\bar{q}g}|^2_{\mathrm{NLO}, 3-\mathrm{jet}}$ is the contribution from the three-jet region. In Fig. 4-8 we plot the ratios $|A^{q\bar{q}g}|^2_{\mathrm{LO}}/|A^{q\bar{q}g}|^2_{\mathrm{QCD}}$ and $(|A^{q\bar{q}g}|^2_{\mathrm{LO}} + |A^{q\bar{q}g}}|^2_{\mathrm{NLO}, 2-\mathrm{jet}})/|A^{q\bar{q}g}}|^2_{\mathrm{QCD}}$ versus $k_{1\perp_{n_0}}$, We notice that including NLO corrections extends the region where tree-level SCET and QCD agree. In Fig. B-3 we plot the the merging of the two-jet and and threejet amplitude squared using the theta function. The theta function defined above smoothly merges the two amplitude squares. In Fig. B-4 we plot $|A^{q\bar{q}g}|^2_{\mathrm{LO}} + |A^{q\bar{q}g}|^2_{\mathrm{NLO}}$ with and without running factors.

B.3.2 Two-Gluon Emissions

We now match $SCET_1$ to $SCET_2$ for two-gluon emissions up to NNLO. In order to do it we first should calculate the matching QCD to $SCET_1$ for two emissions up to NNLO. Besides the operators in (B.26), for two emissions in $SCET_1$ there is the



Figure B-4: Plot of the SCET₂ amplitudes square up to NLO, $|A^{q\bar{q}g}|_{LO}^2 + |A^{q\bar{q}g}|_{NLO}^2$, with (green) and without (red) running factors versus $k_{1\perp_{n_0}}$ for $\bar{k}_1/\bar{q}_0 = 0.4$.

additional NNLO operator

$$\mathcal{O}_{1}^{(2)}(n_{0}, n_{0}, n_{0}) = \bar{\chi}_{n_{0}} g \mathcal{B}_{n_{0} \perp}^{\alpha} g \mathcal{B}_{n_{0} \perp}^{\beta} \chi_{\bar{n}} , \qquad (B.66)$$

but this operators in not relevant for the matching SCET₁ to SCET₂ at NNLO. (B.66) matches to SCET₂ operators with two gluon field, like for example $\mathcal{O}_2^{(2)}(n_1, n'_1, n'_1) = \bar{\chi}_{n_1} \mathcal{B}_{n'_1 \perp}^{\alpha} \mathcal{K}_{\bar{n}}^{\beta}$, with Wilson coefficient $C_2^{(2)}(n_1, n_1, n'_1)$ of order λ^0 , that is (B.66) matches to a Wilson coefficient×operators in SCET₂ of order λ^8 that is N³LO according to the power counting in (B.47). Thus the SCET₁ operators (B.26) are enough for the matching to SCET₂ up to two emissions.

Besides the operators necessary for the matching for one emission, the $SCET_2$ basis for two emissions has the additional two gluons operators

$$\mathcal{O}_{2}^{(2)}(n_{2}, n_{2}, n_{1}') = \bar{\chi}_{n_{2}} g \mathcal{B}_{n_{2}\perp}^{\alpha} g \mathcal{B}_{n_{1}'\perp}^{\beta} \chi_{\bar{n}}, \qquad (B.67)$$

$$\mathcal{O}_{2}^{(2)}(n_{2}, n_{1}', n_{1}') = \bar{\chi}_{n_{2}} g \mathcal{B}_{n_{1}'\perp}^{\alpha} g \mathcal{B}_{n_{1}'\perp}^{\beta} \chi_{\bar{n}}, \qquad (B.67)$$

$$\mathcal{O}_{2}^{(2)}(n_{2}, n_{1}', n_{2}') = \bar{\chi}_{n_{2}} g \mathcal{B}_{n_{1}'\perp}^{\alpha} g \mathcal{B}_{n_{2}'\perp}^{\beta} \chi_{\bar{n}}. \qquad (B.67)$$

In the first operator in (B.67) the quark and the gluon are collinear and it does



Figure B-5: Matching SCET₁ to SCET₂ to SCET₃ for two emissions for the two-jet configuration. In the first column there are the QCD Feyman diagrams; in the second column the SCET₁ diagrams from the operator $\mathcal{O}_1^{(0)}(n_0)$; in the third column the SCET₁ diagrams from the operator $\mathcal{O}_1^{(1)}(n_0, n_0)$ and $\mathcal{T}_1^{(1)}(n_0, n_0)$; in the fourth column the SCET₁ diagram from the operator $\mathcal{O}_1^{(2)}(n_0, n_0, n_0)$, this operator contributes only at N³LO to the SCET₂ matching; in the fifth column the SCET₂ diagram from the operator $\mathcal{O}_1^{(1)}(n_2, n_1')$; In the sixth column the SCET₂ diagrams from the operators $\mathcal{O}_2^{(2)}(n_2, n_1', n_2')$, $\mathcal{O}_2^{(2)}(n_2, n_2, n_1')$ and $\mathcal{O}_2^{(2)}(n_2, n_1', n_1')$; in the seventh column the SCET₃ diagram from the operator $\mathcal{O}_3^{(2)}(n_1, n_1', n_2')$.

annihilate the state $|q_{n_2}g_{n_2}g_{n'_1}\bar{q}_{\bar{n}}\rangle_2$, in the second the two gluons are collinear and it does annihilate the state $|q_{n_2}g_{n'_1}g_{n'_1}\bar{q}_{\bar{n}}\rangle_2$, and in the third all the particles are far apart and it does annihilate the state $|q_{n_2}g_{n'_1}g_{n'_2}\bar{q}_{\bar{n}}\rangle_2$. The last SCET₂ operator in (B.67) is not necessary for the matching at NNLO. It can only be closed with the state $|q_{n_0}g_{n_0}g_{n_0}\bar{q}_{\bar{n}}\rangle_2$ that describes two emissions both collinear in SCET₂, his coefficient can only come form the SCET₁ operator $\mathcal{O}_1^{(2)}(n_0, n_0, n_0)$ and so it only contributes at N³LO. The Wilson coefficients of the operators (B.67) are defined such that

$$J_{\text{QCD}}^{\mu} = C_{1,\text{LO}}^{(0)}(n_0)\mathcal{O}_1^{(0)}(n_0) + C_1^{(1)}(n_0, n_0)\mathcal{O}_1^{(1)}(n_0, n_0) + C_{1,\mathcal{T}}^{(1)}\mathcal{T}_1^{(1)}(n_0, n_0) \qquad (B.68)$$

+ $C_1^{(1)}(n_1, n_1')\mathcal{O}_1^{(1)}(n_1, n_1') + \cdots$
= $C_2^{(0)}(n_0)\mathcal{O}_2^{(0)}(n_0) + C_1^{(1)}(n_0, n_0)\mathcal{O}_2^{(2)}(n_0, n_0) + C_{2,\mathcal{T}}^{(2)}\mathcal{T}_2^{(2)}(n_0, n_0) \qquad (B.69)$
+ $C_2^{(1)}(n_1, n_1')\mathcal{O}_2^{(1)}(n_1, n_1') + C_2^{(2)}(n_2, n_2, n_1')\mathcal{O}_2^{(2)}(n_2, n_2, n_1')$
+ $C_2^{(2)}(n_2, n_1', n_1')\mathcal{O}_2^{(2)}(n_2, n_1', n_1') + C_2^{(2)}(n_2, n_1', n_2')\mathcal{O}_2^{(2)}(n_2, n_1', n_2') + \cdots$

where in the first equation we have written the original QCD current in terms of $SCET_1$ operators and in the second equation in terms of $SCET_2$ operators. The ellipses indicate higher order operators.

We divide the Wilson coefficients in two category: jet and hard. We labeled the jet coefficients with the upper script J, and the hard with H. As already mention in the previous section, the jet coefficients are those whose structures do not depend on the QCD current. That is, if we had starting the matching from a general QCD current, $\bar{q} \Gamma^{\mu} q$ instead of $J^{\mu}_{\text{QCD}} = \bar{q} \gamma^{\mu} q$, the jet coefficient would be the same upon the substitution in Eq. (B.54). The corrections that come from these coefficients are universal in the sense that they are independent of the information about the hard scattering that happens at the beginning of the shower. Instead the hard coefficients are those who depend on the QCD current, that it they give corrections that depend on the hard scattering. All the jet coefficients come from the matching with the SCET₁ operators $\mathcal{O}_1^{(0)}(n_0)$ because it can emits two gluons only through the Largangian insertion and the Lagrangian does not dependent on the initial QCD current.

We have seen in the previous section that the LO coefficient×operator that we need for the parton shower is $C_2^{(1)}(n_1)\mathcal{O}_2^{(1)}(n_1,n_1')$, that is of order λ^5 , Eq. (B.47). We are interested to calculate the amplitude squared at NNLO, that is we only need to calculate the NNLO amplitudes that interference with the LO amplitude. The LO amplitude for two emissions is

$$A_{\rm LO}^{qgg\bar{q}} = C_{2,\rm LO}^{(1)}(n_1, n_1') \int dx \langle 0|T\{\mathcal{L}_{\rm SCET_2}(x)\mathcal{O}_2^{(1)}(n_1, n_1')\}|q_{n_2}g_{n_2}g_{n_1'}\bar{q}_{\bar{n}}\rangle_2.$$
(B.70)

Because $A_{\rm LO}^{qgg\bar{q}}$ comes from closing the operator with the state $|q_{n_2}g_{n_2}g_{n_1}\bar{q}_{\bar{n}}\rangle_2$, it can only interferences with the amplitude that comes from $\mathcal{O}_2^{(2)}(n_2, n_2, n_1')$. Thus we will only calculate the component of the coefficients $C_2^{(2)}(n_2, n_1', n_2')$, and $C_2^{(2)}(n_2, n_1', n_1')$ that give NLO contribution, that is such that $C_2^{(2)}\mathcal{O}_2^{(2)}(n_2, n_1', n_2')$ and $C_2^{(2)}\mathcal{O}_2^{(2)}(n_2, n_1', n_2')$ is of order λ^6 . Knowing that $\mathcal{O}_2^{(2)} \sim \lambda^8$, we have to calculate $C_2^{(2)}(n_2, n_1', n_2')$, and $C_2^{(2)}(n_2, n_1', n_1')$ only up to λ^{-2} . For $\mathcal{O}_2^{(2)}(n_2, n_2, n_1')$ instead we have also to calculate his coefficient up to λ^{-1} . We will use the lower script NLO/NNLO/N³LO if the coefficient scales as $\lambda^{-2}/\lambda^{-1}/\lambda^0$.

We now turn to calculate the coefficient of (B.67), it can be defined by closing Eq. (B.69) with the state $|q_{n_2}g_{n'_1}\bar{q}_{n'_2}\bar{q}_{\bar{n}}\rangle_2$. We decompose $C_2^{(2)}(n_2, n'_1, n'_2)$ as

$$C_2^{(2)}(n_2, n_1', n_2') = C_{2, \text{NLO}}^{(2)J}(n_2, n_1', n_2') + C_2^{(2)H}(n_2, n_1', n_2'), \qquad (B.71)$$

where

$$C_{2,\text{NLO}}^{(2)J}\langle 0|\mathcal{O}_{2}^{(2)}(n_{2},n_{1}',n_{2}')|q_{n_{2}}g_{n_{1}'}g_{n_{2}'}\bar{q}_{\bar{n}}\rangle_{2} =$$

$$C_{1,\text{LO}}^{(0)}(n_{0})\int dx_{1}dx_{2}\langle 0|T\{\mathcal{L}_{\text{SCET}_{1}}(x_{1})\mathcal{L}_{\text{SCET}_{1}}(x_{2})\mathcal{O}_{1}^{(0)}(n_{0},n_{0})\}|q_{n_{2}}g_{n_{1}'}g_{n_{2}'}\bar{q}_{\bar{n}}\rangle_{2}$$
(B.72)

and

$$C_{2}^{(2)H}(n_{2},n_{1}',n_{2}')\langle 0|\mathcal{O}_{2}^{(2)}(n_{2},n_{1}',n_{2}')|q_{n_{2}}g_{n_{1}'}g_{n_{2}'}\bar{q}_{\bar{n}}\rangle_{2} =$$
(B.73)
+ $C_{1}^{(1)}(n_{0},n_{0})\int dx \langle 0|T\{\mathcal{L}_{\text{SCET}_{1}}(x)\mathcal{O}_{1}^{(1)}(n_{0},n_{0})\}|q_{n_{2}}g_{n_{1}'}g_{n_{2}'}\bar{q}_{\bar{n}}\rangle_{2}$
+ $C_{1,\mathcal{T}}^{(1)}(n_{0},n_{0})\int dx \langle 0|T\{\mathcal{L}_{\text{SCET}_{1}}(x)\mathcal{T}_{1}^{(1)}(n_{0},n_{0})\}|q_{n_{2}}g_{n_{1}'}g_{n_{2}'}\bar{q}_{\bar{n}}\rangle_{2}$
+ $C_{1}^{(1)}(n_{1},n_{1}')\int dx \langle 0|T\{\mathcal{L}_{\text{SCET}_{1}}(x)\mathcal{O}_{1}^{(1)}(n_{1},n_{1}')\}|q_{n_{2}}g_{n_{1}'}g_{n_{2}'}\bar{q}_{\bar{n}}\rangle_{2}.$

We decompose $C_2^{(2)H}(n_2, n'_1, n'_2)$ as

$$C_{2}^{(2)H}(n_{2}, n_{1}', n_{2}') = C_{2,\text{NNLO}}^{(2)H,a}(n_{2}, n_{1}', n_{2}') + C_{2,\text{N}^{3}\text{LO}}^{(2)H}(n_{2}, n_{1}', n_{2}') + C_{2,\text{NNLO}}^{(2)H,b}(n_{2}, n_{1}', n_{2}'),$$
(B.74)

where $C_{2,\text{NNLO}}^{(2)H,a}(n_2,n'_1,n'_2)$ is the coefficient that multiplied by $\langle 0|\mathcal{O}_2^{(2)}(n_2,n'_1,n'_2)|q_{n_2}g_{n'_1}g_{n'_2}\bar{q}_{\bar{n}}\rangle_2$ reproduces the the second line in the Eq. (B.73), $C_{2,\text{N}^3\text{LO}}^{(2)H}(n_2,n'_1,n'_2)$ the third line and so on. In Eqs. (B.71-B.74) we put the labels in the coefficients to indicate their power counting. Because the operator $\mathcal{O}_2^{(2)}(n_2,n'_1,n'_2)$ does not interference with the LO operator we only need the coefficient $C_{2,\text{NLO}}^{(2)J}(n_2,n'_1,n'_2)$ to calculate the amplitude squared at NLO. We calculate $C_{2,\text{NLO}}^{(2)J}(n_2,n'_1,n'_2)$ and also $C_{2,\text{NNLO}}^{(2)H,a}(n_2,n'_1,n'_2)$ because it will be useful later, and we prove that are of order λ^{-2} and λ^{-1} respectively. We



Figure B-6: Feynman diagrams for two emissions in SCET₁ from the operator $\mathcal{O}_1^{(0)}$.

will not calculate here the other coefficients, however it is easy to do it just repeating the method we will show below and one can prove that they have the power counting labelled in Eq. (B.74). $C_{2,\text{NLO}}^{(2)J}(n_2, n'_1, n'_2)$ and $C_{2,\text{NNLO}}^{(2)H,a}(n_2, n'_1, n'_2)$ come from two-jet operators in SCET₁, thus they are defined in a phase space region where $n_2 \cdot n'_1, n_2 \cdot n'_2, n'_1 \cdot n'_2 \leq \lambda^2$. As we have done for the coefficients calculated in the one gluon matching, we have to encode this information in the Wilson coefficients because the SCET₂ operator $\mathcal{O}_2^{(2)}(n_2, n'_1, n'_2)$ only knows that $n_2 \cdot n'_1, n_2 \cdot n'_2, n'_1 \cdot n'_2 > \lambda^4$. We will do it inserting Θ functions in the coefficients like for the one gluon matching.

To calculate the coefficients we proceed like we have done for one-gluon emission: on the LHS of Eqs. (B.72) and (B.73) we calculate the SCET₁ amplitude, we rewrite it along the direction n_2 , n'_1 , n'_2 where the quark and the two gluons are aligned using the finite RPI defined in Appendix (B.1); on the RHS we write the SCET₂ amplitude and calculate the Wilson coefficient necessary for the matching. We decompose the SCET₁ amplitude

$$A_{\rm NLO}^{q\bar{q}gg} = C_1^{(0)} \int dx_1 dx_2 \langle 0 | T \{ \mathcal{L}_{\rm SCET_1}(x_1) \mathcal{L}_{\rm SCET_1}(x_2) \mathcal{O}_1^{(0)} \} | q_{n_2} g_{n_1'} g_{n_2'} \bar{q}_{\bar{n}} \rangle_2 \,, \qquad (B.75)$$

in

$$A_{\text{NLO}}^{q\bar{q}gg} = A_{\text{NLO},A}^{q\bar{q}gg} + A_{\text{NLO},B}^{q\bar{q}gg} + A_{\text{NLO},C}^{q\bar{q}gg} , \qquad (B.76)$$

where A, B, C correspond to the three graphs in Fig. B-6. Using the SCET Feynman

rules we have

$$\begin{aligned} A_{\text{NLO},A}^{q\bar{q}gg} &= U^{(0)}(n_{0})(Q,(k_{1})_{n_{0}\perp}) g^{2} \,\bar{u}_{n_{0}} \left[n_{0}^{\beta} + \gamma_{n_{0}\perp}^{\beta} \frac{(\not{q}_{1})_{n_{0}\perp}}{\bar{q}_{1}} + \frac{(\not{q}_{2})_{\bar{n}_{0}\perp}}{\bar{q}_{2}} \gamma_{n_{0}\perp}^{\beta} \right] \\ &\times \left[n_{0}^{\alpha} + \frac{(\not{q}_{1})_{n_{0}\perp}}{\bar{q}_{1}} \gamma_{n_{0}\perp}^{\alpha} \right] \frac{\bar{q}_{1}}{q_{1}^{2}} \frac{\bar{q}_{0}}{q_{0}^{2}} \Gamma^{\mu} v_{\bar{n}} \,, \\ A_{\text{NLO},B}^{q\bar{q}gg} &= U^{(0)}(n_{0})(Q,(k_{1})_{n_{0}\perp}) g^{2} \,\bar{u}_{n_{0}} \left[n_{0}^{\alpha} + \gamma_{n_{0}\perp}^{\alpha} \frac{(\not{q}_{2} + \not{k}_{1})_{n_{0}\perp}}{\bar{q}_{2} + \bar{k}_{1}} + \frac{(\not{q}_{2})_{n_{0}\perp}}{\bar{q}_{2}} \gamma_{n_{0}\perp}^{\alpha} \right] \frac{\bar{q}_{2} + \bar{k}_{1}}{(q_{2} + k_{1})^{2}} \\ &\times \left[n_{0}^{\beta} + \frac{(\not{q}_{2} + \not{k}_{1})_{n_{0}\perp}}{\bar{q}_{2} + \bar{k}_{1}} \gamma_{n_{0}\perp}^{\beta} \right] \frac{\bar{q}_{0}}{q_{0}^{2}} \Gamma^{\mu} v_{\bar{n}} \,, \\ A_{\text{NLO},C}^{q\bar{q}gg} &= U^{(0)}(n_{0})(Q,(k_{1})_{n_{0}\perp}) g^{2} \,\bar{u}_{n_{0}} \left[\frac{1}{\bar{q}_{0} - \bar{k}_{2}} \gamma_{n_{0}\perp}^{\alpha} \gamma_{n_{0}\perp}^{\beta} + \frac{1}{\bar{k}_{2} + \bar{q}_{2}} \gamma_{n_{0}\perp}^{\beta} \gamma_{n_{0}\perp}^{\alpha} \right] \frac{\bar{q}_{0}}{q_{0}^{2}} \Gamma^{\mu} v_{\bar{n}} \,, \\ (B.77) \end{aligned}$$

where $q_1 = q_2 + k_2$ and $q_0 = q_1 + q_2 + k_2$. As before we do not write terms with \bar{n}^{α} and \bar{n}^{β} , they are not necessary for the matching because the operator $\bar{n} \cdot A_n$ is constrained by gauge invariant to be only in the Wilson line. Now we rotate the amplitude (B.77) to the directions n_2 and n'_1 and n'_2 parallel to the quark and the two gluons, as described in Eq. (B.16)

 $n_1 \cdot n'_1$ is defined in terms of n_2 , n'_1 and n'_2 in Eqs. (B.20),

 v_1 and v_2 are defined in Eqs.(B.14) and (B.19), and the values of q_0^2 , q_1^2 and

 $(q_1 + k_2)^2$ are given in Eqs.(B.22) and (B.23). As for the one-gluon emission, we can neglect the term with $n_1^{\prime \alpha}$ and $n_2^{\prime \beta}$ because not physical. The SCET₂ amplitude for $\langle 0|\mathcal{O}_2^{(2)}(n_2, n_1^{\prime}, n_2^{\prime})|q_{n_2}g_{n_1^{\prime}}g_{n_2^{\prime}}\bar{q}_{\bar{n}} \rangle$ is

$$\langle 0 | \bar{\chi}_{n_2} g \mathcal{B}^{\alpha}_{n'_2 \perp} g \mathcal{B}^{\beta}_{n'_1 \perp} \chi_{\bar{n}} | q_{n_2} g_{n'_1} g_{n'_2} \bar{q}_{\bar{n}} \rangle = g^2 \bar{u}_{n_2} \epsilon^{\alpha}_{n'_1 \perp} \epsilon^{\beta}_{n'_2 \perp} v_{\bar{n}} \,. \tag{B.79}$$

In Eq. (B.79) we have expicitly written the polarizations vectors of the external gluons in the state. Thus the Wilson coefficients $C_{2,\text{NLO}}^{(2)J}(n_2, n'_1, n'_2)$ is

$$C_{2,\text{NLO}}^{(2)J}(n_2, n_1', n_2') = U^{(0)}(n_0)(Q, \mu_1) d_1^{J\,\alpha\beta}(n_2, n_1', n_2') \Theta_{\lambda^2}[n_2 \cdot n_1'] \Theta_{\lambda^2}[n_2 \cdot n_2'] \Theta_{\lambda^2}[n_2' \cdot n_1'],$$
(B.80)

where

$$d_1^{J\,\alpha\beta}(n_2,n_1',n_2') = d_{1,A}^{J\,\alpha\beta}(n_2,n_1',n_2') + d_{1,B}^{J\,\alpha\beta}(n_2,n_1',n_2') + d_{1,C}^{J\,\alpha\beta}(n_2,n_1',n_2'), \quad (B.81)$$

with

$$d_{1,A}^{J\,\alpha\beta}(n_2,n_1',n_2') = \left[\frac{\bar{q}_2}{\bar{q}_2 + \bar{k}_2}\sqrt{|n_2 \cdot n_2'|} v_2^\beta + \frac{\bar{k}_2}{\bar{q}_2 + \bar{k}_2}\sqrt{|n_2 \cdot n_2'|} \frac{\psi_2}{2}\gamma_{n_2'\perp}^\beta\right] \\ \times \left[\frac{\bar{q}_1}{\bar{q}_1 + \bar{k}_1}\sqrt{|n_1 \cdot n_1'|} v_1^\alpha + \frac{\bar{k}_1}{\bar{q}_1 + \bar{k}_1}\sqrt{|n_1 \cdot n_1'|} \frac{\psi_1}{2}\gamma_{n_1'\perp}^\alpha\right] \frac{\bar{q}_1}{q_1^2} \frac{\bar{q}_0}{q_0^2}\gamma_{n_0\perp}^\mu,$$
(B.82)

$$\begin{split} d_{1,B}^{J\,\alpha\beta}(n_{2},n_{1}',n_{2}') &= \left[\frac{\bar{q}_{1}}{\bar{q}_{1}+\bar{k}_{1}}\sqrt{|n_{1}\cdot n_{1}'|} v_{1}^{\alpha} + \frac{\bar{k}_{2}}{\bar{q}_{2}+\bar{k}_{2}}\sqrt{|n_{2}\cdot n_{2}'|} \frac{\not{p}_{2}}{2}\gamma_{n_{1}'\perp}^{\alpha} \right. \\ &+ \gamma_{n_{1}'\perp}^{\alpha} \frac{\bar{k}_{2}\,\bar{q}_{2}}{(\bar{q}_{2}+\bar{k}_{1})(\bar{q}_{2}+\bar{k}_{2})}\sqrt{|n_{2}\cdot n_{2}'|} \frac{\not{p}_{2}}{2} - \gamma_{n_{1}'\perp}^{\alpha} \frac{\bar{k}_{1}\,\bar{q}_{1}}{(\bar{q}_{2}+\bar{k}_{1})(\bar{q}_{1}+\bar{k}_{1})}\sqrt{|n_{1}\cdot n_{1}'|} \frac{\not{p}_{1}}{2} \right] \\ &\times \left[\frac{\bar{q}_{2}}{\bar{q}_{2}+\bar{k}_{2}}\sqrt{|n_{2}\cdot n_{2}'|} v_{2}^{\beta} - \frac{\bar{k}_{1}}{\bar{q}_{1}+\bar{k}_{1}}\sqrt{|n_{1}\cdot n_{1}'|} v_{1}^{\beta} + \frac{\bar{k}_{1}}{\bar{q}_{1}+\bar{k}_{1}}\sqrt{|n_{1}\cdot n_{1}'|} \frac{\not{p}_{1}}{2}\gamma_{n_{2}'\perp}^{\beta} \right] \\ &\times \frac{\bar{q}_{2}+\bar{k}_{1}}{(q_{2}+k_{1})^{2}} \frac{\bar{q}_{0}}{q_{0}^{2}}\gamma_{n_{0}\perp}^{\mu}, \\ d_{1,C}^{J\,\alpha\beta}(n_{2},n_{1}',n_{2}') &= \left[\frac{1}{\bar{q}_{0}-\bar{k}_{2}}\gamma_{n_{1}'\perp}^{\alpha}\gamma_{n_{2}'\perp}^{\beta} + \frac{1}{\bar{k}_{2}+\bar{q}_{2}}\gamma_{n_{2}'\perp}^{\beta}\gamma_{n_{1}'\perp}^{\alpha}\right] \frac{\bar{q}_{0}}{q_{0}^{2}}\gamma_{n_{0}\perp}^{\mu}. \end{split}$$

The Θ functions in Eq. (B.80) are necessary to encode the information that $C_{2,\text{NLO}}^{(2)J}(n_2, n'_1, n'_2)$



Figure B-7: Feynman diagrams for two emissions in SCET₁ from the operator $\mathcal{O}_1^{(1)}$.

comes from the two-jet SCET₁ operators. To calculate the power counting of $C_{2,\text{NLO}}^{(2)J}(n_2, n'_1, n'_2)$, we have to consider that this coefficient comes from matching SCET₁ to SCET₂ in the region where $n_2 \cdot n'_1 \sim n_2 \cdot n'_2 \sim n'_1 \cdot n'_2 \sim \lambda^2$, thus we have

$$C_{2,\text{NLO}}^{(2)J}(n_2, n_1', n_2') \sim \lambda^{-2}$$
. (B.83)

We proceed similarly to calculate the coefficient $C_{2,\text{NNLO}}^{(2),H,a}(n_2,n'_1,n'_2)$ and prove that it has power counting λ^{-1} . We decompose the SCET₁ amplitude

$$A_{\rm NNLO}^{q\bar{q}gg} = C_1^{(1)}(n_0, n_0) \int dx \langle 0|T\{\mathcal{L}_{\rm SCET_1}(x)\mathcal{O}_1^{(1)}(n_0, n_0)\}|q_{n_2}g_{n_1'}g_{n_2'}\bar{q}_{\bar{n}}\rangle_2,$$

in

$$A_{\rm NNLO}^{q\bar{q}gg} = A_{\rm NLO,A}^{q\bar{q}gg} + A_{\rm NLO,B}^{q\bar{q}gg} , \qquad (B.84)$$

where A, B correspond to the two graphs in Fig. B-7. We have

$$\begin{aligned} A_{\text{NNLO},A}^{q\bar{q}gg} &= U^{(2,1,0)}(n_0,n_0)(Q,\mu_1) g^2 \,\bar{u}_{n_2} \Big[\frac{\bar{q}_2}{\bar{q}_2 + \bar{k}_2} \sqrt{|n_2 \cdot n'_2|} \, v_2^\beta \\ &+ \frac{\bar{k}_2}{\bar{q}_2 + \bar{k}_2} \sqrt{|n_2 \cdot n'_2|} \, \frac{\psi_2}{2} \gamma_{n'_2 \perp}^\beta \Big] \gamma_{n'_1 \perp}^\alpha \times \frac{\bar{q}_1}{q_1^2} \frac{\bar{q}_0}{q_0^2} \, \frac{n^\mu - \bar{n}^\mu}{Q} \gamma_{n_0 \perp}^\mu v_{\bar{n}} \,, \\ A_{\text{NNLO},B}^{q\bar{q}gg} &= U^{(2,1,0)}(n_0,n_0)(Q,\mu_1) \, g^2 \, \bar{u}_{n_2} \Big[\frac{\bar{q}_1}{\bar{q}_1 + \bar{k}_1} \sqrt{|n_1 \cdot n'_1|} \, v_1^\alpha \\ &+ \frac{\bar{k}_2}{\bar{q}_2 + \bar{k}_2} \sqrt{|n_2 \cdot n'_2|} \, \frac{\psi_2}{2} \gamma_{n'_1 \perp}^\alpha + \gamma_{n'_1 \perp}^\alpha \frac{\bar{k}_2 \, \bar{q}_2}{(\bar{q}_2 + \bar{k}_1)(\bar{q}_2 + \bar{k}_2)} \sqrt{|n_2 \cdot n'_2|} \, \frac{\psi_2}{2} \end{aligned}$$

$$-\gamma^{\alpha}_{n_{1}^{\prime}\perp}\frac{\bar{k}_{1}\bar{q}_{1}}{(\bar{q}_{2}+\bar{k}_{1})(\bar{q}_{1}+\bar{k}_{1})}\sqrt{|n_{1}\cdot n_{1}^{\prime}|}\frac{\psi_{1}}{2}\Big]\gamma^{\beta}_{n_{1}^{\prime}\perp}\frac{\bar{q}_{1}}{q_{1}^{2}}\frac{\bar{q}_{0}}{q_{0}^{2}}\frac{n^{\mu}-\bar{n}^{\mu}}{Q}\gamma^{\mu}_{n_{0}\perp}v_{\bar{n}}\,,$$

where in Eq. (B.85) we have already rotate the amplitude to the directions n_2 , n'_1 and n'_2 . From Eqs. (B.79) and (B.85) we can see that the Wilson coefficient $C_{2,\text{NNLO}}^{(2)H,a}(n_2,n'_1,n'_2)$ is

$$C_{2,\text{NNLO}}^{(2)H,a}(n_2,n_1',n_2') = U^{(2,1,0)}(n_0,n_0)(Q,\mu_1)d_1^{H\,\alpha\beta}(n_2,n_1',n_2')$$
$$\times \Theta_{\lambda^2}[n_2 \cdot n_1']\Theta_{\lambda^2}[n_2 \cdot n_2']\Theta_{\lambda^2}[n_2' \cdot n_1'], \qquad (B.86)$$

where

$$d_1^{H\,\alpha\beta}(n_2, n_1', n_2') = d_{1,A}^{H\,\alpha\beta}(n_2, n_1', n_2') + d_{1,B}^{H\,\alpha\beta}(n_2, n_1', n_2') \tag{B.87}$$

with

$$\begin{aligned} d_{1,A}^{H\,\alpha\beta}(n_2,n_1',n_2') &= \left[\frac{\bar{q}_2}{\bar{q}_2 + \bar{k}_2} \sqrt{|n_2 \cdot n_2'|} \, v_2^\beta + \frac{\bar{k}_2}{\bar{q}_2 + \bar{k}_2} \sqrt{|n_2 \cdot n_2'|} \, \frac{\psi_2}{2} \gamma_{n_2'\perp}^\beta \right] \gamma_{n_1'\perp}^\alpha \frac{\bar{q}_1}{q_1^2} \frac{\bar{q}_0}{q_0^2} \, \frac{n^\mu - \bar{n}^\mu}{Q} \gamma_{n_0\perp}^\mu \,, \\ d_{1,B}^{H\,\alpha\beta}(n_2,n_1',n_2') &= \left[\frac{\bar{q}_1}{\bar{q}_1 + \bar{k}_1} \sqrt{|n_1 \cdot n_1'|} \, v_1^\alpha + \frac{\bar{k}_2}{\bar{q}_2 + \bar{k}_2} \sqrt{|n_2 \cdot n_2'|} \, \frac{\psi_2}{2} \gamma_{n_1'\perp}^\alpha \right] \\ &+ \gamma_{n_1'\perp}^\alpha \frac{\bar{k}_2 \, \bar{q}_2}{(\bar{q}_2 + \bar{k}_1)(\bar{q}_2 + \bar{k}_2)} \sqrt{|n_2 \cdot n_2'|} \, \frac{\psi_2}{2} \, (B.88) \\ &- \gamma_{n_1'\perp}^\alpha \frac{\bar{k}_1 \, \bar{q}_1}{(\bar{q}_2 + \bar{k}_1)(\bar{q}_1 + \bar{k}_1)} \sqrt{|n_1 \cdot n_1'|} \, \frac{\psi_1}{2} \right] \gamma_{n_2'\perp}^\beta \frac{\bar{q}_2 + \bar{k}_1}{(q_2 + k_1)^2} \frac{\bar{q}_0}{q_0^2} \frac{n^\mu - \bar{n}^\mu}{Q} \gamma_{n_0\perp}^\mu \,. \end{aligned}$$

For the power counting of $C_{2,\text{NNLO}}^{(2)H,a}(n_2,n_1',n_2')$, as for the previous case, we have to consider that the matching is done in a region where $n_2 \cdot n_1' \sim n_2 \cdot n_2' \sim n_1' \cdot n_2' \sim \lambda^2$, this implies

$$C_{2,\text{NNLO}}^{(2)H,a}(n_2,n_1',n_2') \sim \lambda^{-1}$$
. (B.89)

We now turn to calculate the coefficient $C_2^{(2)}(n_1, n_1', n_1')$. We will follow the same

path done above. We decompose $C_2^{(2)}(n_1, n_1', n_1')$ as

$$C_2^{(2)}(n_2, n'_1, n'_1) = C_{2,\text{NLO}}^{(2)J}(n_2, n'_1, n'_1) + C_2^{(2)H}(n_2, n'_1, n'_1), \qquad (B.90)$$

where

$$C_{2,\text{NLO}}^{(2)J}(n_2, n'_1, n'_1) \langle 0 | \mathcal{O}_2^{(2)}(n_2, n'_1, n'_1) | q_{n_2} g_{n'_1} g_{n'_1} \bar{q}_{\bar{n}} \rangle_2$$

$$= C_{1,\text{LO}}^{(0)} \int dx_1 dx_2 \langle 0 | T \{ \mathcal{L}_{\text{SCET}_1}(x_1) \mathcal{L}_{\text{SCET}_1}(x_2) \mathcal{O}_1^{(0)}(n_0, n_0) \} | q_{n_2} g_{n'_1} \bar{q}_{\bar{n}} \rangle_2 ,$$
(B.91)

and

$$C_{2}^{(2)H}(n_{2}, n_{1}', n_{1}') \langle 0 | \mathcal{O}_{2}^{(2)}(n_{2}, n_{1}', n_{1}') | q_{n_{2}} g_{n_{1}'} \bar{q}_{\bar{n}} \rangle_{2}$$

$$= C_{1}^{(1)} \int dx \langle 0 | T \{ \mathcal{L}_{\text{SCET}_{1}}(x) \mathcal{O}_{1}^{(1)}(n_{0}, n_{0}) \} | q_{n_{2}} g_{n_{1}'} \bar{q}_{\bar{n}} \rangle_{2}$$

$$+ C_{1,\mathcal{T}}^{(1)} \int dx \langle 0 | T \{ \mathcal{L}_{\text{SCET}_{1}}(x) \mathcal{T}_{1}^{(1)}(n_{0}, n_{0}) \} | q_{n_{2}} g_{n_{1}'} \bar{q}_{\bar{n}} \rangle_{2}.$$
(B.92)

We decompose $C_2^{(2)}(n_2, n'_1, n'_1)$ as

$$C_2^{(2)H}(n_2, n_1', n_1') = C_{2,\text{NNLO}}^{(2)H}(n_2, n_1', n_1') + C_{2,\text{N}^3\text{LO}}^{(2)H}(n_2, n_1', n_1'), \qquad (B.93)$$

where $C_{2,\text{NLO}}^{(2)H}(n_2, n'_1, n'_1)$ is the coefficient that multiplied by $\langle 0|\mathcal{O}_2^{(2)}(n_2, n'_1, n'_2)|q_{n_2}g_{n'_1}g_{n'_2}\bar{q}_{\bar{n}}\rangle_2$ reproduces the the second line in the Eq. (B.92) and $C_{2,\text{NLO}}^{(2)H}(n_2, n'_1, n'_1)$ the third line. We will only calculate $C_{2,\text{NLO}}^{(2)J}(n_2, n'_1, n'_1)$ and prove that scales as λ^{-2} . This is the only operator that we need to calculate the amplitude squared at NLO. We will not calculate $C_{2,\text{NNLO}}^{(2)H}(n_2, n'_1, n'_1)$ and $C_{2,\text{NNLO}}^{(2)H}(n_2, n'_1, n'_1)$ here, but it is not difficult to do it and to prove that they scale as λ^{-1} and λ^0 .

To calculate the amplitude in the second line in Eq. (B.91), we can use Eqs. (B.78) that is written in terms of the directions n_2 , n'_1 and n'_2 that are the directions parallel to the external particles and take the limit $n'_2 \cdot n'_1 \rightarrow \lambda^4$. This is because in this case the two gluons are collinear in SCET₂, that is we have $k_1 \cdot k_2 \sim \lambda^4 Q$ or equivalently $n'_1 \cdot n'_2 \sim \lambda^4$ with $n_1 \cdot n'_1$ and $n_1 \cdot n'_2$ still of order λ^2 . Thus we can define $C_{2, \text{LO}}^{(2)J}(n_2, n'_1, n'_1)$

$$C_{2,\text{NLO}}^{(2)J}(n_2, n_1', n_1') = U^{(0)}(n_0)(Q, \mu_1) d_2^{J\,\alpha\beta}(n_2, n_1', n_2') \Theta_{\lambda^2}[n_2 \cdot n_1'] \Theta_{\lambda^2}[n_2 \cdot n_2'], \quad (B.94)$$

where

as

$$\begin{split} d_{2}^{J\,\alpha\beta}(n_{2},n_{1}',n_{2}') &= \lim_{n_{2}'n_{1}'\to\lambda^{4}} d_{1}^{\alpha\beta}(n_{2},n_{1}',n_{2}') \quad (B.95) \\ &= \left(\left[\frac{\bar{q}_{2}}{\bar{q}_{2}+\bar{k}_{2}} \sqrt{|n_{2}\cdot n_{2}'|} v_{2}^{\beta} + \frac{\bar{k}_{2}}{\bar{q}_{2}+\bar{k}_{2}} \sqrt{|n_{2}\cdot n_{2}'|} \frac{\psi_{2}}{2} \gamma_{n_{2}'\perp}^{\beta} \right] \quad (B.96) \\ &\times \left[\frac{\bar{q}_{1}}{\bar{q}_{1}+\bar{k}_{1}} \sqrt{|n_{1}\cdot n_{1}'|} v_{1}^{\alpha} + \frac{\bar{k}_{1}}{\bar{q}_{1}+\bar{k}_{1}} \sqrt{|n_{1}\cdot n_{1}'|} \frac{\psi_{1}}{2} \gamma_{n_{1}'\perp}^{\alpha} \right] \\ &\quad \frac{\bar{q}_{1}}{\bar{q}_{1}^{2} \bar{q}_{2} \bar{k}_{1}(n_{2}\cdot n_{1}') + \bar{q}_{2} \bar{k}_{2}(n_{2}\cdot n_{2}')} \gamma_{n_{0}\perp}^{\mu}, \\ &+ \left[\frac{\bar{q}_{1}}{\bar{q}_{1}+\bar{k}_{1}} \sqrt{|n_{1}\cdot n_{1}'|} v_{1}^{\alpha} + \frac{\bar{k}_{2}}{\bar{q}_{2}+\bar{k}_{2}} \sqrt{|n_{2}\cdot n_{2}'|} \frac{\psi_{2}}{2} \gamma_{n_{1}'\perp}^{\alpha} \right] \\ &+ \gamma_{n_{1}'\perp}^{\alpha} \frac{\bar{k}_{2} \bar{q}_{2}}{(\bar{q}_{2}+\bar{k}_{1})(\bar{q}_{2}+\bar{k}_{2})} \sqrt{|n_{2}\cdot n_{2}'|} \frac{\psi_{2}}{2} - \gamma_{n_{1}'\perp}^{\alpha} \frac{\bar{k}_{1} \bar{q}_{1}}{(\bar{q}_{2}+\bar{k}_{1})(\bar{q}_{1}+\bar{k}_{1})} \sqrt{|n_{1}\cdot n_{1}'|} \frac{\psi_{1}}{2} \\ &\times \left[\frac{\bar{q}_{2}}{\bar{q}_{2}+\bar{k}_{2}} \sqrt{|n_{2}\cdot n_{2}'|} v_{2}^{\beta} - \frac{\bar{k}_{1}}{\bar{q}_{1}+\bar{k}_{1}} \sqrt{|n_{1}\cdot n_{1}'|} v_{1}^{\beta} + \frac{\bar{k}_{1}}{\bar{q}_{1}+\bar{k}_{1}} \sqrt{|n_{1}\cdot n_{1}'|} \frac{\psi_{1}}{2} \gamma_{n_{2}'\perp}^{\beta} \right] \\ &\times \frac{\bar{q}_{2}+\bar{k}_{1}}{(q_{2}+\bar{k}_{1})^{2}} \frac{4\bar{q}_{0}}{\bar{q}_{2}\bar{k}_{1}(n_{2}\cdot n_{1}') + \bar{q}_{2}\bar{k}_{2}(n_{2}\cdot n_{2}')} \gamma_{n_{0}\perp}^{\mu}, \\ &+ \left[\frac{1}{\bar{q}_{0}-\bar{k}_{2}} \gamma_{n_{1}'\perp}^{\alpha} \gamma_{n_{2}'\perp}^{\beta} + \frac{1}{\bar{k}_{2}+\bar{q}_{2}} \gamma_{n_{2}'\perp}^{\beta} \gamma_{n_{1}'\perp}^{\beta} \right] \frac{4\bar{q}_{0}}{\bar{q}_{2}\bar{k}_{1}(n_{2}\cdot n_{1}') + \bar{q}_{2}\bar{k}_{2}(n_{2}\cdot n_{2}')} \gamma_{n_{0}\perp}^{\mu}) \\ &\times \Theta_{\lambda^{4}}[n_{1}'\cdot n_{2}']. \end{split}$$

In Eq. (B.94) there is a mismatch of notation between the LHS and RHS. In the LHS we have the quark labeled with n_2 and the two gluons with n'_1 because the coefficient (B.80) is for the operator $\mathcal{O}_2^{(2)}(n_2, n'_1, n'_1)$, where the tow gluons are labeled with the same direction n'_1 because they are collinear in SCET₂. In the RHS of (B.80) n_2 , n'_1 and n'_2 are the directions parallel to the quarks and the two gluons as defined in Appendix B.1. We encode the information that the two gluons are collinear using the theta function in the RHS of equation Eq. (B.96) that carries the information that $n'_1 \cdot n'_2 \lesssim \lambda$. In the RHS of Eq. (B.96) we could decompose n_2 in terms of n_1 and \bar{n} , writing everything only in terms of this two directions, and in this way we can avoid to insert the theta function, however it is convenient to leave explicitly n_2 , n'_1 and n'_2 , because it will make easier for the matching to SCET₃. We notice that the RHS of Eq. (B.96) is just equal to the coefficient $C_{2,\text{NLO}}^{(2)J}(n_2, n'_1, n'_2)$ defined in Eq. (B.80) with the substitution $q_0^2 \rightarrow \bar{q}_2 \bar{k}_1(n_2 \cdot n'_1)/4 + \bar{q}_2 \bar{k}_2(n_2 \cdot n'_2)/4$. Knowing that $n'_1 \cdot n'_2 \sim \lambda^4$, $n_1 \cdot n'_2 \sim \lambda^2$ and $n_1 \cdot n'_1 \sim \lambda^2$, it is easy to check that Eq. (B.96) scales as λ^{-2} . Because $C_{2,\text{LO}}^{(2)J}(n_2, n'_1, n'_1)$ comes from a two-jet SCET₁ operator, we have encoded the information the $n_2, \cdot n'_1 \lesssim \lambda^2$ in Eq. (B.94) on the Θ functions.

For the coefficient $C_2^{(2)}(n_2, n_2, n_1')$, we decompose it as

$$C_2^{(2)}(n_2, n_2, n_1') = C_{2,\text{NLO}}^{(2)J}(n_2, n_2, n_1') + C_2^{(2)H}(n_2, n_2, n_1'), \qquad (B.97)$$

where

$$C_{2,\text{NLO}}^{(2)J}(n_2, n_2, n_1') \langle 0 | \mathcal{O}_2^{(2)}(n_2, n_2, n_1') | q_{n_2} g_{n_2} g_{n_1'} \bar{q}_{\bar{n}} \rangle_2$$
(B.98)
= $C_{1,\text{LO}}^{(0)} \int dx_1 dx_2 \langle 0 | T \{ \mathcal{L}_{\text{SCET}_1}(x_1) \mathcal{L}_{\text{SCET}_1}(x_2) \mathcal{O}_1^{(0)}(n_0, n_0) \} | q_{n_2} g_{n_2} g_{n_1'} \bar{q}_{\bar{n}} \rangle_2$
- $C_{2,\text{LO}}^{(1)}(n_2, n_1') \int dx \langle 0 | T \{ \mathcal{L}_{\text{SCET}_2}(x) \mathcal{O}_2^{(1)}(n_2, n_1') \} | q_{n_2} g_{n_2} g_{n_1'} \bar{q}_{\bar{n}} \rangle_2 ,$

and

$$C_{2}^{(2)H}(n_{2}, n_{2}, n_{1}')\langle 0|\mathcal{O}_{2}^{(2)}(n_{2}, n_{2}, n_{1}')|q_{n_{2}}g_{n_{2}}g_{n_{1}'}\bar{q}_{\bar{n}}\rangle_{2}$$
(B.99)
$$= C_{1}^{(1)} \int dx \langle 0|T\{\mathcal{L}_{\text{SCET}_{1}}(x)\mathcal{O}_{1}^{(1)}(n_{0}, n_{0})\}|q_{n_{2}}g_{n_{2}}g_{n_{1}'}\bar{q}_{\bar{n}}\rangle_{2} - C_{2,\text{NLO}}^{(1)}(n_{2}, n_{1}') \int dx \langle 0|T\{\mathcal{L}_{\text{SCET}_{2}}(x)\mathcal{O}_{2}^{(1)}(n_{2}, n_{1}')\}|q_{n_{2}}g_{n_{2}}g_{n_{1}'}\bar{q}_{\bar{n}}\rangle_{2} + C_{1,\mathcal{T}}^{(1)} \int dx \langle 0|T\{\mathcal{L}_{\text{SCET}_{1}}(x)\mathcal{T}_{1}^{(1)}(n_{0}, n_{0})\}|q_{n_{2}}g_{n_{2}}g_{n_{1}'}\bar{q}_{\bar{n}}\rangle_{2} - C_{2,\text{NNLO}}^{(1)}(n_{2}, n_{1}') \int dx \langle 0|T\{\mathcal{L}_{\text{SCET}_{2}}(x)\mathcal{O}_{2}^{(1)}(n_{2}, n_{1}')\}|q_{n_{2}}g_{n_{2}}g_{n_{1}'}\bar{q}_{\bar{n}}\rangle_{2},$$

We decompose $C_2^{(2)H}(n_2, n_2, n_1')$ as

$$C_2^{(2)}(n_2, n_2, n_1') = C_{2,\text{NNLO}}^{(2)H}(n_2, n_2, n_1') + C_{2,\text{N}^3\text{LO}}^{(2)H}(n_2, n_2, n_1'), \qquad (B.100)$$

where $C_{2,\text{NNLO}}^{(2)H}(n_2, n_2, n_1')$ is the coefficient that multiplied by $\langle 0|\mathcal{O}_2^{(2)}(n_2, n_1', n_2')|q_{n_2}g_{n_1'}g_{n_2'}\bar{q}_{\bar{n}}\rangle_2$ reproduces the the second and third line in the Eq. (B.99), and $C_{2,\text{N}^3\text{LO}}^{(2)H}$ the forth and fifth line. As for the previous cases, the coefficient $C_{2,\text{NLO}}^{(2)J}(n_2, n_2, n_1')$ scale as $\lambda^{-2}, C_{2,\text{NNLO}}^{(2)H}(n_2, n_2, n_1')$ as λ^{-1} and $C_{2,\text{N}^3\text{LO}}^{(2)H}(n_2, n_2, n_1')$ as λ^0 . Because the operator $\mathcal{O}_2^{(2)}(n_2, n_2, n_1')$ interferences with the LO operator $\mathcal{O}_2^{(1)}(n_1, n_1')$, to have the amplitude squared up to NLO we need both $C_{2,\text{NLO}}^{(2)J}$ and $C_{2,\text{NNLO}}^{(2)H}$. Let us star with $C_{2,\text{NLO}}^{(2)J}$. To calculate the amplitude in the second line in Eq. (B.98), we use Eq. (B.78) and take the limit $n_2 \cdot n_2' \to \lambda^4$ with $n_2 \cdot n_1' \sim n_1' \cdot n_2' \sim \lambda^2$ because now we have the quark and a gluon are collinear in SCET₂.⁴ It is easy to check that

$$\lim_{n_2 \cdot n'_2 \to \lambda^4} C^{\alpha\beta}_{\text{LO},A} \mathcal{O}^{(2)}_2(n_2, n_2, n'_1) = C^{(1)}_{2,\text{LO}} \int dx \langle 0| T \{ \mathcal{L}_{\text{SCET}_2}(x) \mathcal{O}^{(1)}_2 \} |q_{n_2} g_{n_2} g_{n'_1} \bar{q}_{\bar{n}} \rangle_2 .$$
(B.101)

Thanks to Eq. (B.101), we can write $C_{2,\text{NLO}}^{(2)J}(n_2, n_2, n_1')$ as

$$C_{2,\text{NLO}}^{(2)J}(n_2, n_2, n_1') = U^{(0)}(n_0)(Q, \mu_1) d_3^{J \alpha \beta}(n_2, n_1', n_2') \Theta_{\lambda^2}[n_2 \cdot n_2'] \Theta_{\lambda^2}[n_1' \cdot n_2'] \quad (B.102)$$

where

$$d_{3}^{J\alpha\beta}(n_{2},n_{1}',n_{2}') = \lim_{n_{2}\cdot n_{2}'\to\lambda^{4}} (d_{1B}^{J\alpha\beta}(n_{2},n_{1}',n_{2}') + d_{1C}^{J\alpha\beta}(n_{2},n_{1}',n_{2}'))$$
(B.103)
$$= \left(\left[\frac{\bar{q}_{1}}{\bar{q}_{1} + \bar{k}_{1}} v_{1}^{\alpha} - \gamma_{n_{1}'\perp}^{\alpha} \frac{\bar{k}_{1} \bar{q}_{1}}{(\bar{q}_{2} + \bar{k}_{1})(\bar{q}_{1} + \bar{k}_{1})} \frac{\not{p}_{1}}{2} \right]$$
$$+ \left[\frac{1}{\bar{q}_{0} - \bar{k}_{2}} \gamma_{n_{1}'\perp}^{\alpha} \gamma_{n_{2}'\perp}^{\beta} + \frac{1}{\bar{k}_{2} + \bar{q}_{2}} \gamma_{n_{2}'\perp}^{\beta} \gamma_{n_{1}'\perp}^{\alpha} \right]$$
$$\times \frac{4\bar{q}_{0}}{\bar{k}_{1}(\bar{k}_{2}(n_{2}'\cdot n_{1}') + \bar{q}_{2}(n_{2}\cdot n_{1}'))} \gamma_{n_{0}\perp}^{\mu} \right) \Theta_{\lambda^{4}}[n_{2}\cdot n_{1}'] .$$

The limit (B.102) is define for $(n_2n'_1) \sim (n'_2n'_1) \sim \lambda^2$, thus the coefficient $C_{2,\text{NLO}}^{(2)J}(n_2, n_2, n'_1)$ is of order λ^{-2} . As previously for $C_{2,\text{NLO}}^{(2)J}(n_1, n'_1, n'_1)$, we prefer leaving (B.102) in terms of the direction n_2 , n'_1 and n'_2 . To calculate $C_{2,\text{NNLO}}^{(2)H}(n_2, n_2, n'_1)$ we proceed in the same

⁴We could alternatively take the limit $n_2 \cdot n_1' \to \lambda^4$ with $n_2 \cdot n_2' \sim n_2' \cdot n_1' \sim \lambda^2$

:

way. We have

$$C_{2,\text{NNLO}}^{(2)H}(n_2, n_2, n_1') = U^{(2,1,0)}(n_0, n_0)(Q, \mu_1) d_3^{H\,\alpha\beta}(n_2, n_1', n_2') \Theta_{\lambda^2}[n_2 \cdot n_2'] \Theta_{\lambda^2}[n_1' \cdot n_2'],$$
(B.104)

where

$$d_{3}^{H\,\alpha\beta}(n_{2},n_{1}',n_{2}') = \lim_{n_{2}\cdot n_{2}'\to\lambda^{4}} d_{1B}^{H\,\alpha\beta}(n_{2},n_{1}',n_{2}')$$

$$= \left(\left[\frac{\bar{q}_{1}}{\bar{q}_{1}+\bar{k}_{1}}\sqrt{|n_{1}\cdot n_{1}'|} v_{1}^{\alpha} - \gamma_{n_{1}'\perp}^{\alpha} \frac{\bar{k}_{1}\,\bar{q}_{1}}{(\bar{q}_{2}+\bar{k}_{1})(\bar{q}_{1}+\bar{k}_{1})} \sqrt{|n_{1}\cdot n_{1}'|} \frac{\not{p}_{1}}{2} \right]$$

$$\times \gamma_{n_{2}'\perp}^{\beta} \frac{\bar{q}_{2}+\bar{k}_{1}}{(q_{2}+k_{1})^{2}} \frac{\bar{q}_{0}}{q_{0}^{2}} \frac{n^{\mu}-\bar{n}^{\mu}}{Q} \gamma_{n_{0}\perp}^{\mu} \right) \Theta_{\lambda^{4}}[n_{2}\cdot n_{1}'].$$
(B.105)

In Eq. (B.104) we have use the fact that

$$\lim_{n_2 \cdot n_2' \to \lambda^4} C_{\text{NNLO},A}^{\alpha\beta} \mathcal{O}_2^{(2)}(n_2, n_2, n_1') = C_{2,\text{NLO}}^{(1)} \int dx \langle 0 | T \{ \mathcal{L}_{\text{SCET}_2}(x) \mathcal{O}_2^{(1)} \} | q_{n_2} g_{n_2} g_{n_1'} \bar{q}_{\bar{n}} \rangle_2 \,.$$
(B.106)

In Eqs. (B.102, B.104) there is a mismatch of notation between the LHS and RHS of the same kind of Eq. (B.94). In the LHS we have the quark and one gluon labeled with n_2 and the other gluon with n'_1 because the coefficient (B.100) is for the operator $\mathcal{O}_2^{(2)}(n_2, n_2, n'_1)$, where the quark and a gluon are labeled with the same direction n_2 because they are collinear in SCET₂. In the RHS of (B.80) n_2 , n'_1 and n'_2 are the directions parallel to the quarks and the two gluons as defined in Appendix B.1. We encode the information that the quark and a gluon are collinear in the theta function in the RHS of equation Eqs. (B.103, B.104) that carries the information that $n_2 \cdot n'_1 \lesssim \lambda^4$. Because $C_{2,\text{LO}}^{(2)J}(n_2, n_2, n'_1)$ and $C_{2,\text{NNLO}}^{(2)H}(n_2, n_2, n'_1)$ comes from two-jet SCET₁ operators, we encoded the information that $n_2, \cdot n'_1 \lesssim \lambda^2$ in Eqs. (B.102,B.104) in the Θ functions.

We have that all the NLO terms from the two gluon matching come from the $SCET_1$ operator $\mathcal{O}_1^{(0)}(n_0)$ and are jet corrections and at NNLO we have only hard corrections. Before matching $SCET_2$ to $SCET_3$, we have to insert in the coefficients

the $SCET_2$ running factors. Below we list all the $SCET_2$ coefficients at NLO that we have calculated with the appropriated $SCET_2$ running factors. From the matching of one-gluon emission we have the coefficients:

$$C_{2,\text{LO}}^{(1)}(n_1, n_1') = U^{(2,1,0)}(n_1, n_1')(\mu_1, \mu)U^{(0)}(n_0)(Q, \mu_1)c_{\text{LO}}(n_0)\frac{\bar{q}_0}{q_0^2}\gamma_{n_0\perp}^{\mu}, \quad (B.107)$$

$$C_{2,\text{NLO}}^{(1)H,a}(n_1, n_1') = U^{(2,1,0)}(n_1, n_1')(\mu_1, \mu)U^{(2,1,0)}(n_0, n_0)(Q, \mu_1) \otimes c_{2,\text{NLO}}^{H,a}, \quad (B.107)$$

$$C_{2,\text{NNLO}}^{(1)H}(n_1, n_1') = U^{(2,1,0)}(n_1, n_1')(\mu_1, \mu)U^{(2,1,0)}(n_0, n_0)(Q, \mu_1) \otimes c_{2,\text{NNLO}}^{H}, \quad (C_{2,\text{NNLO}}^{(1)H,b}(n_1, n_1') = U^{(2,1,0)}(n_1, n_1')(\mu_1, \mu)C_1^{(1)}(n_1, n_1')\tilde{\Theta}_{\lambda^2}[n_1 \cdot n_1'], \quad (B.107)$$

where the first coefficient in (B.107) is defined in Eq. (B.51), the second and third in Eqs. (B.57), and the last in (B.59). From the matching of two-gluon emission we have the coefficients:

$$C_{2,\text{NLO}}^{(2)J}(n_{2},n'_{2},n'_{1}) = U^{(2,1,0)}(n_{2},n'_{2},n'_{1})(\mu_{1},\mu) U^{(0)}(n_{0})(Q,\mu_{1}) \qquad (B.108)$$

$$\times d_{1}^{J}(n_{2},n'_{1},n'_{2})\Theta_{\lambda^{2}}[n_{2}\cdot n'_{1}]\Theta_{\lambda^{2}}[n_{2}\cdot n'_{2}]\Theta_{\lambda^{2}}[n'_{2}\cdot n'_{1}],$$

$$C_{2,\text{NLO}}^{(2)J}(n_{2},n'_{1},n'_{1}) = U^{(2,1,0)}(n_{2},n'_{1},n'_{1})(\mu_{1},\mu) \otimes U^{(0)}(n_{0})(Q,\mu_{1})$$

$$\times d_{2}^{J}(n_{2},n'_{1},n'_{2})\Theta_{\lambda^{2}}[n_{2}\cdot n'_{1}]\Theta_{\lambda^{2}}[n_{2}\cdot n'_{2}],$$

$$C_{2,\text{NLO}}^{(2)J}(n_{2},n_{2},n'_{1}) = U^{(2,1,0)}(n_{2},n_{2},n'_{1})(\mu_{1},\mu) \otimes U^{(0)}(n_{0})(Q,\mu_{1})$$

$$\times d_{3}^{J}(n_{2},n'_{1},n'_{2})\Theta_{\lambda^{2}}[n_{2}\cdot n'_{2}]\Theta_{\lambda^{2}}[n'_{1}\cdot n'_{2}],$$

$$\times d_{1}^{H}(n_{2},n'_{1},n'_{2})\Theta_{\lambda^{2}}[n_{2}\cdot n'_{1}]\Theta_{\lambda^{2}}[n_{2}\cdot n'_{2}]\Theta_{\lambda^{2}}[n'_{2}\cdot n'_{1}],$$

$$C_{2,\text{NNLO}}^{(2)H}(n_{2},n_{2},n'_{1}) = U^{(2,1,0)}(n_{2},n_{2},n'_{1})(\mu_{1},\mu) \otimes U^{(2,1,0)}(n_{0},n_{0})(Q,\mu_{1})$$

$$\times d_{3}^{H}(n_{2},n'_{1},n'_{2})\Theta_{\lambda^{2}}[n_{2}\cdot n'_{2}]\Theta_{\lambda^{2}}[n'_{1}\cdot n'_{2}],$$

where the coefficients are defined in Eqs. (B.80, B.94, B.102, B.104).

B.4 Matching SCET₂ to SCET₃ to SCET_N

In this Appendix we match SCET₂ to SCET₃. All the SCET₃ operators necessary for the matching up to two-gluon emission are: $\mathcal{O}_3^{(0)}(n_0)$, $\mathcal{O}_3^{(1)}(n_0, n_0)$, $\mathcal{O}_3^{(1)}(n_1, n_1')$, $\mathcal{O}_{3}^{(2)}(n_{2},n_{1}',n_{2}'), \mathcal{O}_{3}^{(2)}(n_{1},n_{1},n_{1}'), \mathcal{O}_{3}^{(2)}(n_{1},n_{1}',n_{1}')$. We have seen that to describe parton shower for one emission we only need the information of the coefficient of the SCET₂ operator $\mathcal{O}_{2}^{(1)}(n_{1},n_{1}')$. We can apply the same argument here and infer that in SCET₃ the only information that we need, is in the the coefficient of the operator $\mathcal{O}_{3}^{(2)}(n_{2},n_{2}',n_{1}')$, that is the only one that we calculate. We can follow the same steps made for matching SCET₁ to SCET₂ to calculate the Wilson coefficients $C_{3}^{(2)}(n_{2},n_{2}',n_{1}')$. In this way it is no difficult to show that

$$C_{3}^{(2)}(n_{2}, n_{2}', n_{1}') = C_{3, \text{LO}}^{(2)} + C_{3, \text{NLO}}^{(2)H,a} + C_{3, \text{NLO}}^{(2)H,b} + C_{3, \text{NLO}}^{(2)Ja} + C_{3, \text{NLO}}^{(2)Jb} + C_{3, \text{NLO}}^{(2)Jc} + C_{3, \text{NNLO}}^{(2)H,a} + C_{3, \text{NNLO}}^{(2)H,b},$$
(B.109)

where

$$C_{3,\text{LO}}^{(2)}(n_{2},n_{1}',n_{2}') = C_{3,\text{LO}}^{(1)}(n_{2},n_{2}')C_{2,\text{LO}}^{(1)}(n_{1},n_{1}'), \qquad (B.110)$$

$$C_{3,\text{NLO}}^{(2)H,a}(n_{2},n_{1}',n_{2}') = C_{3,\text{LO}}^{(1)}(n_{2},n_{2}')C_{2,\text{NLO}}^{(1)H,a}(n_{1},n_{1}'), \qquad (B.110)$$

$$C_{3,\text{NLO}}^{(2)H,b}(n_{2},n_{1}',n_{2}') = C_{3,\text{LO}}^{(1)}(n_{2},n_{2}')C_{2,\text{NLO}}^{(1)H,b}(n_{1},n_{1}'), \qquad (B.110)$$

$$C_{3,\text{NLO}}^{(2)J,1}(n_{2},n_{1}',n_{2}') = C_{2,\text{NLO}}^{(1)}(n_{2},n_{2}')C_{2,\text{NLO}}^{(1)H,b}(n_{1},n_{1}'), \qquad (C_{3,\text{NLO}}^{(2)J,1}(n_{2},n_{1}',n_{2}') = C_{2,\text{NLO}}^{(2)J}(n_{2},n_{1}',n_{2}')\tilde{\Theta}_{\lambda^{4}}[n_{2}\cdot n_{2}']\tilde{\Theta}_{\lambda^{4}}[n_{2}\cdot n_{1}']\tilde{\Theta}_{\lambda^{4}}[n_{2}'\cdot n_{1}'], \qquad (C_{3,\text{NLO}}^{(2)J,2}(n_{2},n_{1},n_{2}') = C_{2,\text{NLO}}^{(2)J}(n_{2},n_{1}',n_{1}')\tilde{\Theta}_{\lambda^{4}}[n_{2}\cdot n_{2}']\tilde{\Theta}_{\lambda^{4}}[n_{2}\cdot n_{1}']\tilde{\Theta}_{\lambda^{4}}[n_{2}'\cdot n_{1}'], \qquad (C_{3,\text{NLO}}^{(2)J,3}(n_{2},n_{1}',n_{2}') = C_{2,\text{NLO}}^{(2)J}(n_{2},n_{2},n_{1}')\Theta_{\lambda^{4}}[n_{2}\cdot n_{2}']\tilde{\Theta}_{\lambda^{4}}[n_{2}\cdot n_{1}']\tilde{\Theta}_{\lambda^{4}}[n_{2}'\cdot n_{1}'], \qquad (C_{3,\text{NLO}}^{(2)H,a}(n_{2},n_{1}',n_{2}') = C_{3,\text{LO}}^{(1)}(n_{2},n_{2}')C_{2,\text{NNLO}}^{(1)}(n_{1},n_{1}'), \qquad (C_{3,\text{NNLO}}^{(2)H,b}(n_{2},n_{1}',n_{2}') = C_{2,\text{NNLO}}^{(2)H}(n_{2},n_{2},n_{1}')\Theta_{\lambda^{4}}[n_{2}\cdot n_{2}']\tilde{\Theta}_{\lambda^{4}}[n_{2}\cdot n_{1}']\tilde{\Theta}_{\lambda^{4}}[n_{2}'\cdot n_{1}'], \qquad (C_{3,\text{NNLO}}^{(2)H,b}(n_{2},n_{1}',n_{2}') = C_{3,\text{LO}}^{(2)H}(n_{2},n_{2},n_{1}')\Theta_{\lambda^{4}}[n_{2}\cdot n_{2}']\tilde{\Theta}_{\lambda^{4}}[n_{2}\cdot n_{1}']\tilde{\Theta}_{\lambda^{4}}[n_{2}'\cdot n_{1}'], \qquad (C_{3,\text{NNLO}}^{(2)H,b}(n_{2},n_{1}',n_{2}') = C_{2,\text{NNLO}}^{(2)H}(n_{2},n_{2},n_{1}')\Theta_{\lambda^{4}}[n_{2}\cdot n_{2}']\tilde{\Theta}_{\lambda^{4}}[n_{2}\cdot n_{1}']\tilde{\Theta}_{\lambda^{4}}[n_{2}'\cdot n_{1}'], \qquad (C_{3,\text{NNLO}}^{(2)H,b}(n_{2},n_{1}',n_{2}') = C_{2,\text{NNLO}}^{(2)H}(n_{2},n_{2},n_{1}')\Theta_{\lambda^{4}}[n_{2}\cdot n_{2}']\tilde{\Theta}_{\lambda^{4}}[n_{2}\cdot n_{1}']\tilde{\Theta}_{\lambda^{4}}[n_{2}'\cdot n_{1}'], \qquad (C_{3,\text{NNLO}}^{(2)H,b}(n_{2},n_{1}',n_{2}') = C_{2,\text{NNLO}}^{(2)H}(n_{2},n_{2},n_{1}')\Theta_{\lambda^{4}}[n_{2}\cdot n_{2}']\tilde{\Theta}_{\lambda^{4}}[n_{2}'\cdot n_{1}']\tilde{\Theta}_{\lambda^{4}}[n_{2}'\cdot n_{1}'], \qquad (C_{3,\text{NNLO}}^{(2)H,b}(n_{2},n_{1}',n_{2}') = C_{2,\text{NNLO}}^{(2)H}(n_{2},n_{2},n_{1}')$$

and

$$C_{3,\text{LO}}^{(1)}(n_2, n_2') = \left(2\frac{(q_2)_{n_1\perp}^{\beta}}{\bar{k}_2} + \frac{(q_2)_{n_0\perp}\gamma_{n_2'\perp}^{\beta}}{\bar{q}_2}\right)\frac{\bar{q}_1}{q_1^2}\frac{\not{n}n_1}{4}\Theta_{\lambda^4}[n_2 \cdot n_2'].$$
(B.111)

In the LHS of the equations in the first, second and third line of (B.110) we can write n_1 in terms of n_2 , n'_2 and n'_1 using the formulas in (B.18). The SCET₂ coefficients

 $C_{3,\text{NLO}}^{(2)J}$ and $C_{3,\text{NNLO}}^{(2)H,b}$ are defined in Eqs. (B.108) and thy were already defined in terms of n_2 , n'_2 and n'_1 (see Eqs. (B.80, B.94, B.102, B.104). As for the SCET₁ to SCET₂ matching, there is the problem of encoding additional information into the Wilson coefficient. The operator $\mathcal{O}_3^{(2)}(n_2, n'_2, n'_1)$ only knows that $n_2 \cdot n'_1, n_2 \cdot n'_2, n'_1 \cdot n'_2 > \lambda^6$, but we can see that the Wilson coefficients defined in Eqs. (B.110) come from matching SCET₂ to SCET₃ in different regions of the phase space. $C_{3,\text{LO}}^{(2)H,a}, C_{3,\text{NLO}}^{(2)H,a}$ are defined for $n_2 \cdot n'_2 \lesssim \lambda^4$ and $\lambda^4 < n_2 \cdot n'_1, n'_1 \cdot n'_2 \lesssim \lambda^2$, $C_{3,\text{NLO}}^{(2)J,a}$ for $\lambda^4 < n_2 \cdot n'_1, n_2 \cdot n'_2, n'_1 \cdot n'_2 \lesssim \lambda^2$; $C_{3,\text{NLO}}^{(2)J,b}$ for $n'_1 \cdot n'_2 \lesssim \lambda^4$ and $\lambda^4 < n_2 \cdot n'_1, n_2 \cdot n'_2 \lesssim \lambda^2$; $C_{3,\text{NLO}}^{(2)J,c}$ for $n_2 \cdot n'_2 \lesssim \lambda^4$ and $\lambda^4 < n_2 \cdot n'_1, n'_1 \cdot n'_2 \lesssim \lambda^2$. Some information about the scalar product of n_3 is already present in the theta functions of the SCET₂ coefficients, we add the missing information in the Θ functions present in Eqs. (B.110). All the coefficients SCET₃ defined above are for the same SCET₃ operator, thus now all the power counting is in the Wilson coefficients. The LO coefficient scales as λ^{-3} , the NLO coefficients as λ^{-2} and the NNLO coefficients as λ^{-1} .

At LO the coefficient×operator in SCET₃ is given by the LO coefficient×operator in SCET₂, $C_{2,LO}^{(1)}\mathcal{O}_2^{(1)}(n_1, n_1')$ multiplying it by the running function $U^{(2,1,0)}(n_1, n_1')(\mu_1, \mu)$ and applying the replacement

$$(\bar{\chi}_{n_2})_i \to (c_{\rm LO}^{\alpha})_{ji}(n_1)(\bar{\chi}_{n_1})_j g \mathcal{B}_{\alpha}^{n_1' \perp},$$
 (B.112)

where $c_{\rm LO}^{\alpha}(n_1)$ is

Eq. (B.112) has the same structure than Eq. (B.55). If we go on with the matching down to SCET_N we find that the LO result would be given applying the above replacement N-1 times. At SCET_N we could match everything to the operator $\mathcal{O}_N^{(N-1)}(n_{N-1}, n'_1, \ldots, n'_{N-1})$, and the LO coefficient is

$$C_{N, \text{LO}}^{(N-1)} = \prod_{k=1}^{N-1} U^{(k-1)}(\mu_{k-1}, \mu_k) c_{\text{LO}}^{\alpha_k}(n_{k-1}) \Gamma^{\mu}, \qquad (B.114)$$

with $\mu_k = \sqrt{|n_{k-1} \cdot n'_{k-1}|}Q$ and

$$c_{\rm LO}^{\alpha}(n_k) = \left(2\frac{(q_{k+1})_{n_0\perp}^{\alpha}}{\bar{k}_{k+1}} + \frac{(\not{q}_{k+1})_{n_k\perp}\gamma_{n'_{k+1}\perp}^{\alpha}}{\bar{q}_{k+1}}\right)\frac{\not{n}\not{n}_k}{4}\Theta_{(\lambda^{(k+1)})^2}[n_{k+1}\cdot n'_{k+1}]. \quad (B.115)$$

At NLO we have two kinds of corrections: hard and jet. The hard coefficients in Eqs. (B.110) depend on the QCD current, that is on the hard process at the top of the shower. We notice that the coefficient×operator in SCET₃ is just given by the coefficient×operator of the hard corrections in SCET₂ with the application of the replacement rule (B.112). If we go on with the matching down to SCET_N we find that the NLO hard correction would be given applying the above replacement rules N - 2 times to the SCET₂ hard correction operators. Thus we can consider this correction as a correction on the matrix elements from whom we start the shower:

$$C_{N,\text{NLO}}^{(N-1)H} = \left(C_{2,\text{NLO}}^{(1)H,a}(n_1, n_1') + C_{2,\text{NLO}}^{(1)H,b}(n_1, n_1') \right) \left(\prod_{k=2}^{N-1} U^{(k-1)}(\mu_{k-1}, \mu_k) c_{\text{LO}}^{\alpha_k}(n_{k-1}) \right).$$
(B.116)

The NLO jet corrections do not depend on the QCD current. In this case the SCET₃ coefficient×operators $\hat{C}_{3,\text{NLO}}^{(2)J,I}\mathcal{O}_{3}^{(2)}(n_{2},n_{1}',n_{2}')$, where $I = \{1,2,3\}$, are given by the LO SCET₁ operator $\bar{\chi}_{n_{0}}\gamma^{\mu}\chi_{\bar{n}}$ in three steps. First we multiply it for the running factor $U^{(2,1,0)}(n_{0})(Q, \mu_{1})$, second we apply the replacements

$$(\bar{\chi}_{n_2})_i \to (h_I^{\alpha\beta})_{ji}(n_2, n_1', n_2')(\bar{\chi}_{n_1})_j g \mathcal{B}_{\alpha}^{n_1' \perp} g \mathcal{B}_{\beta}^{n_2' \perp},$$
 (B.117)

where

$$h_{1}^{\alpha\beta}(n_{2},n_{1}',n_{2}') = d_{1}^{\alpha\beta}(n_{2},n_{1}',n_{2}')\tilde{\Theta}_{\lambda^{4}}[n_{2}\cdot n_{2}']\tilde{\Theta}_{\lambda^{4}}[n_{2}\cdot n_{1}']\tilde{\Theta}_{\lambda^{4}}[n_{2}'\cdot n_{1}'], \qquad (B.118)$$

$$h_{1}^{\alpha\beta}(n_{2},n_{1}',n_{2}') = d_{2}^{\alpha\beta}(n_{2},n_{1}',n_{2}')\tilde{\Theta}_{\lambda^{4}}[n_{2}\cdot n_{2}']\tilde{\Theta}_{\lambda^{4}}[n_{2}\cdot n_{1}']\Theta_{\lambda^{4}}[n_{2}'\cdot n_{1}'], \qquad (B.118)$$

$$h_{3}^{\alpha\beta}(n_{2},n_{1}',n_{2}') = d_{3}^{\alpha\beta}(n_{2},n_{1}',n_{2}')\Theta_{\lambda^{4}}[n_{2}\cdot n_{2}']\tilde{\Theta}_{\lambda^{4}}[n_{2}\cdot n_{1}']\tilde{\Theta}_{\lambda^{4}}[n_{2}'\cdot n_{1}'].$$

The $d_I^{\alpha\beta}$ coefficients are defined in Eqs.(B.81, B.96, B.103). Third we multiply the

opeartors that come from applying Eqs. (B.118) by the second running factor. This depends on the SCET₂ operator so each replacement rule (B.117) is followed by a different factor: $h_1^{\alpha\beta}$ by $U^{(2,1,0)}(n_2, n'_1, n'_2)(\mu_1, \mu_2)$, $h_2^{\alpha\beta}$ by $U^{(2,2,0)}(n_2, n'_1, n'_1)(\mu_1, \mu_2)$ and $h_3^{\alpha\beta}$ by $U^{(2,2,0)}(n_2, n_2, n'_1)(\mu_1, \mu_2)$. Because these corrections are independent of the hard scattering, we would encounter the same calculations we have done in the previous section for SCET₁ to SCET₂, at any matching SCET_i to SCET_{i+1}. Thus the NLO jet coefficients for the SCET_N final operator is

$$C_{N,\text{NLO}}^{(N-1)J} = \sum_{l=1}^{N-2} C_{N,\text{NLO}}^{(N)J}(l) , \qquad (B.119)$$

where

$$C_{N,\text{NLO}}^{(N-1)J}(l) = \sum_{I=1}^{3} \left[\left(\prod_{k=1}^{l-1} U^{(k-1)}(\mu_{k-1}, \mu_{k}) c_{\text{LO}}^{\alpha_{k}}(n_{k-1}) \right) U^{(l-1)}(\mu_{k-1}, \mu_{k}) \otimes h_{I}^{\alpha\beta}(n_{l+1}, n_{l}', n_{l+1}') \right] \times \left(\prod_{k=l+1}^{N-1} U^{(k-1)}(\mu_{k-1}, \mu_{k}) c_{\text{LO}}^{\alpha_{k}}(n_{k-1}) \right) \Gamma^{\mu}.$$
(B.120)

and

$$\begin{aligned} h_{1}^{\alpha\beta}(n_{l+1},n_{l}',n_{l+1}') &= d_{1}^{\alpha\beta}(n_{l+1},n_{l}',n_{l+1}')\tilde{\Theta}_{(\lambda^{l+1})^{2}}[n_{l+1}\cdot n_{l+1}']\tilde{\Theta}_{(\lambda^{l+1})^{2}}[n_{l+1}\cdot n_{l}']\tilde{\Theta}_{(\lambda^{l+1})^{2}}[n_{l+1}\cdot n_{l}']\tilde{\Theta}_$$

The coefficients $d_I^{\alpha\beta}(n_{l+1}, n'_l, n'_{l+1})$ are equal to the coefficients $d_I^{\alpha\beta}(n_2, n'_1, n'_2)$ defined in Eqs.(B.81, B.96, B.103) upon the substitution $(n_2, n'_1, n'_2) \rightarrow (n_{l+1}, n'_l, n'_{l+1})$ and $\lambda^2 \rightarrow \lambda^{2l}$. At NLO we calculated only the coefficients for the operators that interference with the LO operator, and we found only hard corrections. The considerations we have done above for the NLO hard corrections apply also to the NNLO hard correction, and we can write for the NNLO hard coefficient at SCET_N

$$C_{N,\text{NNLO}}^{(N-1)H} = C_{2,\text{NNLO}}^{(1)H.a}(n_1, n_1') \Big(\prod_{k=2}^{N-2} U^{(k-1)}(\mu_{k-1}, \mu_k) c_{\text{LO}}^{\alpha_k}(n_{k-1}) \Big)$$

$$+ C_{2,\text{NNLO}}^{(1)H.b}(n_2, n_1', n_2') \Big(\prod_{k=3}^{N-3} U^{(k-1)}(\mu_{k-1}, \mu_k) c_{\text{LO}}^{\alpha_k}(n_{k-1}) \Big),$$
(B.122)

where the coefficients $C_{2,\text{NNLO}}^{(1)H.a}(n_1,n_1')$ and $C_{2,\text{NNLO}}^{(1)H.b}(n_2,n_1',n_2')$ are defined in Eqs. (B.108).

Appendix C

$\mathcal{O}(\alpha_s^2)$ Correction to Splitting Function

In this appendix we rederive the (Abelian part of) the $\mathcal{O}(\alpha_s^2)$ correction to the $q \rightarrow qg$ splitting function, $P_{qq}^{(1)}$. This follows from obtaining the NLO (in λ , the SCET power counting parameter) correction to two-gluon emission. For comparison, we have chosen the classic result of Curci *et al.* [39]. The full expression for $P_{qq}^{(1)}$ involves many real and virtual contributions. We only calculated the $\sim C_F^2$ component of $P_{qq}^{(1)}$, and found that it agrees. Fortunately, [39] tabulated the different terms by graph, and so we were able to determine that we matched exactly for those calculations we did. Obtaining the full result would be a straightforward calculation of various one-loop renormalizations and non-Abelian emissions in SCET and combining them with our result. Ref. [39] splits the Abelian, two-gluon, real emission contributions to $P_{qq}^{(1)}$ into two topologically inequivalent diagrams, the box and crossed graphs, Fig. C-1. We calculated each of these individually.

The SCET₁ amplitude contains three graphs for two-gluon emission. These are shown in Fig. B-6, and we give the corresponding amplitudes in Eqs. (B.77). In order to obtain $P_{qq}^{(1)}$, we will need to square the amplitudes and integrate over phase space. Thus, we need to choose an explicit kinematics. We redraw, in figure (C-2), our vector labels for two-gluon emission. We choose a somewhat nonstandard assignment for our variables. This is to aid in the comparison with [39]. The final state parton



Figure C-1: Two distinct real emission contributions to $P_{qq}^{(1)}$ drawn as amplitudes squared. They are referred to as the box (**L**) and crossed (**R**) contributions.



Figure C-2: Kinematics for double gluon emission. This particular diagram corresponds to the "A" graph of figure (B-6).

shower occurs for timelike virtual particles, and momentum fractions decrease the farther we are from the initial hard scattering. By contrast, [39] considered a DIS-type interaction where the shower is spacelike. Since the radiation in that case comes from initial states, the momentum fractions decrease toward the hard interaction. Only at LO in α_s are the spacelike and timelike splitting functions equal. This is the Gribov-Lipatov relation [56]. At higher orders, this gets violated, but there is a straightforward conversion procedure, detailed in [39, 94]. We, however, choose our kinematics such that our variable relations are equivalent to those for a spacelike process. For example, $P_{qq}^{(1)}$ is a function of $x \equiv \bar{q}_0/\bar{q}_2$. In a spacelike process, $x \in [0, 1]$. Rather than convert our answer, we will also define x as above, even though this means for us $x \in [1, \infty)$. Other integration variables will have their ranges shifted so that they have the same relation with x as in DIS, and thus they

enter into our expression in the same way. Lastly, we do not do the phase space integration for q_2 . While this *is* necessary for the timelike splitting function, the analogous particle for a spacelike process is a fixed initial state. Thus, for comparison purposes, we can leave it undone. Our vectors are as follows (note that this is a different frame from the one used previously for matching):

$$q_{2} = \{p, 0, 0, p\}$$

$$k_{1} = \left\{-z_{1}p - \frac{k_{1\perp}^{2}}{4pz_{1}}, k_{1\perp}\cos(\phi_{1}), k_{1\perp}\sin(\phi_{1}), -z_{1}p + \frac{k_{1\perp}^{2}}{4pz_{1}}\right\}$$

$$k_{2} = \left\{-z_{2}p - \frac{k_{2\perp}^{2}}{4pz_{2}}, k_{2\perp}, 0, -z_{2}p + \frac{k_{2\perp}^{2}}{4pz_{2}}\right\}$$

$$q_{0} = \left\{xp + \frac{q_{0}^{2} + |\vec{\mathbf{k}_{1\perp}} + \vec{\mathbf{k}_{2\perp}}|^{2}}{4px}, \vec{\mathbf{k}_{1\perp}} + \vec{\mathbf{k}_{2\perp}}, xp - \frac{q_{0}^{2} + |\vec{\mathbf{k}_{1\perp}} + \vec{\mathbf{k}_{2\perp}}|^{2}}{4px}\right\}. (C.1)$$

Before proceeding, we wish to note some things about our assignment. First of all, while it is redundant to include $q_0 = k_1 + k_2 + q_2$, we will integrate over d^4q_0 and wanted to present our parametrization. We see that $x = 1 - z_1 - z_2$. This is for consistency with the spacelike case, but here, $z_1, z_2 \in (-\infty, 0]$, hence the minus signs in k_1 and k_2 . Additionally, only the relative azimuthal angle between k_1 and k_2 is physical. Thus, to simplify our formulas, we fix k_2 in the x - z plane. A trivial factor of 2π will enter from the k_2 phase space once we integrate.

As a last step before squaring and integrating, we will introduce our measure and integral parametrization. While one could integrate the full final state phase space, we found such an approach prohibitively difficult. Instead, we can exploit the factorization of the the cross-Subsection into a hard interaction and a jet-function for an appropriate definition of each. We need only integrate the latter, and it will remain independent of the details of the former. We split up the cross-Subsection as follows ($d \equiv 4 + \epsilon$):

$$\sigma = \int \frac{dx}{x} \mathcal{H}(x) \left(\mathcal{J}_{\text{LO}}(x, q^2) + \mathcal{J}_{J, \text{NLO}}(x, q^2) + \ldots \right) q_{B, F}(x), \quad (C.2)$$

where

$$\mathcal{J}_{\text{LO}} = \sum_{i} \frac{2}{\bar{q}_{i}} \int \prod_{j=1}^{i} \frac{d^{d-1}k_{j}}{z_{j}} d^{d}q_{0} dq^{2} \operatorname{PP} \left[C_{N,\text{LO}}^{(N)} |\langle 0|\mathcal{O}_{N}^{(N)}|q(Ng)\bar{q}\rangle |^{2} \right] \quad (C.3)$$

$$\times \delta(x - \bar{q}_{0}/\bar{q}_{i})\delta(q^{2} - (q_{i} + \sum_{j=1}^{i} k_{j})^{2}) (2\pi)^{m} \delta^{(m)}(q_{0} - \sum_{j=1}^{i} k_{j})),$$

$$\mathcal{J}_{J,\text{NLO}} = \sum_{i} \frac{2}{\bar{q}_{i}} \int \prod_{j=1}^{i} \frac{d^{d-1}k_{j}}{z_{j}} d^{d}q_{0} dq^{2} \operatorname{PP} \left[C_{N,\text{NLO}}^{(N)J} |\langle 0|\mathcal{O}_{N}^{(N)}|q(Ng)\bar{q}\rangle |^{2} \right] \quad (C.4)$$

$$\times \delta(x - \bar{q}_{0}/\bar{q}_{i})\delta(q^{2} - (q_{i} + \sum_{j=1}^{i} k_{j})^{2}) (2\pi)^{m} \delta^{(m)}(q_{0} - \sum_{j=1}^{i} k_{j})),$$

and the q_i phase space and spin-sum gets moved into \mathcal{H} . We define z_j analogously to Eqs. (C.1) and (C.1). The function $q_{B,F}(x)$ is the bare fragmentation function which determines how the partons arrange themselves into hadrons. We call it 'bare' as it contains singularities necessary to cancel the collinear divergences from parton splitting to give a finite observable. The \mathcal{J} terms defined above consist of only the pole portions of the corresponding operator expectation values. The reason we extract only the pole terms is that these are precisely what give the expression for $P_{qq}^{(1)}$.

By fixing the virtuality of $q_0^2 \equiv q^2$, we can obtain an expression without having to know its exact limits, which will depend on the details of the hard scattering. For $P_{qq}^{(1)}$, one only needs to calculate one-loop corrections to single emission and double emission, and we now specialize to the latter case. We perform the *d*-dimensional δ -function over $d^d q_0$ and rewrite the integral in terms of $k_{1\perp}$ and $k_{2\perp}$ dependent functions with $z_{1,2}$ -dependent coefficients. Using the parametrization of [45], we can write:

$$\mathcal{J}_{J,\text{NLO}}^{q \to qgg} = \frac{1}{(16\pi^2)^2} \int dq^2 \frac{dz_1}{z_1} \frac{dz_2}{z_2} \frac{d^{d-2}\mathbf{k_{1\perp}}}{\pi} \frac{d^{d-2}\mathbf{k_{2\perp}}}{\pi} \,\delta(1-x-z_1-z_2) \\ \times \delta\left(q^2 - (a_1\,\mathbf{k_{1\perp}}^2 + a_2\,\mathbf{k_{2\perp}}^2 - \mathbf{k_{1\perp}}\cdot\mathbf{k_{2\perp}})\right) \\ \times \left(A(z_1, z_2) + B(z_1, z_2) \frac{\mathbf{k_{1\perp}}\cdot\mathbf{k_{2\perp}}}{\mathbf{k_{1\perp}}^2} + C(z_1, z_2) \frac{\mathbf{k_{1\perp}}\cdot\mathbf{k_{2\perp}}}{\mathbf{k_{2\perp}}^2} \right) \\ + D(z_1, z_2) \frac{(\mathbf{k_{1\perp}}\cdot\mathbf{k_{2\perp}})^2}{\mathbf{k_{1\perp}}^2\mathbf{k_{2\perp}}^2} + E(z_1, z_2) \frac{\mathbf{k_{1\perp}}^2}{\mathbf{k_{2\perp}}^2} + F(z_1, z_2) \frac{\mathbf{k_{2\perp}}^2}{\mathbf{k_{1\perp}}^2}\right) \frac{1}{q^4} \quad (C.5)$$

$$- [LO],$$
 (C.6)

where

$$a1 = -\frac{1-z_2}{z_1}, \ a2 = -\frac{1-z_1}{z_2}.$$
 (C.7)

The functions A, B, C, D are defined in [45], and their corresponding $\mathbf{k}_{i\perp}$ integrals are finite. We can check the intermediate step of their integration with [45]. The terms in our $q^2 \delta$ -function have a relative sign compared to theirs, as our $q^2 > 0$. We found it simplest to calculate in SCET₃ where only $C_{3,\text{NLO}}^{(2)J,1}$ in Eq. (B.110) contributes. This corresponds to taking limits such that only its Θ -function gives support, while the jet-structure coeffecients are zero. By doing this, we wind up integrating over the strongly-ordered region of phase space, but since we know the LO contribution from $C_3^{(2)}$, we can formulate the subtraction in Eq. (C.6) at the operator level. There is just one subtlety, which we describe below, having to do with the appropriate treatment of dim reg in the standard scheme for calculating $P_{qq}^{(1)}$.

As a computational aside, we found it easiest to pass to a change of variables: $(u \equiv k_{1\perp}k_{2\perp}, w \equiv k_{1\perp}/k_{2\perp})$. Then the δ -function just enforces:

$$u = u_0 \equiv \frac{q^2 w}{a_1 w^2 + a_2 - 2 w \cos(\phi_1)}.$$
 (C.8)

Performing all but the dz_i integrals in \mathcal{J} , we get Table C.1, which corresponds to [45]'s Table 5. We thus reproduce the earlier result except for what we believe is a

Function of $\mathbf{k}_{\mathbf{i}\perp}$	Contribution to $\mathcal J$ multiplying	
in integrand of equation (C.6)	$\frac{q^2}{(16\pi^2)^2 x} \int dz_1 dz_2 \delta(1-z_1-z_2-x)$	
1	$A(z_1,z_2)$	
$rac{\mathbf{k_1}_{\perp}\cdot\mathbf{k_2}_{\perp}}{\mathbf{k_1}_{\perp}^2}$	$-rac{z_2}{1-z_1}B(z_1,z_2)$	
$rac{\mathbf{k_1}_{\perp}\cdot\mathbf{k_2}_{\perp}}{\mathbf{k_2}_{\perp}^2}$	$-rac{z_1}{1-z_2}C(z_1,z_2)$	
$rac{(\mathbf{k_1}_{\perp}\cdot\mathbf{k_2}_{\perp})^2}{\mathbf{k_1}_{\perp}^2\mathbf{k_2}_{\perp}^2}$	$\left(1 + \frac{x}{2z_1 z_2} \ln\left[\frac{x}{(1-z_1)(1-z_2)}\right]\right) D(z_1, z_2)$	

Table C.1: Purely finite contributions to ${\mathcal J}$

typo on their part swapping the B and C entries.

The E, F functions multiply integrals that lead to single ϵ poles after the $dk_{i\perp}$ integral (and double poles after integrating q^2), and so we must be more careful in treating them. These double poles correspond to the LO contribution, which we are explicitly subtracting as it does not contribute to $P_{qq}^{(1)}$. We discuss it in detail below Eq. (C.17). For now we concentrateon the divergent integrals multiplying E and F. When we did our computations for Table (C.1), we were helped by the finiteness of the expressions under the $dk_{i\perp}$ integration. We could thus take $\epsilon \to 0$ for these terms, which greatly simplifies their integrals. By contrast, we will need to keep the ϵ -dependence of the E, F terms, which results in an intractable computation. To get around this, one can introduce subtraction functions, which simply reproduce the ϵ poles (these are merely a computational aid and are not related to the subtraction of LO). We will need to take care that they do not remove any finite pieces. Secondly, since their full contribution to \mathcal{J} is $\propto 1/\epsilon^2$, we will need to include for E and Fany terms $\propto \epsilon$ that multiply $\frac{\mathbf{k}_1^2}{\mathbf{k}_2^2}$ or $\frac{\mathbf{k}_1^2}{\mathbf{k}_2^2}$. These arise from doing Dirac algebra in m-dimensions.

To do the integrals in \mathcal{J} multiplying and E and F, we will change variables to u, w, and perform the u integration as well as the trivial ϕ_2 azimuthal one. We get for this contribution to \mathcal{J} :

$$\mathcal{J}|_{E,F} = \frac{1}{(16\pi^2)^2} \frac{2}{\pi} \int dq^2 \frac{dz_1}{z_1} \frac{dz_2}{z_2} d\phi_1 \, dw \, \delta(1 - x - z_1 - z_2) \\ \times \left(\frac{w \, u_0^{2+\epsilon}}{2 \, q^2} E(z_1, z_2) + \frac{u_0^{2+\epsilon}}{2 \, w \, q^2} F(z_1, z_2)\right) \frac{1}{q^4}, \tag{C.9}$$

where u_0 is defined by equation (C.8). Unfortunately, we could not manage to perform the w and ϕ_1 integrals for the functions multiplying E, F and obtain a result in terms of elementary functions. However, we only need the leading poles in ϵ , and so we will define subtraction functions to reproduce the poles of $\frac{w u_0^{2+\epsilon}}{2q^2}$, $\frac{u_0^{2+\epsilon}}{2wq^2}$, respectively:

$$S_E = \frac{q^2}{2a_1^2} \frac{w^{-\epsilon}}{(w+1)}$$

$$S_F = \frac{q^2}{2a_2^2} \frac{w^{\epsilon}}{(w+w^2)}.$$
(C.10)

Integrating these in w gives us a pure $1/\epsilon$ term. Subtracting them from the functions in equation (C.9):

$$\mathcal{A}_{E} \equiv \frac{w \, u_{0}^{2+\epsilon}}{2 \, q^{2}} = \frac{q^{2} \, w^{3}}{2(a_{2}+a_{1}w^{2}-2w\cos(\phi_{1}))^{2}} \left(\frac{w \, q^{2}}{a_{2}+a_{1}w^{2}-2w\cos(\phi_{1})}\right)^{\epsilon} (C.11)$$
$$\mathcal{A}_{F} \equiv \frac{u_{0}^{2+\epsilon}}{2 \, w \, q^{2}} = \frac{q^{2}}{2 \, w(a_{2}+a_{1}w^{2}-2w\cos(\phi_{1}))^{2}} \left(\frac{w \, q^{2}}{a_{2}+a_{1}w^{2}-2w\cos(\phi_{1})}\right)^{\epsilon},$$

leads to finite integrals, allowing us to pass to the $\epsilon \to 0$ limit prior to integration making the calculation tractable. Note that the subtraction terms are purely computational aides and are different from the factorization scheme mentioned above, which is designed to absorb a collinear divergence into a fragmentation function. After integrating w and ϕ_1 , we want the $\epsilon^{-1,0}$ pieces as these turn into the single and double poles upon doing the q^2 integral and contribute to \mathcal{J} . The ϵ^0 piece, has one contribution besides that from $(\mathcal{A}_{E,F} - \mathcal{S}_{E,F})|_{\epsilon=0}$ ($\mathcal{S}_{E,F}$ contributes a pure $1/\epsilon$ pole). Our w integration goes from 0 to ∞ , and we obtained $\mathcal{S}_{E,F}$ by expanding $\mathcal{A}_{E,F}$ in the appropriate $w \to 0, \infty$ limit to pick up the pole, while carefully regulating the other integration limit so as not to contribute its own spurious divergence or any subleading terms. However, we see that in equation (C.12), taking these limits actually results in factors $(a_1w)^{-\epsilon}$ and $(w/a_2)^{\epsilon}$. Expanding the $a_i^{\pm\epsilon}$ to LO in ϵ does not affect $\mathcal{S}_{E,F}$. Nonetheless, since the subtraction functions have $1/\epsilon$ poles, including the NLO piece will yield an ϵ^0 contribution. We do not have it in $\mathcal{A}_{E,F}|_{\epsilon=0}$ since that sends $u_0^{\epsilon} \rightarrow$ 1. Thus, we have the following addition to the contributions from the integration of $\mathcal{J}|_{E,F}$:

$$\mathcal{B}_E = -\epsilon \ln(a_1) \frac{q^2}{2 a_1^2} \frac{w^{-\epsilon}}{(w+1)} \tag{C.12}$$

$$\mathcal{B}_F = -\epsilon \ln(a_2) \frac{q^2}{2 a_2^2} \frac{w^\epsilon}{(w+w^2)}.$$
(C.13)

In the end, our $\epsilon^{-1,0}$ contributions after w and ϕ_1 integration come from: $S_{E,F} + \mathcal{B}_{E,F} + (\mathcal{A}_{E,F} - \mathcal{S}_{E,F})|_{\epsilon=0}$. For integrating the first two terms, we leave the full ϵ dependence as this is tractable. Once we have accounted for where it is needed, we can set $\epsilon = 0$ in the last term to pick up the remaining finite result. Collecting

Eunction of k :	Contribution to 7 multiplying
$\mathbf{I} = \mathbf{I} = $	a^2 () a^2 ()
in integrand of equation $(C.6)$	$\frac{q}{(16\pi^2)^2 x} \int dz_1 dz_2 \delta(1-z_1-z_2-x)$
$\frac{\mathbf{k_1}_{\perp}^2}{\mathbf{k_2}_{\perp}^2}$	$\left[\frac{2xz_1}{(1-z_2)^2z_2\epsilon}\left(1-\epsilon\ln\left[-\frac{1-z_2}{z_1}\right]\right)\right]$
	$\left[+ \frac{z_1}{z_2(1-z_2)^2} \left(z_1 z_2 + x \left(\ln \left[\frac{z_2(1-z_2)}{z_1 x} \right] - 1 \right) \right) \right] E(z_1, z_2)$
$\frac{\mathbf{k_2}_{\perp}^2}{\mathbf{k_1}_{\perp}^2}$	$\left[\frac{2xz_2}{(1-z_1)^2 z_1\epsilon}\left(1-\epsilon\ln\left[-\frac{1-z_1}{z_2}\right]\right)\right]$
	$\Big + \frac{z_2}{z_1(1-z_1)^2} \left(z_1 z_2 + x \left(\ln \left[\frac{z_1(1-z_1)}{z_2 x} \right] - 1 \right) \right) \Big] F(z_1, z_2)$

all terms, we can obtain the counterpart to Table C.1 for E, F. The origin of the contributions in terms of S, B, and S - A should be straightforward.

Table C.2: Contributions to $\mathcal{J}|_{E,F}$

Having set up this much of the integration, we can take the amplitude squared from the process of interest and decompose it in terms of the $A(z_1, z_2)$, $B(z_1, z_2)$, etc. basis. We then simply have to read off the results from Tables C.1 and C.2, and perform the $z_{1,2}$ integrals. One of these is made trivial by the remaining x-dependent δ -function. As mentioned at the beginning of this Appendix, Ref. [39] recognizes two topologically distinct contributions, which we shall refer to as box and crossed, Fig. C-1, because of their appearance as cut two-loop diagrams. We can identify them in our calculation by their color structure (C_F^2 and $C_F^2 - \frac{1}{2}C_FC_A$, respectively). In fact, we can already calculate the entire crossed contribution as it only involves terms from Table C.1, having no double pole contribution to \mathcal{J} . Determining the box graph, however, involves treating the LO subtraction properly.

As this subtraction is one of the more subtle points of the computation, we will present it in some detail. Its handling is tied up with what one means precisely by a "subleading splitting function." At LO in α_s , the definition is clear. The same splitting function that gives us the probability for a $1 \rightarrow 2$ radiation also determines the running with scale of parton densities:

$$Q^2 \frac{\partial}{\partial Q^2} f(x, Q^2) = \int_x^1 \frac{dz}{z} P_{qq}\left(\frac{x}{z}, \alpha_s(Q^2)\right) f(z, Q^2), \qquad (C.14)$$

where the $\mathcal{O}(\alpha_s)$ part of P_{qq} , $P_{qq}^{(0)}$ is given by Eq. (4.3). To determine $P_{qq}^{(1)}$, we have had to calculate a 1 \rightarrow 3 splitting, thus the probabilistic interpretation in terms of radiation is nontrivial as it involves a mix of $1 \rightarrow 2$ and $1 \rightarrow 3$ processes. At the level of Eq. (C.14) though, we see that we are just correcting PDF (or fragmentation function) evolution. In addition to the real-emission calculation that we are pursuing, one can alternatively determine P_{qq} from the anomalous dimension of certain twist-2 operators [50, 51]. Ref. [39] made a comparison to this approach and found agreement to $\mathcal{O}(\alpha_s^2)$. Since $P_{qq}^{(1)}$ is thus a two-loop object, it has the scheme dependence one would expect at this order, and so we need to make sure that we compute in the same one. In SCET, one could attempt the same cross-check from a straightforward two-loop calculation after fixing the renormalization scheme.

We will now show how to subtract the LO portion in the calculation of $\mathcal{J}_{J,\text{NLO}}^{q \to qgg}$ We get a double collinear pole associated with the strongly-ordered emission of two gluons. We want to write this as removing the emission from our LO operator, $C_{3,LO}^{(2)}\mathcal{O}_3^{(2)}$. As with any subtraction scheme, while the pole is unambiguous, we need to make sure to remove the appropriate finite pieces. We note that $c_{\rm LO}^{\alpha}$ defined by Eq. (4.47) contains NLO pieces (in $SCET_3$ power counting) which come from the offshellness of the intermediate quark. It is true that the LO replacement rule, Eq. (4.46), gives only the splitting function times the logarithmic, collinear divergence. Nonetheless, the Wilson coefficients given by Eq. (4.53) for offshell quarks have additional terms. From the point of view of amplitdue matching, this poses no problem. However, if we want to copy [39]'s scheme, then we can only subtract poles associated with the pure LO result. As an operator subtraction in $SCET_3$, this means we need to change $\mathcal{O}_3^{(2)}$. In order to recover the correct splitting function with no NLO contribution, we will need to project the offshell quark momentum to an onshell one with the same \bar{p} fraction. This alone, though, does not specify the spatial orientation of the vector and will not necessarily kill the subleading terms. To do that, we write the replacement rule, but in the limit that the offshell quark's daughters are exactly collinear with it. Equivalently, if we are in the frame determined by $\bar{n} = \{1, 0, 0, -1\}$, we can project the quark momentum along $n = \{1, 0, 0, 1\}$, *i.e.* $q_i \rightarrow \frac{\bar{q}_i}{2}n = q'_i$. Since the replacement rule also makes reference to the quark's parent's momentum, we also need to project it to what it would be if it had emitted an onshell quark with q'_i .

Thus, $q_{i-1} \rightarrow k_i + q'_i = q'_{i-1}$. In the end, this changes our replacement rule coefficient for the j^{th} quark to:

$$c_{\rm LO}^{\prime \alpha_{j+1}} = \frac{\bar{q}_j}{{q'_j}^2} \left(n_{q'_j}^{\alpha} + \frac{{\not\!\!\!\!\!/}_{j+1\perp_{n_{q'_j}}}^{\gamma} \gamma_{\perp j+1}^{\alpha}}{\bar{q}_{j+1}} \right) \frac{{\not\!\!\!/} {n_{q_j}}}{4} \,. \tag{C.15}$$

This changes the expression for $C_{3,LO}^{(2)}\mathcal{O}_3^{(2)}$ to involve c'_{LO} instead of c'_{LO} (cf. Eq. 4.53)

After the dq^2 integration, the $1/\epsilon$ term in $\mathcal{J}_{J,\text{NLO}}^{q \to qgg}$ will allow us to read off $P_{qq}^{(1)}$. The subtraction operator lives in SCET₃ as both gluons have their own direction and Wilson coefficient in that theory. We need it because our NLO operator, $C_{3,\text{NLO}}^{(2)J,1}\mathcal{O}_{3}^{(2)}$ is supported over all of phase space, and there contains LO portions. We therefore have

$$\mathcal{J}_{J,\,\rm NLO}^{q \to qgg} = \int d\Pi_{k_1,\,k_2,\,q_0} \left[|C_{3,\,\rm NLO}^{(2)J,\,1}\langle 0|\mathcal{O}_3^{(2)}|qgg\bar{q}\rangle|^2 - \left(|c_{\rm LO}^{\prime \alpha_1} c_{\rm LO}^{\prime \alpha_2}\langle 0|\mathcal{O}_3^{(2)}\Gamma^{\mu}|qgg\bar{q}\rangle|^2 \right)_{\overline{\rm MS}} \right]$$
(C.16)

The $\overline{\text{MS}}$ indicates that we are only subtracting pole parts of the LO contribution, with no finite pieces. However, there is still an amibiguity over *which* pole parts we subtract, as even though the LO contribution has a double pole from its two collinear divergences, we are at some liberty to decide which single pole parts we remove as well. As we expect, this subtraction operator squared takes the form of a convolution of two splitting functions:

$$\int d\Pi \left(|c_{\rm LO}^{\prime \alpha_1} c_{\rm LO}^{\prime \alpha_2} \langle 0 | \mathcal{O}_3^{(2)} \Gamma^{\mu} | q \bar{q} g g \rangle |^2 \right)_{\overline{\rm MS}} = 2 \int dq^2 \, dy \, x \, p \, (1-y)^{\frac{\epsilon}{2}} (q^2)^{-1+\epsilon/2} \frac{\alpha^2}{2\pi^2} \\ \times \frac{1}{y} P_{d,qq}^{(0)}(y) \frac{P_{4,qq}^{(0)}(x/y)}{\epsilon}.$$
(C.17)

What may seem surprising is that different splitting functions live in different dimensions. The reason for this particular scheme for regluating phase space has to do with the alternate, two-loop method for calculating $P_{qq}^{(1)}$, which was the original approach. For that result, in $\overline{\text{MS}}$ we would subtract a simple pole counterterm, regulate the loop integral in *d*-dimensions, and leave external particles in 4d. Since the phase space integrals are related to loops by cuts, we see above that our y-integral is, in fact, in d-dimensions, but the splitting involving two external particles is left simply in four.

Looking at the SCET₁ diagrams for the process (Fig. B-6), the amplitude $c'_{\rm LO}{}^{\alpha_1}c'_{\rm LO}{}^{\alpha_2}\langle 0|\mathcal{O}_3^{(2)}\Gamma^{\mu}|qgg\bar{q}\rangle$ comes from a subset of diagrams A^2 and B^2 . The expression for subtraction is thus:

$$\int d\Pi \left(|c_{\text{LO}}^{\prime \alpha_{1}} c_{\text{LO}}^{\prime \alpha_{2}} \langle 0 | \mathcal{O}_{\text{S}_{3}}^{(2)} \Gamma^{\mu} | q \bar{q} g g \rangle |^{2} \right)_{\overline{\text{MS}}} = \int dq^{2} dz_{1} x \, p \, \left(\frac{z_{1}}{x + z_{1}} \right)^{\frac{\epsilon}{2}} \\ \times \frac{(q^{2})^{-1 + \epsilon/2}}{\epsilon} \frac{\alpha^{2}}{2\pi^{2}} \frac{1}{x + z_{1}} \left[\frac{1 + \left(\frac{x}{x + z_{1}} \right)^{2}}{\frac{x}{x + z_{1}} - 1} + \frac{\epsilon}{2} \left(1 - \frac{x}{x + z_{1}} \right) \right] \left(\frac{1 + (x + z_{1})^{2}}{x + z_{1} - 1} \right) \\ + z_{1} \leftrightarrow z_{2}.$$
(C.18)

We can note several things about this expression. For concreteness, we discuss the z_1 -dependent term corresponding to graph A^2 , Fig. B-6. The \bar{p} fraction of q_0 relative to q_1 is $x/(x+z_1)$, and that of q_1 to q_2 is $x+z_1$, in terms of the variables in Eq. (C.17), $y' = x/(x+z_1)$. Performing the integrals leads to double and single poles. For later use, we write down the result of doing the dq^2 , dz_i integrals, where one of latter is trivial since we have $\delta(1-x-z_1-z_2)$ sitting inside Π (*cf.* Eq. C.6).

$$\int d\Pi \left(|c_{\text{LO}}^{\prime} \alpha^{\prime} c_{\text{LO}}^{\prime} \alpha^{\prime} \langle 0 | \mathcal{O}_{3}^{(2)} \Gamma^{\mu} | q \bar{q} g g \rangle |^{2} \right)_{\overline{\text{MS}}} = 2x p \frac{\alpha^{2}}{2\pi^{2}} \left(\frac{1}{\epsilon^{2}} \frac{2}{x-1} \left[\left(-2 \left(2 \left(x^{2}+1 \right) \log(\lambda) + (x-1)^{2} \right) + 4 \left(x^{2}+1 \right) \log(x-1) - \left(x^{2}-1 \right) \log(x) \right) \right] + 4 \left(x^{2}+1 \right) \log(x-1) - \left(x^{2}-1 \right) \log(x) \right) \right] + \frac{2}{\epsilon} \frac{1}{(x-1)^{2}x} \left[2x(x-1) \left(2 \left(x^{2}+1 \right) \left(\text{Li}_{2}(1-x) - \text{Li}_{2} \left(\frac{1}{x} \right) \right) + (x^{2}-1) \text{Li}_{2} \left(\frac{x-1}{x} \right) \right) \right) + x \left(-2 \left(x^{2}+1 \right) \left(x-1 \right) \log^{2}(\lambda) + 4 \left(\left(x^{2}+1 \right) \left(\log \left(\frac{x-1}{x} \right) + x \log \left(\frac{x}{x-1} \right) \right) \right) - (x-1)^{3} \log(\lambda) - \left(3x^{2}+5 \right) (x-1) \log^{2}(x) + 2(x-1)^{2} \log(x) \right) + 6x \left(x^{2}+1 \right) (x-1) \log^{2}(x-1) - 2x(x-1)^{3} - 2x(x+1)(x-1)^{2} \log(x-1) \log(x)((x-1)^{2}x) \right] \right) \tag{C.19}$$

where we have done the dz_i integrals between $1 - x + \lambda$ and $-\lambda$ to regulate soft

divergences. All λ -dependence cancels out of the final answer, which gives us a consistency check on the scheme.

Before comparing $P_{qq}^{(1)}$, we can check our setup with $P_{qq}^{(0)}$, by looking at the $\mathcal{O}(\alpha_s)$ contribution to \mathcal{J}_{LO} We see that [39] gets the following contribution:

$$P_{qq}^{(0)} = \left(\frac{\alpha_s}{2\pi}\right) \frac{2}{\epsilon} \frac{1+x^2}{1-x}.$$
 (C.20)

Calculating in $SCET_1$, we get the following amplitude squared:

$$A_{q \to qg} = \frac{\bar{q}_0}{q_0^2} \left(\frac{2n \cdot k_1}{\bar{k}_1} + \frac{2k_{1\perp} \cdot q_{1\perp}}{\bar{q}_1 \bar{k}_1} - \frac{q_{1\perp}^2}{\bar{q}_1^2} \right) \operatorname{Tr}[\not n \Omega \Omega^{\dagger}]$$
(C.21)

With our definition of \mathcal{J}_{LO} in Eq. (C.6), we get:

$$P_{qq}^{(0)} = \left(\frac{\alpha_s}{2\pi}\right) \frac{2}{\epsilon} \frac{1+x^2}{x-1}.$$
 (C.22)

The overall minus sign between eqns. (C.20) and (C.22) is due to the difference between the spacelike and timelike processes. It arises in the dz_i integral. Even though the z_i dependence is the same in the two calculations, and the integration limits are the same, 0 and 1 - x. For us, 1 - x < 0, but in Ref. [39], it is positive.

We will compare the different contributions to double emission separately. In SCET₁, the C graph in Fig. B-6 will give box and crossed terms when interfered with itself and the A and B ones. We identify the crossed contribution by inserting the color structure and taking those terms proportional to $C_F^2 - \frac{1}{2}C_F C_A$. As mentioned above, it only contains the integrals in Table C.1. In terms of its notation, we have: The

Function defined in Eq. (C.6)	Value in crossed diagram
$A(z_1,z_2)$	$-rac{16x \left(x^2+x z_1+(z_1-1) z_1+1 ight)}{z_1 (x+z_1-1)}$
$B(z_1, z_2)$	$rac{8 \left(x^2 (z_1 - 2) - x z_1 + z_1 - 1 ight)}{x + z_1 - 1}$
$C(z_1, z_2)$	$\frac{8(x(x^2+(x-1)z_1+2)+z_1)}{z_1}$
$D(z_1, z_2)$	$16(x^2+1)$

Table C.3: Contributions to crossed amplitude squared diagram

box contribution additionally contains the functions in Table C.2, though we are only
Function defined in Eq. (C.6)	Value in box diagram
$A(z_1, z_2)$	$12x^2 + 8xz_1 + 8(z_1 - 1)z_1 + 12$
$B(z_1, z_2)$	$\frac{8(z_1-1)\left(x^2+(z_1-2)z_1+2\right)}{x+z_1-1}$
$C(z_1, z_2)$	$\frac{8(x+z_1)\left(2x^2+2xz_1+z_1^2+1\right)}{z_1}$
$D(z_1, z_2)$	0
$E(z_1,z_2)$	$4\left[\frac{\left(2x^{4}+6x^{3}z_{1}+x^{2}\left(7z_{1}^{2}+2\right)+2x\left(2z_{1}^{3}+z_{1}\right)+z_{1}^{4}+z_{1}^{2}\right)}{z_{1}^{2}}\right] +4\epsilon\left[\frac{\left(x^{2}(x+z_{1}-1)^{2}+z_{1}^{2}\left((x+z_{1}-1)^{2}+x+z_{1}\right)+xz_{1}(x+z_{1}-1)^{2}\right)}{z_{1}^{2}}\right]$
$F(z_1, z_2)$	$4\left[\frac{\left(z_{1}^{2}-2z_{1}+2\right)\left(x^{2}+(z_{1}-1)^{2}\right)}{(x+z_{1}-1)^{2}}\right] + 4\epsilon\left[\frac{\left(x^{2}((z_{1}-1)z_{1}+1)+x(z_{1}-1)((z_{1}-2)z_{1}+2)+(z_{1}-1)^{2}((z_{1}-1)z_{1}+1)\right)}{(x+z_{1}-1)^{2}}\right]$

interested in the finite parts. Their \boldsymbol{z}_i dependence is: For the crossed contribution,

Table C.4: Contributions to box amplitude squared diagram

we perform the multiplication in Table C.1 with the functions defined in Table C.3 and integrate dz_1 , having already done the trivial dz_2 integral. We again use a cutoff to avoid soft divergences, thus its range is between $1 - x + \lambda$ and $-\lambda$. In the end, we obtain:

$$P_{qq\,\text{crossed}}^{(1)} = \left(\frac{\alpha_s}{2\pi}\right)^2 \left[\left(\frac{1+x^2}{x-1}\right) \left(4\ln(x-1) - \ln^2(x) - \ln(\lambda)\right) - 2(x+1)\ln(x)\right].$$
(C.23)

The λ -dependent pieces will cancel against those from the box contribution. The other terms agree with [39] up to the previously discussed minus sign, and wherever $\ln(1-x)$ appears in the spacelike calculation, we get $\ln(x-1)$. Since our integrand and integration region are real, the imaginary pieces generated by $\ln(1-x)$ when making x > 1 all must cancel.

The box calculation proceeds similarly except that we also include the terms proportional to $E(z_1, z_2)$ and $F(z_1, z_2)$ and perform the internal collinear subtraction, which changes the overall single pole term. Doing all this, we get:

$$P_{qq\,box}^{(1)} = \left(\frac{\alpha_s}{2\pi}\right)^2 \left[\left(\frac{1+x^2}{x-1}\right) \left(\ln(\lambda) - \ln(x-1)\right) + 2(2x-1)\ln(x) \right].$$
(C.24)

The soft divergent pieces cancel against the crossed contribution, and once again we agree with [39] up to an overall sign, and the continuation $\ln(1-x) \rightarrow \ln(x-1)$.

In addition to these real emission contributions to the C_F^2 portion of $P_{qq}^{(1)}$, there are also single-emission, one-loop diagrams, Fig. C-3. We can account for their con-



Figure C-3: Single emission, one-loop contributions to $P_{qq}^{(1)}$.

tributions in SCET easily. We have already derived the the same expression for single emission (Eqs. C.20 and C.22). Furthermore, both the quark wavefuncton renormalization and the vertex renormalization are the same in SCET as in QCD [10]. Thus, we recover the entire, gauge-invarant, $\propto \alpha_s^2 C_F^2$ contribution to the splitting function [39],

$$P_{qq\,\text{abelian}}^{(1)} = C_F^2 \frac{\alpha_s^2}{2\pi} \left[(1-x)\ln(x) - \frac{3}{2}\frac{1+x^2}{1-x}\ln(x) - 2\frac{1+x^2}{1-x}\ln(x)\ln(1-x) - \frac{1}{2}(1+x)\ln^2(x) - 5(1-x) - \frac{5}{2}(1+x)\ln(x) \right], \quad (C.25)$$

where we have written it with its usual sign conventions for spacelike evolution.

Bibliography

- Valentin Ahrens, Thomas Becher, Matthias Neubert, and Li Lin Yang. Origin of the Large Perturbative Corrections to Higgs Production at Hadron Colliders. *Phys. Rev.*, D79:033013, 2009.
- [2] Charalampos Anastasiou, Lance J. Dixon, Kirill Melnikov, and Frank Petriello. High precision QCD at hadron colliders: Electroweak gauge boson rapidity distributions at NNLO. *Phys. Rev. D*, 69:094008, 2004.
- [3] Charalampos Anastasiou, Kirill Melnikov, and Frank Petriello. Fully differential Higgs boson production and the di- photon signal through next-to-next-toleading order. Nucl. Phys., B724:197–246, 2005.
- [4] Christian M. Arnesen, Joydip Kundu, and Iain W. Stewart. Constraint equations for heavy-to-light currents in scet. *Phys. Rev. D*, 72:114002, 2005.
- [5] M. Bahr et al. Herwig++ Physics and Manual. Eur. Phys. J. C, 58:639–707, 2008.
- [6] Jochen Bartels, Claas Bontus, and Hubert Spiesberger. Factorization of twistfour gluon operator contributions. 1999.
- [7] Sergei Bashinsky and R. L. Jaffe. Quark and gluon orbital angular momentum and spin in hard processes. *Nucl. Phys.*, B536:303–317, 1998.
- [8] Christian W. Bauer, Sean Fleming, and Michael E. Luke. Summing Sudakov logarithms in $B \to X_s \gamma$ in effective field theory. *Phys. Rev. D*, 63:014006, 2000.
- [9] Christian W. Bauer, Sean Fleming, Dan Pirjol, Ira Z. Rothstein, and Iain W. Stewart. Hard scattering factorization from effective field theory. *Phys. Rev. D*, 66:014017, 2002.
- [10] Christian W. Bauer, Sean Fleming, Dan Pirjol, and Iain W. Stewart. An effective field theory for collinear and soft gluons: Heavy to light decays. *Phys. Rev. D*, 63:114020, 2001.
- [11] Christian W. Bauer, Andrew Hornig, and Frank J. Tackmann. Factorization for generic jet production. *Phys. Rev. D*, 79:114013, 2009.

- [12] Christian W. Bauer, Christopher Lee, Aneesh V. Manohar, and Mark B. Wise. Enhanced nonperturbative effects in Z decays to hadrons. *Phys. Rev. D*, 70:034014, 2004.
- [13] Christian W. Bauer, Aneesh V. Manohar, and Mark B. Wise. Enhanced nonperturbative effects in jet distributions. *Phys. Rev. Lett.*, 91:122001, 2003.
- [14] Christian W. Bauer, Dan Pirjol, and Iain W. Stewart. Soft-collinear factorization in effective field theory. *Phys. Rev. D*, 65:054022, 2002.
- [15] Christian W. Bauer, Dan Pirjol, and Iain W. Stewart. On power suppressed operators and gauge invariance in SCET. *Phys. Rev.*, D68:034021, 2003.
- [16] Christian W. Bauer and Matthew D. Schwartz. Improving jet distributions with effective field theory. *Phys. Rev. Lett.*, 97:142001, 2006.
- [17] Christian W. Bauer and Matthew D. Schwartz. Event generation from effective field theory. *Phys. Rev.*, D76:074004, 2007.
- [18] Christian W. Bauer and Iain W. Stewart. Invariant operators in collinear effective theory. *Phys. Lett. B*, 516:134–142, 2001.
- [19] Christian W. Bauer, Frank J. Tackmann, and Jesse Thaler. talk by C.W. Bauer at SCET 2010, http://wwwth.mppmu.mpg.de/members/scet2010/.
- [20] Christian W. Bauer, Frank J. Tackmann, and Jesse Thaler. GenEvA. I. A New framework for event generation. *JHEP*, 12:010, 2008.
- [21] M. Baumgart, C. Marcantonini, and I. W. Stewart. Parton shower to NLO in the power expansion. To appear.
- [22] Thomas Becher and Matthew D. Schwartz. A Precise determination of α_s from LEP thrust data using effective field theory. *JHEP*, 07:034, 2008.
- [23] A. V. Belitsky and A. V. Radyushkin. Unraveling hadron structure with generalized parton distributions. *Phys. Rept.*, 418:1–387, 2005.
- [24] M. Beneke, A. P. Chapovsky, M. Diehl, and T. Feldmann. Soft-collinear effective theory and heavy-to-light currents beyond leading power. *Nucl. Phys.*, B643:431– 476, 2002.
- [25] M. Beneke and T. Feldmann. Multipole-expanded soft-collinear effective theory with non-abelian gauge symmetry. *Phys. Lett.*, B553:267–276, 2003.
- [26] C. F. Berger et al. Next-to-Leading Order QCD Predictions for W+3-Jet Distributions at Hadron Colliders. *Phys. Rev.*, D80:074036, 2009.
- [27] Axel Bredenstein, Ansgar Denner, Stefan Dittmaier, and Stefano Pozzorini. NLO QCD corrections to top anti-top bottom anti-bottom production at the LHC: 2. full hadronic results. 2010.

- [28] Stanley J. Brodsky, Dae Sung Hwang, and Ivan Schmidt. Final-state interactions and single-spin asymmetries in semi-inclusive deep inelastic scattering. *Phys. Lett.*, B530:99–107, 2002.
- [29] S. Catani, Yuri L. Dokshitzer, M. Olsson, G. Turnock, and B. R. Webber. New clustering algorithm for multi - jet cross-sections in e+ e- annihilation. *Phys. Lett.*, B269:432–438, 1991.
- [30] S. Catani, G. Ferrera, and M. Grazzini. W boson production at hadron colliders: the lepton charge asymmetry in NNLO QCD. 2010.
- [31] S. Catani, F. Krauss, R. Kuhn, and B. R. Webber. QCD Matrix Elements + Parton Showers. *JHEP*, 11:063, 2001.
- [32] S. Catani and M. H. Seymour. A general algorithm for calculating jet cross sections in NLO QCD. Nucl. Phys., B485:291–419, 1997.
- [33] Stefano Catani, Stefan Dittmaier, Michael H. Seymour, and Zoltan Trocsanyi. The dipole formalism for next-to-leading order QCD calculations with massive partons. *Nucl. Phys.*, B627:189–265, 2002.
- [34] Junegone Chay, Chul Kim, and Adam K. Leibovich. Quark mass effects in the soft-collinear effective theory and $\overline{B} \to X_s \gamma$ in the endpoint region. *Phys. Rev.*, D72:014010, 2005.
- [35] Jui-yu Chiu, Frank Golf, Randall Kelley, and Aneesh V. Manohar. Electroweak Sudakov Corrections using Effective Field Theory. *Phys. Rev. Lett.*, 100:021802, 2008.
- [36] John C. Collins. Leading-twist Single-transverse-spin asymmetries: Drell- Yan and Deep-Inelastic Scattering. *Phys. Lett.*, B536:43–48, 2002.
- [37] John C. Collins, Davison E. Soper, and George Sterman. Factorization of hard processes in qcd. Adv. Ser. Direct. High Energy Phys., 5:1–91, 1988.
- [38] G. Corcella et al. Herwig 6: An event generator for hadron emission reactions with interfering gluons (including supersymmetric processes). *JHEP*, 01:010, 2001.
- [39] G. Curci, W. Furmanski, and R. Petronzio. Evolution of Parton Densities Beyond Leading Order: The Nonsinglet Case. *Nucl. Phys.*, B175:27, 1980.
- [40] M. A. Dobbs et al. Les Houches guidebook to Monte Carlo generators for hadron collider physics. 2004.
- [41] R. Keith Ellis, W. Furmanski, and R. Petronzio. Power Corrections to the Parton Model in QCD. Nucl. Phys., B207:1, 1982.
- [42] R. Keith Ellis, W. Furmanski, and R. Petronzio. Unraveling Higher Twists. Nucl. Phys., B212:29, 1983.

- [43] R. Keith Ellis and J. C. Sexton. Qcd radiative corrections to parton parton scattering. Nucl. Phys. B, 269:445, 1986.
- [44] R. Keith Ellis, W. James Stirling, and B. R. Webber. QCD and collider physics. Camb. Monogr. Part. Phys. Nucl. Phys. Cosmol., 8:1–435, 1996.
- [45] R. Keith Ellis and W. Vogelsang. The evolution of parton distributions beyond leading order: the singlet case. 1996.
- [46] Markus Finkemeier, Howard Georgi, and Matt McIrvin. Reparameterization invariance revisited. *Phys. Rev.*, D55:6933–6943, 1997.
- [47] Sean Fleming, Andre H. Hoang, Sonny Mantry, and Iain W. Stewart. Jets from massive unstable particles: Top-mass determination. *Phys. Rev. D*, 77:074010, 2008.
- [48] Sean Fleming, Andre H. Hoang, Sonny Mantry, and Iain W. Stewart. Top Jets in the Peak Region: Factorization Analysis with NLL Resummation. *Phys. Rev.* D, 77:114003, 2008.
- [49] Sean Fleming, Adam K. Leibovich, and Thomas Mehen. Resumming the coloroctet contribution to e+ e- -¿ J/psi + X. Phys. Rev. D, 68:094011, 2003.
- [50] E. G. Floratos, D. A. Ross, and Christopher T. Sachrajda. Higher Order Effects in Asymptotically Free Gauge Theories: The Anomalous Dimensions of Wilson Operators. *Nucl. Phys.*, B129:66–88, 1977.
- [51] E. G. Floratos, D. A. Ross, and Christopher T. Sachrajda. Higher Order Effects in Asymptotically Free Gauge Theories. 2. Flavor Singlet Wilson Operators and Coefficient Functions. *Nucl. Phys.*, B152:493, 1979.
- [52] Stefano Frixione and Bryan R. Webber. Matching NLO QCD computations and parton shower simulations. *JHEP*, 06:029, 2002.
- [53] Bodo Geyer and Markus Lazar. Parton distribution functions from nonlocal light-cone operators with definite twist. *Phys. Rev.*, D63:094003, 2001.
- [54] Tanju Gleisberg et al. SHERPA 1.alpha, a proof-of-concept version. *JHEP*, 02:056, 2004.
- [55] Tanju Gleisberg and Stefan Hoche. Comix, a new matrix element generator. JHEP, 12:039, 2008.
- [56] V. N. Gribov and L. N. Lipatov. e⁺ e⁻ pair annihilation and deep inelastic e p scattering in perturbation theory. Sov. J. Nucl. Phys., 15:675–684, 1972.
- [57] D. J. Gross and Frank Wilczek. Asymptotically Free Gauge Theories. 1. Phys. Rev., D8:3633–3652, 1973.

- [58] Andri Hardmeier, Enrico Lunghi, Dan Pirjol, and Daniel Wyler. Subleading collinear operators and their matrix elements. *Nucl. Phys.*, B682:150–182, 2004.
- [59] Andre H. Hoang and Stefan Kluth. Hemisphere Soft Function at $O(\alpha_s^2)$ for Dijet Production in e^+e^- Annihilation. 2008.
- [60] S. Jadach and M. Skrzypek. QCD evolution in the fully unintegrated form. Acta Phys. Polon., B40:2071–2096, 2009.
- [61] S. Jadach, M. Skrzypek, A. Kusina, and M. Slawinska. Exclusive Monte Carlo modelling of NLO DGLAP evolution. 2010.
- [62] R. L. Jaffe and M. Soldate. Twist Four in the QCD Analysis of Leptoproduction. *Phys. Lett.*, B105:467–472, 1981.
- [63] R. L. Jaffe and M. Soldate. Twist Four in Electroproduction: Canonical Operators and Coefficient Functions. *Phys. Rev.*, D26:49–68, 1982.
- [64] Robert L. Jaffe. Spin, twist and hadron structure in deep inelastic processes. 1996.
- [65] Ambar Jain, Ignazio Scimemi, and Iain W. Stewart. Two-loop Jet-Function and Jet-Mass for Top Quarks. *Phys. Rev.*, D77:094008, 2008.
- [66] Xiang-dong Ji, Jian-ping Ma, and Feng Yuan. QCD factorization for semiinclusive deep-inelastic scattering at low transverse momentum. *Phys. Rev.*, D71:034005, 2005.
- [67] R. Keith Ellis, Kirill Melnikov, and Giulia Zanderighi. W+3 jet production at the Tevatron. *Phys. Rev.*, D80:094002, 2009.
- [68] Nikolaos Kidonakis, Gianluca Oderda, and George Sterman. Threshold resummation for dijet cross sections. Nucl. Phys. B, 525:299–332, 1998.
- [69] Wolfgang Kilian, Thorsten Ohl, and Jurgen Reuter. WHIZARD: Simulating Multi-Particle Processes at LHC and ILC. 2007.
- [70] John B. Kogut and Davison E. Soper. Quantum Electrodynamics in the Infinite Momentum Frame. Phys. Rev., D1:2901–2913, 1970.
- [71] Christopher Lee and George Sterman. Universality of nonperturbative effects in event shapes. 2006.
- [72] Christopher Lee and George Sterman. Momentum flow correlations from event shapes: Factorized soft gluons and soft-collinear effective theory. *Phys. Rev. D*, 75:014022, 2007.
- [73] Clarence L. Y. Lee. Reparameterization invariance in heavy particle effective field theories. 1997.

- [74] Adam K. Leibovich, Zoltan Ligeti, and Mark B. Wise. Comment on quark masses in SCET. *Phys. Lett.*, B564:231–234, 2003.
- [75] Michelangelo L. Mangano, Mauro Moretti, Fulvio Piccinini, Roberto Pittau, and Antonio D. Polosa. ALPGEN, a generator for hard multiparton processes in hadronic collisions. *JHEP*, 07:001, 2003.
- [76] Aneesh V. Manohar. Effective field theories. 1996.
- [77] Aneesh V. Manohar, Thomas Mehen, Dan Pirjol, and Iain W. Stewart. Reparameterization invariance for collinear operators. *Phys. Lett. B*, 539:59–66, 2002.
- [78] Aneesh V. Manohar and Iain W. Stewart. The zero-bin and mode factorization in quantum field theory. *Phys. Rev. D*, 76:074002, 2007.
- [79] Aneesh V. Manohar and Mark B. Wise. Heavy quark physics. Camb. Monogr. Part. Phys. Nucl. Phys. Cosmol., 10:1–191, 2000.
- [80] Claudio Marcantonini and Iain W. Stewart. Reparameterization invariant collinear operators. 2008.
- [81] G. Marchesini et al. HERWIG: A Monte Carlo event generator for simulating hadron emission reactions with interfering gluons. Version 5.1 - April 1991. *Comput. Phys. Commun.*, 67:465–508, 1992.
- [82] Paolo Nason. A new method for combining NLO QCD with shower Monte Carlo algorithms. *JHEP*, 11:040, 2004.
- [83] C. G. Papadopoulos and M. Worek. HELAC: A Monte Carlo generator for multijet processes. 2006.
- [84] Michael Edward Peskin and Daniel V. Schroeder. An Introduction to quantum field theory. Reading, USA: Addison-Wesley (1995) 842 p.
- [85] Dan Pirjol and Iain W. Stewart. A complete basis for power suppressed collinearultrasoft operators. *Phys. Rev.*, D67:094005, 2003.
- [86] H. David Politzer. Power Corrections at Short Distances. Nucl. Phys., B172:349, 1980.
- [87] A. Retey and J. A. M. Vermaseren. Some higher moments of deep inelastic structure functions at next-to-next-to leading order of perturbative QCD. *Nucl. Phys.*, B604:281–311, 2001.
- [88] Ira Z. Rothstein. TASI lectures on effective field theories. 2003.
- [89] Ira Z. Rothstein. Factorization, power corrections, and the pion form factor. *Phys. Rev.*, D70:054024, 2004.

- [90] Matthew D. Schwartz. Resummation and NLO Matching of Event Shapes with Effective Field Theory. *Phys. Rev. D*, 77:014026, 2008.
- [91] Torbjörn Sjöstrand, Stephen Mrenna, and Peter Skands. Pythia 6.4 physics and manual. JHEP, 05:026, 2006.
- [92] Torbjörn Sjöstrand, Stephen Mrenna, and Peter Skands. A Brief Introduction to PYTHIA 8.1. Comput. Phys. Commun., 178:852–867, 2008.
- [93] Maciej Skrzypek and Stanislaw Jadach. The Exclusive NLO DGLAP Kernels for Non-Singlet Evolution. 2009.
- [94] M. Stratmann and W. Vogelsang. Next-to-leading order evolution of polarized and unpolarized fragmentation functions. *Nucl. Phys.*, B496:41–65, 1997.
- [95] Frank J. Tackmann. Full-phase-space twist expansion in semileptonic and radiative B-meson decays. *Phys. Rev.*, D72:034036, 2005.
- [96] Michael Trott. Jets in effective theory: Summing phase space logs. Phys. Rev. D, 75:054011, 2007.
- [97] Steven Weinberg. The quantum theory of fields. Vol. 2: Modern applications. Cambridge, UK: Univ. Pr. (1996) 489 p.
- [98] Jui yu Chiu, Frank Golf, Randall Kelley, and Aneesh V. Manohar. Electroweak corrections in high energy processes using effective field theory. *Phys. Rev. D*, 77:053004, 2008.
- [99] Jui yu Chiu, Randall Kelley, and Aneesh V. Manohar. Electroweak Corrections using Effective Field Theory: Applications to the LHC. *Phys. Rev. D*, 78:073006, 2008.